

The impact, assessment, and longitudinal course of negative symptoms of schizophrenia.

Savill, Mark

The copyright of this thesis rests with the author and no quotation from it or information derived from it may be published without the prior written consent of the author

For additional information about this publication click this link. http://qmro.qmul.ac.uk/xmlui/handle/123456789/12926

Information about this research object was correct at the time of download; we occasionally make corrections to records, please therefore check the published record when citing. For more information contact scholarlycommunications@qmul.ac.uk

The impact, assessment, and longitudinal course of negative symptoms of schizophrenia

Mark Savill

Submitted in partial fulfilment of the requirements of the Degree of Doctor of Philosophy

I. Contents

II. Statement of originality	7
III. Acknowledgements	9
IV. Abstract	10
V. List of Abbreviations	11

Chapter 1. Introduction
1.1. Introducing schizophrenia13
1.1.1. What is schizophrenia and why is it important?
1.1.2. The typical course of schizophrenia14
1.1.3. Factors associated with the development of schizophrenia
1.1.4. The current diagnostic criterion for schizophrenia and implications for research 18
1.1.5. A history of the conceptual model of schizophrenia
1.2. The symptoms of schizophrenia
1.2.1. Positive symptoms 31
1.2.2. Cognitive symptoms
1.2.3. Negative symptoms 34
1.3. Primary and secondary negative symptoms of schizophrenia
1.3.1. Factors that induce secondary negative symptoms
1.3.3. Deficit syndrome and persistent negative symptoms
1.3.4. Persistent negative symptoms as an alternative to the deficit syndrome
1.5. Longitudinal course of negative symptoms of schizophrenia
1.5.1. Longitudinal course of negative symptoms in patients with deficit syndrome 51
1.6. The impact of negative symptoms on quality of life
1.7. The treatment of negative symptoms53
1.8. The NIMH-MATRICs consensus statement on negative symptoms of schizophrenia 54
1.8.1. The ISCTM and NEWMEDS update on clinical trials in negative symptoms
1.9. Assessing negative symptoms
1.9.1. Structured interviews of negative symptoms59
1.9.2. Pooling negative symptom assessments72

Chapter 2. Rationale and research questions	73
2.1. Rationale for investigation	73
2.2. Study aims and research questions	75
Chapter 3: The longitudinal course of negative symptoms	80
3.1. Introduction	80
3.2. Research questions	82
3.3. Methods	82
3.3.1. Search strategy	82
3.3.2. Eligibility criteria	84
3.3.4. Analysis plan	84
3.4. Results	86
3.4.1. Summary of articles selected	86
3.4.2. Longitudinal course of negative symptoms	90
3.4.3. Change in negative symptoms reported by different assessment measures	93
3.4.4. Comparing control arms between drug and non-drug trials	95
3.4.5. Examination of individual negative symptoms	96
3.4.6. Comparing those that do and do not adopt criteria in line with the diagnosis of	:
persistent negative symptoms	96
3.4.7. Examination of publication bias	99
3.4.8. Eligible studies not pooled into the main analysis	99
3.5. Discussion	99
3.5.1. Main findings	99
3.5.2. Strengths and limitations.	100
3.5.4. Summary	102

Chapter 4. Association between negative symptoms and positive and depressive symptoms
after adopting different study inclusion criteria103
4.1. Introduction
4.2. Research Questions 106
4.3. Methods
4.3.1. Sample
4.3.2. Scales
4.3.3. Prominent and predominant symptom inclusion criteria under evaluation: 107
4.3.4. Analysis plan 110

4.4. Results	110
4.4.1. Summary of sample	110
4.4.2. Baseline and change in symptoms over time	112
4.4.3. Association between negative symptoms, and positive and depressive sym	nptoms
after adopting different eligibility criteria	114
4.4.4. Prominent negative symptom inclusion criteria:	116
4.4.5. Predominant negative symptom exclusion criteria	117
4.5. Discussion	124
4.5.1. Main findings	124
4.5.2. Strengths and limitations	125
4.5.4. Summary	127
Chapter 5. Comparing the CAINS and the PANSS as a measure of negative symptoms	128
5.1. Introduction	128
5.2. Research questions	131
5.3. Method	131
5.3.1. Sample	131
5.3.2. Scales	132
5.3.3. Analysis plan	133
5.4. Results	136
5.4.1. Description of sample	136
5.4.2. Association between variables	137
5.4.3. Relationship between the PANSS negative and the CAINS total subscales	138
5.4.4. Degree of agreement between the PANSS negative subscale and the CAIN	S
experiential and expressive subscales	139
5.4.5. A comparison of the change scores in the PANSS negative and the CAINS t	otal score
over time	141
5.4.6. A comparison of the PANSS negative subscales and the CAINS experiential	and
expressive subscales over time	144
5.4.7. An assessment of the incremental validity of the CAINS over the PANSS ne	gative
subscales	149
5.5. Discussion	156
5.5.1. Main findings	156
5.5.2. Strengths and limitations:	156
5.5.4. Summary	159

Chapter 6. Association between negative symptoms and subjective quality of life
6.1. Introduction
6.2. Research questions
6.3. Method
6.3.1. Sample
6.3.2. Assessment tools
6.3.3. Analysis
6.4. Results
6.4.1. Description of sample
6.4.2. The association between negative symptoms and subjective quality of life 166
6.4.3. The association between the changes in negative symptoms and subjective quality
of life over time
6.5. Discussion
6.5.1. Main findings
6.5.2. Strengths and weaknesses168
6.5.4. Conclusion

Chapter 7. The impact of negative symptoms on the subjective initial appraisal of psychiatric	2
inpatient treatment	170
7.1. Introduction	170
7.2. Research questions	172
7.3. Methods	172
7.3.1. Design	172
7.3.2. Sample	172
7.3.3. Measures	173
7.3.4. Analysis	175
7.4. Results	176
7.4.1. Sample characteristics	176
7.4.2. Symptoms and other patient characteristics associated with initial appraisal of	
treatment	179
7.5. Discussion	182
7.5.1. Main findings	182
7.5.2. Strengths and limitations	183
7.5.3. Summary	184

Chapter 8. Discussion	186
8.1. Thesis aims	186
8.2. Summary of findings and comparisons to the literature	187
8.2.1. The change in negative symptoms over time in outpatients	187
8.2.2. The impact of adopting different symptom inclusion criteria in negative sympton	n
trials	189
8.2.3. A comparison of the different negative symptom assessment scales	191
8.2.4. The Impact of adopting different eligibility criteria on the sample pool size	192
8.2.5. What is nature of the relationship between the CAINS and PANSS negative	
subscale?	193
8.2.6. The association between the negative symptom construct and other variables w	hen
measured by the CAINS, PANSS and BPRS	194
8.2.7. The association between negative symptoms and other constructs when express	sive
and experiential constructs are considered separately	196
8.3. Project Strengths and Limitations	197
8.4. Implications	198
8.4.1. Implications on research	198
8.4.2. Implications on policy	200
8.4.3. Implications on practice	201
8.5. Future directions	202
8.6. Concluding Statement	204
References	206
Appendices	241

II. Statement of originality

I, Mark Savill, confirm that the research included within this thesis is my own work or that where it has been carried out in collaboration with, or supported by others, that this is duly acknowledged below and my contribution indicated. Previously published material is also acknowledged below.

I attest that I have exercised reasonable care to ensure that the work is original, and does not to the best of my knowledge break any UK law, infringe any third party's copyright or other Intellectual Property Right, or contain any confidential material. I accept that the College has the right to use plagiarism detection software to check the electronic version of the thesis. I confirm that this thesis has not been previously submitted for the award of a degree by this or any other university. The copyright of this thesis rests with the author and no quotation from it or information derived from it may be published without the prior written consent of the author.

Signature:

Date:

Details of collaboration and publications:

The main findings of chapter 3 and chapter 7 have been published. In all chapters I was responsible for study design, analysis, interpretation, and manuscript preparation. Aspects of study design, interpretation, and presentation of the findings was supported by my primary supervisor Professor Stefan Priebe. Dr Stephen Bremner provided statistical advice for chapters 3 and 7, and Dr Ceire Costelloe provided statistical advice for chapter 4.In chapter 3 I completed all aspects the systematic search, study selection and data extraction procedure. For purposes of reliability, Ciara Banks duplicated 25% of the abstract screening and 50% of the full paper screening, whilst Husnara Khanom duplicated the remaining 50% of the full paper screen. In chapter 4, the original data was obtained from a previously published trial evaluating the effectiveness of a structured communication intervention (Priebe et al., 2007). The data analysed in chapters 5 and 6 was obtained from a trial evaluating the effectiveness of body psychotherapy as a treatment for negative symptoms of schizophrenia (Priebe et al.,

2013), for which I have been the Trial Manager for the entire duration of the study. In this role I was involved in study design, and the management of participant recruitment and data collection. In chapter 7, a pooled analysis of data from three previous studies was completed (Kallert et al., 2005; Kallert et al., 2007; Priebe et al., 2010)

Published findings:

Appendix I: Savill, M., C. Banks, H. Khanom, and S. Priebe (2014). Do negative symptoms of schizophrenia change over time? A meta-analysis of longitudinal data. *Psychological Medicine 45*, 1613-1627.

Appendix II: Savill, Mark, Jelena Jankovic, Christina Katsakou, Thomas Kallert, and Stefan Priebe (2012). Symptom levels and initial appraisal of hospital treatment in patients with schizophrenia. *Psychiatry Research* 199, 2, 79-83.

III. Acknowledgements

Firstly I would also like to thank my supervisor, Professor Stefan Priebe, for the chance to work on projects which have cemented my dedication to pursuing a career in psychiatric research, and for the opportunity to undertake a PhD under his expert guidance. During this period I have learnt a huge amount about the research process for which I will always be grateful. I would also like to thank all the members of the Unit for Social and Community Psychiatry for their moral support and wonderful insight over the past five years. It was a real pleasure working with so many bright, committed, and interesting people. In particular, I would like to thank Dr Stephen Bremner and Dr Ceire Costelloe for their statistical advice (and their patience!), and Ciara Banks and Husnara Khanom for their assistance in the systematic review.

Many of the ideas explored in this thesis were developed through ongoing discussions with colleagues involved in the NESS project. As a consequence, I would like to take the opportunity to thank Professor Til Wykes, Professor Richard Bentall, Professor Christoph Lauber, Professor Frank Rohricht, Nina Papadopoulos, Stavros Orfanos and my second supervisor, Dr. Ulrich Reininghaus for their input and expertise. I would also like to thank Professor Thomas Kallert for the use of the EDEN dataset in the pooled analysis.

I would like to convey my gratitude to my parents, John Savill and Hilary Lucas, in addition to Alan Lucas, for their continual patience and support both during this period and throughout my education. It is their support that has enabled me to pursue a career in research for which I will always be indebted.

Finally, I dedicate this thesis to my wife, Dr. Kristin Savill, for her unending support, encouragement, kindness, patience, advice, and love. Here's to many exciting opportunities in the future.

IV. Abstract

The negative symptoms of schizophrenia include impoverished speech, affective blunting, an inability to anticipate pleasure, asociality, and amotivation. Current effective treatment options are limited, which is a significant issue given their strong association to functional impairment. The primary aims of this thesis are to explore areas which may inform the design of clinical trials addressing these symptoms, build upon recent advances in negative symptom assessment, and further understand the impact of negative symptoms on outcomes.

A meta-analysis examining the longitudinal course of negative symptoms demonstrated a consistent reduction in negative symptoms over the study period in all treatment conditions examined, including in studies which adopted different eligibility criteria to minimise the presence of secondary negative symptoms. A comparison of different eligibility criteria revealed that whilst these criteria reduce the association between negative and depressive symptoms, their association to positive symptoms remains largely unaffected, and can substantially reduce the proportion of available participants.

Given existing negative symptom assessment tools have been identified as a significant barrier in the development of new treatments, a comparison of assessment methods was performed. Results revealed the Clinical Assessment Interview for Negative Symptoms to be a more sensitive instrument, and a better predictor of functional impairments, relative to the negative subscale of the Positive and Negative Syndrome Scale. Findings described here also demonstrate that the link between negative symptoms and subjective quality of life may be stronger than previously assumed, relating exclusively to experiential deficits. Finally, manic and positive symptoms, rather than negative symptoms, were found to be associated with inpatient treatment appraisal, an important factor in treatment outcomes.

Overall, this thesis describes key findings which demonstrate the importance of the appropriate assessment of negative symptoms of schizophrenia, shine new light on symptom progression over time, and highlight a need to further refine clinical trial inclusion criteria.

V. List of Abbreviations

- APA: American Psychiatric Association
- ATC: Anatomical Therapeutic Chemical Classification
- **BDI: Beck Depression Inventory**
- **BNSS: Brief Negative Symptom Scale**
- **BPRS: Brief Psychiatric Rating Scale**
- BSA: Broad Spectrum Antipsychotics
- CAINS: Clinical Assessment Interview for Negative Symptoms
- CANSAS: Collaboration to Advance Negative Symptom Assessment in Schizophrenia
- CAT: Client Assessment of Treatment scale
- **CFA: Confirmatory Factor Analysis**
- CGI-SCH: Clinical Global Impressions Scale Schizophrenia
- CI: Confidence Interval
- CPRS: Comprehensive Psychopathological Rating Scale
- DDD: Defined Daily Dose
- DSM: Diagnostic and Statistical Manual of Mental Disorders
- EFA: Exploratory Factor Analysis
- EPS: Extrapyramidal Symptoms
- ES: Effect Size
- FGA: First Generation Antipsychotic
- HDRS: Hamilton Depression Rating Scale
- ICC: Intraclass Correlation Coefficient
- ICD: International Classification of Diseases
- LOCF: Last Observation Carried Forward
- MANSA: Manchester Short Assessment of Quality of life

MATRICS: Measurement and Treatment Research to Improve Cognition in Schizophrenia

MCCB: MATRICS Consensus Cognitive Battery

NIMH: National Institute for Mental Health

NIHR: National Institute for Health Research

NSA: Negative Symptom Assessment Scale

OCCPI: Operational Criteria Checklist of Psychotic Illness

OQOL: Objective Quality of Life

PANSS: Positive and Negative Syndrome Scale

SANS: Scale to Assess Negative Symptoms

SAPS: Scale to Assess Positive Symptoms

SCID: Structured Clinical Interview for DSM Disorders

SD: Standard Deviation

SDS: Schedule for the Deficit Syndrome

SE: Standard Error

SGA: Second Generation Antipsychotics

SIR: Subjective Initial Response

SNS: Social Network Scale

SQOL: Subjective Quality of Life

TUS: Time Use Survey

WHO: World Health Organisation

Chapter 1. Introduction

1.1. Introducing schizophrenia

1.1.1. What is schizophrenia and why is it important?

Schizophrenia is a serious mental health disorder characterised by a disruption in thinking and emotional responses. Whilst the point incidence is relatively low (2-10/10,000), the prevalence is much higher due to the chronic nature of the illness (70/10,000) (Saha et al., 2005). Despite recent evidence which suggests that the long term negative impact of the illness may have been previously overestimated due to sample biases (Morgan et al., 2014), it is clear that schizophrenia is a highly disabling condition in terms of physical, social, and functional outcomes.

Mortality and suicide rates in those diagnosed with schizophrenia is disproportionately high relative to the general population (Hor and Taylor, 2010, McGrath et al., 2008, Reininghaus et al., 2015), and co-morbid disorders, particularly substance use disorders, are common (Buckley et al., 2009). Patients diagnosed with schizophrenia are more likely to suffer chronic physical conditions such as obesity, type II diabetes and coronary heart disease, and medications taken for schizophrenia have been associated with sexual dysfunction, obesity, movement-disorders, and cataracts (Marder et al., 2014). In the AESOP-10 study, almost three quarters of the participants assessed remained unemployed for at least 75% of the time over a period of ten years, and further psychiatric hospital admissions occurred in 88% of cases (Morgan et al., 2014). Impairments in work and academic performance, interpersonal relations, and self-care are so intrinsic to our current definition of schizophrenia that these features form part of the diagnostic criterion for the illness (APA, 2013). In a cohort study spanning 15 years, 80% of participants were found to have a persistent impairment in social functioning (Wiersma et al., 2000), and the disorder can result in substantial caregiver burden (McDonell et al., 2003). Following the first psychotic episode, only 13.7% of patients (95% CI= 6.4% - 20.9%) were found to meet criteria for full recovery for at least two of the following 5 years (Robinson et al., 2014).

In addition to the significant impact that the disorder can have on both the individual and caregivers, schizophrenia is also recognised to have a major economic cost. In 2012, the estimated annual cost of schizophrenia to English society was placed was £11.8 billion, £7.2

billion of which costed to the public sector (Andrew et al., 2012). In summary, schizophrenia is recognised to have a substantial negative impact on the individual, families and care givers, and to society at large, and therefore presents a need for future research to help alleviate the devastating impact of this disorder.

1.1.2. The typical course of schizophrenia

Prior to the onset of schizophrenia individuals typically experience a period known as the prodromal phase, which can last for a number years before acute psychotic symptoms surface (Häfner et al., 1999). In his stage, which usually occurs in late teens or early adulthood, it is common for people to experience a decline in global functioning. This can include various forms of cognitive impairments such as deficits in executive function, general intelligence, verbal and visual memory, verbal fluency, attention, working memory, and social cognition (Fusar-Poli et al., 2012), in addition to depressed mood or anxiety, infrequent or attenuated psychotic symptoms, or negative symptoms such as social withdrawal (Häfner et al., 1999). During this period it is also common for patients to experience a decline in social, occupational or academic performance (Cornblatt et al., 2007). In a longitudinal study following those that qualify for prodromal syndrome, 35% of cases were found to go on to develop psychosis within a period of 30 months (Cannon et al., 2008). In those that do go on to develop schizophrenia, patients typically experience acute psychotic symptoms such as delusions, hallucinations, thought disorder, and a further deterioration in cognitive and social functioning. Following treatment, most patients transition on to the chronic phase of the disorder where psychotic symptoms diminish, but significant impairments in functioning and negative symptoms remain. Approximately 80% of patients in the acute phase of the disorder go on to experience a substantial reduction in psychotic symptoms, although many (~70%) will go on to experience a relapse into the acute phase again within 5-7 years (Wiersma et al., 1998). The antecedents of regressing back to the acute phase of the disorder are not entirely clear, however nonadherence to mediation and social withdrawal have been found to be associated (Robinson et al., 1999). Other predictors identified include younger age, earlier illness onset, substance use disorder, and prior episodes of relapse (Ascher-Svanum et al., 2010). Whilst complete remission from the disorder is relatively rare (Jääskeläinen et al., 2013), in a more recent study there is evidence to suggest that periods of remission are more common than was previously believed (Morgan et al., 2014).

1.1.3. Factors associated with the development of schizophrenia

Whilst the causes of schizophrenia are undetermined, a number of risk factors have been identified. Of these, having a positive family history has been recognised as the most influential factor. Lifetime risk is 6.5 times higher in people who have first degree relatives diagnosed with the disorder (Kendler and Diehl, 1993). Twin studies suggest a strong genetic component for this association, with the risk factor as high as 40% in monozygotic twins where one has been diagnosed with the disorder (Cardno et al., 1999).

Two systematic reviews suggest that the incidence rate of schizophrenia is higher in males in comparison to females, with a male to female ratio of 1.4:1 (95% CI: 1.3 - 1.6) (Aleman et al., 2003, McGrath et al., 2008). In addition, males appear to have an earlier age of onset, experience more severe negative symptoms, and report a worse outcome (Jablensky, 2000). However, as described in a review by McGrath et al. (McGrath et al., 2008) this higher incidence rate did not appear to translate to a higher prevalence rate in males. The reason for this is not entirely clear, however it is possible that this could be explained at least in part by the higher mortality rates in male schizophrenia patients in comparison to females sufferers (standardized mortality ratio's, male=2.8, females=2.4) (Ösby et al., 2000).

Another factor associated with developing psychosis is migrant status, with both the incidence and prevalence of schizophrenia significantly higher in migrants relative to native-born individuals (Fearon et al., 2006, McGrath, 2006). This higher incidence rate of psychosis was found in both first and second generation immigrants, which suggests that the difference is not attributable to events which occurred prior to immigration (Bourque et al., 2011). In addition, this risk appears to be inversely related to the proportion of migrants within their local community (Boydell et al., 2001). In the UK, higher incidence rates of psychosis in Caribbean migrants have been consistently reported (Fearon et al., 2006, Sharpley et al., 2001). However, with rates not found to be higher in the countries of origin, this suggests that the issue is related to being an immigrant, rather than having any genetic predisposition towards the illness (Messias et al., 2007).

Despite the risk being relatively small, there has been consistent evidence which suggests that people that are born during the winter months are significantly more likely to develop psychosis (Davies et al., 2003). The reasons for this are not entirely clear, however a number of theories have been proposed. One hypothesis states that influenza infections during gestation may be related to increased rates of schizophrenia, however no evidence for this was found in a large epidemiological study (Morgan et al., 1997). Another theory proposes that this may be

associated with biometeorologic variables such as temperature, ultra-violet radiation and precipitation, which is supported by the finding that people born at higher latitudes have both higher prevalence and incidence rates (Saha et al., 2006). Related to this, McGrath went on to propose that low pre-natal vitamin D may be a risk modifying factor for schizophrenia (McGrath, 1999). To complicate the issue of birth seasonality and its impact on schizophrenia further, there is evidence to suggest that this effect is not consistent in all types of the illness. In a pooled analysis of 6 countries, patients diagnosed with the deficit syndrome of schizophrenia (see section 1.3.3 for a summary) were significantly more likely to have been born in the months of June-July, relative to non-deficit syndrome schizophrenia patients (Messias et al., 2004).

In addition to the season in which the gestational period occurs, prenatal and obstetric complications have also long been implicated in increase rates of schizophrenia (Kraepelin, 1971). In a systematic review by Geddes and colleagues (Geddes et al., 1999) the diagnosis of schizophrenia was associated with premature rupture of membranes, a gestational age below 37 weeks, and the use of resuscitation or an incubator. There was also evidence of a weak effect between low birth weight and a forceps delivery. In a review by Cannon and colleagues (Cannon et al., 2002), different risk complications were grouped into three different factors; complications in pregnancy, abnormal foetal growth and development; and delivery complications are associated with higher incidence of schizophrenia have been proposed, including the effect of hypoxia and or Ischemia (Geddes et al., 1999, Zornberg et al., 2000), malnutrition (Susser and Lin, 1992), or developmental impairments caused by premature birth (Nosarti et al., 2012).

Another possible risk factor is paternal age, where there is evidence to suggest that this is positively associated with an increased risk of schizophrenia (Brown et al., 2014, Zammit et al., 2003). In one cohort study an increase in relative risk by a factor of 2.96 (95% CI 1.60-5.47) was detected in the offspring of fathers over 55, in comparison to those aged 20-24 (Malaspina et al., 2001). Malaspina postulated that this effect may result from *de novo* mutations in sperm cells. In support of this argument, it has since been found that the positive association with paternal age is only significant among those without any family history of the disease (Sipos et al., 2004).

In addition to obstetric complications and the link between paternal age and schizophrenia, there is evidence of link between the mother experiencing traumatic event up to six months

before or during pregnancy and increased later incidence of schizophrenia in the child (Khashan et al., 2008). Following birth, Larssen (Larsson et al., 2013) found that 85% of people assessed with psychosis reported experiencing childhood trauma, with emotional abuse and neglect particularly common (63% and 67% respectively). In a review by Read (Read et al., 2005) the authors suggested a causal factor between childhood experience of abuse and schizophrenia, however Morgan and Fisher (Morgan and Fisher, 2007) urged caution in determining such a link based on the current evidence base.

Since the 1930's there has been evidence which suggests that the prevalence of schizophrenia is higher in urban, as opposed to semi-urban or rural sites (Faris and Dunham, 1939). Given the cross-sectional design of the analysis however it was argued that this difference may be attributable to a selection bias, where better access to healthcare in urban sites may result in improved illness detection, and encourage migration to major cities. In more recent studies this has been examined prospectively (Lewis et al., 1992, Marcelis et al., 1998), which again supported a higher prevalence in urban sites, leading to the conclusion that the higher prevalence is independent of selection (Krabbendam and Van Os, 2005). A number of possible reasons for this difference have been postulated, such as environmental pollutants (Pedersen and Mortensen, 2006) and overcrowding stresses (Cougnard et al., 2007). In the review by McGrath and colleagues (McGrath et al., 2008) a higher incidence of the disorder was detected, though no changes in prevalence were observed.

Finally, there is evidence to suggest a link between schizophrenia and smoking cannabis. In a large cohort study in Sweden, cannabis was associated with an increased risk of developing schizophrenia, with a dose response effect present (Zammit et al., 2002). In prospective longitudinal study, Arseneault and colleagues (Arseneault et al., 2002) also found a higher risk of developing psychosis with taking cannabis prior to experiencing psychotic symptoms, which suggests the association is not attributable to self-medicating against positive symptoms. In a meta-analysis examining overall effect size and consistency of the association between cannabis and psychosis, (Henquet et al., 2005) an effect was found to be consistent after controlling for any effects of self-medication, with a pooled odds ratio of 2.1 (95% Cl 1.7-2.5). Finally, in an overview of 5 systematic reviews (Minozzi et al., 2010) a consistent association between cannabis use and psychotic symptoms was detected.

Overall, the evidence suggests that biological, psychological and societal factors all have a significant influence on the development of psychosis (Broome et al., 2005). Many of these factors are likely to be inter-related, and further work on the interaction between these

differing risk factors is needed. Such work may advance our understanding of how to detect those at most risk of developing the disorder, which is important given there is evidence to suggest early treatment of at-risk groups may help reduce the likelihood of transitioning to psychosis (Morrison et al., 2004). However, whilst such work may help reduce the proportion of people developing the disorder, a significant proportion of people at present still go on to develop the illness. Consequently, there is still an urgent need to develop new treatments, and to develop an appropriate framework in which therapeutics can be evaluated in the most efficient manner possible.

1.1.4. The current diagnostic criterion for schizophrenia and implications for research

In schizophrenia, diagnosis is determined by the presence of particular symptoms. Despite the existence of a number of diagnostic systems (for a review, see (Berner, 1983), the two most commonly implemented diagnostic tools at present are published by the ICD (International Classification of Diseases) and the DSM (Diagnostic and Statistical Manual of Mental Health Disorders). The DSM is the manual published periodically by the American Psychiatric Association (APA) and primarily adopted in the USA, whilst the ICD is the criterion adopted by the WHO and is most commonly used in the UK and throughout Europe. In both manuals, the concept of schizophrenia is defined as a spectrum disorder, including a number of variations of psychotic illnesses and different disorder subtypes. In order to qualify for a diagnosis of schizophrenia utilising the ICD-10 criteria, the following conditions must be met:

At least one of the following "first rank" symptoms must be present:

a) Thought echo, thought insertion or withdrawal, or thought broadcasting.

b) Delusions of control, influence or passivity, clearly referred to body or limb movements or specific thoughts, actions, or sensations; delusional perception.

c) Hallucinatory voices giving a running commentary on the patient's behaviour, or discussing him between themselves, or other types of hallucinatory voices coming from some part of the body. d) Persistent delusions of a different kind which are culturally inappropriate and completely impossible (e.g. being able to control the weather, or being in communication with aliens from another world).

OR, at least two of the following "second rank" symptoms must be present:

e) Persistent hallucinations in any modality, when occurring every day for at least one month, when accompanied by delusions (which may be fleeting or half-formed) without clear affective content, or when accompanied by persistent over-valued ideas.

f) Neologisms, breaks or interpolations in the train of thought, resulting in incoherence or irrelevant speech.

g) Catatonic behaviour, such as excitement, posturing or waxy flexibility, negativism, mutism and stupor.

h) "Negative" symptoms such as marked apathy, paucity of speech, and blunting or incongruity of emotional responses.

The symptoms should be present most of the time during an episode of psychotic illness, for a duration of at least one month. In addition, the symptoms must have occurred independently of a depressive or manic episode, and not be attributable to either an organic brain disease, or substance abuse withdrawal or intoxication.

In the ICD-10 the diagnosis of schizophrenia is broken down into different subtypes, including paranoid, hebephrenic, catatonic, residual, simple and undifferentiated schizophrenia. Between these subtypes, differences in the presentation of symptoms are evident. In paranoid schizophrenia, delusions and hallucinations must be prominent, and negative symptoms (see section 1.2.3.) must not dominate the clinical picture. In hebephrenic schizophrenia, thought disorder and negative symptoms are dominant, relative to hallucinations and delusions. In catatonic schizophrenia, extremes of behaviour must be prominent for at least two weeks. These can include severe rigidity, stupor, bizarre posturing, command automatism or negativism, or, at the other end of the spectrum, extreme excitement and constant motor activity. In residual schizophrenia, the general criteria must have been met at some point in the past, and negative symptoms must have been present continuously over the past 12 months. In simple schizophrenia a progressive deterioration of functioning and an increase in

negative symptoms must have occurred over at least 12 months, whilst delusions, thought disorder and hallucinations are absent. Lastly, in undifferentiated schizophrenia patients either meet none of the outlined subtypes, or would qualify for multiple subtypes.

In the DSM-V (APA, 2013), the criteria for the diagnosis of schizophrenia include:

At least two of the following symptoms (with at least one of them being A-C) being present for a significant proportion of time during a 1-month period.

- a) Delusions
- b) Hallucinations
- c) Disorganised speech
- d) Grossly disorganised or catatonic behaviour
- e) Negative symptoms

In addition, some form of social or occupational dysfunction must be present for a significant portion of time since the onset of the disturbance, occurring over a period of at least 6 months. Prior to receiving a diagnosis of schizophrenia, both schizoaffective and mood disorder must be ruled out. Furthermore, the disturbance must not be due to physiological effects of substance abuse, withdrawal, or intoxication.

Whilst there is a significant degree of overlap between the diagnostic criteria for the DSM-V and the ICD-10, some differences do exist. In contrast to the ICD-10 criteria, in the DSM-V formulation no distinction between subtypes of the illness is made. In the DSM-V, evidence of social and occupational dysfunction from the point of disturbance onset is required. In the DSM-V, no distinction between Schneiderian first and second rank positive symptoms is made (see section 1.1.5 for a summary). Finally, in the DSM-V recent amendments to the criteria for schizophreniform disorder, schizoaffective disorder, and schizotypal disorder have implications on the specificity of the schizophrenia diagnosis. Given these differences are likely to have implications on who may and may not qualify for a diagnosis of schizophrenia, this may influence the characteristics of the sample evaluated. Consequently, specifying which criteria have been adopted for each part of the investigation should be considered important. In the meta-analysis completed in chapter 3 the aim was to pool a broad range of samples from as many countries as possible. As a result, patients diagnosed in accordance to either ICD or DSM criteria were considered. However in chapters 4-7, where all participants were recruited within Europe, all participants were diagnosed in accordance with ICD-10 criteria. As specified previously, the ICD-10 criteria make distinctions between different subtypes of schizophrenia such as paranoid, hebephrenic, catatonic and residual schizophrenia. As a result, schizophrenia can either be evaluated as a single concept, or with the subtypes each evaluated separately. However, as reviewed by Tandon and colleagues (Tandon et al., 2013), a wealth of evidence exists which suggests that subtype classifications exhibit poor reliability, longitudinal stability, and poor prognostic value (Deister and Marneros, 1993) resulting in little distinction between subtypes (Linscott et al., 2010). Given these criticisms, subtypes were not considered in any of the analysis completed in this investigation. Whilst this is a divergence from the criterion from which all of the patients would have been diagnosed, this approach is reflective of current research practice, as indicated in a review by Braff (Braff et al., 2013).

Related to diagnosis it is important to consider whether schizophrenia alone should be evaluated (ICD-10 criteria F20.0-F20.9), or if other psychotic disorders such as schizotypal disorder (F21), delusional disorders (F22-24), or schizoaffective disorder (F25) should also be included. In the analysis which aimed evaluate negative symptoms in relatively stable outpatients over different timepoints (chapter 3-6) only patients with schizophrenia were included. In the analysis of acute patients at the point of hospital admission (chapter 7) participants with any psychotic disorder were included.

In schizotypal disorder (F21) negative and cognitive symptoms are typically less severe than in schizophrenia (Handest and Parnas, 2005), and patients are not required to exhibit any level of functional impairment to qualify for the diagnosis. In addition, the diagnosis of schizotypal disorder has been found to have relatively poor diagnostic stability over time (Shea et al., 2002). Therefore, patients with schizotypal disorders were excluded from samples evaluating stable outpatients order to avoid over-stating the variability of negative symptoms, and/or under-reporting their severity in clinical samples. In delusional disorders (F22-24), patients do not typically present with negative symptoms, and they are not included as part of the diagnostic criteria. Given these symptoms were the primary focus of the analysis in chapters 3-6, participants with this diagnosis were excluded. In chapter 7, other symptoms such as positive, depressive and manic symptoms were considered, therefore meriting the inclusion of other psychotic disorder as part of the investigation.

Schizoaffective disorder (F25) shares many of the features of schizophrenia and have been included in trials evaluating the effectiveness of negative symptom treatments, such as the CONSIST study (Buchanan et al., 2007). In addition, patients diagnosed with schizoaffective disorder have been included in studies which evaluate primary negative symptoms and the

"deficit syndrome" (Amador et al., 1999, Carpenter et al., 1988) which is an important feature of any investigation into negative symptoms. However, patients with schizoaffective disorder were omitted from all analyses evaluating stable outpatients (i.e. chapters 3-6) for a number of reasons. Firstly, there is evidence to suggest that there are a range of symptomatic, clinical and demographic differences between schizophrenic and schizoaffective clinical samples, with negative symptoms in particular more severe in patients with schizophrenia (Cheniaux et al., 2008). Secondly, schizoaffective disorder appears to follow a different longitudinal course, with significantly better long terms outcomes (Harrow et al., 1997). Thirdly, schizoaffective disorder has also been found to have poor diagnostic stability (Malhi et al., 2008), (Jäger et al., 2011). Consequently, patients with schizoaffective disorder were omitted from the analysis completed in chapters 3-6. In chapter 7 the stability of symptoms over time was not a feature of the analysis, and so therefore patients with schizoaffective disorder were included in this part of the investigation.

1.1.5. A history of the conceptual model of schizophrenia

With no diagnostic physiological markers found in schizophrenia, our concept of the disorder has been built up by the presence and absence of particular behaviours and experiences. Over the years this concept has been repeatedly revised. Given the extensive re-conceptualisations of the illness, providing a full historical account of the disorder is beyond the scope of this thesis. Instead, with the focus of this project limited to the deficient (or negative) features of the illness, the aim is to explore the evolution of how these symptoms have been considered as part of the wider schizophrenia complex. From negative symptoms being defined as the central feature of the disorder at the turn of the 20th century, greater emphasis has been placed on the positive psychotic symptoms of the illness over time. However, in recent years there has been an increasing focus on negative symptoms again due to their impact on various social and functional outcomes. Understanding the evolution of schizophrenia should be considered important of this investigation as a number of recent developments in negative symptoms are rooted in earlier conceptions of the illness.

In a review by Adityanjee and colleagues (Adityanjee et al., 1999), it was proposed that Phillippe Pinel drafted the first integrated description of schizophrenia in the medical literature (Pinel, 1806). Some 70 years later, Karl Kahlbaum (Kahlbaum, 1874) was one of the first to recognise the importance of course and outcome in the nosology of mental health disorders such as the "degenerative psychoses". Kahlbaums approach was adopted by Emil Kraepelin

(Kraepelin, 1971), whose work has since been recognised as amongst the most significant advances in the field of mental health, forming the basis of modern psychiatry (Eysenck, 1960).

Although not the first, Emil Kraeplin (1856-1926) was a major advocate of classifying mental health disorders into categories (Kraepelin, 1971). Central to his work was the distinction between manic depression, which would now be recognised as encompassing mood disorders, and 'dementia praecox', which forms the basis of what we recognise as schizophrenia today. Dementia praecox was characterised by a significant deterioration in attention, memory, and volition, starting during the individuals late teens or early twenties. In this conception of schizophrenia, symptoms relating to what would now be defined as cognitive and negative symptoms were recognised as central features of the disorder.

Kraepelin linked a number of biological abnormalities to the cause of schizophrenia, such as severe lesions in the cerebral cortex. Originally Kraepelin believed that these symptoms typically followed a course of prolonged and progressive deterioration, however in later work this position shifted to recognise that remission or improvement may be possible in at least some cases. Kraeplin went on to validate the existence of dementia praecox cross culturally, and postulated a number of antecedent factors including genetic factors, family history, and obstetric complications. In later work, Kraeplin integrated hebephrenia, catatonia, and paranoia into the dementia praecox classification, proposing that the subtypes are different expressions of a single disease entity.

Following Kraepelin, Eugene Bleuler (1857-1939) proposed an alternative, broader classification of the illness which focused more on the psychological aspects of the disorder (Bleuler, 1950). Bleuler rejected the idea that the illness follows a typically deteriorating course and so rejected the term dementia praecox, instead preferring schizophrenia, which literally translates to *split-mind*. This newly proposed name reflected the importance Bleuler gave to associative splitting as the central feature of the illness. For Bleuler, this splitting of associations gave rise to four features that he believed were central to the disorder, known as the "four A's". The four A's include the loosening of associations, affective flattening, autism, and ambivalence. Other symptoms of schizophrenia, such as hallucinations and delusions, were believed to be accessory symptoms resulting from an inability to manage the core "four A" symptoms. Therefore, similar to Kraepelin's definition of schizophrenia, Bleuler proposed that the primary features of the illness are related to a deficit in functioning, as opposed to the presence of symptoms such as hallucinations or delusions. Building on Kraepelin's work which incorporated different subtypes with the singular entity of schizophrenia, Bleuler proposed

two other subtypes known as 'latent' and 'simplex' forms of the disorder, in which the accessory features were not typically present. The strength of reconceptualising schizophrenia in this manner meant that the concept was better able to capture the heterogeneity of the disorder. However, this also led to criticism that the concept became too expansive (Van Praag, 1976).

Whilst the focus of his enquires were not specific to schizophrenia, the work of neurologist John Hughlings Jackson (1835-1911) was also highly influential to later conceptualisations of the illness (Jackson, 1958). Hughlings Jackson has been credited with developing the conceptual framework for clinical neurophysiology (York and Steinberg, 2011), framing the nervous system as an exclusively sensorimotor mechanism and rejecting metaphysical input. Hughlings Jackson proposed that the nervous system comprises of three levels; the highest level situated in the pre-frontal region, the mid-levels in the cortex, and lowest, least complex functions situated in the medulla and spinal cord. The higher levels were believed to control and inhibit the lower levels, meaning that cortical disease could potentially result in two distinct types of symptoms; the loss of function of the cortex itself, and/or the loss of the ability to inhibit lower nervous system levels. Following the work of John Russell Reynolds (Eadie, 2007) the loss of cortex function was believed to lead to "negative" symptoms, which reflect the loss of higher level functioning. Alternatively, "positive" symptoms were believed to result from an inability to inhibit lower cortical level functioning, and therefore consisted of the appearance of previously inhibited functions. This theory influenced a later distinction made between "positive" and "negative" symptoms of schizophrenia, and also "subtypes" of schizophrenia (Strauss et al., 1974) (see section 1.2.4).

As psychiatry moved towards the standardisation of diagnostic systems, the classification of different symptoms of schizophrenia proposed by Kurt Schneider (Schneider, 1959) became increasingly influential. Schneider proposed a hierarchy of symptoms based on the work of Karl Jaspers (Jaspers et al., 1997) which focused on the form, as opposed to the content of the symptom. In a departure from Kraepelin and Bleuler, greater emphasis was placed on particular positive symptoms of schizophrenia, such as command hallucinations and persistent delusions given they were easier to observe, and at the time were considered to be specific to psychotic disorders. These symptoms, thought disorders and other hallucinations were considered "second rank symptoms". In the diagnostic criteria, fewer first rank symptoms were required to qualify for a diagnosis of schizophrenia, relative to second rank symptoms. At the time, it was thought that this distinction would improve the sensitivity and specificity of the schizophrenia diagnosis. This shift in emphasis towards positive symptoms was further

justified by the observed response of patients to the newly developed antipsychotic medications such as chlorpromazine, which were found to be particularly effective in the treatment of these particular symptoms.

While Schneider's conception of schizophrenia has underpinned the diagnostic criteria utilised over the past 50 years, the specificity of first rank symptoms to schizophrenia has been increasingly questioned (Crichton, 1996, Ihara et al., 2009, Nordgaard et al., 2008, Peralta and Cuesta, 1999a, Tanenberg-Karant et al., 1995). These limitations highlighted lead to the distinction being dropped in the most recent update of the DSM (APA, 2013), although to date it currently remains in the ICD formulation. In addition, with cognitive and negative symptoms being found to have a greater impact on functioning, relative to positive symptoms (Green et al., 2004a, Hunter and Barry, 2012), in recent years there has been there has been an increasing focus on the symptoms originally identified as the core features of the disorder by Kraepelin and Bleuler.

In 1974, Strauss and colleagues attempted to go beyond the categorisation of symptoms proposed by Schneider by outlining a model to explain the pathological mechanisms of the disorder. In doing so, the aim was to enhance the specificity of diagnostic criterion, provide a framework in which treatments can be evaluated, and account for the significant heterogeneity which has consistently been found in the presentation of the disorder (Strauss et al., 1974). In contrast to the theories proposed both by Kraeplin (1919), and Bleuler (1950), Strauss argued that the presentation of schizophrenia is a consequence of multiple processes, as opposed to a singular process. Drawing on the work by Hughlings Jackson (Jackson, 1958) see section 1.1.4.), Strauss postulated that positive symptoms, negative symptoms, and disorders of relating may represent three semi-autonomous processes with different antecedents. Positive symptoms were considered to be a consequence of biological, family environment, and psychological factors. Negative symptoms were considered either to be related to a loss of functioning of higher order cortical function, or the consequence of prolonged isolation and social stigma that patients with chronic schizophrenia typically experience. Negative symptoms could therefore represent core psychopathology, which may in turn lead to accessory positive symptoms (as proposed by both Hughlings Jackson and Bleuler), or be accessory symptoms themselves resulting from external factors. Finally, disorders of relating were considered to be somewhat different, representing dysfunctions in psychological processes that typically precede the onset of psychiatric symptoms.

Following the model outlined by Strauss and colleagues, Crow (Crow et al., 1980) built upon the idea of multiple processes existing within the schizophrenia complex and proposed the

existence of two syndromes of schizophrenia, called "type 1" and "type 2" schizophrenia. In type 1 schizophrenia patients primarily experienced positive symptoms, and was commonly seen in the acute phase of the disorder. Crow stated that type 1 schizophrenics exhibited a good response to neuroleptics, had a relatively good prognosis, and did not experience accompanying intellectual impairment. In contrast, type 2 schizophrenia was characterised by severe and chronic negative symptoms and intellectual impairment. Patients with type 2 schizophrenia were thought to respond poorly to neuroleptics, and had a much poorer longterm prognosis. Crow proposed differing pathological processes, attributing type I schizophrenia to increased dopamine receptor sensitivity, and type 2 schizophrenia to structural changes in the brain such as ventricular enlargement (Andreasen and Olsen, 1982). However, Crow also recognised that both syndromes can occur in the same patient, often at the same time, and suggested that the two dimensions may have the same aetiology (Crow, 1985). This conclusion was supported by Andreasen and colleagues, who found over that three quarters of participants evaluated were found to present with a mixture of positive and negative symptoms (Andreasen et al., 1990, Peralta et al., 1992b). In later work, many of the neurological differences between the positive and negative subtypes found in earlier studies were not reproduced. Finally, it was evident that the subtypes of schizophrenia were not typically stable over time (Marneros et al., 1992).

Following the work of Crow and Strauss (Crow, 1985, Strauss et al., 1974), a series of factor analytic studies were conducted in an attempt to validate the distinction between positive and negative symptoms as separate dimensions. Such work was conducted both as a means of model building (utilising explanatory factor analysis; EFA) and model validation (utilising confirmatory factor analysis; CFA). Whilst the majority of these studies supported a distinction between positive and negative symptoms, the majority of studies failed to uncover a 2-factor solution, with the positive symptom construct in particular being relatively unstable (Andreasen et al., 1990). As identified in a review of models (Peralta and Cuesta, 2001), a series of factor solutions were proposed, a summary of which is outlined in table 1. Whilst a large number of different factor solutions have been proposed, ranging from between 2-11 factors depending on the statistical methodology employed and sample evaluated, a distinction between positive, negative, and disorganised symptoms were consistently proposed (Bilder et al., 1985, Brekke et al., 1994, Liddle, 1987, Peralta et al., 1992a, Toomey et al., 1997), which was later supported in a meta-analytic confirmatory factor analysis (Smith et al., 1998). In some of the more complex models, such as the 11 factor model proposed by Peralta and Cuesta (Peralta and Cuesta, 1999b), they suggested that the different factors could be combined into 3 super-ordinate themes of positive, negative, disorganisation symptoms.

These findings led to the proposal that schizophrenia disorder is characterised by three separate dimensions of positive, negative and disorganised symptoms.

Whilst the distinction between positive, negative and disorganised symptom clusters were relatively consistent, a number of unresolved issues remained, not least the issue regarding the relative instability of the disorganisation symptom cluster (Peralta and Cuesta, 2001). In different factor analytic studies formal thought disorder, poverty of speech content, bizarre behaviour, alogia, inappropriate affect, conceptual disorganisation, difficulties in abstract thinking, disturbance of volition, stereotypical thinking and poor insight have all been included within the disorganization symptom construct (Hardy-Baylé et al., 2003). In an alternative approach to factor analytic studies, Hardy-Bayle instead proposed taking a pathogenic approach to understanding disorganisation in schizophrenia, suggesting that two distinct cognitive processes underpin the symptom dimension. These two processes included deficits in the integration of contextual information (Cohen et al., 1999, Widlöcher and Hardy-Bayle, 1989), and theory of mind deficits (Corcoran et al., 1995, Frith and Corcoran, 1996). Following this work, Harvey (Harvey et al., 2006) further emphasised the distinction between negative and cognitive symptoms.

The distinctions between cognitive, positive, and negative symptoms have now been consistently adopted in the literature, and this will provide the framework for how symptoms will be evaluated throughout this investigation. An outline of what symptoms these different dimensions are comprised of are presented in section 1.2.

Study	Ν	Scale	Method of Analysis	No of factors	Factor loadings
Chronic patient same	le				
(Lorr et al., 1962)	296	Inpatient Multi- dimensional Rating Scale	EFA	10	Mania, depression, negative, paranoid, hallucinations, catatonia, hostility, megalomania, mania, disorganisation
(Overall and Woodward, 1975)	2000	BPRS	Multi- response analysis of variance	11	Psychosis, disorganisation, hostility, anxiety, agitated, depression, depressive- withdrawal, excitement, positive
(Schuldberg et al., 1990)	399	SAPS-SANS	EFA	2	Positive, negative
(Shtasel et al., 1992)	107	SAPS-SANS, BPRS	EFA	4	Negative, disorganisation, psychotic, paranoid
(Minas et al. <i>,</i> 1992)	114	SAPS-SANS	Multidimensi onal scaling	3	Psychosis, disorganisation, negative
(Malla et al., 1993)	155	SAPS-SANS	EFA	3	Psychosis, disorganisation, negative
(Thompson and Meltzer, 1993)	131	SADS and PSE	EFA	3	Psychosis, disorganisation, negative
(Minas et al., 1994)	114	SAPS-SANS	EFA	5	Negative, social dysfunction, delusions, hallucinations, thought disorder
(Lindenmayer et al., 1994)	240	PANSS	CFA	5	Negative, positive, excitement, cognitive, depression-anxiety
(Peralta et al. <i>,</i> 1994)	253	SAPS-SANS	CFA	4	Psychosis, disorganisation, negative, social dysfunction
(Murphy et al. <i>,</i> 1994)	169	SCID, SANS	EFA	3	Psychosis, disorganisation, negative
(Bell et al. <i>,</i> 1994)	146	PANSS	EFA	5	Positive, negative, excitement, depressive, cognitive
(Andreasen et al., 1995)	243	SAPS-SANS	EFA	3	Psychosis, disorganisation, negative
(Maziade et al. <i>,</i> 1995)	138	SAPS-SANS	EFA	3	Psychosis, disorganisation, negative
(Maziade et al., 1995)	131	SAPS-SANS	EFA	3	Psychosis/disorganisation, apathy/anhedonia, affective blunting/alogia
(Kitamura et al., 1995)	584	Checklist	EFA	5	Psychosis, negative, mania, depression, catatonia
(Johnstone and Frith, 1996)	329	Manchester scale	EFA	3	Psychosis, disorganisation, negative
(Mellers et al., 1996)	114	PSE	EFA	4	Hallucinations, delusions, negative, disorganisation
(Cardno et al., 1996)	102	ОССРІ	EFA	5	Paranoid, disorganisation, negative, bizarre delusions, first-rank hallucinations
(Lenzenweger and Dworkin, 1996)	192	SANS, Phillips scales, Venables- O'Connor scale	CFA	4	Psychosis, disorganisation, negative, poor premorbid socia adjustment

Table 1: Summary of factor solutions conducted examining schizophrenia sympto	Fable 1: Summar	1: Summary of factor soluti	ions conducted ex	xamining schizo	ohrenia symptom:
---	-----------------	-----------------------------	-------------------	-----------------	------------------

	(Harvey et al., 1996)	404	Social Behaviour Schedule	EFA	4	Disorganisation, social withdrawal, depression, anti- social behaviour
	(Serretti et al., 1996)	500	OCCPI	EFA	4	Excitement, depression, disorganisation, delusional
	(Peralta et al., 1997)	314	SAPS-SANS	EFA	3	Psychosis, disorganisation, negative
	(Toomey et al., 1997)	630	SAPS-SANS	EFA	5	Negative, disorganisation, social dysfunction, bizarre delusions, auditory hallucinations
	(White et al. <i>,</i> 1997)	1233	PANSS	CFA	5	Negative, psychosis, excitement, depression, autism
	(Ratakonda et al., 1998)	221	SAPS-SANS	EFA	3	Psychosis, disorganisation, negative
	(van Os et al., 1999)	706	CPRS	EFA	4	Depression, mania, psychosis, negative
	(van Os et al. <i>,</i> 1999)	706	OCCPI	EFA	5	Depression, mania, psychosis, negative, disorganisation
	(Hori et al. <i>,</i> 1999)	258	Manchester scale	EFA	3	Psychosis, disorganisation, negative
	(Peralta and Cuesta, 1999b)	660	SAPS-SANS	EFA	11	Poverty of affect and speech, thought disorder/ inappropriate affect, bizarre delusions, social dysfunction, paranoid delusions, bizarre behaviour, auditory hallucinations, manic thought disorder, attention
Re	cent-onset of psych	nosis sam	ples			
	(Gureje et al., 1995)	60	SANS, BPRS	EFA	3	Psychosis, disorganisation, negative
	(Vazquez- Barquero et al., 1996)	86	SAPS-SANS	EFA	4	negative, disorganisation, paranoid
	(Van Os et al., 1996)	166	OCCPI	EFA	7	Catatonia/ bizarre behaviour, bizarre delusions, mania, negative, depression, insight, other delusions
	(McGorry et al., 1998)	509	Multidiagn ostic instrument for Psychosis	EFA	4	Mania, depression, bleulerian, schneiderian
Fol	low-up studies <1	year ^ь				
	(Jackson et al., 1990)	53	Mancheste r scale	EFA	T0: 3	Depression, mixed, positive
					T1: 3	Negative, positive, depression- psychosis
	(Addington and Addington, 1991)	41	SAPS-SANS	EFA	T0: 4	Avolition, disorganisation, poverty of affect and speech, psychosis
					T1: 3	Negative, psychosis, disorganisation
	(Goldman et al. <i>,</i> 1991)	40	BPRS, SANS	EFA	T0: 3	Negative, psychosis, excitement
					T1: 3	Negative, psychosis, excitement

(Dollfus and Petit, 1995a)	57	SAPS-SANS	EFA	T0: 3	Negative, disorganisation, psychosis
				T1: 3	Negative, disorganisation,
(Dollfus and Petit, 1995a)	57	PANSS	EFA	T0: 5	Negative, disorganisation, excitement, psychosis, anxiety
				T1: 5	Negative, excitement, psychosis, disorganisation, depression
(Nakaya et al. <i>,</i> 1999)	86	PANSS	EFA	T0: 5	Negative, hostility, psychosis, disorganisation, stereotyped thinking
Follow-up studies >1 ve	arb			T1: 3	Negative, mixed, abstraction
(Kulhara and Chandiramani,	89	SAPS-SANS	EFA	T0: 3	Negative, psychosis, disorganisation
1990)				T1: 3	Negative, mixed, disorganisation
(Rey et al., 1994)	163	PSE, SANS, PIRS	EFA	T0: 5	Poverty of affect and speech, anhedonia, neurosis, positive symptoms, psychosis
				T1: 5	Poverty of affect and speech, anhedonia, neurosis, positive
(Eaton et al. <i>,</i> 1995)	90	PSE	EFA	T0: 3	Negative, psychosis, disorganisation
				T1: 3	Negative, mixed, disorganisation
(Van der Does et al., 1995)	65	BPRS-E	EFA	T0: 4	Manic, negative, psychosis, depression
				T1: 4	Disorganisation, negative, manic, depression
(Arndt et al., 1995)	65	SAPS-SANS	FA	T0: 3	Negative, disorganisation, psychosis
				T1: 3	Negative, disorganisation, psychosis
(Salokangas, 1997)	156	CPRS	EFA	T0: 5	Negative, delusions, anxiety, mania, hallucinations
				T1: 5	Depression, negative, disorganisation, psychosis, delusions

^aTable adapted from Peralta and Cuesta, 2001

_

^b In cases where multiple assessment points were evaluated, the first and last stage data are included. CPRS= Comprehensive Psychopathological Rating Scale; SANS= Scale of Assessment for Negative Symptoms; SAPS= Scale for Assessment for Positive Symptoms; PSE= Present State Examination; PIRS= Psychological Impairments Rating Scale; PANSS= Positive and Negative Syndrome Scale; BPRS= Brief Psychiatric Rating Scale; FA= Factor analysis; EFA=Explanatory Factor Analysis; CFA= Confirmatory Factor Analysis; OCCPI, SCID= Structured Clinical Interview for DSM Disorders.

1.2. The symptoms of schizophrenia

The symptoms of schizophrenia as they are currently defined are separated into three distinct clusters, known as positive, negative and cognitive symptoms. The severity of these symptoms is defined by their frequency, persistence, and the degree to which they impact on functioning. An outline of what these different clusters constitute is summarised below.

1.2.1. Positive symptoms

The positive symptoms of schizophrenia include hallucinations, delusions, and thought disorders. These symptoms are known as "positive symptoms" due to the fact that their abnormality lies in their presence, as opposed to an absence, of typical human experience.

Hallucinations are defined as "a sensory perception that has the compelling sense of reality of a true perception, but that occurs without external stimulation of the relevant sensory organ" (APA, 2013). Hallucinations can present in auditory, visual, olfactory, haptic, or gustatory form, with the perceived origin being either internal ("pseudohallucinations") or external ("true hallucinations") in nature (Jaspers et al., 1997). Schneider (Schneider, 1959) made a distinction between different forms of auditory hallucinations, defining hallucinations of voices that communicate between themselves, provide a running commentary, or emanate from other parts of the body as specific to psychotic disorders, leading to an increased emphasis in these particular forms in some diagnostic criterion for schizophrenia (see section 1.1.3.). Approximately 70% of people diagnosed with schizophrenia are thought to experience hallucinations in the acute phase of their illness, with auditory hallucinations most common (Sartorius et al., 1974). In a review by Andreasen and Flaum (Andreasen and Flaum, 1991), the percentage of patients reporting experiencing audible thoughts of at least moderate severity ranged from 1.5% to 28% (Carpenter and Strauss, 1974, Koehler et al., 1977).

A number of factors have been thought to influence the occurrence of hallucinations in people with schizophrenia including stress, predisposing factors, environmental stimulation, and reinforcement (Bentall, 1990, Slade, 1976). In addition, a number of theories that underlie the formulation and maintenance of experiencing hallucinations have been postulated, summarised in a review by Bentall (Bentall, 1990). These included classical conditioning (Davies, 1974), the seepage of preconscious material into consciousness (Frith, 1979, West,

1975), a consequence of vivid mental imagery (Mintz and Alpert, 1972), and the link between sub-vocalisation and hallucinations.

Delusions are bizarre or irrational beliefs which are out of context from the society or culture from which they are part. In assessing the presence of delusions, Oltmanns (Oltmanns, 1988) suggested that six dimensions should be considered, including how implausible it is, how strongly it is held, what it is based upon, whether others also hold it, how pre-occupying it is, and the level of distress it causes. Delusions can take a number of forms, including somatic influence, thought withdrawal, thought insertion, thought broadcasting, projected feelings and impulses, delusions of control, and delusional perceptions. Of these, delusional perceptions relating to reference and persecution have been found to be amongst the most commonly reported psychotic symptoms (Sartorius et al., 1986), with persecutory delusions the most common type of delusion which people act upon (Wessely et al., 1993). The concept of persecutory delusions were operationalised by Freeman and Garety (Freeman and Garety, 2000), suggesting that in order for the delusion to be considered persecutory in nature the individual must believe that harm is either occurring, or else will occur to them (meaning harm to friends and or family and not the individual does not count), and that the persecutor has an intention to cause harm.

In addition to hallucinations and delusions, positive symptoms can take the form of thought disorder, the external manifestation of which presents itself in the form of disordered speech. In a review by Rule (Rule, 2005) a number of different terms were identified which cover the different expressions of disordered speech, with a significant degree of overlap and redundancy. Rule summarised the different terms used to define thought disorder symptoms into four overarching themes, including faults in descriptive terminology, mental processing, idiosyncrasy in the use of words, and intelligible speech.

Faults in descriptive terminology cover different symptoms all related to the formulation of the speech itself, and are not connected to underlying cognitive deficits. These include formal thought disorder, akataphasia, paralogia, alogia, laconic speech, poverty of thought, and bradyphasia. Poverty of thought content and empty speech both relate to the content of speech. Desultory thinking, loosening of associations, knights move thinking, and crowding of thought all relate in some form to a jumping from one theme to another without a logical connection. Derailment, tangentiality, "vorbeirden", and loss of goal refer to an expected change in direction in a thought process. Transitory thinking is similar, albeit with additional words either added or omitted. Paragrammatism, and parasyntax are the inappropriate use of

words and sentence structure. Rule identified numerous terms to define of unintelligible speech including, drivelling, muddling, verbigeration, word salad, speech confusion, incoherence, aculalia glossolalia, catologia, cataphasia, jargon aphasia, and schiozophasia. With regards to the idiosyncratic use of words, Rule made a distinction between the use of words or phrases used in an unusual context, and the invention of new words. Neologisms and word approximations include either the unusual or invented use of words. Metonymy relates to the use of imprecise expressions. Paraphasia, paraphemia, pseudoagrammatism, paraphrasia all relate to the substitution of one word for another with similar form. The last theme identified by Rule included symptoms which imply faults in mental processing. The different terms that are used that relate to this theme include condensation, fusion, displacement, substitution, omission, thought blocking, incoordination, asyndesis, interpenetration, overinclusion, and fragmentation.

1.2.2. Cognitive symptoms

The cognitive symptoms of schizophrenia include impairments in memory, attention, executive functioning, verbal fluency, and psychomotor performance, and are experienced by the majority of those diagnosed with the illness (Marder and Fenton, 2004). Historically, cognitive deficits have been recognised as a key feature of schizophrenia (Kraepelin, 1971), consistently identified both in the acute phase (McGhie and Chapman, 1961), and during periods of remission (Wohlberg and Kornetsky, 1973).

Cognitive deficits are typically present during the prodromal period (Lencz et al., 2006), and unlike positive symptoms, are thought to be a highly stable feature of the disorder (Nopoulos et al., 1994, Rund, 1998). Whilst there is some evidence to suggest that second generation antipsychotics (SGA) may be a more effective treatment for cognitive symptoms, relative to first generation antipsychotics (FGA), their response to treatment is typically poor (Harvey and Keefe, 2001). This this lack of treatment efficacy is seen as an important issue, given cognitive symptoms have been found to be strongly associated with functional impairment (Green et al., 2000, Green et al., 2004a).

In light of the minimal progress in developing treatments for cognitive symptoms, the National Institute for Mental Health (NIMH) initiated the Measurement And Treatment Research to Improve Cognition in Schizophrenia (MATRICS) collaboration. Under the auspices of the MATRICS initiative a multi-stage programme was initiated to improve the assessment of

cognitive symptoms, with the aim of providing a more robust framework by which new treatments could be evaluated. This process led to a systematic review to identify different dimensions of the cognitive symptom construct that are independent, reliable, and underpinned by the neuropsychological and cognitive neuroscience literature (Nuechterlein et al., 2004). In total, 7 different dimensions of cognitive impairments in schizophrenia were identified, including speed of processing, attention/vigilance, working memory, verbal learning and memory, visual learning and memory, reasoning and problem solving, and verbal comprehension. Whilst there was insufficient evidence at the time of the review, social cognition was added the list of constructs evaluated in the MCCB (MATRICS Consensus Cognitive Battery) (Kern et al., 2008, Nuechterlein et al., 2008), given its possible importance as a mediatory between neurocognitive deficits and functional outcomes (Brekke et al., 2005). Social cognition, whilst related to negative symptoms, appears to have a much stronger association with neurocognition and so was not considered part of the negative symptom construct (Kirkpatrick et al., 2006, Sergi et al., 2007). This construct is instead thought to contain 5 distinct domains, covering theory of mind, social perception, social knowledge, attributional bias, and emotional processing (Green et al., 2008).

1.2.3. Negative symptoms

Negative symptoms represent a pathological deficit in typical human experience and have been considered central to schizophrenia since the earliest conceptions of the disorder. Negative symptoms relate specifically to Bleuler's core symptoms of schizophrenia, known as the "4 A's" (loosening of associations, affective flattening, autism, and ambivalence) (Bleuler, 1950) and Kraepelin's "avolitional syndrome" (Kraepelin, 1971). At present, the negative symptoms of schizophrenia include alogia, asociality, anhedonia, avolition, and blunted affect (Kirkpatrick et al., 2006). Recent work suggests that these negative symptoms comprise of at least two distinct subdomains, with alogia and blunted affect representing expressive deficits; and asociality, anhedonia and amotivation representing experiential deficits (Blanchard and Cohen, 2006, Horan et al., 2011, Kring et al., 2013).

Negative symptoms have been found to be highly prevalent in clinical samples, with Bobes et al (Bobes et al., 2010) finding that 57.6% of patients treated for schizophrenia in routine outpatient care present with at least one negative symptom of moderate severity or worse. A summary of the five different types of negative symptoms are outlined below.

1.2.3.1. Alogia

Alogia relates to the reduction in the initiation and production of speech, and is distinct from irrelevant speech and incoherence which can either be attributable to thought disorder or cognitive deficits. In a recent meta-analysis significant deficits in speech production were found in patients with schizophrenia relative to healthy controls, particularly in regards to response latency (Cohen et al., 2014).

A number of different theories have been postulated as an explanation for poverty of speech in schizophrenia. Berenbaum and colleagues (Berenbaum et al., 2008) suggest that alogia is a result of impairments in generating ideas and planning disturbances (Kravariti et al., 2003, Morris et al., 1995), as opposed to impairments in working memory (Barch and Berenbaum, 1994), a general fluency disturbance (Stolar et al., 1994), or word finding difficulties (Alpert et al., 1994).

1.2.3.2. Blunted affect:

Blunted affect is the reduction in emotional expression, variation in vocal tone, and body movement. Different features identified include an unchanging expression, decreased movements, paucity of gestures, poor eye contact, affective non-responsivity, inappropriate affect, and lack of vocal inflection (Andreasen, 1983). This reduction in expressiveness is recognised to relate both to positive and negative emotion expression. In a meta-analysis comparing schizophrenia compared to healthy controls, schizophrenia patients were found to present with significant impairments in emotional expression (d=-1.11, 95% CI = -1.78 to -0.43) (Hoekert et al., 2007).

As summarised by Kring and colleagues (Kring et al., 1993) there are two main theories which explain the reduction of emotional expression in schizophrenia. Bleuler (Bleuler, 1950) proposed the inhibition hypothesis, which suggests that schizophrenia patients experience emotion at a level comparable to controls, but experience significant impairments in the ability to express emotions. Rado (Rado, 1953) suggested that low expressivity was a reflection a reduced ability to experience emotions, and positive emotions in particular. Whilst the terms are sometimes used interchangeably, the former is characterised as blunted affect, whilst the latter is known as emotional blunting. The current evidence base appears to support the
inhibition hypothesis, with a number of laboratory studies finding that patients report experiencing both positive and negative emotional experiences to the same degree as healthy controls (Kring et al., 1993, Berenbaum and Oltmanns, 1992, Cohen and Minor, 2010). In addition, there is evidence to suggest that these deficits may relate to impairments in the amplification of emotional expression of experienced emotions, as opposed to any expression of emotions experienced being suppressed (Henry et al., 2007).

Blunted affect in schizophrenia is associated with abnormalities in social behaviour, and are predicative of poor outcomes (Dworkin et al., 1998). Blunted effect has also been found to be associated in poorer premorbid adjustment, poorer quality of life, and worse outcome at 1-year follow up (Gur et al., 2006). However, blunted affect was not found to be associated with deficits in social skills (Salem and Kring, 1999) or social functioning (Sayers et al., 1996). This suggests that any impairments in outcomes experienced may either be attributable other factors, such as their association with other symptoms (notably amotivation; (Sayers et al., 1996), or the responses from people they are in contact with.

1.2.3.3. Asociality:

Asociality relates to reduced engagement with family members, partners, friends, and/or casual contacts (Blanchard et al., 2011). Asociality is a multifaceted concept in that it can relate both to a desire for relationships and or contact, and actual number of social activities conducted. Therefore, asociality can be evaluated both behaviourally (for example by counting up the amount and frequency of contacts over a specified period) or subjectively, by exploring the individuals interest in engaging with others. Given the fact that behavioural indicators of can be strongly influenced by extrinsic factors, Blanchard and colleagues (Blanchard et al., 2011) emphasise the importance of tapping into the subjective component of this symptom construct.

Understanding why patients with schizophrenia are more isolated appears to be somewhat complex, particularly given patients often report a desire to have friends, a vocation, and a family (Davidson and Stayner, 1997). Part of this is likely to be attributable to overlapping deficits in motivation, which explains why these symptoms are consistently reported to be part of the same overall construct in factor analytic studies (Blanchard and Cohen, 2006). In support of this hypothesis, Sayers and colleagues have found that amotivation is consistently positively associated with social dysfunction in patients with schizophrenia (Sayers et al., 1996).

However, increased asociality may also be attributable to other extrinsic factors, such as environmental under-stimulation and increased positive symptoms. Regarding the impact of positive symptoms on asociality, paranoid ideation has been found to be significantly associated with social anxiety (Rector, 2004), whilst formal thought disorder is linked to greater sensitivity to social rejection (Grant and Beck, 2009), both of which may result in patients being less engaged in their social environment.

Other possible antecedents of asociality include skill deficits and dysfunctional beliefs. There is evidence to suggest that patients with schizophrenia exhibit significant deficits in social skills which can contribute to social isolation (Bellack et al., 1990). Patients with schizophrenia often mistake other people's emotional states, which is associated with poorer social competence (Cramer et al., 1992, Mueser et al., 1996). People with schizophrenia have also been found to a report a greater number of asocial beliefs, which were found to be a strong predictor of asociality, both cross-sectionally and over time (Grant and Beck, 2010). In addition, Rector (Rector, 2004) found a significant association between negative symptoms and dysfunctional social performance beliefs.

1.2.3.4. Anhedonia:

Anhedonia is defined as the inability to experience pleasure from activities which are typically found to be enjoyable. In laboratory-based studies, schizophrenia patients appear to experience positive emotions as intensively as controls (Berenbaum and Oltmanns, 1992, Kring and Earnst, 1999, Kring and Neale, 1996). However, in both structured interviews and selfreport questionnaires patients report experiencing significantly lower levels of social and physical pleasure in comparison to healthy subjects (Blanchard et al., 1994, Blanchard et al., 1998, Herbener and Harrow, 2002). In a review by Cohen and colleagues (Cohen et al., 2011), five different explanations for this apparent disconnect between state and trait experiences of anhedonia was proposed; a specific affective deficit in anticipatory hedonic experience; an affective regulation deficit; an encoding-retrieval deficit; a representational deficit; and a social-specific hedonic deficit.

Of the five theories, a deficit in anticipatory hedonic experience is arguably the one that has received the greatest attention. Gard and colleagues (Gard et al., 2007) suggested that people with schizophrenia may experience deficits exclusively in anticipatory, as opposed to consummatory pleasure. These deficits in anticipatory pleasure extend both to the prediction

of experiencing pleasure in the future, and the experience of pleasure in the moment through the anticipation of future events. In support of this theory, Heerey and colleagues (Heerey et al., 2007) found that patients with schizophrenia are significant more likely to choose smaller immediate rewards over greater, delayed rewards. In addition, there is evidence to suggest that different neural pathways are involved for experiencing anticipatory and consummatory pleasure, supporting a distinction between these different types of experience (Berridge and Robinson, 1998, Berridge and Robinson, 2003). In neurobiology, serotonin and opioid systems are primarily linked with the experience of consummatory pleasure (Schultz, 2002, Wise, 2002), whilst anticipatory pleasure has been found to be linked the dopamine system (Berridge and Robinson, 1998). This is significant, given dopamine dysregulation has long been implicated in the development and maintenance of psychosis and schizophrenia (Howes and Kapur, 2009).

In the second possible explanation, an affective regulation deficit relates to the inability to minimise unpleasant experiences and enhance pleasurable ones, which are automatic strategies people unconsciously employ to enhance psychological wellbeing (Gross, 1998, Westen et al., 1997). Support for this hypothesis comes from the field of neuropsychology, where areas which are implicated in the affective regulation process such as prefrontal and anterior cingulate cortex (Kross et al., 2009) have been found to be substantially impaired in individuals with schizophrenia (Barch et al., 2003, Heckers et al., 2004). In the third explanation, it is proposed that dysfunctional encoding or retrieval of experiences may lead to inaccurate representations, a theory supported by evidence which suggests that people with schizophrenia can experience significant impairments in memory recall (Gold et al., 1992). In the fourth, a representational deficit relates to a deficit in the ability to assign an appropriate value to a stimuli (Gold et al., 2008). The final theory outlined in the review by Cohen and Colleagues (Cohen et al., 2011) relates to a social-specific deficit, where they suggest that hedonic impairments relate only to social experiences, whereas consummatory experiences remain largely intact. In their conclusion, Cohen and colleagues proposed that there is likely to be a significant degree of overlap between the theories, and highlighted the importance of considering the different systems together as a way to develop a fuller understanding of the mechanism of anhedonia in schizophrenia.

Studies into the longitudinal course of symptoms of schizophrenia suggest that anhedonia is a relatively common feature of the disorder, with 76% of patients displaying at least mild anhedonia, and 23% showing marked or severe hedonic deficits (Fenton and McGlashan, 1991). Historically, both Rado (Rado, 1956) and Meehl (Meehl, 1962) identified anhedonia as

the central feature of schizophrenia, and a reliable prognostic indicator for later developing the illness. In support of this, college graduates with high social anhedonia were found to be 24 times more likely to be diagnosed with a schizophrenia-spectrum disorder within 10 years compared to participants who report low social anhedonia (Kwapil, 1998). In addition, family members of those diagnosed with schizophrenia have been found to experience significantly greater levels of trait anhedonia when compared with controls (Franke et al., 1994, Grove et al., 1991, Katsanis et al., 1990).

1.2.3.4. Avolition:

Avolition is defined as the lack of motivation to perform activities and pursue goals. According to Ryan and Deci (Ryan and Deci, 2000), motivation concerns "energy, direction, persistence, and equifinality". Avolition can be characterised into two different components, relating either to a deficit in the individuals internal will to action, or the behaviours they display. In recent review determining how negative symptoms may be assessed, Blanchard and colleagues identified four different areas were avolition may be relevant to schizophrenia patients, including social activity, work/vocation, recreation, and self-care (Blanchard et al., 2011).

Deficits in motivation have been seen as a core feature of schizophrenia since the earliest days of its conception. Emil Kraeplin identified avolition as the key feature of patients' long term decline, whilst Bleuler identified avolition as one of one of the "4-A's". More recently, Foussias and Remmington (Foussias and Remington, 2010) proposed that avolition results in the clinical presentation of negative symptoms, and is the principle driver of functional decline.

That patients with schizophrenia exhibit persistent deficits in goal-seeking behaviours have been recognised in the ealiest models of the disorder (Bleuler, 1950, Kraepelin, 1971). A number of theories have been proposed as to why such deficits occur. Gard and colleagues (Gard et al., 2014) found that schizophrenia patients experience difficulties in perceiving reward outcomes, which results in participants engaging in less effortful goals which earn lower rewards. Related to this, Heerey and Gold (Heerey and Gold, 2007) found that subjects show a deficit ability to associate their behaviour to motivational properties of a stimulus.

1.3. Primary and secondary negative symptoms of schizophrenia

In 1974 Strauss and colleagues (Strauss et al., 1974) proposed that negative symptoms have different antecedents, suggesting that they may either constitute a fundamental feature of the disorder itself, or else may arise as a consequence of dealing with institutionalisation, social stigma, and isolation which can result from the diagnosis. Carpenter and colleagues (Carpenter et al., 1985) further expanded upon the importance of identifying the antecedents of negative symptoms in order to explain the heterogeneity in treatment response. In addition to the primary negative symptoms of the disorder which were considered core psychopathology, Carpenter et al. suggested that negative symptoms can arise as a consequence secondary factors, including positive symptoms, the side-effects of antipsychotic medication, a response to under-stimulating environments (with particular reference to long term institutionalisation), and depressive symptoms. Whilst secondary negative symptoms usually improve relatively quickly once these factors are addressed, primary negative symptoms were considered to be highly stable, have a poor prognosis, and be largely non-responsive to treatment.

Differentiating between primary and secondary symptoms may be important for a number of reasons. Firstly, a distinction between these symptoms may explain the heterogeneity in longitudinal course and treatment response which has been recognised in respect to negative symptoms. Secondly, identifying the aetiology of the negative symptoms can inform the appropriate response with regards to treatment. Third, differentiating between these symptoms may be important with regards to evaluating therapeutics for core psychopathology. Finally, isolating symptoms specific to the core pathology of the disorder may help to uncover possible causes and risk factors of schizophrenia.

In a study of consecutively admitted inpatients, the existence of persistent primary negative symptoms was detected in 25.7% of cases with a diagnosis of schizophrenia (Peralta and Cuesta, 2004), suggesting a significant proportion of patients may experience such symptoms. A summary of the different factors identified to cause secondary negative symptoms are summarised below.

1.3.1. Factors that induce secondary negative symptoms

1.3.1.1. Environmental under-stimulation

The link between under-stimulating institutional care and negative symptoms such as decreased spontaneity, reduced drive, and blunted affect have been recognised in concepts such as "institutional neurosis" (Barton, 1959) and "social breakdown syndrome" (Gruenberg et al., 1966). In a study comparing different mental hospitals, Wing and Brown (Wing and Brown, 1970) found a strong association between the severity of the social environment with what they defined as "clinical poverty", which included symptoms such as blunted affect, poverty of speech, and social withdrawal. In addition, they found that these symptoms reduced relatively quickly once their environment improved. In a replication of the study completed 30 years later (Curson et al., 1992), no association between length of stay and negative symptoms was detected, and only a weak association between environmental poverty and negative symptoms was found. Curson and colleagues attributed this weaker relationship to the improved hospital environment, supporting the conclusion that it is the environmental impoverishment, as opposed to incarceration itself, which exacerbates negative symptoms. In a cross-cultural validation of the original Wing and Brown study (Oshima et al., 2003), the link between environment and negative symptoms was evaluated in 139 Japanese hospitals, where deinstitutionalisation and rehabilitative practices had not been adopted at the time (Mino et al., 1990). Whilst no association was detected between length of stay and negative symptoms, a significant association between negative symptoms and an understimulating environment was found.

Overall, these findings support the link between environmental under-stimulation and negative symptoms, and the importance of providing stimulating environments in order to minimise secondary negative symptoms. However, given only a very low association was detected in the study by Curson (Curson et al., 1992) it suggests that the negative symptoms induced in hospitals with good social and rehabilitative provision may be lower than previously thought. Regardless, given the potential impact of hospitalisation, and the fact that it is beyond the scope of this investigation to evaluate the environmental poverty of each institution, these findings highlight another issue in disentangling primary and secondary negative symptoms in inpatient settings. Consequently, this investigation focuses exclusively on outpatients, apart from the analysis conducted in chapter 7 which considers the impact of symptoms specifically in the acute phase of treatment.

1.3.1.2. Antipsychotic medication side effects:

Since the discovery of antipsychotic medication in the 1950's it has been recognised that these compounds can cause extra-pyramidal side effects (EPS) (Rifkin, 1987). Common extrapyramidal side effects include dystonia, parkinsonism, tardive dyskinesia, and akathisia. Other movement-disorders related to antipsychotic drugs include catatonia, and in very rare cases, neuroleptic malignant syndrome (Caroff, 1980). Akathisia is characterised by restlessness, anxiety, inner tension and the need to be in constant motion. Akathisia is thought to be relatively common, with prevalence estimated at between 20-35% of patients in routine practice (Kane et al., 2009). Dystonia is an acute involuntary movement disorder, characterised by prolonged muscle contractions causing twisting movements and abnormal postures (Berardelli et al., 1998). Drug-induced parkinsonism mimics Parkinson's disease, and can result in tremors, postural or gait disturbances, hypersalivation, rigidity, and bradykinesia (slowness in the execution of movement), and akinesia (loss of movement power). Tardive dyskinesia is characterised by repetitive hyperkinetic movements, which, unlike most EPS symptoms, can be irreversible (Caroff et al., 2011). Catatonic symptoms associated antipsychotic side-effects include stupor, and states where patients find themselves unable to move and/or speak (akinetic mutism). In rarer cases, patients can experience catalepsy (which is a state whereby patients are outwardly unresponsive to external stimuli) and waxy flexibility (which is a rigid maintenance of body position) (Gelenberg and Mandel, 1977). EPS are positively associated with medication non-compliance (Fenton et al., 1997), which in turn is associated with a higher risk of relapse (Gilbert et al., 1995). In a review by Fenton and colleagues (Fenton et al., 1997) it was found that between 25%-66% of patients cite medication side effects as the primary reason for treatment non-compliance in schizophrenia

When second generation antipsychotics (SGA's) first became available, it was believed that a lower risk of EPS was one of their major benefits over the older first generation antipsychotics (FGA's) (Kane et al., 1988, Safferman et al., 1991). However, in a meta-analysis comparing the side effects of typical and atypical antipsychotics (Leucht et al., 2009b) much of the difference previously reported was attributable to high potency FGA's being used in the control arm conditions. When different SGA's were compared to lower potency FGA's (defined as equipotent or less potent than chlorpromazine) the difference in EPS was less marked, and less consistent. In another study comparing EPS levels between atypical antipsychotics and a lower potency FGA (perphenazine), no differences were detected (Lieberman et al., 2005). These findings suggest that issues related to EPS have not been eliminated by the advent of second

generation antipsychotics, and remain an important consideration in the treatment and research of schizophrenia.

In addition to causing significant levels of distress to patients, EPS can mimic negative symptoms which can make assessing core psychopathology difficult (Carpenter et al., 1985). In animal studies, antipsychotic drugs have been found to be associated to behavioural changes comparable to negative symptoms such as apathy, and lack of spontaneity and affective arousal (McKinney et al., 1980, McKinney and Moran, 1981). In addition, both the dose and duration of antipsychotic medication received was found to be positively associated with negative symptoms (Perenyi et al., 1998). When individual negative symptoms were considered separately, Kelley found that medication effects appeared to be related primarily to expressive deficits such as blunted affect (Kelley et al., 1999).

Regarding particular types of EPS, significant overlaps between parkinsonian symptoms such as bradykinesia and akinesia and negative symptoms have been identified (Allan et al., 1998). In a study by Kane (Kane et al., 1994), a significant correlation between akinesia and anergia was detected, which persists even after a washout period of at least two weeks prior to the assessment taking place. Similarly, Prosser reported that negative symptoms have been found to be significantly correlated with the parkinsonian symptoms akinesia and tremor (Prosser et al., 1987). Regarding individual negative symptoms, Allen suggested that the link between facial bradykinesia and blunted affect, and akinesia to avolition may be a considered a particular issue (Allan et al., 1998).

In clinical practice, anticholinergic drugs have typically been prescribed to control or limit the impact of EPS (Ogino et al., 2014). In research, EPS can be measured using validated scales such as the Simpson-Angus Scale (SAS) (Simpson and Angus, 1970) or Barnes Akathisia Rating Scale (BARS) (Barnes, 1989) to either to specify maximum EPS inclusion criteria, or else be included as covariates in the model. Given the content overlap between negative symptoms and akinetic parkinsonian symptoms such as bradykinesia and akinesia, Peralta and Cuesta suggest that focusing on the non-akinetic symptoms such as tremor and hypersalivation may be a more useful in distinguishing primary from drug-induced negative symptoms (Peralta and Cuesta, 1999c).

1.3.1.3 Depression

Depression is a state of low mood that can lead to changes in feelings, thoughts and behaviours. Symptoms include feelings of hopelessness, apathy, lethargy, changes in sleep patterns, guilt, suicidal thoughts and somatic complaints. Depression is considered to be relatively common in patients with schizophrenia. In a review by Siris (Siris, 1991), the prevalence of depression and prominent depressive symptoms ranged from 7%-70%, depending upon the criteria adopted and the sample evaluated. At the point of hospital admission for a first episode, 71% of participants presented with clinically meaningful depressive symptoms, and 23% fulfilled criteria for a diagnosed depressive episode (Häfner et al., 2005). Prior to onset, the lifetime prevalence of depressive mood for a period longer than 2 weeks was found to be particularly high (83%; (Häfner et al., 2005). Overall, depression and depressive symptomology is considered to be highly prevalent in patients with schizophrenia.

Whilst studies have consistently shown that negative and depressive symptoms are distinct constructs (Blanchard and Cohen, 2006), there is a significant degree of overlap with regards to their clinical presentation (Mulholland and Cooper, 2000). For example, both negative symptoms and depression can manifest as a lack of energy and drive, anhedonia, and social withdrawal. However, a number of differences are also apparent. As recognised in in section 1.2.3.4., anhedonic deficits in schizophrenia appear to be confined specifically to anticipatory pleasure (Gard et al., 2007), whilst in depression anhedonia relates to experiencing pleasure overall (Snaith, 1993). In addition, depressive symptoms appear to be more associated to positive, rather than negative symptoms (Lindenmayer et al., 1991). Other differences include the fact that patients with severe negative symptoms often report relatively high subjective quality of life and satisfaction scores (Fitzgerald et al., 2001, Priebe, 2007). This is in contrast with depression, where the negative association between depressive symptoms and subjective quality of life is well documented (Priebe et al., 2000, Reine et al., 2003).

Whilst a number of scales have been developed to measure depressive symptoms such as the Beck depression inventory (BDI) (Beck et al., 1988) and the Hamilton Depression Rating Scale (HRDS) (Hamilton, 1960), disentangling symptoms using these scales is considered problematic. Addington and colleagues (Addington et al., 1996) found significant correlations between the HRDS and assessments of negative symptoms in patients with schizophrenia. In a different study, the HRDS was found to contain a negative symptom factor which correlates strongly with the SANS (Goldman et al., 1992). In addition, there is evidence to suggest that global rating scales such as the Positive and Negative Syndrome Scale (PANSS) (Kay et al.,

1987) are not to be able to distinguish between depression, negative symptoms, and EPS (Collins et al., 1996).

In an attempt to disentangle ratings of depression and negative symptoms, the Calgary Depression Scale for Schizophrenia Scale (CDSS) (Addington et al., 1993, Addington et al., 1992) was designed specifically for use in psychotic populations. In a study examining the divergent validity of the CDSS from negative symptomology (Addington et al., 1996), no significant associations were found both at hospital admission (r^2 =.02, P=.67), or at three months follow-up (r^2 =.06, P=.14). These findings are in contrast to an assessment of the relationship between the HDRS and negative symptoms, where significant associations were detected at both time points (r^2 =.28, P<.01, and r^2 =.23 P<.01 respectively). Overall, these findings highlight the importance and complexity of disentangling depressive and negative symptoms, and the importance of choosing appropriate methods to control for depressive symptoms when controlling for secondary negative symptoms.

1.3.1.4. Positive symptoms:

The positive symptoms of schizophrenia include hallucinations, delusions and thought disorder (for a detailed summary see 1.2.1.). In reformulations of the positive/negative symptoms dichotomy (Crow, 1980, Crow, 1985, Strauss et al., 1974) positive and negative symptoms were understood to be independent, with some studies suggesting they may even be inversely associated once general psychopathology is controlled for (Kay et al., 1987). However, later work has found positive correlations between the two symptom dimensions, both in the acute phase and during periods of remission (Addington and Addington, 1991).

In an evaluation of the positive and negative "subtypes" of schizophrenia, Angrist and colleagues (Angrist et al., 1980) examined the effects of prescribing antipsychotics and amphetamines to patients with schizophrenia. As hypothesised, amphetamines increased the severity of positive symptoms but, contrary to expectations, negative symptoms also increased. When this was examined at an item level, the increase was found to be exclusively attributable to elevated emotional withdrawal. Angrist postulated that the increase in emotional withdrawal may be attributable to autistic preoccupations with increased persecutory delusions or auditory hallucinations, as opposed to any change in core psychopathology. In a more recent study, (Kelley et al., 1999) found that increases in positive symptoms predicted more severe avolition but not blunted affect, supporting the theory that

positive symptoms appear to influence the experiential features of negative symptoms in particular.

Since the study by Angrist, a number of other researchers have reported on the improvement in negative symptoms being linked to improvements in positive symptoms (Tandon et al., 1990, Tandon et al., 1993, van Kammen et al., 1987). Whilst no such association was detected in studies by Brier or Serban (Breier et al., 1987, Serban et al., 1992), Tandon suggested this may be attributable to small sample sizes and insufficient positive symptoms present at baseline to drive changes in negative symptoms (Tandon et al., 1993). In an attempt to model for the impact of positive symptoms on negative symptoms in clinical trials, a number of pathanalytical studies have been completed (Alvarez et al., 2006, Moller et al., 1995, Tollefson and Sanger, 1997). Whilst direct effects of atypical antipsychotics on negative symptoms were detected in all studies, significant improvements in these symptoms were also found to be attributable to improvements in positive symptoms. For example, in the study by Alvarez et al. 42.69% of the improvements detected in negative symptoms in patients prescribed with olanzapine over risperidone was found to be attributable in the improvement in positive symptoms (Alvarez et al., 2006). Such findings highlight the importance of either limiting, or controlling the severity of positive symptoms when attempting to evaluate core negative psychopathology.

1.3.3. Deficit syndrome and persistent negative symptoms

Following the distinction between primary and secondary negative symptoms, Carpenter and colleagues went on to suggest that those who experience primary negative symptoms may constitute a distinct subtype of schizophrenia, known as the "deficit syndrome" (Carpenter et al., 1988). As recognised in a review by Buchanan (Buchanan, 2007) the concept is based upon Kraepelin's "avolitional syndrome" (Kraepelin, 1971). In a departure from the subtype definition proposed by Crow (Crow, 1980, Crow, 1985), a diagnosis of deficit syndrome is determined by the persistence of core psychopathology, as opposed to the presence of symptoms measured at a single time point. The operational criteria for the Deficit syndrome proposed by Carpenter et al., 1988) include:

 At least 2 of the following 6 symptoms present at a clinically significant level: flattened affect, diminished emotional range, alogia, impaired interest, diminished sense of purpose, asociality.

- 2+ symptoms present for minimum of 12 months, present during periods of clinical stability.
- 3) The above symptoms must not be a consequence of secondary factors
- 4) A diagnosis of schizophrenia

The distinction between deficit and non-deficit forms of schizophrenia has been found to be highly stable over time (Amador et al., 1999), with the diagnosis of deficit syndrome consistent in 83% of cases, and a non-diagnosis consistent in 88% of cases 3-4 years later. Over a longer period, Strauss and colleagues (Strauss et al., 2010), found that approximately 70% of deficit syndrome patients at baseline still qualified for the diagnosis 20 years later. The prevalence of the deficit syndrome has been estimated to be 20-30% of schizophrenia in clinical samples, and 14-17% in schizophrenia population samples (Kirkpatrick et al., 2006). The prevalence for the disorder appears to be significantly higher in older patient samples, with 37% of participants found to qualify for deficit syndrome in a sample of patients over 45 years old (Harris et al., 1991).

In a review of deficit syndrome by Kirkpatrick, it was proposed that differences in the signs and symptoms, longitudinal course, risk and antecedent factors, treatment response, and pathophysiological correlates between deficit and non-deficit subtypes all support the theory that they constitute distinct disorders (Kirkpatrick et al., 2001, Kirkpatrick and Galderisi, 2008). Patients diagnosed with deficit syndrome have been found to present with lower positive psychotic symptoms and less severe depression relative to non-deficit patients, in addition to lower rates of less drug abuse, suicidality, anxiety, guilt and hostility (Kirkpatrick et al., 1996a, Kirkpatrick et al., 1994). As identified in section 1.1.3., winter birth is associated with nondeficit schizophrenia, whilst deficit syndrome has a stronger association with summer birth (Messias et al., 2004). Deficit syndrome patients have been found to have a significantly lower age of onset relative to non-deficit patients (Kirkpatrick et al., 2000a). With regards to the course of the illness, the deficit syndrome has been found to be associated with poorer premorbid adjustment (Galderisi et al., 2002). Tek and colleagues found that patients diagnosed with deficit syndrome had a poorer quality of life, experienced a greater impairment in functioning, and had more severe negative symptoms 5 years later (Tek et al., 2001). In studies that compare the symptoms of family members of people with deficit and non-deficit schizophrenia, sub-clinical negative symptoms are more severe in the families of

people with deficit syndrome (Hong et al., 2003), whilst symptoms of dysphoria and positive symptoms are higher in family members of non-deficit patients (Kirkpatrick et al., 2000b).

Regarding any neurological differences between deficit and non-deficit syndrome there is evidence to suggest that non-deficit patients experience a greater reduction in frontal lobe white matter relative to both deficit patients and healthy controls (Buchanan et al., 1993), whilst there is also some evidence of electrophysiological differences between deficit and nondeficit patients (Mucci et al., 2007, Turetsky et al., 1998). In addition, there is evidence to suggest that eye-tracking dysfunction appears to be significantly more impaired in deficit patients, relative to non-deficit patients (Ross et al., 1997, Ross et al., 1996). Overall, there is a substantial (and still emerging) evidence base to suggest that deficit syndrome may represent a distinct subtype of schizophrenia which emphases the importance of considering this distinction in any investigation of negative symptoms.

1.3.4. Persistent negative symptoms as an alternative to the deficit syndrome

Whilst assessment tools such as the Schedule for the Deficit Syndrome (SDS) have been found to be a reliable tool for identifying deficit patients (Kirkpatrick et al., 1989), Buchanan suggested that making such a distinction may be problematic in the context of a clinical trial (Buchanan, 2007). Firstly, accurate information regarding the presentation of symptoms is required over a period of 12 months, which is likely to be difficult to obtain in the majority of cases. Secondly, determining the aetiology of negative symptoms (i.e. whether they are primary in origin or attributable to secondary factors) requires a high degree of clinical expertise that assessors in clinical trials may not typically have. As an alternative, Buchanan proposed that clinical trials designed to evaluate treatments for negative symptoms should instead include those that experience what he defined as "persistent negative symptoms".

The criteria that Buchanan proposed to define what constitutes persistent negative symptoms differ from the criteria for deficit syndrome in two important respects. Firstly, the requirement that the negative symptoms considered must be present for 12 months is omitted. No predefined time period is specified, although 6 months is generally recommended. Secondly, patients with secondary negative symptoms may still qualify for eligibility so long as the negative symptoms experienced are persistent, and have failed to respond to usual treatments for causes of secondary symptoms which are adequately controlled for. To operationalise the

concept of persistent negative symptoms, Buchanan (Buchanan, 2007) proposed the following criteria:

- 1) At least moderate severity of negative symptoms, measured on a validated symptom rating scale
- Demonstrated clinical stability for an extended period of time prior to the start of study
- A defined maximum threshold of positive symptoms as measured on a validated symptom rating scale
- 4) No (or low level of) depressive symptoms, as measured on a validated rating scale
- 5) No (or low level of) extrapyramidal symptoms, as measured on a validated rating scale

(Buchanan, 2007)p.1016)

Adopting the criteria of persistent negative symptoms over deficit syndrome criteria has a number of advantages. By not having to disentangle the aetiology of the negative symptoms adopting persistent negative symptom criteria should be easier to implement, relative to deficit syndrome criteria. In addition, utilising the persistent negative symptom criteria should mean that a substantially larger pool of potential participants would be available, easing the challenge to recruit eligible participants. Related to this, with the eligible population being much closer to the clinical population, any subsequent findings could be considered to have greater ecological validity. As recognised in the NIMH-MATRICS consensus statement on negative symptoms (Kirkpatrick et al., 2006) symptoms that are highly persistent, irrespective of the aetiology, represent an unmet therapeutic need and so merit inclusion in clinical trials designed to treatment these symptoms.

Whilst it may appear attractive to adopt the criteria of persistent negative symptoms in clinical trials, further work on how these criteria should be implemented is required. At present the criteria are broad, and it is not clear how the different ways in which they can be implemented may influence the association between negative symptoms and causes of secondary negative symptoms such as depression, positive symptoms, and EPS. In addition, recent work has suggested that the manner in which these criteria are adopted can have a substantial impact on the size of the potential sample pool (Dunayevich et al., 2014, Rabinowitz et al., 2013),

which can impact both the validity of the findings and the ease of recruitment. This area will be explored in greater depth in chapter 4.

1.5. Longitudinal course of negative symptoms of schizophrenia

A number of different conceptual models of schizophrenia have suggested that negative symptoms are either highly stable, or get progressively more severe over time. Kraepelin (Kraepelin, 1971) defined schizophrenia as a degenerative disorder, with all symptoms getting more severe over time. In contrast, Bleuler (Bleuler, 1950) adopted a much broader conception of schizophrenia arguing that outcome can be much more heterogeneous, albeit with negative symptoms being more stable than the "accessory" positive symptoms. More recently, Crow (Crow, 1980, Crow, 1985) thought negative symptoms were permanent, believing they were a result of morphological changes in the brain.

In longitudinal assessments of symptoms over time, Pfhol and Winokur (Pfohl and Winokur, 1983) catalogued the presence and absence of symptoms over 25 years in 52 chronic institutionalised hebephrenic and catatonic schizophrenia patients. In this study they found that the prevalence of avolition, impaired social interaction and flat affect increased over time, whilst positive symptoms reduced. In a series of investigations as part of the Chestnut Lodge study, Fenton and McGlashan found that that negative symptoms "accrue severity, stability and prognostic weight over time" (Fenton and McGlashan, 1991, McGlashan and Fenton, 1992, McGlashan and Fenton, 1993). However, there is also a significant body of literature which suggests that negative symptoms are largely stable, with a minority of patients exhibiting small improvements over time. Pogue-Geile & Harrow (Pogue-Geile and Harrow, 1985) found that the severity of negative symptoms were largely stable over a period of 5 years, with a tendency towards remission. In a study spanning 4 years, Johnstone and colleagues (Johnstone et al., 1987) found that whilst negative are much more stable than positive symptoms, and that the deficits were not irreversible with all of features of negative symptoms improving in at least some cases. Over a period of 10 years, Eaton (Eaton et al., 1995) found that the prevalence of both positive and negative symptoms reduced in the year following first hospitalisation, and were then stable hereafter. In a sample of neuroleptic-naïve patients at admission, Arndt (Arndt et al., 1995) found that negative symptoms were prominent at the baseline stage, and remained largely stable over a period of 2 years. In an evaluation of community outpatients over a period of 2 years, Quinlan (Quinlan et al., 1995) found that negative symptoms significantly reduced, whilst positive symptoms increased.

More recently, Whitty (Whitty et al., 2008) reported significant improvements in both positive and negative symptoms over a period of 4 years from first presentation, suggesting that the outcome may not be as pessimistic as once thought.

In addition to examining the longitudinal course of negative symptoms as a singular construct, a number of studies have also examined the course of individual negative symptoms (Dollfus and Petit, 1995b, Eaton et al., 1995, Fenton and McGlashan, 1991, Pogue-Geile and Harrow, 1985). There is some evidence to suggest that alogia and blunted affect are more stable than other negative symptoms (Dollfus and Petit, 1995b, Johnstone et al., 1987, Kelley et al., 2008), which may be due to these symptoms being less influenced by improvements in positive symptoms (Kelley et al., 1999).

In the current investigation, the longitudinal course of negative symptoms will be examined by way of meta-analysis in an attempt to uncover broader trends regarding how these symptoms may change over time in chapter 3.

1.5.1. Longitudinal course of negative symptoms in patients with deficit syndrome

Whilst it is important to determine the longitudinal course of negative symptoms that patients typically experience irrespective of their aetiology, disentangling their origin may be helpful as a way to explain the heterogeneous findings highlighted in section 1.5. As stated previously, Carpenter (Carpenter et al., 1985) proposed that while secondary negative symptoms may fluctuate in accordance with the extrinsic factors which cause the symptoms change, primary negative symptoms should be considered a highly stable feature of the disorder. In order to evaluate the severity and stability of primary negative symptoms over time, a number of longitudinal studies have been completed in patients that qualify for the deficit syndrome.

In one of the earliest evaluations of negative symptoms in deficit and non-deficit schizophrenia patients, Fenton and McGlashan (Fenton and McGlashan, 1994) found that negative and cognitive symptoms get significantly worse in patients that qualify for deficit syndrome over a period of 5 years. In addition, periods of symptomatic remission were found to be a lot less frequent in patients in these patients, with 78% showing a continuing course of illness, relative to 28% of non-deficit patients. These findings were similar to those reported in a later study by Kirkpatrick and colleagues (Kirkpatrick et al., 1996b) who found that deficit patients presented with more severe negative symptoms two years later, in addition to reporting poorer psychosocial functioning and greater global impairment.

In a prospective study evaluating the longitudinal course of negative symptoms over a much longer period (20 years) (Strauss et al., 2010), the results were broadly consistent with earlier studies. Deficit syndrome patients were found to experience significantly fewer periods of global recovery, and a greater degree of social and occupational impairment. However, in contrast to previous studies negative symptoms were not found to get significantly worse over time. Instead, negative symptoms were found to follow a fluctuating course in a similar manner to non-deficit patients.

1.6. The impact of negative symptoms on quality of life

Considerable attention has been paid to the impact of negative symptoms of schizophrenia on quality of life, despite the fact that there is no precise definition of what constitutes quality of life in this context (Priebe et al., 2010b). For the purposes of this investigation, quality of life is defined to include subjective components such as wellbeing and satisfaction with life, and objective components including daily life functioning and external resources, both material and social (Katschnig, 2000, Lehman et al., 1982). Consequently, a number of studies which do not specifically mention quality of life, but do evaluate concepts such as functioning, wellbeing, and satisfaction have all been deemed as relevant to this area of interest.

With regards to the objective components of quality of life (OQOL), negative symptoms have been found to be significantly associated with less frequent social contacts, lower quality of social interaction, and poorer community participation (Lysaker and Davis, 2004). Hunter and Barry (Hunter and Barry, 2012) found a strong association between negative symptoms and impairments in various measures of functioning, including the quality of interpersonal relations, vocational role, intrapsychic foundations, and personal and social performance. In longitudinal studies, higher negative symptoms at baseline have been found to be associated with more substantial impairments in OQOL 2 to 7 years later in areas such as global functioning, psychosocial functioning, impairment in recreational activities and relationships, and work performance (Ho et al., 1998, Milev et al., 2005, Pogue-Geile and Harrow, 1985, Whitty et al., 2008). Such quality of life deficits have been found to be particularly severe in patients that qualify for deficit syndrome (Fenton and McGlashan, 1994, Tek et al., 2001). In an evaluation of individual symptoms, anhedonia in particular has been found to be a predictor of various features of lower quality of life, including psychological, social, and spiritual domains (Ritsner et al., 2011), whilst apathy has been found to be associated with poorer individual living skills (Kiang et al., 2003).

Whilst a strong association between various domains of OQOL and negative symptoms have been consistently detected, the association between negative symptoms and subjective quality of life (SQOL) is less clear. Somewhat counter-intuitively, patients with negative symptoms tend to report relatively high SQOL, despite experiencing a poor OQOL (Arns and Linney, 1993, Katschnig et al., 2006). As a result, the association between objective and subjective quality of life have been found to be only weak-to-moderate in schizophrenia, ranging from 0.04 to 0.57 (Priebe and Fakhoury, 2007). Fitzgerald and colleagues (Fitzgerald et al., 2001) found that whilst various observer-rated quality of life domains were found to be significantly associated to negative symptoms, subjective life satisfaction was not. In a metaanalysis examining the cross-sectional association between symptoms of schizophrenia and SQOL, only a weak association was detected with negative symptoms (Eack and Newhill, 2007). In an assessment of the association between the change in subjective quality of life and anergia symptoms over time only a weak association was detected, and this was not significant after controlling for other symptoms, notably depression (Priebe et al., 2011b).

Subjective quality of life can cover a number of concepts, such as ones satisfaction with their employment situation, finances, recreational activities, friendships, safety, housing, health, sex-life, family and overall life satisfaction (Priebe et al., 1999). Determining ones subjective appraisal of their life is thought to be influenced by three different factors, including a comparison between expectations and aspirations, a comparison with others, and adaptation over time (Priebe, 2007). As a result, Priebe suggested that patients may report relatively high levels of SQOL due to the fact that schizophrenia is a highly chronic illness, meaning patients may have had a long time to adapt to their present situation, in addition to peer comparison. However, an alternative explanation is that many scales which have been designed to measure negative symptoms focus primarily on behavioural referents (Blanchard et al., 2011), which may result in the association between SQOL and symptoms being under-reported. Regardless of the explanation, subjective quality of life reports are an important indicator of patients' views and experiences which cannot be disqualified by any external measure, and so should be respected (Priebe, 2007). This relationship is further explored in chapter 6.

1.7. The treatment of negative symptoms

Whilst negative symptoms have been found to improve as a result of antipsychotic treatment (Goldberg, 1985, Kane et al., 1988, Marder and Meibach, 1994), this is considered primarily to be a consequence of improvement in the causes of secondary negative symptoms, such as

positive symptoms (Buchanan, 2007, Tandon et al., 1993). To date, any advances in the treatment of schizophrenia have been found to provide only limited benefits to negative symptoms directly. In a meta-analysis by Leucht which examined the efficacy of individual second generation antipsychotics, most were found not to provide a significant improvement relative to first generation drugs, and in those that did the effect sizes found were small (Leucht et al., 2009a). Meta-analyses into the efficacy of adjuntive medications such as alpha-2 receptor antagonists (Hecht and Landy, 2012) and glutamatergic compounds (Tuominen et al., 2005) show some promise, although since these publications a large-scale trial found no impact of either glycine or D-cycloserine (Buchanan et al., 2007). There is some evidence to suggest that adjunctive antidepressant medication may have some limited benefit (Singh et al., 2010). In a broader review evaluating the different pharmacological approaches in treating negative symptoms (Arango et al., 2013), promising new drugs that act on the NMDA and alpha-7 nicotinic receptors are highlighted, but further work is required.

In evaluations of psychotherapeutic approaches, a meta-analysis of CBT reported a small effect on negative symptoms (Jauhar et al., 2014), a meta-analysis on social skills training reported a moderate effect size (Kurtz and Mueser, 2008), whilst a meta-analysis on social cognitive training found no effect (Kurtz and Richardson, 2012). In a meta-analysis of cognitive remediation therapy which evaluated overall symptoms, only a small effect was detected (Wykes et al., 2011). In the UK, NICE have recommended arts therapies as a treatment which has shown consistent efficacy in reduction of negative symptoms (NCCfMH, 2010), however this has since been challenged by the non-significant result of the MATISSE trial (Crawford et al., 2012b). In a broader meta-analysis looking at the effectiveness of a number of different treatment for negative symptoms, some treatments were found to result in statistically significant improvements, but none were considered to result in clinically meaningful improvements (Fusar-Poli et al., 2015). Overall, this current lack of treatment efficacy has led to negative symptoms being recognised as an unmet therapeutic need, and an important target for new interventions (Kirkpatrick et al., 2006).

1.8. The NIMH-MATRICs consensus statement on negative symptoms of schizophrenia

Given the devastating impact of negative symptoms, combined with the limited progress in treatment development, in 2005 a consensus development conference was held on behalf of the National Institute of Mental Health (NIMH). The aim of this meeting was to further develop

the measures and methodologies adopted in negative symptom clinical trials (Kirkpatrick et al., 2006). The meeting followed the success of the MATRICs conference on cognitive symptoms of schizophrenia (Green et al., 2004b), which helped form the basis for standardising the cognitive assessment battery.

The consensus statement concluded that negative and cognitive symptoms are separate domains, with negative symptoms representing an unmet therapeutic need in a large proportion of cases. There was an agreement that blunted affect, alogia, asociality, anhedonia, and avolition should all be considered part of the negative symptom construct. In addition, they proposed that this construct may comprise of at least two distinct factors; expressive deficits, which include alogia and blunted affect, and experiential deficits, which include asociality, anhedonia and avolition (Blanchard and Cohen, 2006, Horan et al., 2011). Despite recognising that patients who experience primary and secondary negative symptoms can be differentiated with relatively good reliability (Kirkpatrick et al., 2001), it was suggested that making a distinction between these types of symptoms was not essential, so long test subjects presented with persistent negative symptoms with sources of secondary negative symptoms controlled for through study design. However, no specific details of what constitutes an appropriate strategy to limit the impact of positive, depressive or medication side effects were presented, an issue which will be further addressed in chapter 4.

In research trials which evaluate the effectiveness of medications taken specifically for negative symptoms, it was agreed that the ideal design would be to conduct a double-blind, placebo-controlled comparison of parallel groups, including clinically stable patients maintained on second-generation antipsychotics. In the case of broad spectrum antipsychotics there was a consensus that it was not possible to establish superior efficacy for negative symptoms due to the "pseudospecificity problem", which is the inability to disentangle improvements in negative symptoms from improvements in sources of secondary negative symptoms (see section 1.3.1.). There was also a consensus that whilst proof of concept studies may be relatively brief (4-12 weeks), full-scale trials should be much longer (6-months) in order to evaluate the persistent efficacy of treatments. The final point of agreement was that whilst the Scale for the Assessment of Negative Symptoms (SANS) (Andreasen, 1983) was preferable to single symptom-item scales such as the Positive and Negative Symptom Scale (Kay et al., 1987), the information obtained from single-item scales should still be considered important and valid.

In addition to the points of agreement, the workshop also outlined recommendations for future research. Of these, the key recommendation was the need to develop of new instruments designed to measure negative symptoms which address the limitation of existing scales. This recommendation resulted in the development of the Collaboration to Advance Negative Symptom Assessment for Schizophrenia (CANSAS) (Blanchard et al., 2011), which in turn led to the develop of two new scales; the Brief Negative Symptom Scale (BNSS) (Kirkpatrick et al., 2011), and the more comprehensive Clinical Assessment Interview for Negative Symptoms (CAINS) (Forbes et al., 2010, Horan et al., 2011, Kring et al., 2013). Whilst the CAINS has since been found to have excellent psychometric properties (Kring et al., 2013), the report highlights the need to evaluate the sensitivity of the scale, which is explored in chapter 5 of this investigation.

Other areas for research identified included the need for more research into the prevalence and longitudinal course of negative symptoms (which will be examined in chapter 3 of this investigation), and determining what improvement in negative symptoms constitutes a clinically meaningful effect size improvement. Whilst it has been suggested that a reduction of 10 to 15 points on the PANSS corresponds to a Clinical Global Impressions (CGI) change of "minimally improved" (Leucht et al., 2006), this figure is dependent upon the baseline severity of symptoms, and more importantly only corresponds to the PANSS total score, as opposed to the PANSS negative subscale alone. In recent work regarding the SANS, a reduction of symptoms by 13-21% was found to correspond to "minimally improved" on the CGI, 42-50% reduction to "much improved" and 67-90% to "much improved" (Levine and Leucht, 2013).

1.8.1. The ISCTM and NEWMEDS update on clinical trials in negative symptoms

Following the original NIMH-MATRIC meeting for negative symptoms, two further workshops were held, one held by the International Society for Central Nervous System Clinical Trials and Methodology, (ISCTM) (Marder et al., 2011), and other held by the Novel methods leading to New MEdications in Depression and Schizophrenia collaboration (NEWMEDS) (Marder et al., 2013). These updates were deemed necessary for a number of reasons. Firstly, concerns were raised that the original 2006 consensus statement (Kirkpatrick et al., 2006) did not include sufficient representation from the pharmaceutical industry, which is important given their involvement in producing and evaluating new pharmacological treatments for negative symptoms. Secondly, there were concerns at the time that the Food and Drug Administration (FDA) were considering amending their guidelines, requiring clinical trials of treatments for

negative symptoms to include a co-primary outcome measure evaluating functional outcomes. Thirdly, it was felt that further consensus regarding trial design would be helpful, particularly given the developments in negative symptoms and lessons learnt from recent clinical trials. The meeting included individuals from academia, pharmacological industry, and from licensing bodies covering both the USA (FDA) and Europe (EMA; the European Medicines Agency).

During the ISCTN meeting, one the main conclusions that was drawn was the recognition that given severe negative symptoms include functional impairment, improvements in negative symptoms alone should be deemed sufficient for drug approval. This conclusion therefore supports research which focuses on negative symptoms specifically as an outcome for research.

In the ISCTN consensus statement, it was agreed that clinical trials need to be no less than 2 months, and ideally at least 6 months in duration (not including pre-randomisation). Clinically meaningful change was defined as a d=.05, which Cohen described as an effect "visible to the naked eye". Regarding the evaluation of negative symptom severity, the SANS (Andreasen, 1982), PANSS negative subscale (Kay et al., 1987), and the NSA-16 (Axelrod et al., 1993) were all identified as appropriate for use in clinical trials. However, it was noted that the original PANSS negative symptom subscale does not provide sufficient coverage of the negative construct, which lead to the recommendation that alternative factor solutions, such as the one proposed by Marder and colleagues (Marder et al., 1997), would be preferred (see section 1.9.1.3.). The promising psychometric properties of the scales developed as a consequence of the CANSAS (Blanchard et al., 2011) were noted, and were identified as a potentially importance advance in the field. Last point of consensus was that in order to receive approval as a treatment for negative symptoms then a global improvement in negative symptoms must occur. Therefore, whilst there is evidence to suggest that negative symptoms may consist of at least 2 distinct constructs (Blanchard and Cohen, 2006) evaluating negative symptoms as a singular construct is still the preferred method in clinical trials.

During the NEWMEDS meeting there was considerable debate regarding whether negative symptoms should be prominent or predominant at the point of recruitment. Predominance specifies that the negative symptoms of schizophrenia alone should be present to at least a moderate degree of severity, whilst participants who experience prominent negative symptoms may also experience other symptoms of schizophrenia, with particular reference to positive symptoms. Whilst there was an agreement that participants should present with negative symptoms above a minimum degree of severity and a maximum level of EPS and

depression, no consensus was reached on whether a maximum positive symptom threshold should also be implemented. This is an important issue in the design of clinical trials, and is further explored in chapter 4 of this investigation.

1.9. Assessing negative symptoms

Since positive and negative symptoms of schizophrenia have been identified as distinct constructs, a number of scales have been developed to measure the negative symptom domain. In the MATRICs consensus statement, the manner in which negative symptoms are assessed was recognised as a significant barrier in the development of new treatments (Kirkpatrick et al., 2006).

In this section the scales which are currently used to evaluate the negative symptom construct will be summarised. Alternative assessment methods of negative symptoms which utilise self-report, video, or laboratory assessments of negative symptoms are not included, however details of these are available elsewhere (Horan et al., 2006, Kupper et al., 2010). Older assessment scales which are no longer adopted in negative symptom research, such as the Krawiecka-Manchester Scale (Krawiecka et al., 1977), the Negative Symptom Scale produced by Lewine (Lewine et al., 1983), or the Negative Symptom Scale produced by Pogue-Geile and Harrow (Pogue-Geile and Harrow, 1985) are also not evaluated. Finally, only tools which evaluate the whole negative symptom construct will be summarised, meaning single symptom scales such as the Chapman physical and social anhedonia scales (Chapman et al., 1976), the Temporal Experience of Pleasure Scale (TEPS) (Gard et al., 2006), and the emotional blunting scale (Abrams and Taylor, 1978), have also not been considered.

In an evaluation of any assessment tool it is important to consider which symptoms are being evaluated. This is particularly important in regards to negative symptoms, given the construct has undergone a number of substantial revisions over time (see section 1.1.5.). In addition, it is important to consider how these scales evaluate the different domains. For example, Blanchard and colleagues (Blanchard et al., 2011) suggest that for symptoms such as asociality, anhedonia and avolition it is important to evaluate the experiential components of these symptoms, rather than just relying on the behavioural aspects. This is because behaviours may be influenced by other factors such as finances, housing arrangements, and available support from family and friends, unlike an internal drive to action which may better represent core pathology. Lastly, dependent upon the nature of the enquiry it may be important to determine

the aetiology of symptoms, i.e. determining whether symptoms represent core pathology, or whether they are a consequence of other factors such as depression.

1.9.1. Structured interviews of negative symptoms

1.9.1.1. CGI-SCH: Clinical Global Impressions Scale (schizophrenia version)

The Clinical Global Impressions scale (Guy, 1976) is a brief, 3-item questionnaire designed to evaluate symptom severity, change in symptoms over time, and therapeutic response. Each item relates to a global assessment of multiple constructs, rating symptoms, behaviour, and the impact of illness on functioning over a set period (typically the past 2 days). The symptom severity item is rated on a 7 item scale, ranging from 1 "not at all ill", to 7 "Amongst the most extremely ill patients". The change items range from 1- "very much improved" to 7- "very much worse".

In a later adaptation, Haro et al. (Haro et al., 2003) modified the scale to assess positive, negative, depressive, and cognitive symptoms of schizophrenia, in addition to providing a global assessment of the illness. The scale assesses the severity of the illness over previous week, and degree of change relative to the previous assessment. The convergent validity of the CGI-SCH with the PANSS was found to be high with the cognitive, negative, positive, and global items (ICC= .78, .80, .86 and .75 respectively), and moderate with the depressive item (ICC=.61) (Haro et al., 2003). In addition, the inter-rater reliability of the scale was high for the cognitive, positive, negative, and global items (ICC range= .73-.82), and moderate for the depressive item (ICC=.64).

Given the scale only measures negative symptoms using a single item it was not identified in the MATRICS or ISCTM consensus report as an appropriate primary outcome for use in clinical trials for negative symptoms (Kirkpatrick et al., 2006, Marder et al., 2011). In addition, the CGI has been criticised for lacking standardised definitions given the scale scoring is framed in the context of other patients (Beneke and Rasmus, 1992). One significant advantage of the CGI is that it provides a result that translates into an easily interpretable clinical assessment (Nierenberg and DeCecco, 2000). Consequently, the scale has been used as a way to provide anchor points for measuring how changes in other measures such as the BPRS (Overall and Gorham, 1962) and the PANSS (Kay et al., 1987) translate into clinical response (Leucht and Engel, 2006, Leucht et al., 2005, Levine et al., 2008, Mortimer, 2007).

1.9.1.2 BPRS: Brief Psychiatric Rating Scale

The Brief Psychiatric Rating Scale (BPRS) was developed by Overall and Gorham (Overall and Gorham, 1962) as an assessment tool to evaluate symptom change. The original model comprises of 16 items, measuring somatic concern, anxiety, emotional withdrawal, conceptual disorganisation, feelings of guilt, tension, mannerisms and posturing, grandiosity, depression, hostility, suspiciousness, hallucinations, motor retardation, uncooperativeness, unusual thought content, and blunted affect. Each item is rated on a scale of 1 (absent) to 7 (extremely severe). Severity is defined both by the frequency of symptoms experienced, and their impact on functioning over the past week. Five items are rated based on observations made during the interview and information from clinicians and caregivers, whilst 11 are based primarily on the verbal report of the interviewee. In an assessment of the item-level rater reliability, moderate to high correlations have been detected (r=.56 - .86) (Overall and Rhoades, 1982).

Since the original model the scale has been expanded to include 18 items, adding disorientation and excitement (Overall and Klett, 1972). The scale was then later expanded to 24 items, adding motor hyperactivity, elevated mood, distractibility, self-neglect, bizarre behaviour, and suicidality (Lukoff et al., 1986). Subsequent evaluations of the expanded version of the scale found the interview is a sensitive measure with good inter-rater reliability (Ventura et al., 1993)

Whilst the original scale was designed to provide a single summary score, since its inception a series of factor analytic studies have been completed with schizophrenia patients which suggest the scale comprises of multiple symptom subdomains (Burger et al., 1997, Dingemans et al., 1995, Long and Brekke, 1999, Malla et al., 1993, Mueser et al., 1997, Overall and Beller, 1984, Ruggeri et al., 2005, Van der Does et al., 1993, Velligan et al., 2005, Ventura et al., 2000). A summary of the different factor loadings of the 24-item version of the scale are presented in table 2. These studies appear to suggest a 4 or 5 factor solution, covering negative/anergia symptoms, positive/psychosis symptoms, anxiety/depressive/affective symptoms, and hostility/activation symptoms. For the purposes of this investigation four factor solution proposed by Velligan (Velligan et al., 2005) was adopted for two reasons. Firstly, this study was completed on a significantly larger sample size than the others (1440 participants), which is important given all but the Ruggeri study included less than 200 participants, which has been deemed inadequate for factor-analytic studies (Comrey and Lee, 2013). Secondly, in the

Velligan study both exploratory and confirmatory factor analysis were conducted, meaning the findings are more robust.

	Study							
BPRS Items	(Velligan et al., 2005)	(Dingemans et al., 1995)	(Van der Does et al., 1995)	(Burger et al., 1997)	(Ventura et al., 2000)	(Ruggeri et al., 2005)		
Somatic Concern			5	5		5		
Anxiety	5	5	5	5	5	5		
Emotional Withdrawal	1	1	1	1	1	1		
Conceptual disorganisation	2	4	4	2		2		
Guilt	5	5	5	5	5	5		
Tension	4	4	4	4		5		
Mannerisms and Posturing			4	4		4		
Grandiosity		2		2	4	2		
Depression	5	5	5	5	5	5		
Hostility			4	3	4	4		
Suspiciousness		2	2	3	2	2		
Hallucinations	2	2	2	2	2	2		
Motor retardation	1	1	1	1	1	1		
Uncooperativeness		2		3		4		
Unusual thought content	2	2	2	2	2	2		
Blunted Affect	1	1	1	1	1	1		
Disorientation				1	2	1		
Excitement	4	4	4	4	4	4		
Motor hyperactivity	4	4	4	4	4	4		
Elevated mood			4		4	4		
Distractibility		4	4		4	4		
Self-neglect		1	1	1	1	4		
Bizarre behaviour	2	2	4	2	2	4		
Suicidality	5	5	5	5	5	5		

Table 2: Different configurations of the 24-Item BPRS-E from different factor analysis studies^a

Factor loadings: 1= Negative / Anergia symptoms, 2= Positive symptoms/ thinking disorder 3= Hostility / suspiciousness 4=Excitement/activation symptoms 5= Depression/anxiety / affective symptoms

^aHighest factor loadings presented, with those lower than <.4 omitted.

Grey highlighting indicates the negative symptom construct adopted in chapter 7.

Despite being extensively used in clinical and research settings, the BPRS has a number of limitations with regards to negative symptom assessment. One significant issue is that the scale does not appear to tap into a number of domains which are considered part of the

negative symptom construct, such motivational and anhedonic deficits, as outlined in section 1.2.3. In addition, whilst the item "emotional withdrawal" is included, it appears to relate specifically to how emotionally engaged the participant is during the interview, and so does not represent the broader concept of asociality. As a result it has been suggested that the BPRS insufficiently evaluates the experiential features of negative symptoms (Blanchard et al., 2011). This was evident during the development of the Clinical Assessment Interview for Negative Symptoms (CAINS) (Kring et al., 2013) which found a moderate correlation between the BPRS and expressive symptoms (r=.52), but only a low correlation with experiential features of negative symptoms (r=.28). A second issue is that in a comparison to the Negative symptom Assessment scale (NSA) (Axelrod et al., 1993), and the Scale to Assess Negative Symptoms (SANS) (Andreasen, 1983) the BPRS was found to be a much less sensitive instrument in detecting symptom change (d=.32, in comparison to d=.70 and d=.78 for the SANS and NSA respectively) (Eckert et al., 1996). This is a significant issue given the lack of instrument sensitivity has been identified as one of the major barriers to treatment development in negative symptoms (Kirkpatrick et al., 2006).

1.9.1.3 PANSS: Positive and Negative Symptom Scale

The Positive and Negative Syndrome Scale (PANSS) (Kay et al., 1987) is a structured clinical interview designed to measure the symptoms of schizophrenia, and along with the CGI and the BPRS is one of the most extensively used assessment tools in the field of schizophrenia research (Mortimer, 2007). The PANSS is based on items from the BPRS (Overall and Gorham, 1962) and the Psychopathology Ratings Scale (Singh and Kay, 1975). In a similar fashion to the BPRS (Overall and Gorham, 1962), symptom severity is defined both by the frequency of symptoms experienced, and their impact on functioning over the past week The scale is composed of 30 items; 7 positive symptom items, 7 negative, and 16 general symptoms which include, amongst other things, affective symptoms and cognitive deficits. Each item is rated from 1 to 7, giving a range of 7-49 for the positive and negative subscales, and 16-112 for the general subscale. Obermeier and colleagues (Obermeier et al., 2010) recommend transforming the data to be re-scaled at ratio level (i.e. 1 subtracted from each item so they are scaled from 0-6), which was completed in chapter 6 in sections where percentage changes were considered. However, this was not adopted in other sections given this has not consistently been adopted in the literature, meaning it would be harder to interpret any findings from this investigation in the context of other studies.

An early evaluation of the scale suggested it is possible to achieve high levels of inter-rater reliability (ICC>.8) (Kay et al., 1988, Peralta and Cuesta, 1994), which has been since been replicated in a series of major trials. The scale has also been found to have high convergent validity to the SANS and the SAPS (Andreasen, 1983, Andreasen, 1984).

Since the publication of the PANSS, a series of factor analytic studies have suggested that the positive, negative and general symptoms subscale structure is not sufficiently stable. Instead, the majority of studies appear to support a 5 factor solution comprising of negative, positive, disorganised/cognitive, excited, and depression/anxiety symptoms (see table 3 for a summary).

These five domains have been found to be consistent over age, symptom severity, chronicity, and in patients both in the chronic and first-episode of the disorder (Emsley et al., 2003). However, as indicated in table 3 there has been some disagreement with regards to the different items which correspond to each factor. Given the negative symptom construct is the principle focus or this investigation, the priority was to adopt the most consistent factor solution for this domain. In the 9 studies evaluated (Emsley et al., 2003, Kay and Sevy, 1990, Lancon et al., 2000, Lancon et al., 1999, Lindenmayer et al., 1994, Lykouras et al., 2000, Marder et al., 1997, Mass et al., 2000, White et al., 1997) there was a consensus that five of the original negative items; affective blunting, emotional withdrawal, apathetic social withdrawal, poor rapport, and impaired conversation flow all related to the negative symptom construct. Of the remaining two, deficits in abstract thinking appeared to correspond to cognitive deficits, whilst stereotypical thinking appeared somewhat unstable and was found to relate to positive, cognitive, manic-excitement factors, or else did not load onto any factor at all. Instead, the general psychopathology items active social avoidance and motor retardation were found to consistently load on to the negative factor. Other items, such as preoccupation, disturbance of volition, mannerisms and posturing and poor attention were not found to load consistently onto negative symptoms and so were not considered. Following the results of these factor analytic studies, the 7-item negative subscale comprising of affective blunting, emotional withdrawal, apathetic social withdrawal, poor rapport, impaired conversation flow, motor retardation and active social avoidance was adopted throughout this investigation. This model, whilst proposed by a number of studies (Lancon et al., 2000, Lindenmayer et al., 1994, Marder et al., 1997), has been most commonly referred to as the Marder configuration, the term adopted in this thesis.

	Study										
PANSS Items (configured in standard factor loadings)	Marder et al., 1997	White et al., 1997	Linden- mayer et al., 1994	Emsley et al., 2003	Kay & Sevy, 1990	Lancon et al., 1999	Lancon et al., 2000	Mass et al., 2000	Lykouras et al., 2000		
Negative subscale:											
N1: Blunted affect	1	1	1	1	1	1	1	1	1		
N2: Emotional	4	4						4			
withdrawal	T	T	T	T	T	1	T	T	T		
N3: Poor rapport	1	1	1	1	1	1	1	1	1		
N4: Passive social	1	1	1	1	1	1	1	1	1		
withdrawal	T	T	T	T	T	T	T	T	T		
N5: Difficulty in	3	З	3	2	3	3	3	З	з		
abstract thinking	5	5	5	5	5	5	J	5	5		
N6: Lack of	1	1	1	1	1	1	1	1	1		
spontaneity											
N7: Stereotypical	2	3	4	3	-	4	-	-	3		
UIIIKIIIg											
Positive subscale:	2	2	2	-	2	-	2	-	2		
P1: Delusions	2	2	2	2	2	2	2	2	2		
disorganisation	3	-	3	3	3	3	3	3	3		
P3. Hallucinations	2	2	2	2	2	2	2	2	2		
P4: Excitation	4	4	4	4	4	4	4	4	4		
P5: Grandiosity	2	2	2	2	2	2	2	4	2		
P6: Suspiciousness	2	_	2	2	-	2	2	-	2		
P7: Hostility	4	4	4	4	4	4	4	4	4		
General Psychonathology	subscale										
G1: Somatic concern	2	5	5	5	5	2	-	-	4		
G2: Anxiety	5	5	5	5	5	5	5	5	5		
G3: Guilt Feelings	5	5	5	5	5	5	5	5	5		
G4: Tension	5	5	4	5	4	4	4	4	4		
G5: Mannerisms	n	1	2	2	1	2		4	1/2		
and posturing	2	T	3	3	T	Z	-	4	1/3		
G6: Depression	5	5	5	5	5	5	5	5	5		
G7: Motor	1	1	1	1	1	1	1	-	1		
retardation											
G8: Uncoonorativonoss	4	4	4	4	4	4	-	4	4		
G9. Unusual											
thought content	2	2	2	2	2	2	2	2	2		
G10: Disorientation	3	-	3	3	3	3	3	-	3		
G11: Poor attention	3	3	3	3	1	3	3	3	3		
G12: Lack of											
judgement	2	-	2	2	2	2	2	-	2/5		
G13: Disturbance of	C	C	2	1	1	1			1		
volition	3	3	3	T	T	T	-	-	T		
G14: Impulsivity	4	4	4	4	4	4	4	4	4		
G15: Preoccupation	3	3	5	3	5	1	-	-	1		
G16: Active social	1	-	1	1	1	1	1	1	1		

Table 3: Different configurations of the PANSS present from factor analytic studies according to highest factor loadings^a

Factor loadings: 1= Negative symptoms, 2=Positive symptoms 3: Disorganised/cognitive symptoms 4=Excited symptoms 5= Anxiety/ depressive symptoms

^aTable adapted from Emsley et al., 2003. Grey highlighting represents the negative symptom construct adopted in this investigation

Despite the widespread use of the scale, the PANSS has been recognised to have a number of limitations in measuring the negative symptom construct (Blanchard et al., 2011). Firstly, it has been suggested that the PANSS relies primarily on behavioural referents such as the range and frequency of activities, interviewer observations, and reports from family members or carers, and insufficiently taps into what the interviewees themselves experience. The problem of basing results exclusively on behavioural referents is the fact that other factors such as finances, depressive and positive symptoms, lack of social skill (Bellack et al., 1994), and social rejection (Corrigan, 2004) may contribute to a lack of social activity, independent of the will to engage in such activities. Secondly, by focusing on behavioural referents, the scale may underrepresent deficits in interviewee experiences which has been found constitute a distinct construct in negative symptoms (Blanchard and Cohen, 2006).

1.9.1.4 SANS: Scale for the Assessment of Negative Symptoms

Unlike the previous tools outlined, the Scale for the Assessment of Negative Symptoms (SANS) (Andreasen, 1983) is designed exclusively to measure the negative symptoms of schizophrenia. The scale comprises of 30 items covering five domains, with each symptom measured on a 6point scale ranging from 0 (none) to 5 (severe). The five domains evaluate affective flattening, anhedonia-asociality, alogia, avolition-apathy, and attention rated over the previous month. The affective flattening subscale comprises of 9 items, measuring reductions in facial, vocal, and gestural expressiveness; a decrease in spontaneous movements; poor eye contact; inappropriate affect; affective non-responsivity; and a subjective rating of affective flattening. The alogia subscale consists of 6 items, and rates the poverty of the quantity and content of speech; increased latency of response; the presence of blocking; and a subjective rating of alogia. Avolition-apathy includes 4 items, assessing motivation to address grooming and hygiene; vocational pursuits; physical anergia; and a subjective assessment of avolition-apathy. The anhedonia-asociality subscale measures deficits relating to interest and pursuit in recreational activities; sexual interest and activity; ability to feel intimacy; relationships with friends; and a measure of subjective awareness of anhedonia-asociality. The final subscale, attentional impairments, measures inattentiveness during work and assessment; in addition to a subjective measure of attentiveness deficits.

In the original study the Interrater reliability of single items (ICC=.701-.926), subscale scores (ICC=.860-.925) and the overall summary score (ICC=.838) were all found to be high (Andreasen, 1982). In addition, the internal consistency of each of the subscales were all high (α =.799 - .844), other than the anhedonia-asociality subscale score (α =.632). In an examination

of the correlation of individual items, the subjective assessments were all found to correlate relatively poorly to the subscale scores, leading to Andreasen to suggest that these should be omitted in calculating the summary ratings.

One major limitation of the SANS is the fact that some of the symptoms measured are now not considered to represent part of the negative symptom construct. For example, the SANS subscale "attention" is now considered to evaluate cognitive, rather than negative symptoms (Blanchard et al., 2011, Harvey et al., 2006, Milev et al., 2005). Consequently, a number of studies using the SANS have removed this subscale (Rabany et al., 2011), supported by evidence which suggests that omitting attention improves the internal consistency of the scale overall (Peralta et al., 1992b). In addition, a number of individual items have also been found to be problematic such as poor eye contact, which does not load on to any factor (Rabany et al., 2011), blocking, which is an expression of thought disorder and so therefore a positive symptom (Rule, 2005), and poverty of content of speech, which is understood to represent symptoms of cognitive impairment (Liddle, 1987). Another significant issue with the SANS is the fact that individual subscales can relate to multiple, conceptually distinct domains (Blanchard et al., 2011). The anhedonia-asociality subscale can reflect frequency of social contacts, interest, experience of pleasure and hostility, meaning it is unclear what psychological processes are reflected in the ratings given. Lastly, whilst the scale was designed to focus on behavioural referents due to evidence at the time which suggested this is a more reliable approach to measuring negative symptoms (Andreasen, 1979), this could also be considered somewhat problematic. Conceptually, symptoms such as avolition and asociality are defined by internal states, such as ones desire for relationships and contact, whilst the behavioural measures of these constructs, (i.e. the amount of friend contacts they have had) are representations of these internal states. However, these behaviours can potentially be influenced by a number of extrinsic factors such as social skill (Bellack et al., 1990), opportunities for social contacts, and the level of support they are given to encourage social encounters which do not represent core negative symptoms (Blanchard et al., 2011).

1.9.1.5. NSA: Negative Symptoms Assessment scale

In its original format the Negative Symptom Assessment scale (NSA) (Alphs et al., 1988) is a 26item tool evaluating 5 distinct domains, including impaired communication, disturbed affect/emotion, reduced social involvement, reduced motivation, and impaired gross cognition. Following a subsequent factor analytic study of the scale a number of minor adjustments were made, including dropping the item "slowed movements", and splitting the impaired communication domain into two distinct clusters, covering impaired communication and retardation (Axelrod et al., 1993). The inter-rater reliability and test-retest reliability of the scale was found to be excellent (Alphs et al., 1988), whilst the internal consistency was also found to be high (α =0.94) (Axelrod et al., 1993). In addition, after only a 30 minute training session novice raters were able to complete assessments consistent with expert ratings (Axelrod and Alphs, 1993).

In response to concerns regarding the length of the scale, a shorter 16-item version of the scale was devised (Axelrod et al., 1993). In this scale the gross cognition domain was dropped, whilst 4 other items were removed due to redundancy. In the alternative 16-item version impaired communication was assessed over 4 items, measuring quantity, content, and latency of speech. Three items assessed affect/emotion, measuring the range of emotions experienced, reduced emotional display, and reduced modulation. Social involvement was measured by 3 items assessing social drive, rapport, and sexual interest. Four items are included in the motivation domain, covering a reduced sense of purpose, daily activity, hobbies and interests, and poor grooming and hygiene. Finally, retardation was measured by 2 items, evaluating reduced expressive gestures and slow movements.

In this amended version of the scale the internal consistency was found to be high (a=.85). In addition, the scale was found to have good convergent validity with the PANSS negative subscale, and good discriminant validity from scales assessing depression, positive symptoms and quality of life (Alphs et al., 2011). In a further shortening of the scale, a 4-item version of the NSA was devised, designed for routine clinical use (Alphs et al., 2011). This scale was found have good convergent validity with the NSA-16, the PANSS negative subscale, and the Strauss-Carpenter Level of Functioning scale (Strauss and Carpenter, 1977), in addition good discriminant validity from scales assessing depression, positive symptoms and quality of life scales.

As with the PANSS and the BPRS, the NSA has been criticised for not representing what is currently understood to constitute the negative symptom construct, in addition to other conceptual issues (Blanchard et al., 2011). For example, the item "reduced emotional range" can theoretically lead to conflating anhedonic response with a lack of experiences, both positive and negative. In addition, this item can also result in the counter-intuitive position where people report experiencing more negative experiences at follow-up, relative to baseline, which would result in a lower symptom severity score. In addition, the scale also

explicitly instructs raters to focus on behaviours, meaning the experiential features of negative symptoms are under-represented.

1.9.1.6. CAINS: Clinical Assessment Interview for Negative Symptoms of schizophrenia

In the 2005 NIMH-MATRICS consensus meeting on negative symptoms of schizophrenia, issues regarding current negative symptom assessment scales were identified as a significant barrier to the development of new treatments (Kirkpatrick et al., 2006). In recognition of this problem, the Collaboration to Advance Negative Symptom Assessment Scales (CANSAS) (Blanchard et al., 2011) was established, which in turn developed two assessment scales; the Brief Negative Symptom Scale (BNSS) (Kirkpatrick et al., 2011) and the Clinical Assessment Interview for Negative Symptoms of Schizophrenia (CAINS) (Forbes et al., 2010, Horan et al., 2011, Kring et al., 2013).

The CANSAS followed a process modelled on the original MATRICS collaboration that developed the MCCB for cognitive symptoms (Nuechterlein et al., 2008), which was similar to the method outlined by Clark and Watson (Clark and Watson, 1995). Clark and Watson emphasised the importance of developing a detailed conception of the construct, and its theoretical context, as the first step in the scale development process. In order to achieve this, the CANSAS group conducted a series of literature reviews and biweekly conference calls between members in order to develop the theoretical concept of the negative symptom construct that the scale would be designed to measure. As identified in the MATRICS consensus statement, the negative symptom construct was defined to include alogia, blunted affect, anhedonia, avolition and asociality (Kirkpatrick et al., 2006), with the possibility that the construct comprises of two distinct subdomains of expressive and experiential deficits. How these symptoms were evaluated was guided by related current neurobiological models, such as the recognition that different hedonic experiences have distinct neural circuits (Berridge and Robinson, 1998, Knutson et al., 2001), with negative symptoms related specifically to anticipatory pleasure (Gard et al., 2007). In a departure from earlier scales, the decision was taken that the interview would aim to evaluate interviewees' descriptions of their internal states, in addition to behaviours and environmental factors. While it was originally considered that assessing behavioural rather than internal states may be a more reliable approach to measuring negative symptoms (Andreasen, 1979), later work has suggested that the subjective experiences of schizophrenia patients can be both reliable and valid (Jaeger et al., 1990, Liddle and Barnes, 1988). One significant advantage of assessing internal states is that they are

considered to be less influenced by external factors such as finances and input from family and carers, meaning this may represent a more accurate measure of core psychopathology (Blanchard et al., 2011).

In the next stage, a pool of initial items for a clinical rating scale were generated, and were evaluated by group members, other (unspecified) clinical trial researchers, presented at national conferences, and posted to a public website where other academics and researchers from industry could provide further feedback. As recommend by Clark and Watson, in the initial stages the aim was to be over-inclusive in terms of the number of items, with particular attention paid to the wording of the questions presented (Clark and Watson, 1995). Following this feedback, a beta version of the scale was produced and validation of the scale was performed (Forbes et al., 2010).

In the initial feasibility study of the CAINS scale by Forbes and colleagues (Forbes et al., 2010), a 23-item beta-version of the interview was evaluated. Items were rated on a 7-point scale (scored 0-6) with higher scores reflecting greater psychopathology. The scale was designed to cover the 5 symptoms of schizophrenia as specified in the MATRICs consensus statement review (Kirkpatrick et al., 2006), including avolition (4 items), asociality (3 items), anhedonia (9 items), alogia (2 items) and blunted affect (5 items). The internal consistency, discriminant validity, and convergent validity of the overall scale appeared good. However, relatively poor interrater agreement with the avolition (ICC=.53) and alogia (ICC=.48) items, issues with the anchoring of the anhedonia items, and low internal consistency in the asociality (α =.32) and avolition (α =.47) items called for a significant amendments to the scale. In a revised version, outlined by Horan and colleagues (Horan et al., 2011), 16 of 23 items were retained within a 2factor solution, with each item rated on a scale of 0-4. The first factor covered experiential items, relating to anhedonia, amotivation and asociality, and the second factor expressive symptoms, including items relating to alogia and blunted affect. In this second iteration the items retained were found to have a better distribution, high interrater agreement, and good convergent and discriminant validity (Horan et al., 2011). An assessment of the categories using item response theory suggested that most of the anchor points for each item were able to provide useful distinctions (particularly with the expressive items), but did identify that revisions would be necessary to better demarcate symptom thresholds at lower levels.

The final development and validation stage was outlined by Kring and colleagues (Kring et al., 2013). In this iteration, a 13 item scale comprising to two factors, measuring emotional experience and expression was proposed. The experiential subscale comprises of nine items,

assessing the frequency, motivation and anticipated pleasure in activities and relationships around work and school, friends, and the family. The emotional expression subscale consists of four items, measuring the vocal and gestural features of expression. The subscales can be combined to provide one summary score of negative symptoms, thus adhering to the recommendations outlined in the ISCTM consensus statement (Marder et al., 2011). However, the authors recommend reporting the scores separately given the evidence suggests they appear to be represent distinct constructs (Horan et al., 2011).

In the final version, the interrater reliability for the experiential and expressive subscales were high (ICC=.93 and ICC=.77 respectively). The test-retest reliability of the two subscales also demonstrated adequate reliability (ICC=.69), comparable to those found with the negative symptom subscale of the PANSS. In an assessment of the convergent validity of the scale, both CAINS subscales were positively correlated with the BPRS anergia subscale and the SANS, and negatively associated with the desire for close relationships and social engagement. In an assessment of the scales discriminant validity, both CAINS subscales were found not to be associated with depression, medication side effects, or cognitive impairment. A moderate association was detected between the experiential subscale and positive and agitation symptoms, however this association was significantly weaker than that detected between positive and agitation symptoms and both the SANS and BPRS anergia subscale.

Despite the extensive validation process which went into the production of the CAINS, it is important to note the steps in scale construction which were not adopted. For example, while the opinions of a number of clinical and industry experts in the field were sought, this was not formalised using a process such as the Delphi technique. The Delphi technique, originally developed by the RAND corporation (Dalkey and Helmer, 1963) is a structured, iterative process by which to aggregate the opinions of experts in any particular field. Some of the main advantages of this process are that it allows a diverse panel of participants to be recruited, it is conducted anonymously which can reduce any possible "halo effect" of prominent participants, and that controlled, staged feedback process can reduce any possible effect of noise which may distort the data by focusing on individual interests (Hsu and Sandford, 2007). As a result, adopting this method may have allowed for a consensus to be reached by incorporating a broader range of opinions than may otherwise have been the case.

Conceptually, the CAINS appears to follow a much closer representation of what we currently understand negative symptoms to be, assessing anhedonia, asociality, avolition, alogia and blunted affect (Blanchard et al., 2011). However, at present it is not clear if the scale is more

sensitive in determining change relative to existing scales such as the SANS, or the PANSS. The lack of sensitivity of existing scales has been recognised as significant issue which impedes the evaluation new therapeutic interventions designed to treat these symptoms (Kirkpatrick et al., 2006). In recognition of this current gap in the literature, the relationship between the CAINS and the PANSS negative subscale will be explored in greater depth in chapter 5.

1.9.1.7. BNSS: Brief Negative Symptom Scale

The Brief Negative Symptom Scale (BNSS) (Kirkpatrick et al., 2011) is the second scale produced as part of the CANSAS collaboration (Blanchard et al., 2011). Relative to the CAINS, the BNSS is a relatively brief instrument designed for routine clinical use, in addition to clinical trials. The scale itself consists of 13 items, covering the five domains of alogia, anhedonia, avolition, blunted affect and asociality recognised to cover negative symptoms. One of the main advantages of this particular scale is that given its brevity, the scale can be completed in approximately 15 minutes (Strauss et al., 2012). As with the CAINS, an examination of the factor structure of the scale yielded two distinct clusters reflecting expressive and experiential symptoms (Strauss et al., 2012).

In an initial evaluation of the psychometric properties of the scale , the inter-rater reliability of the 5 subscales in addition to the global score were found to be high (global score ICC=.96, individual subscales ICC=.89 - .95). An assessment of the internal consistency of the scale was very high (α =.93), and all items were moderately to highly correlated to the global BNSS score (r=.53 - .85), suggesting the scales measures a singular latent construct. The test-retest reliability over a period of 2 weeks was also found to be relatively high, both for the total score (r=.81), and the subscales (r=.76 - r=.90). In an assessment of the concurrent and discriminant validity of the scale, the BNSS was found to be highly correlated with the PANSS negative subscale and the SANS (r=.80 and r=.84 respectively), whilst not being correlated with the PANSS depression item (r=.14), or cognitive deficits. In a final validation of the scale (Strauss et al., 2012) internal consistency, temporal stability, and appropriate convergent and divergent validity were high.
1.9.2. Pooling negative symptom assessments

In this investigation negative symptoms will be measured by a number of different scales. Given the constructs evaluated in each scale are slightly different, combining scores evaluating individual negative symptoms is not straight forward. One way to pool different items measured by the SANS, PANSS and BPRS is the method proposed by Lyne (Lyne et al., 2012) (see table 4), which was adopted as part of the investigation completed in chapter 3.

SANS	PANSS	BPRS
Affective flattening	Blunted Affect	Blunted Affect
Alogia	Lack of Spontaneity/flow of conversation	-
Avolition/Apathy	Passive social withdrawal	Self-Neglect
Anhedonia-Asociality	-	-

|--|

*Method of combining items proposed by Lyne et al., 2012

Chapter 2. Rationale and research questions

2.1. Rationale for investigation

The negative symptoms of schizophrenia have been found to be largely resistant to current pharmacological and psychosocial treatments (Fusar-Poli et al., 2015). This is a significant issue, given these symptoms are associated with substantial impairments in functioning (Hunter and Barry, 2012). In an attempt to address a lack of progress in treatment development there have been considerable efforts to improve our understanding of the negative symptom construct, their assessment, and the manner in which clinical trials in this area are conducted (Kirkpatrick et al., 2006, Marder et al., 2013, Marder et al., 2011). The aim of this thesis will be to address a series of questions which either may inform the design of clinical trials, or else builds upon recent advances in negative symptom assessment to revisit areas of interest.

Regarding the longitudinal course of negative symptoms, the evidence suggests that whilst secondary negative symptoms improve relatively quickly once the underlying causes are addressed, primary negative symptoms are highly stable. A number of observational studies and randomised controlled trials have reported differing trajectories in negative symptom course, with the heterogeneity likely to be a consequence of factors such as symptom antecedents, clinical variables, socio-demographic differences, and variations in the methodologies employed. The first aim of this investigation will be to uncover broad trends in how these symptoms may change over time in schizophrenia outpatients by pooling studies which utilise different treatment options and exploring the impact of factors considered to influence symptom course. Such an investigation could potentially be informative to future trial design, in addition to clinically relevant.

Whilst conducting a meta-analysis may be useful in uncovering broader trends of negative symptoms over time, methodological limitations mean it it is difficult to determine what degree of change is attributable to transient secondary symptoms. In clinical trials there have been a number of different methods proposed to limit the variability of negative symptoms attributable to secondary sources (Buchanan, 2007, Kirkpatrick et al., 2006, Marder et al., 2013, Marder et al., 2011). However, at present there does not appear to be a consensus on how these criteria should be implemented, particularly with regards to limiting the impact of positive symptoms. There is evidence to suggest that adopting different criteria can result in

large differences in the potential sample pool (Dunayevich et al., 2014, Rabinowitz et al., 2013), whilst it is unclear whether adopting predominant criteria reduces the impact of positive symptoms on negative symptoms (Dunayevich et al., 2014, Stauffer et al., 2012). In this investigation a series of study inclusion criteria will be compared to determine their impact on the association between negative symptoms, and positive and depressive symptoms. Such an investigation may prove helpful in standardising the inclusion criteria used in negative symptom trials.

In the MATRICs consensus statment (Kirkpatrick et al., 2006) the limitations of existing negative symptom assessment tools were identified as a significant barrier in the evaluation of new treatments. The report from this meeting recommended the development of new assessment tools which are both in line with our current conception of the negative symptom construct, and are more sensitive to detect symptom change compared to existing scales. This led to the development of the CAINS (Horan et al., 2011) which has been found to have excellent psychometric properties (Kring et al., 2013). However, at present it is not clear how the sensitivity of the CAINS compares with existing negative symptom scales. In this investigation the sensitivity of the CAINS will be compared to the PANSS negative subscale, both in terms of how they measure the change in symptoms over time. In addition, the incremental validity of the CAINS will be compared to the PANSS negative subscale to determine which scale is a stronger predictor of indicators of social impoverishment. This investigation of the CAINS could be considered the important next step in validating the scale for use in clinical trials evaluating treatments for negative symptoms.

Recent developments in our understanding of negative symptoms suggest that expressive and experiential deficits represent two distinct constructs (Blanchard and Cohen, 2006). However, at present the relationship between these separate subdomains and functional outcomes have not been fully explored, possibly due to limitations in earlier scales which have only recently been addressed by the development of the CAINS (Blanchard et al., 2011, Kring et al., 2013). One such area that may benefit from examining negative symptoms in this manner is reviewing the relationship between negative symptoms and quality of life (QOL).

Whilst a strong inverse relationship between negative symptoms and OQOL has been consistently been reported, only weak associations have been detected between negative symptoms and SQOL (Fitzgerald et al., 2001). It has been proposed that a weak relationship between negative symptoms and SQOL may be attributable to the chronicity of the disorder or

peer group comparison (Priebe, 2007). However, another possibility is that the relationship may have been previously under-reported by earlier assessment tools insufficiently evaluating the experiential features of negative symptoms. In this investigation the relationship between SQOL, negative symptoms, and experiential and expressive deficits will be evaluated, both cross-sectionally and over time. These relationships will be compared to those reported by PANSS negative subscale, which measures negative symptoms as a singular construct.

In addition to SQOL, other subjective reports from patients such as initial appraisal of inpatient treatment are important given they are associated with outcomes long after the patient has been discharged (Priebe et al., 2009). Therefore, it is important to understand the impact of symptoms on initial appraisal of treatment in the acute phase, regardless of whether they can be considered primary or secondary in origin. In this final section of the study the aim will be to look at the impact of negative symptoms, in addition to other symptoms, at the point of admission to examine their impact on subjective initial appraisal of inpatient.

2.2. Study aims and research questions

In order to address the above questions, five different investigations were completed. A summary of each study is presented below.

2.2.1. Examination of the longitudinal course of negative symptoms in schizophrenia outpatients

To evaluate the course of negative symptoms over time a systematic search will be conducted to identify all studies which evaluated negative symptoms over at least two time points in schizophrenic outpatient samples. The identified studies will then be pooled together by way of meta-analysis to examine trends in negative symptom change over time. Next, a series of meta-regressions will be conducted in order to examine the impact of different eligibility criteria, study duration, and assessment scales used. In addition to the evaluation of negative symptoms as a single construct, individual symptoms were also examined. Finally, planned sub-group analyses will be conducted looking at negative symptom change between different assessment scales, in addition to studies which specify minimum negative and maximum positive and depressive symptoms as part of their eligibility criteria. Given the investigation will be exploratory in nature no specific hypothesis will be tested in the main part of the analysis. In the subgroup analyses it is predicted that the SANS will be a more sensitive instrument to detect negative symptom change relative to the PANSS, and that a smaller change over time will be detected in studies which used criteria which controlled for sources or secondary negative symptoms, in comparison to those that did not.

The principle aim of this chapter is to examine whether negative symptoms change over time, and if so, how. In order to answer this 4 different questions will be addressed:

- 1a) What is the course of negative symptoms in schizophrenia outpatients?
- 1b) Is the change of negative symptoms over time lower in samples that control for sources of secondary negative symptoms?
- 1c) Does the SANS report a greater change in negative symptoms, relative to the PANSS and the BPRS?
- 1d) Is there any variation in the longitudinal course of individual negative symptoms?

2.2.2. Association between positive and depressive symptoms with negative symptoms after adopting different symptomatic inclusion criteria: A further consideration of the pseudospecificity problem

In the second part of this investigation the impact of adopting different inclusion criteria in clinical trials for negative symptom treatments will be explored. Inclusion criteria identified in the systematic review, and those considered by Dunayevich and colleagues (Dunayevich et al., 2014), will be used to generate different subsamples to assess their impact on the potential sample pool size, and changes in negative symptoms over time. In the second part of the analysis a series of longitudinal regressions will be conducted to evaluate the association between negative symptoms, and depression and positive symptoms in the different subsamples generated. Following earlier findings (Dunayevich et al., 2014, Rabinowitz et al., 2013) it is hypothesised that increasingly restrictive eligibility criteria would result in a substantial reduction in the number of eligible participants. The second part of the analysis is exploratory in nature, and so specific hypothesis will be tested.

In this chapter of the thesis, the impact of adopting different symptom inclusion criteria on the association between negative, and positive and depressive symptoms will be evaluated. In order to answer this question, 3 different areas will be addressed:

- 2a) What is the change in negative symptoms over time in stable schizophrenia outpatients after excluding participants with prominent positive and depressive symptoms?
- 2b) What is the impact of adopting different inclusion and exclusion criteria on the proportion of eligible participants?
- 2c) What is the impact of adopting different symptom inclusion and exclusion criteria on the association between negative symptoms, and positive and depressive symptoms?

2.2.3. Comparing the CAINS and the PANSS; how they relate to indicators of social impoverishment

In this part of the investigation, the first aim will be to assess the convergent validity of the CAINS subscales and the PANSS negative subscales, building upon the previous findings which have evaluated the relationship of this scale with the BPRS (Kring et al., 2013). The second aim will be to assess whether there are any systematic differences in how the CAINS and the PANSS capture negative symptoms. This includes a comparison of how the two scales differentiate high and low negative symptoms using Tukey-mean difference plots, and a comparison of how the scales measure symptom change over time. The third aim will be to examine the incremental validity of the CAINS by comparing its predictive ability of objective and subjective measures of social impoverishment with the PANSS negative subscale using dominance analysis. Negative symptoms relate specifically to deficits in motivation to take part in recreational activities and impairments in the quality and quantity of social networks, so the scale which better predicts these outcomes is likely to be a better indicator of the negative symptom construct. This investigation will test the hypothesis that the CAINS is a more sensitive instrument in detecting negative symptom change, and that the experiential subscale is a significantly better predictor of indicators of social impoverishment.

In an exploration of the relationship between the CAINS and the PANSS negative subscale, 4 subordinate questions will be addressed:

- 3a) What is the association between the CAINS and the PANSS negative subscale?
- 3b) Does the CAINS provide a greater differentiation between participants that report high and low symptoms, relative to the PANSS negative subscale?
- 3c) Is the CAINS more sensitive in detecting symptom change over time, relative to the PANSS negative subscale?
- 3d) Is the CAINS a better predictor of social outcomes, relative to the PANSS negative subscale?

2.2.4. The relationship between negative symptoms and quality of life.

The aim of this investigation will be to re-evaluate the link between negative symptoms of schizophrenia and subjective quality of life, using both the PANSS and the experiential and expressive subscales of the CAINS. A series of univariate regression analyses will be conducted to examine the relationship between SQOL and expressive and experiential deficits, in addition negative symptoms measured by the PANSS, and depressive symptoms measured by the Calgary depression scale. This analysis will be conducted both cross-sectionally, and in changes over time, with any predictors approaching significance included in a multivariate model. The aim of this investigation will be to test the hypothesis that the association between negative symptoms and subjective quality of life will relate specifically to experiential deficits, and not expressive deficits. In addition, it is hypothesised that this relationship will remain after controlling for depressive symptoms.

In order to explore the association between expressive and experiential features of negative symptoms and subjective quality of life, 2 subordinate questions will be addressed:

4a) What is the cross-sectional association between experiential and expressive deficits and subjective quality of life, and how does this compare to an assessment of negative symptoms as a single construct using the PANSS negative subscale? 4b) What is the association in changes over time between experiential and expressive deficits and subjective quality of life, and how does this compare to an assessment of negative symptoms as a single construct using the PANSS subscale?

2.2.5. The Impact of negative symptoms on appraisal of treatment in acute services.

In the final part of this investigation the relationship between subjective initial appraisal of acute inpatient treatment and negative, manic, positive, and affective symptoms will be explored. Using a pooled sample from three separate studies, the association between these symptoms and the appraisal of treatment will be evaluated in a multivariate regression analysis, controlling for various socio-demographic and clinical variables. All symptoms which approach significance will then be analysed in a single multivariate model to determine their individual impact on treatment appraisal. In the final part of the analysis, the impact of individual symptoms will be compared between those who are admitted voluntarily and involuntarily.

In order to evaluate the association between symptoms and the initial appraisal of inpatient treatment, 2 subordinate research questions will be addressed:

- 5a) What is the association between negative symptoms and initial appraisal of inpatient treatment, and how does this compare to other symptoms of schizophrenia?
- 5b) Is the association between symptoms and initial appraisal of treatment different in those that were voluntarily and involuntarily admitted?

Chapter 3: The longitudinal course of negative symptoms

3.1. Introduction

Historically, schizophrenia symptoms relating to a deficit in typical human experience were believed to increase over time as patients experience a progressive deterioration in functioning (Kraepelin, 1971). Such theories were underpinned by the idea that these symptoms were attributable to morphological changes in the brain such as lesions, ventricular enlargement, or cortical atrophy (Andreasen et al., 1982, Crow, 1980, Crow, 1985, Kraepelin, 1971). In observational studies which evaluated the progressive course of these symptoms however, the evidence suggested that negative symptoms are highly stable, with a slight tendency to improve over time in some cases (Dollfus and Petit, 1995b, Eaton et al., 1995, Fenton and McGlashan, 1991, Pogue-Geile and Harrow, 1985). Following the distinction between primary and secondary negative symptoms (Carpenter et al., 1985) it was suggested that secondary symptoms improve relatively quickly once the causes are addressed, whilst primary negative symptoms are thought to be highly persistent (Möller, 2007). Initially the evidence suggested that these primary negative symptoms may increase over time (Fenton and McGlashan, 1994, Kirkpatrick et al., 1996b), however this has since been challenged by a longer term follow-up study (Strauss et al., 2010).

Broadly defined, primary negative symptoms are those that relate to core psychopathology, rather than as a consequence of depressive, psychotic or environmental factors. However, distinguishing between primary and secondary negative symptoms can be a complex undertaking in a research context, given the difficulties in obtaining sufficient historical information on the participant and the level of clinical expertise required by the assessors. In light of this, Buchanan (Buchanan, 2007) suggested the alternative broader distinction of "persistent negative symptoms", which include negative symptoms which remain persistent after treatments for secondary negative symptoms have failed. In the NIMH-MATRICS consensus statement for negative symptoms, it was proposed that distinguishing between primary and secondary negative symptoms with persistent symptoms and control for secondary sources of negative symptoms (Kirkpatrick et al., 2006). This position has since been supported by the ISCTM consensus statement (Marder et al., 2011), and adopted by the EMA (EMA., 2012). However, to date there appears to be no consensus on how these

recommendations should be implemented (Rabinowitz et al., 2013), nor it is clear how these criteria impact how these symptoms change over time. Given the current focus on developing new interventions for negative symptoms, understanding their longitudinal course may be important for future study design, as well as being clinically informative.

In addition to identifying appropriate participants for clinical trials evaluating treatments for negative symptoms, issues around symptom assessment were also recognised as a significant issue in the MATRICS consensus statement (Kirkpatrick et al., 2006). As highlighted in section 1.9., many of the scales designed to measure negative symptoms such as the PANSS (Kay et al., 1987), SANS (Andreasen, 1983), and NSA-16 (Axelrod et al., 1993) have a number of conceptual and methodological problems, and may not be sufficiently sensitive to detect clinical meaningful change (Kirkpatrick et al., 2006). The MATRICS report suggests that a multi-item scale (i.e. the SANS) may be preferable to tools which measure individual symptoms on a single item, such as the PANSS and the BPRS (Kirkpatrick et al., 2006). A comparison of the change in negative symptoms between trials which adopted different assessment tools would allow for a comparison between the sensitivity of these scales.

The objective of this study was to examine how negative symptoms change over time, whilst exploring the impact of various study level factors. By pooling a wide variety of studies by way of meta-analysis, the aim was uncover broader trends in how these symptoms may change, as opposed to attempting to identify an estimate of effect size for a particular type of treatment, as has been done previously (ie. (Jauhar et al., 2014, Leucht et al., 2009a, Wykes et al., 2011). Many of the earlier observational studies which examined this question included inpatients, which is problematic given this population would typically receive far higher doses of antipsychotic medication, experience higher positive symptoms such as paranoid delusions (Kasckow et al., 2001), and may potentially be held in an underestimating social environment (Oshima et al., 2003), all of which may induce temporary secondary negative symptom. In addition, a number of the earlier studies included other illnesses such as schizoaffective disorder, which follows a different longitudinal course and can have poorer diagnostic stability, potentially influencing the reported change over time (Malhi et al., 2008).

Following a systematic search, we conducted a meta-analysis of the within-group mean changes in negative symptoms. Following the reasons outlined in section 1.1.3., only samples comprising exclusively of schizophrenia patients were considered. Due to the expected heterogeneity between different interventions, separate effect size estimates were calculated for each treatment type. In the next stage, a series of planned meta-regressions were

conducted to explore any impact of factors which may lead to secondary negative symptoms (Carpenter et al., 1985), and possible sources of methodological bias. Following evidence which suggests that alogia and blunted affect may be more stable than other negative symptoms (Dollfus and Petit, 1995b, Johnstone et al., 1987, Kelley et al., 1999), the changes in individual symptoms were also evaluated.

In the final part of the analysis, pre-planned sub group analyses were conducted to examine the impact of adopting inclusion criteria which specify a minimum level of negative symptoms, a maximum level of positive symptoms, and some form of restriction on either depressive symptoms or those who qualify for a diagnosis of depression. In doing so, the aim was to evaluate whether adopting such criteria reduces the change in negative symptoms course over time. Adopting such criteria may result in less change, given the impact of secondary negative symptoms should in theory be minimised. However, at present it is not clear whether this would be the case. In addition, it is possible that criteria specifying a minimum level of negative symptoms may result in a greater reduction through the removal of floor effects and a greater degree of regression to the mean.

3.2. Research questions

In order to answer the principle question "Do negative symptoms change over time and if so, how?" 4 subordinate questions were addressed:

- 1) What is the course of negative symptoms in stable schizophrenia patients over a standard study period (i.e. 10 weeks-36 months)?
- 2) Is the change of negative symptoms over time lower in samples which attempt to control for sources of secondary negative symptoms?
- 3) Does the SANS report a greater change in negative symptoms, relative to the PANSS and the BPRS?
- 4) Is there any variation in the longitudinal course of individual negative symptoms?

3.3. Methods

3.3.1. Search strategy

The systematic review was conducted following PRISMA statement guidelines (Liberati et al., 2009). An electronic search using the MEDLINE, PsycINFO, EMBASE and CENTRAL databases

was conducted dating back to 1962, which was when the Brief Psychiatric Rating Scale (BPRS) was first published (Overall and Gorham, 1962). The search was conducted on 26th April 2014 and contained three parameters. The first related to diagnosis, the second to negative symptoms, and the third an indicator that the study took place over at least two time points. The terms used for the MEDLINE electronic search are presented in table 5. Minor variations were adopted in the PsychINFO, EMBASE and CENTRAL electronic searches, depending upon the available search options. The protocol for the search procedure is presented in Appendix III.

1. Disorder	2. Symptoms	3. Treatment /Duration
*Schizophren\$	Negative symptoms	Change\$
Deficit syndrome	Reduced affect	Effect\$
	Flattened affect	Therap*
	Blunted affect	Intervention /Intervention studies
	Emotional experience	Efficacy
	Emotional expression	Impact
	Alogia	Treatment / *therapeutics
	Anhedonia	Medication
	Avolition	*Longitudinal
	Asociality	*Follow up /*Follow up studies
	Motor retardation	course
	Amotivation	Stability
	Apathy	Time
		Progress\$
		Persist\$
		Year\$ */treatment outcome

Table 5: Search terms used for the MEDLINE search

* denotes an exploded term

\$ denotes an open term

/ denotes a meshed term

A hand-search of the American Journal of Psychiatry, Acta Scandinavica Psychiatrica, British Journal of Psychiatry, Schizophrenia Bulletin, JAMA-Psychiatry, the Lancet, and Schizophrenia Research was conducted, either from 1962 or the date of first issue. Reference lists from all selected papers were hand-searched. Screening was conducted in three stages. At the first stage a title screen was conducted with the aim of removing studies that were clearly not relevant (for example, case studies, non-human studies, and those which include nonpsychiatric samples), at the second stage an abstract screen was conducted, and at the third a full paper screen was conducted. Twenty percent of the abstract screen was independently replicated by a colleague in order to test the reliability of the original screening. At the full paper screening stage, two colleagues duplicated 50% of the screening each. All discrepancies were resolved without the need for the lead supervisor to adjudicate as planned. During the data extraction phase all assessments of negative symptoms, study inclusion/exclusion criteria, demographic details, industry sponsorship, and study methodology details were extracted into a piloted extraction sheet (see Appendix IV). When necessary, corresponding authors were contacted for further information. In the case of missing standard deviations, a mean from the existing sample was imputed when possible.

3.3.2. Eligibility criteria

During the screening process studies were excluded if they were clearly not relevant, did not have repeated assessments of negative symptoms at set time points, included children (<18 years) or older adults (>65 years), contained no usable data on an exclusively schizophrenic sample, were under ten weeks in length, or did not include a follow-up assessment within 3 years. Studies which included inpatients were considered, so long as the study included one time-point where the sample was exclusively outpatients and was then followed up from a standardised time-point from this assessment. Symptoms were required to be measured on a validated scale. Qualitative studies, case reports, letters to the editor, conference abstracts, and book chapters were excluded. In order to aid translation, all articles were required to be published in a language which used Latin-based characters. Due to the analytic strategy adopted and the risk of small samples leading to biased estimates (Morris, 2000), studies with fewer than 50 participants were excluded.

3.3.4. Analysis plan

In the pooled analysis, the measure of effect size for each study was calculated using the standardised mean change (SMC; (Becker, 1988, Morris, 2000). The estimation of the variance was calculated using the large-sample approximation method recommended by Becker (Becker, 1988) (see figure 1), which has been found to provide reasonably accurate estimates provided the sample sizes are adequately sized (Morris, 2000). The estimate of the correlation between the baseline and end of study scores was set at 0.633, based upon datasets held at our research group (Priebe et al., 2007). In order to test the impact of using this estimate on the findings five other figures, ranging from P=.600 and P=.700, were also evaluated as a part of a sensitivity analysis.

Figure 1: Method used to determine the estimation of the variance.

$$\sigma_{A3}^2 = \frac{2(1-\rho)}{n} + \frac{\delta^2}{2n}.$$

(Formula taken from Morris, 2000)

In deciding the appropriate effects model to adopt, the decision was complicated by the likelihood that multiple arms of the same study would be separately eligible for inclusion. One method of addressing this which was recognised in the Cochrane Handbook (Higgins and Green), is to conduct a two-level, fixed-effect meta-analysis across arms within studies, followed by a random-effects meta-analysis across studies, as a way to account for the mix in fixed and random effects that are likely to be present. However, this model adds considerable complexity to the analysis, whilst the handbook itself acknowledges that *"in practice the difference between different analyses is likely to be trivial"* (section 16.5.5). This being the case, the method was not used, and the DerSimonion and Laird random effects model was adopted (DerSimonian and Laird, 1986). All analysis was completed using STATA version 11 (StataCorp, 2009).

In cases where multiple scales were used to measure negative symptoms, the primary outcome measure was selected. When negative symptoms were measured over more than two time points, only the baseline and the study endpoint data were selected.

In the first stage of the analysis samples were grouped according to whether the intervention involved testing second generation antipsychotics (SGA), first generation antipsychotics (FGA), adjunctive medications, non-drug interventions, or placebo/treatment as usual arms. In the next stage a series of planned univariate meta-regressions were conducted, with those found to approach significance (p<.10) entered into a multivariate model. First, the impact of length of treatment was examined in order to assess whether there was any trend over time. Next, whether there was any difference between studies which incorporated a minimum negative and maximum positive and depressive symptom threshold was tested, compared to those that did not, in order to assess whether the degree of change in negative symptoms varied dependent upon how studies dealt with factors which can cause secondary negative symptoms. Finally, the impact of assessor blinding, a minimum negative symptom inclusion criterion, and whether the study received industry sponsorship was examined.

In any examination of the change in a continuous variable over time which only includes two time points (which is a requirement for meta-analysis) the issue of regression to the mean is an important one that must be considered (Chiolero et al., 2013). This being the case, the mean negative symptoms at baseline were added to the final multivariate model to determine the degree of additional variance which may be explained by a greater reduction in negative symptoms being caused by a higher initial level of symptoms.

3.4. Results

3.4.1. Summary of articles selected

A flow diagram depicting the search strategy for studies is outlined in figure 2. Of the 9480 articles screened, 49 eligible articles were found and 41 were included in the final analysis (see table 6). From these, a total of 89 separate samples were obtained. Of the 41 studies, five came from the USA; four each from Canada, Germany and the UK; three each from India, Spain and Turkey; two each from China, France and Italy; and one each from Brazil, Finland, Israel, Nepal, Poland and Serbia. Four studies were conducted in multiple countries, with sites in Northern America, Europe and Asia. Based on 51 samples, the median of study mean illness duration was 12.4 years (range 0.6 - 27.5 years). Twenty three studies measured negative symptoms using the Positive and Negative Syndrome Scale (PANSS) (Kay et al., 1987), 14 used the Scale to Assess Negative Symptoms (SANS) (Andreasen, 1983), and four used the BPRS (Overall and Gorham, 1962). Studies which used alternative scales were screened, but none met eligibility criteria. After pooling all 89 samples, a final total of 5,944 participants were included in the meta-analysis.

Figure 2: Flow diagram outlining study selection procedure.



Authors	Year	Country	Study duration (weeks)	Outcome measure ^a	Individual symptoms reported	Intervention type ^b	n
(Addington and	2000	Canada	120	DANSS	No	TALL	65
Addington, 2000)	2000	Canada	150	1 41055			05
(Aguglia et al., 2002)	2007	Italy	52	SANS	Yes	Non-drug intervention: Psychoeducation TAU	69 66
(Alptekin et al., 2005)	2005	Turkey	52	BPRS	No	TAU	382
(Alvarez et al., 2006)	2006	Spain	48	SANS	Yes	SGA: Olanzapine	120
						SGA: Risperidone	115
(Cirici and Obiols, 2008a)	2008	Spain	46	PANSS	Yes	Non-drug intervention:	35
						Skills training group TAU	22
(Bales et al., 2009)	2009	Nepal	18	PANSS	No	TAU	30
						TAU + Betel nuts	30
(Behere et al., 2011)	2011	India	16	PANSS	No	Non-drug intervention:	34
						Yoga group Non-drug intervention: Exercise Group	31
						TAU	26
(Bhowmick et al., 2010)	2010	India	12	SANS	No	SGA: Amisulpride	40
						SGA: Olanzapine	40
(Bio and Gattaz, 2011)	2011	Brazil	26	PANSS	No	TAU	57
(Bobes et al., 2009)	2009	Spain	34	BPRS	No	SGA: Risperidone	362
(Bodkin et al., 2005)	2005	USA	12	SANS	Yes	Adjunctive: Selegiline	33
						Placebo	34
(Crawford et al., 2012a)	2012	UK	52	PANSS	No	TAU	137
						Non-drug intervention: Activity group	140
						Non-drug intervention: Art	140
(Fleischhacker et al., 2003)	2003	Multi	52	PANSS	No	therapy group SGA: Risperidone	120
						SGA: Risperidone	228
						SGA: Risperidone	267
(Gaebel et al., 2007)	2007	Germany	52	PANSS	No	SGA: Risperidone	77
						FGA: Haloperidol	74
(Gorna et al., 2008)	2008	Poland	52	PANSS	No	TAU	88
(Hirsch et al., 2002)	2002	Germany	28	PANSS	No	SGA: Ziprasidone	110
						FGA: Haloperidol	117
(Kane et al., 2011)	2011	USA	26	PANSS	No	TAU: Remained on same drug	194
						Placebo: Switched to	192
(Kane et al., 2012)	2012	USA	24	PANSS	No	Adjunctive: Armodafinil	70
						Adjunctive: Armodafinil	69
						Adjunctive: Armodafinil	71
						Placebo	70

Table 6: Summary of eligible studies

(Kaphzan et al., 2014)	2014	Israel	12	PANSS	No	Placebo	22
						Adjunctive: Entacapone	23
(Klingberg et al., 2011)	2011	Germany	52	PANSS	Yes	Non-drug intervention:	99
						CBT Non-drug intervention:	99
(Lasser et al., 2013)	2013	USA	10	SANS	No	Adjunctive: lisdexamfetamine	92
(Lecrubier et al., 2006)	2006	France	26	SANS	No	dimesylate Placebo	34
						SGA: Olanzapine	70
						SGA: Olanzapine	70
						SGA: Amisulpride	70
(Liu et al., 2014)	2014	China	16	PANSS	No	Placebo	40
						Adjunctive: Minocycline	39
(Loebel et al., 2007)	2007	India	64	PANSS	No	SGA: Ziprasidone	32
						SGA: Ziprasidone	30
(Loo et al., 1997)	1997	France	26	SANS	Yes	Placebo	72
						SGA: Amisulpride	69
(Meltzer et al., 2010)	2010	USA	52	BPRS	No	SGA: Clozapine	40
						FGA: Various first	45
(Olia at al. 2000)	2000	N 4 I.E.:	12	DANCE	No	generation drugs	50
(Olle et al., 2006)	2006	wutt	12	PANSS	NO	SGA: Ziprasidone	59
(Death at at 1000)	1000	C	52	CANC	N	SGA: Amisuipride	63
(Pach et al., 1998)	1998	Germany	52	SANS	Yes	FGA: Flupenthixol Decanoate	63
(Peet and Horrobin, 2002)	2002	UK	12	PANSS	No	Placebo	31
						Adjunctive:	32
						Eicosapentaenoic acid Adjunctive:	32
						Eicosapentaenoic acid	-
						Adjunctive: Eicosapentaenoic acid	27
(Purdon et al., 2000)	2000	Canada	54	PANSS	No	SGA: Olanzapine	21
						FGA: Haloperidol	23
						SGA: Risperidone	21
(Ravanic et al., 2009)	2009	Serbia	52	PANSS	No	FGA: Haloperidol	70
						FGA: Haloperidol	35
						FGA: Chlorpromazine	65
						FGA: Chlorpromazine	40
						SGA: Clozapine	65
						SGA: Clozapine	50
(Richardson et al., 2007)	2007	UK	38	SANS	No	TAU	46
						Non-drug intervention: Art therapy group	43
(Semiz et al., 2007)	2007	Turkey	12	SANS	No	SGA: Clozapine	97
(Schoemaker et al., 2014)	2014	Multi	12	SANS	Yes	Adjunctive: Org25935 low dose	71
						Adjunctive: Org25935 high dose	73
						Placebo	70
(Sumiyoshi et al., 2007)	2007	USA	26	BPRS	No	Adjunctive: Buspirone	30
						TAU	29
(Taiminen et al., 1997)	1997	Finland	12	PANSS	No	TAU	39
						Adjunctive: Citalopram	36

(Turkington et al., 2008)	2008	UK	78	SANS	No	Non-drug intervention: CBT	46
						Non-drug intervention: Befriending	44
(Ucok et al., 2011)	2011	Turkey	52	SANS	No	TAU	52
						TAU	41
(Voruganti et al., 2007)	2007	Canada	52	PANSS	No	SGA: Olanzapine	42
						SGA: Quetiapine	43
(Xiang et al., 2006)	2006	China	34	PANSS	No	Non-drug intervention: Community re-entry	48
(Zoccali et al., 2007)	2007	Italv	24	SANS	Yes	Non-drug intervention: Counselling Adiunctive: Lamotrigine	48 26
(, ,						Placebo	25
List of studies not included	l in the ma	in analysis du	e to insuffi	cient data			
(Adams et al., 2013)	2013	Multi	24	NSA-16	No	SGA: multiple types	130
						SGA: LY2140023	131
(Chouinard et al., 1975)	1975	Canada	12	BPRS	No	Placebo	24
						FGA: Amitriptyline Hydrochloride	24
						FGA: Perphenazine	24
(0. (2005		26	64116		FGA: Amitriptyline Perphenazine	24
(Goff et al., 2005)	2005	USA	26	SANS	Yes	Adjunctive: D-cycloserine	26
						Placebo	25
(Hayes et al., 1995)	1995	Australia	44	SANS	No	Non-drug intervention: Skills training group	n/s
						Non-drug intervention: Discussion group	n/s
(Liberman et al., 1998)	1988	USA	156	BPRS	No	Non-drug intervention: Occupational therapy	n/s
						Non-drug intervention: Skills training group	n/s
(Lieberman et al., 2013)	2013	Multi	12	SANS	No	Adjunctive: TC-5619	94
						Placebo	91
(Marder et al., 2003)	2003	USA	104	SANS	Yes	SGA: Risperidone + Skills	33
						training FGA: Haloperidol + Skills training	30
(Pinto et al., 1979)	1979	UK	78	BPRS	No	FGA: Flupenthixol decanoate	34
						FGA: Fluphenazine decanoate	30

3.4.2. Longitudinal course of negative symptoms

As indicated in the forest plot (see figure 3), in all five intervention types a significant reduction in negative symptoms was found between the baseline and follow-up. SGA's were found to result in the greatest reduction (ES 1.09, 95% CI's 0.86 to 1.32, l^2 = 95.3%) whilst the smallest change was found in placebo/treatment as usual samples (ES 0.33, 95% CI's 0.16 to 0.50, l^2 = 92.1%). Substantial heterogeneity was found, as would be expected given the clinical and methodological differences which exist between the studies. In a sensitivity analysis examining the impact of different estimates of the correlation between the baseline and end of study scores included in the model, minimal differences in the overall effect size was detected (see table 7).

study	year	duration		ES (95% CI)
drug- Augmentati Lasser Bodkin Kaphzan Peet Peet Schoemaker Schoemaker Liu Kane Kane Kane Sumiyoshi Subtotal	ion 2013 2005 2014 2002 2002 2002 2002 2002 2002 2014 2014	10 12 12 12 12 12 12 12 12 12 12	**************************************	2.96 (2.40, 3.52) 0.98 (0.60, 1.35) 0.71 (0.30, 1.11) 0.37 (-0.00, 0.75) 0.90 (0.51, 1.30) 0.54 (0.20, 0.88) 0.49 (0.14, 0.83) 1.08 (0.81, 1.36) 1.26 (0.95, 1.57) 1.13 (0.75, 1.51) 2.69 (2.03, 3.34) 0.53 (0.31, 0.75) 0.52 (0.30, 0.73) 0.43 (0.21, 0.64) 1.17 (0.71, 1.63) -0.10 (-0.41, 0.21) 0.93 (0.66, 1.21)
Drug-SGA Bhowmick Bhowmick Ollie Semiz Lecrubier Lecrubier Lecrubier Loo Hirsch Bobes Alvarez Alvarez Fleischhacker Fleischhacker Fleischhacker Gaebel Ravanic Ravanic Vorganti Vorganti Vorganti Derdon Purdon Loebel Loebel Meltzer Subtotal	2010 2006 2006 2006 2006 2006 2006 2006	12 12 12 12 26 26 26 26 28 34 48 52 52 52 52 52 52 52 52 52 52	++++++++++++++++++++++++++++++++++++++	$\begin{array}{c} 2.04 \ (1.51, 2.57) \\ 1.91 \ (1.40, 2.41) \\ 1.08 \ (0.79, 1.37) \\ 1.22 \ (0.92, 1.52) \\ 1.04 \ (0.81, 1.26) \\ 1.39 \ (1.08, 1.69) \\ 1.39 \ (1.08, 1.69) \\ 1.27 \ (0.98, 1.56) \\ 2.50 \ (2.02, 2.98) \\ 0.56 \ (0.39, 0.74) \\ 1.04 \ (0.90, 1.18) \\ 2.39 \ (2.05, 2.73) \\ 1.97 \ (1.67, 2.27) \\ 0.44 \ (0.27, 0.61) \\ 0.23 \ (0.52, 73) \\ 1.97 \ (1.67, 2.27) \\ 0.44 \ (0.27, 0.61) \\ 0.23 \ (0.52, 1.29) \\ 1.04 \ (0.28, 0.52) \\ 0.03 \ (-1, 1.29) \\ 1.06 \ (1.15, 1.29) \\ 1.66 \ (1.15, 1.29) \\ 1.66 \ (1.15, 1.29) \\ 1.66 \ (1.15, 1.29) \\ 1.66 \ (1.15, 1.98) \\ 0.52 \ (0.11, 0.93) \\ 0.14 \ (-0.24, 0.52) \\ 0.71 \ (0.35, 1.06) \\ 1.40 \ (0.94, 1.35) \\ 0.71 \ (0.40, 1.02) \\ 1.09 \ (0.87, 1.32) \\ \end{array}$
Control/TAU/Plac Bodkin Peet Schoemaker Taiminen Behere Liu Bales Cirici Kane Zoccali Bio Kane_2 Kane_2 Lecrubier Loo Sumiyoshi Richardson Aguglia Alptekin Gorna Crawford Addington Ucok Subtotal	2005 2005 2014 1997 2014 2014 2009 2008 2007 2007 2007 2007 2007 2007 2007	12 12 12 16 16 16 24 24 24 24 26 26 26 26 26 26 26 26 26 26 26 26 26	+++++ ++++++++++++++++++++++++++++++++	$\begin{array}{c} 0.73 \ (0.39, 1.07) \\ 0.52 \ (0.19, 0.86) \\ 0.94 \ (0.67, 1.21) \\ 0.88 \ (0.55, 1.21) \\ 0.16 \ (-0.20, 0.52) \\ 1.28 \ (0.88, 1.67) \\ 0.24 \ (-0.80, 0.56) \\ -0.34 \ (-0.71, 0.03) \\ 0.48 \ (0.26, 0.69) \\ -0.01 \ (-0.34, 0.33) \\ 0.39 \ (0.13, 0.64) \\ -0.41 \ (-0.54, -0.28) \\ 0.00 \ (-0.12, 0.12) \\ 1.03 \ (0.65, 1.41) \\ 1.22 \ (0.93, 1.50) \\ -0.03 \ (-0.26, 0.13) \\ 0.03 \ (0.17, 0.24) \\ 0.17 \ (0.02, 0.33) \\ 0.30 \ (0.15, 0.58) \\ 0.30 \ (0.16, 0.55) \\ 0.33 \ (0.16, 0.50) \\ \end{array}$
non-drug interver Behere Bales Cirici Xiang Richardson Aguglia Klingberg Turkington Turkington Turkington Crawford Crawford Subtotal	ntion 2011 2009 2008 2006 2006 2007 2007 2011 2011 2011 2008 2012 2012	16 16 20 34 35 52 52 52 52 78 104 104	**************************************	$\begin{array}{c} 0.63 \ (0.27, \ 1.00) \\ 0.33 \ (-0.09, \ 0.76) \\ 0.31 \ (-0.05, \ 0.67) \\ 0.38 \ (-0.68) \\ 0.42 \ (0.15, \ 0.68) \\ -0.26 \ (-0.51, \ -0.01) \\ -0.19 \ (-0.59, \ 0.21) \\ 0.62 \ (0.39, \ 0.85) \\ 0.57 \ (0.39, \ 0.76) \\ 0.43 \ (0.25, \ 0.61) \\ 0.23 \ (-0.03, \ 0.49) \\ 0.74 \ (0.45, \ 1.03) \\ 0.33 \ (0.17, \ 0.49) \\ 0.23 \ (0.17, \ 0.49) \\ 0.25 \ (0.10, \ 0.41) \\ 0.25 \ (0.10, \ 0.41) \\ 0.35 \ (0.22, \ 0.48) \end{array}$
Drug- FGA Hirsch Gaebel Pach Ravanic Ravanic Ravanic Ravanic Purdon Meltzer Subtotal NOTE: Weights a	2002 2007 1998 2009 2009 2009 2009 2000 2000 2000	28 52 52 52 52 52 52 54 104 random effects analysis	**************************************	0.58 (0.40, 0.75) 0.17 (-0.03, 0.36) 0.37 (0.07, 0.67) 0.38 (0.07, 0.63) 0.28 (0.07, 0.50) 0.38 (0.10, 0.66) 0.62 (0.39, 0.84) 0.24 (-0.20, 0.68) 0.75 (0.45, 1.04) 0.42 (0.29, 0.55)

Figure 3: Forest plot of negative symptom change by treatment type

standar discu med	in change score			
Р	Effect size	95% Cl's	²	
.600	0.572	0.462 - 0.681	93.7%	
.625	0.571	0.463 - 0.680	93.9%	
.633*	0.571	0.463 - 0.679	94.0%	
.650	0.571	0.463 - 0.678	94.1%	
.675	0.570	0.464 - 0.677	94.4%	
.700	0.569	0.464 - 0.675	94.6%	

Table 7: Impact of different imputed correlations between T1 and T2 used in the calculation of the standardised mean change score

*figure used in the model

In order to explore the heterogeneity between studies a series of meta-regressions were conducted. In the univariate analyses the scale used, the intervention type evaluated, the study duration, and a minimum threshold of negative symptoms as an inclusion criterion were all associated with a larger reduction in negative symptom (see table 8). A maximum level of positive symptoms and previous non-response to treatment as exclusion criteria were found to approach significance (p<.10), whilst other variables were non-significant. In the multivariate model, only the type of scale used and the type of intervention received remained significant. Overall, the between-model variance accounted for by the covariates was 43.88%. In a sensitivity analysis the sample-level baseline negative symptoms were added to the model which was found to be a significant predictor (B= .010, SE=.004, 95% CI's .002 to .018). However, the additional variance explained was relatively small (3.57%).

Predictor of negative	Univariate analysis				Multiv	Multivariate analysis				
symptom change	В	SE	95% C	l's	Р	В	SE	95% C	ľs	Р
Study Duration	-0.01	0.00	-0.01	0.00	.035	0.00	0.00	-0.01	0.00	.421
Scale used (compared to PANSS)					.002					
SANS	0.49	0.14	0.21	0.77		0.43	0.13	0.16	0.70	.002
BPRS	-0.12	0.26	-0.63	0.40		-0.07	0.23	-0.54	0.39	.760
Intervention (compared to SGA)					<.001					
Non-drug intervention	-0.75	0.18	-1.11	-0.38		-0.67	0.19	-1.04	-0.30	.001
TAU/Placebo	-0.76	0.16	-1.07	-0.45		-0.73	0.16	-1.06	-0.41	<.001
Drug- FGA	-0.68	0.18	-1.10	-0.25		-0.73	0.16	-0.93	-0.12	.011
Drug- Augmentation	-0.16	0.18	-0.52	-0.19		-0.21	0.18	-0.57	0.16	.265
Min negative symptoms	0.37	0.13	0.11	0.63	.007	0.10	0.12	-0.15	0.35	.413
Max positive symptoms	0.26	0.14	-0.02	0.54	.071	0.07	0.13	-0.20	0.33	.608
Industry sponsorship	0.15	0.14	-0.14	0.43	.309					
Exclusion: Prev. non-response	0.42	0.18	0.07	0.78	.019	0.05	0.17	-0.29	0.40	.765
Raters blinded to allocation ^b	-0.18	0.15	-0.47	0.12	.234					
Exc: mod levels of depression	0.13	0.17	-0.21	0.46	.452					

Table 8: Meta-regression exploring impact of clinical and socio-demographic study level factors on variability in negative symptom change

^b One study not included due to lack of data

Although a significant proportion of the studies evaluated extra-pyramidal symptoms (EPS) as part of their analysis (53.7%), only three studies specified an EPS maximum threshold as an exclusion criterion (Klingberg et al., 2011, Lasser et al., 2013, Schoemaker et al., 2014). Given the lack of data, this was not included in the meta-regression analysis. Of those studies that did report EPS, they were generally considered to be in the low range at study intake, suggesting that the impact of EPS on negative symptoms was likely to be small.

3.4.3. Change in negative symptoms reported by different assessment measures

In order to examine the impact of the different assessment tools used on negative symptom change, studies were grouped by the scale used and the effect sizes were compared. Twenty two Studies (56 samples) used the PANSS negative subscale, 15 studies (29 samples) used the SANS, and four studies (six samples) used the BPRS anergia subscale. A forest plot depicting the different estimates for each of the scales used is presented in figure 4. Studies which used the SANS found a significantly greater reduction in negative symptoms relative to those that used the PANSS or the BPRS (SANS: ES=1.02, 95% Cl's 0.77 to 1.28; PANSS: ES= 0.66, 95% Cl's 0.56 to 0.77; BPRS: ES=0.41, 95% Cl's -0.03 to 0.85).

Figure 4: Forest plot comparison negative symptom change between different assessment tools used

PANSS Bales 2009 130 Bales 2009 18 Behere 2011 16 Behere 2011 16 Bio 2011 26 Cirici 2008 20 Cirici 2009 52 Ravanic 2009 52 Cirici 2006 28 Cirici 2007 52 Cirici 2006 28 Cirici 2007 52 Cirici 2007 52 Cirici 2008 20 Cirici 2008 20			-
Anuglia 2007 52 Aquglia 2007 52 Alvarez 2006 48 Bhowmick 2010 12 Bhowmick 2010 12 Lecrubier 2006 26 Lecrubier 2007 38 Lecrubier 2007 38 Le		$\begin{array}{c} 0.36 & (0.15, 0.58) \\ 0.31 & (-0.08, 0.56) \\ 0.16 & (-0.09, 0.76) \\ 0.33 & (0.027, 1.00) \\ 0.39 & (0.13, 0.64) \\ 0.38 & (0.08, 0.68) \\ 0.33 & (0.02, 0.33) \\ 0.33 & (0.17, 0.49) \\ 0.34 & (0.02, 0.33) \\ 0.33 & (0.17, 0.49) \\ 0.35 & (0.17, 0.49) \\ 0.25 & (0.17, 0.49) \\ 0.25 & (0.17, 0.49) \\ 0.25 & (0.17, 0.49) \\ 0.25 & (0.17, 0.49) \\ 0.25 & (0.17, 0.49) \\ 0.25 & (0.17, 0.49) \\ 0.25 & (0.17, 0.49) \\ 0.30 & (0.28, 0.52) \\ 0.31 & (0.28, 0.52) \\ 0.31 & (0.28, 0.52) \\ 0.31 & (0.28, 0.52) \\ 0.31 & (0.28, 0.52) \\ 0.41 & (0.27, 0.61) \\ 0.30 & (-0.12, 0.12) \\ 0.56 & (0.40, 0.75) \\ 0.56 & (0.40, 0.75) \\ 0.56 & (0.49, 0.74) \\ 0.30 & (-1.19, 0.30) \\ 0.51 & (0.39, 0.74) \\ 1.28 & (0.38, 1.67) \\ 0.71 & (0.30, 0.74) \\ 1.28 & (0.38, 1.67) \\ 0.71 & (0.30, 0.74) \\ 1.28 & (0.38, 1.67) \\ 0.71 & (0.30, 0.74) \\ 1.28 & (0.39, 0.74) \\ 1.28 & (0.39, 0.74) \\ 1.28 & (0.39, 0.74) \\ 1.28 & (0.39, 0.74) \\ 1.28 & (0.39, 0.74) \\ 1.28 & (0.39, 0.74) \\ 1.28 & (0.39, 0.74) \\ 1.28 & (0.39, 0.74) \\ 1.28 & (0.39, 0.74) \\ 1.28 & (0.39, 0.74) \\ 1.28 & (0.39, 0.74) \\ 1.28 & (0.39, 0.74) \\ 1.28 & (0.39, 0.74) \\ 1.28 & (0.39, 0.74) \\ 1.28 & (0.39, 0.74) \\ 1.28 & (0.39, 0.74) \\ 1.28 & (0.29, 0.88) \\ 0.52 & (0.19, 0.28) \\ 0.52 & (0.11, 0.93) \\ 0.33 & (0.04, 0.63) \\ 0.54 & (0.20, 0.68) \\ 0.52 & (0.39, 0.74) \\ 0.52 & (0.39, 0.74) \\ 0.52 & (0.39, 0.74) \\ 0.52 & (0.39, 0.75) \\ 0.38 & (0.77, 0.50) \\ 0.53 & (0.31, 0.75) \\ 0.53 & (0.31, 0.75) \\ 0.53 & (0.31, 0.75) \\ 0.54 & (0.22, 0.64) \\ 0.55 & (0.31, 0.75) \\ 0.54 & (0.25, 0.61) \\ 0.57 & (0.38, 0.76) \\ 0.51 & (0.41, 0.61) \\ 0.51 & (0.51, 0.51) \\ 0.51 & (0.$	$\begin{array}{c} 1.18\\ 1.09\\ 1.09\\ 1.09\\ 1.09\\ 1.08\\ 1.16\\ 1.13\\ 1.21\\ 1.22\\ 1.19\\ 1.222\\ 1.199\\ 1.222\\ 1.199\\ 1.222\\ 1.085\\ 1.085\\ 1.014\\ 1.110\\ 1.060\\ 1.072\\ 1.025\\ 1.18\\ 1.20\\ 1.22\\ 1.05\\$
Turkington 2008 78 Ucok 2011 156 Ucok 2011 156 Zoccali 2007 24 Bodkin 2005 12 Bodkin 2005 12 Bodkin 2005 12 Schoemaker 2014 12 Schoemaker 2014 12 Subtotal BPRS	•	$\begin{array}{c} 0.62 \ (0.39, 0.85)\\ 0.03 \ (-0.17, 0.24)\\ 2.39 \ (205, 2.73)\\ 1.97 \ (1.67, 2.27)\\ 1.91 \ (1.40, 2.41)\\ 1.91 \ (1.40, 2.41)\\ 1.91 \ (1.40, 2.41)\\ 1.91 \ (1.40, 2.41)\\ 1.91 \ (1.40, 2.41)\\ 1.92 \ (1.92 \ (1.91)\\ 1.92 \ (1.92 \ (1.91)\\ 1.92 \ (1.92$	1.18 1.19 1.13 0.975 1.13 1.15 1.07 1.14 1.15 1.07 1.14 1.15 1.08 1.16 1.16 1.15 1.16 1.15 1.16 1.15

symptoms increase over time symptoms reduce over time

3.4.4. Comparing control arms between drug and non-drug trials

The larger effect size improvements in second generation antipsychotics and adjunctive medications, relative to other types of treatment, appear to contradict earlier findings which suggest minimal advantages of these types of treatments (Leucht et al., 2009b, Tuominen et al., 2005). In order to examine whether this reduction over time is a feature of the studies they were part of, as opposed to a specific action of the intervention itself, in a post-hoc analysis the effect size reductions in control arms of studies which were and were not part of an SGA or adjunctive drug study were compared. In total, 12 non- SGA or adjunctive drug study control arms and 11 SGA or adjunctive drug study control arms were compared, with the findings presented in figure 5. Significantly larger reductions in the control arms of studies which were part of an SGA or adjunctive drug study, relative to those control arms which formed part of non-drug studies were detected (ES=0.65, 95% CI 0.56 to 0.75, l^2 =86.9%, in comparison to ES=0.15, 95% CI 0.08 to 0.21, l^2 =60.8%).

Figure 5: Forest plot comparing the change in second generation antipsychotic drug studies and non-drug study control arms.

study	year	duration	ES (95% CI)	Weigh
Non-Drug Stu	dy			
Addington	2000	130	0.36 (0.15, 0.58)	5.81
Aguglia	2007	52	0.03 (-0.17, 0.24)	6.43
Alptekin	2005	52	0.12 (-0.01, 0.25)	16.23
Bales	2009	18	0.24 (-0.08, 0.56)	2.63
Behere	2011	16	0.16 (-0.20, 0.52)	2.11
Bio	2011	26	0.39 (0.13, 0.64)	4.16
Cirici	2008	20	-0.34 (-0.71, 0.03)	1.99
Crawford	2012	104	0.17 (0.02, 0.33)	11.56
Gorna	2008	52	-0.05 (-0.25, 0.15)	7.20
Richardson	2007	38	-0.23 (-0.60, 0.13)	2.07
Ucok	2011	156	0.38 (0.11, 0.65)	3.82
Ucok	2011	156	0.30 (0.06, 0.55)	4.59
Subtotal (I-so	uared	= 60.8%, p = 0.003)	0.15 (0.08, 0.21)	68.60
Drug Study				
Drug Sludy Bodkin	2005	10		0 40
Kono	2005	12		Z.43
Kanbzon	2012	24		1.00
Raprizari	2014	12		1.90
	2000	20		1.92
Liu	2014	10		1.80
LUU	1997	20		3.39
Peel	2002	12		2.40
Sumivoshi	2014	12		3.11
Jurniyosni	2007	∠0 10		2.82
i aiminen	1997	12		2.49
	2007	24		2.44
Subtotal (I-so	luared	= 86.9%, p = 0.000)	0.65 (0.56, 0.75)	31.40
Heterogeneity	/ betwe	en groups: p = 0.000		
Overall (I-squ	ared =	87.9%, p = 0.000)	0.31 (0.25, 0.36)	100.0

3.4.5. Examination of individual negative symptoms

In 18 samples over nine studies the change in individual negative symptoms were also evaluated, with the results presented in figure 6. In seven studies the individual items were rated using the SANS, and in two studies symptoms were rated using the PANSS. Scores from different scales were combined using the method proposed by Lyne and colleagues (Lyne et al., 2012) (see section 1.9.2.). A significant reduction was found in all four of the symptoms measured (blunted affect, alogia, avolition and anhedonia). Of the 4, alogia appeared to reduce the least (ES= 0.64, 95% Cl's 0.45 to 0.83, l^2 = 90.4%) and avolition the most (ES= 0.77, 95% Cl's 0.53 to 1.01, l^2 =93.0%), however the difference between the symptoms were minimal.



3.4.6. Comparing those that do and do not adopt criteria in line with the diagnosis of persistent negative symptoms

Twelve samples over five different studies adopted eligiblity criteria which specified a minimum level of negative symptoms, a maximum positive symptom eligiblity critierion, and some form of restriction on the severity of depressive symptoms. The different studies and the criteria they adopted are presented in table 9. In the comparison of studies which did and did not adopt such criteria, a slightly larger reduction in negative symptoms over time was detected in patients that did (Criteria adopted ES= 0.85, 95% Cl's 0.56 - 1.08, l^2 = 90.2%; criteria

not adopted ES= 0.63, 95% CI's 0.51 – 0.75, l^2 = 94.2%). As clear from the forest plot presented in figure 7, the difference between the two groups appeared to be largely attributable to one observational study (Lasser et al., 2013). When this study was omitted, the effect size change was more similar to those that do not adopt the criteria (ES=0.70, CI's 0.54 - 0.85, l^2 = 79.1%) Overall, these findings corroborate with the results from the meta-regression (see table 8) which together suggest that adopting such criteria does not appear to reduce the reduction in negative symptoms over time.

Table 9: Summary of different symptom inclusion/exclusion criteria

Stud	у	Year	Scales used	Symptom inclusion and exclusion criterion
1	Bodkin	2005	SANS, BPRS	Inclusion: SANS >12, two subscales >3. Exclusion: ≥5 on the BPRS thinking disturbance subscale, mood disorder
2	Kane	2012	PANSS	Inclusion: ≥15 on the negative subscale. Exclusion: positive items ≥4, >10 Calgary
3	Lasser	2013	SANS	Inclusion: >54 on SANS 18, <2 on <1 SANS global score. Exclusion: >7 Calgary
4	Schoemaker	2014	PANSS	Inclusion: ≥4 on at least 3 items, ≥21 overall on the negative subscale. Exclusion: No more than ≤19 on the PANSS positive, <9 Calgary
5	Klingberg	2011	PANSS	Inclusion: At least 1 PANSS negative item ≥4. Exclusion: no positive or depressive item ≥6

PANSS: Positive and Negative Syndrome Scale; SANS: Scale to Assess Negative Symptoms of Schizophrenia; CALGARY: Calgary depression scale for psychosis

	study	year	duration			ES (95% CI)
	Addington Aguglia Aguglia Alptekin Alvarez Bales Behere Behere Behere Behere Bowmick Bio Obes Cirici Cirici Crawford Crawford Crawford Crawford Crawford Crawford Crawford Crawford Crawford Crawford Fleischhacker Fleischhacker Fleischhacker Fleischhacker Fleischacker Gaebel Gaebel Garna Hirsch Hirsch Hirsch Hirsch Kaphzan Lecrubier Lecrubier Lecrubier Lecrubier Lecrubier Lecrubier Lecrubier Lecrubier Lecrubier Lecrubier Cabel Loobel Loo Meltzer Meltzer Ollie Pach Peet Peet Peet Peet Peet Peet Peet Pee	20007 20077 2005 2006 2009 20111 20110 20112 20112 20102 20022 20012 20012 20012 20003 20007 20072 2008 20092 20012 20003 20072 2008 20092 20012 20003 20072 2008 20092 20092 20014 20066 20099 20092 20072 20082 20092 20072 20082 20097 200777 20077 20077 20077 20077 2007772 2007772 2007772 2007772 2007772 2007772 2007772 2007772 2007772 2007772 2007772 2007772 2007772 2007772 2007772 20077720777720777720777720777707777707777707777707777077777077777	130 52 <td></td> <td>- -</td> <td>0.36 (0.15, 0.58) 0.62 (0.39, 0.85) 0.03 (-0.17, 0.24) 0.12 (-0.01, 0.25) 2.39 (2.05, 2.73) 1.97 (1.67, 2.27) 0.31 (-0.05, 0.67) 0.24 (-0.08, 0.56) 0.16 (-0.20, 0.52) 0.33 (-0.09, 0.76) 0.63 (0.27, 1.00) 1.91 (1.40, 2.41) 2.04 (1.51, 2.57) 0.39 (0.13, 0.64) 1.04 (0.90, 1.18) -0.34 (-0.71, 0.03) 0.38 (0.08, 0.68) 0.17 (0.02, 0.33) 0.33 (0.17, 0.49) 0.25 (0.10, 0.41) 0.44 (0.27, 0.61) 0.23 (0.12, 0.34) 0.40 (0.28, 0.52) 0.17 (-0.03, 0.36) 0.03 (-0.16, 0.22) -0.05 (-0.25, 0.15) 0.58 (0.40, 0.75) 0.56 (0.39, 0.74) -0.41 (-0.54, -0.28) 0.00 (-0.12, 0.12) 0.71 (0.30, 1.11) 0.37 (-0.00, 0.75) 1.39 (1.08, 1.69) 1.27 (0.98, 1.56) 1.03 (0.76, 1.29) 1.03 (0.76, 1.29) 1.03 (0.76, 1.29) 1.03 (0.76, 1.29) 1.03 (0.76, 1.29) 1.22 (0.92, 1.50) 0.71 (0.33, 1.06) 1.40 (0.94, 1.85) 2.50 (2.02, 2.98) 1.22 (0.93, 1.50) 0.71 (0.45, 1.04) 1.28 (0.88, 1.67) 0.71 (0.44, 0.83) 0.90 (0.51, 1.30) 0.75 (0.45, 1.04) 1.28 (0.28, 1.50) 0.71 (0.45, 1.02) 0.75 (0.45, 1.04) 1.28 (0.27, 0.67) 0.52 (0.19, 0.86) 0.44 (-0.24, 0.52) 0.24 (-0.20, 0.88) 0.14 (-0.24, 0.52) 0.24 (-0.20, 0.88) 0.14 (-0.24, 0.52) 0.24 (0.07, 0.57) 0.28 (0.07, 0.57) 0.28 (0.07, 0.57) 0.28 (0.07, 0.50) -0.23 (-0.60, 0.13) -0.19 (-0.41, 0.21) 0.08 (-0.39, 0.24) 0.88 (0.55, 1.21) 1.13 (0.75, 1.51) 0.23 (-0.60, 0.13) -0.24 (-0.20, 0.88) 0.14 (-0.24, 0.52) 0.24 (0.03, 0.34) 0.74 (0.45, 1.03) 0.38 (0.11, 0.65) 0.30 (0.06, 0.55) 1.48 (1.07, 1.89) 1.56 (1.15, 1.98) 0.42 (0.15, 0.68) -0.26 (-0.51, -0.01) -0.26 (-0.51, -</td>		- -	0.36 (0.15, 0.58) 0.62 (0.39, 0.85) 0.03 (-0.17, 0.24) 0.12 (-0.01, 0.25) 2.39 (2.05, 2.73) 1.97 (1.67, 2.27) 0.31 (-0.05, 0.67) 0.24 (-0.08, 0.56) 0.16 (-0.20, 0.52) 0.33 (-0.09, 0.76) 0.63 (0.27, 1.00) 1.91 (1.40, 2.41) 2.04 (1.51, 2.57) 0.39 (0.13, 0.64) 1.04 (0.90, 1.18) -0.34 (-0.71, 0.03) 0.38 (0.08, 0.68) 0.17 (0.02, 0.33) 0.33 (0.17, 0.49) 0.25 (0.10, 0.41) 0.44 (0.27, 0.61) 0.23 (0.12, 0.34) 0.40 (0.28, 0.52) 0.17 (-0.03, 0.36) 0.03 (-0.16, 0.22) -0.05 (-0.25, 0.15) 0.58 (0.40, 0.75) 0.56 (0.39, 0.74) -0.41 (-0.54, -0.28) 0.00 (-0.12, 0.12) 0.71 (0.30, 1.11) 0.37 (-0.00, 0.75) 1.39 (1.08, 1.69) 1.27 (0.98, 1.56) 1.03 (0.76, 1.29) 1.03 (0.76, 1.29) 1.03 (0.76, 1.29) 1.03 (0.76, 1.29) 1.03 (0.76, 1.29) 1.22 (0.92, 1.50) 0.71 (0.33, 1.06) 1.40 (0.94, 1.85) 2.50 (2.02, 2.98) 1.22 (0.93, 1.50) 0.71 (0.45, 1.04) 1.28 (0.88, 1.67) 0.71 (0.44, 0.83) 0.90 (0.51, 1.30) 0.75 (0.45, 1.04) 1.28 (0.28, 1.50) 0.71 (0.45, 1.02) 0.75 (0.45, 1.04) 1.28 (0.27, 0.67) 0.52 (0.19, 0.86) 0.44 (-0.24, 0.52) 0.24 (-0.20, 0.88) 0.14 (-0.24, 0.52) 0.24 (-0.20, 0.88) 0.14 (-0.24, 0.52) 0.24 (0.07, 0.57) 0.28 (0.07, 0.57) 0.28 (0.07, 0.57) 0.28 (0.07, 0.50) -0.23 (-0.60, 0.13) -0.19 (-0.41, 0.21) 0.08 (-0.39, 0.24) 0.88 (0.55, 1.21) 1.13 (0.75, 1.51) 0.23 (-0.60, 0.13) -0.24 (-0.20, 0.88) 0.14 (-0.24, 0.52) 0.24 (0.03, 0.34) 0.74 (0.45, 1.03) 0.38 (0.11, 0.65) 0.30 (0.06, 0.55) 1.48 (1.07, 1.89) 1.56 (1.15, 1.98) 0.42 (0.15, 0.68) -0.26 (-0.51, -0.01) -0.26 (-0.51, -
	Bodkin Bodkin Kane Kane Kingberg Klingberg Lasser Schoemaker Schoemaker Schoemaker Subtotal	2005 2005 2012 2012 2012 2012 2011 2011	12 24 24 24 24 52 52 10 12 12 12	┿┿╪╪╪╸╴╴╴┿╸ ╵		$\begin{array}{c} 0.98 & (0.60, 1.35) \\ 0.73 & (0.39, 1.07) \\ 0.52 & (0.30, 0.73) \\ 0.43 & (0.21, 0.64) \\ 0.48 & (0.26 & 0.69) \\ 0.53 & (0.31, 0.75) \\ 0.43 & (0.25, 0.61) \\ 0.57 & (0.38 & 0.76) \\ 2.96 & (2.40, 3.52) \\ 1.08 & (0.81, 1.36) \\ 1.26 & (0.95, 1.57) \\ 0.94 & (0.67, 1.21) \\ 0.85 & (0.62, 1.08) \end{array}$
_	Overall NOTE: Weights	are fror	m random effects analysis	•		0.66 (0.56, 0.77)
				I I I I -1 0 1 2	1 3	

Figure 7: Comparison of trial arms which did and did not specify a minimum negative and maximum negative and depressive symptom criterion.

3.4.7. Examination of publication bias

Publication bias was examined by intervention group using Duval and Tweedie's 'trim and fill' method via the STATA *metatrim* command (Duval and Tweedie, 2000). Possible bias was detected in the non-drug intervention group and the adjunctive medication group, suggesting a small over-estimation of the effect size. Two imputed studies in the non-drug intervention group and one in the adjunctive medication group reduced the effect sizes, but both remained significant (Non drug group: ES= 0.27, 95% Cl's 0.12 - 0.41; adjunctive medication group: ES= 0.58, 95% Cl's 0.38 - 0.79). However, given the fixed effect differences between samples from the same study resulted in within-group clustering, it is important to recognise that this reported bias is difficult to interpret, and may be an artefact of the study design.

3.4.8. Eligible studies not pooled into the main analysis

In addition to those pooled, eight studies were found to be eligible but could not be included in the main analysis (see table 6). In line with the main results, 11 samples found some form of reduction in negative symptoms from baseline to end of study, five saw no change, and in two the change was not specified.

3.5. Discussion

3.5.1. Main findings

The meta-analysis provided a clear result; negative symptoms of schizophrenia tend to improve significantly in an outpatient setting. A reduction in negative symptoms was found across all intervention types, with the effect sizes ranging from small to large depending upon the intervention type. In addition, a significant reduction was found in all four of the separate negative symptoms examined, representing both experiential and expressive features of the disorder. Whilst substantial heterogeneity was present in the sample, a series of planned meta-regressions indicated that there was no difference in the reduction between studies which did and did not exclude participants with at least moderate levels of positive or depressive symptoms. In addition, study-level methodological differences such as whether assessors were blinded, the symptom eligibility criteria, and whether the study received industry sponsorship also did not appear to influence the result. The largest changes in negative symptoms were found in trial arms where participants were randomised to receive SGA's or adjunctive medications. However, in these studies a significantly larger reduction was also found in the control arms of studies, suggesting there may be an unspecified feature of the trial itself which may drive at least part of this change. In studies which used the SANS (Andreasen, 1983) as opposed to the PANSS (Kay et al., 1987) or the BPRS (Overall and Gorham, 1962) a significantly larger change in negative symptoms was found, highlighting the impact of adopting different methods of negative symptom assessment.

3.5.2. Strengths and limitations.

One of the main strengths of the study is that, despite the broad range of study interventions considered, the findings are consistent. Of the 89 study arms included, only one found a clear significant increase in symptoms. In this case, the sample was part of a continuation study where patients who had previously responded well to their SGA medication were then switched to a placebo (Kane et al., 2011). When testing for the effect of regression to the mean, adding baseline negative symptoms to the multivariate model appeared to add relatively little additional explanatory power of the variance (3.87%), suggesting the findings are relatively robust to this effect. A further strength of this study is that despite the broad study inclusion criteria, removing samples which included inpatients at baseline, and other psychotic diagnoses, meant the participant inclusion criteria were relatively stringent in comparison to other observational studies that have looked at how negative symptoms change over time (Dollfus and Petit, 1995b, Eaton et al., 1995, Fenton and McGlashan, 1991, Pogue-Geile and Harrow, 1985).

One limitation of the study is that due to the variance estimation method adopted, a number of otherwise eligible studies were excluded due to them containing an insufficient number of participants (Morris, 2000). However, given there is evidence to suggest that smaller studies can often present larger effect sizes (i.e. (Zhang et al., 2013), our findings may have led to a more conservative estimate, with the reduction in negative symptoms possibly being even more substantial than reported here. Another limitation is that, despite the fairly large number of studies included in the analysis (41), the final sample of 5,944 patients was smaller than what was originally anticipated. This was due to the fact that a number of the larger studies in the field either contained inpatients (Lieberman et al., 2005), used a single item to measure negative symptoms rather than a validated scale (Dossenbach et al., 2004), or included patients with other psychotic disorders.

Another limitation is that due to the study design it is difficult to determine whether the changes detected are attributable to an improvement in primary negative symptoms, persistent negative symptoms, or secondary symptoms which one would expect to improve relatively quickly over time once the causes are addressed. No difference in the change in negative symptoms over time was detected between those that did and did not include some form eligibility criteria required to determine persistent negative symptoms (i.e. including a minimum level of negative symptoms, and a maximum level of positive and depressive symptoms in the eligibility criteria; (Buchanan, 2007). Consistent with the results by Stauffer et al. (Stauffer et al., 2012), these findings suggest that the change in negative symptoms are the same in samples that have attempted to minimise the presence of secondary negative symptoms, relative to those that have not. However, examining the change in symptoms between studies with very different criteria adopted is not a recognised method to determine persistent negative symptoms, and cannot provide any insight related to primary negative symptoms (Carpenter and Kirkpatrick, 2015). Consequently, further work is required to disentangle the both how primary negative symptoms change over time, and what the impact of different predominant inclusion criteria may be.

Another important consideration is that it is difficult to determine the nature of the change detected. In the meta-regression the duration of the study was not found to be a predictor of greater negative symptom reduction which suggests that these symptoms do not gradually improve over time. This finding is supported by longitudinal studies over very long periods which have found that negative symptoms follow a fluctuating course and do not typically lead towards recovery (Jääskeläinen et al., 2013, Strauss et al., 2010). One possible reason for the detected improvement may be the impact of the participating in multiple assessments. In subsequent assessments the interviewee may be more relaxed if a rapport has been previously established with the interviewer, and they are likely to have a much better idea of the nature of the questions they would be asked. This could potentially lead to participants being more expressive in the interview, resulting in a less severe symptom rating.

It is important to note that given the limitations of the within-group design, the results should not be interpreted as an assessment of treatment effectiveness. As highlighted earlier, a series of meta-analyses have been conducted to evaluate various treatments for schizophrenia (Jauhar et al., 2014, Kurtz and Mueser, 2008, Leucht et al., 2009a, Wykes et al., 2011) using between-group designs which offer a more appropriate assessment for treatment effectiveness. Overall, these reviews have detected relatively limited benefits for negative symptoms, contrasting with the large within-group effect sizes noted here. At least part of this

difference could be attributable to the fact that effect sizes in control arms of studies evaluating drug treatments appear to be significantly larger in comparison to non-drug studies, suggesting there is something inherent to the methodologies employed which makes drug studies more likely to detect symptoms improvements.

Finally, due to the lack of available data, "no-medication" as a therapeutic option could not be evaluated, meaning it is not clear whether this reduction over time would also occur in nonmedicated patients. Given there is evidence to suggest that patients who do not immediately relapse upon termination of their antipsychotic regimen may experience improved global functioning over time (Harrow and Jobe, 2007), it is conceivable that improvements in this participant group may be possible also.

3.5.4. Summary

Based on the available data of almost 6,000 outpatients, negative symptoms of schizophrenia do not tend to be stable or deteriorate, but are instead likely to improve over time. This finding offers a further critique of the historical understandings of schizophrenia originally proposed by Kraepelin and Bleuler, and instead provides further support to the recovery model which suggests the functioning of patients with a diagnosis of schizophrenia can improve over time (Warner, 2009). Overall, these findings suggest that negative symptoms may not be as resistant to change as what has previously been assumed, and perhaps offer new hope to those who may experience such symptoms.

Chapter 4. Association between negative symptoms and positive and depressive symptoms after adopting different study inclusion criteria

4.1. Introduction

In the previous chapter it was found that negative symptoms of schizophrenia appear to significantly reduce in stable outpatients, including in samples which exclude patients with high depressive and positive symptoms. Whilst conducting a meta-analysis may be helpful in understanding broad trends regarding how symptoms may change over time, inherent to the study design exists a number of limitations. Most significantly, using heterogenous study-level criteria is not a recognised method to control for secondary sources of symptoms (Carpenter and Kirkpatrick, 2015), and as such it is unclear to what extent any change in negative symptoms detected may be secondary, rather than primary in origin.

Differentiating between symptoms that are a direct manifestation of the disorder itself, as opposed to being a consequence of positive symptoms, depressive symptoms or medication side effects is seen as an important issue in the testing of new therapeutics for negative symptoms (Kirkpatrick et al., 2006, Marder et al., 2013, Marder et al., 2011). Whilst primary negative symptoms of schizophrenia are typically seen as highly enduring, those resulting from secondary sources are thought to improve relatively quickly if the causes of these symptoms can be addressed (Möller, 2007). Consequently, it is the chronic, persistent negative symptoms, as opposed to transient secondary symptoms, which are regarded as an unmet therapeutic need (Kirkpatrick et al., 2006). To determine the effectiveness of treatments for negative symptoms, it is important to filter out those that would experience improvements due to changes in secondary sources of these symptoms.

Of the different methods proposed to minimise the impact of transient secondary negative symptoms, the criteria of persistent negative symptoms (Buchanan, 2007) is advocated by the NIMH-MATRICS consensus statement (Kirkpatrick et al., 2006). Persistent negative symptoms are defined by the presence of at least a moderate severity of negative symptoms; low positive symptoms, depressive symptoms, and EPS; and evidence of clinical stability. The main differences between these criteria and the deficit syndrome is that less historical information is required (evidence of clinical stability is only recommended for six months, as opposed required for one year), and the distinction between primary and secondary negative symptoms

is dropped with maximum limits to depressive, positive and EPS proposed as an alternative. By using these criteria it means that persistent negative symptoms can be attributable to secondary, as well as primary causes. However, the stability componant ensures that symptoms are not transitory in nature, whilst the maximum EPS, positive and depressive symptoms should theoretically limit the impact of possible causes of secondary negative symptoms.

While there appears to be a consensus that participants should be excluded from negative treatment clinical trials if they are either in the acute phase of the disorder, or present with notable EPS or depressive symptoms, there is still some debate regarding whether those that present with positive symptoms should also be omitted (Marder et al., 2013). Studies which specify maximum positive and minimum negative symptom levels are considered to adopt predominent eligiblity criteria, whilst studies which only specify a minimum level of negative symptoms are defined are considered to adopt prominent eligiblity criteria. Adopting predominent eligiblity criteria can be considered problematic given the association between positive and negative symptoms is relatively low (Horan et al., 2011), they have different neurobiological origins (Berman et al., 1997), follow different trajectories (Ventura et al., 2004), and impact functional outcomes independently (Rabinowitz et al., 2012). In addition, there are concerns that participants that present with negative, but not positive symptoms may only consistitute a relatively small sample of patients, distinct from the broader population that such treatments would be designed for. However, by not specifying a maximum level of positive symptoms any issues related to the pseudospecificity problem may remain.

Aside from whether it is more appropriate to adopt prominent or predominant symptom criteria in negative symptom trials, the issue is further complicated by the fact that there appears to be no clear consensus on how these criteria should be operationalised. For example, in the systematic review completed in chapter 3, three of the 41 selected studies included contraints on the minimum level of negative symptoms, and eight implimented both a minimum negative and maximum positive symptom inclusion criterion (Alvarez et al., 2006, Bodkin et al., 2005, Cirici and Obiols, 2008b, Kane et al., 2012, Kaphzan et al., 2014, Klingberg et al., 2011, Lasser et al., 2013, Lecrubier et al., 2006, Loo et al., 1997, Olie et al., 2006, Schoemaker et al., 2014). The criteria were specified using the PANSS (Kay et al., 1987), SANS (Andreasen, 1983), SAPS (Andreasen, 1984), or a mixture of these scales. In all 11 cases different inclusion and exclusion criteria were adopted. In a recent study by Dunayevich which explored the impact of different inclusion criteria (Dunayevich et al., 2014), a further five

different methods which used the PANSS to determine dominance or predominance were identified (Kinon et al., 2006, Moller et al., 2004, Rabinowitz et al., 2013, Riedel et al., 2005, Stauffer et al., 2012). In all, the 16 studies identified used 16 different ways to determine either the prominance or predominence of negative symptoms. Furthermore, it is important to note that this is not an exhaustive list, with a literature search suggesting that other major trials may have adopted different critiera again (Buchanan et al., 2007, Priebe et al., 2013b).

Adopting such a broard range of dominant and predominant inclusion criteria in negative symptom trials could be considered problematic. Rabinowitz (Rabinowitz et al., 2013) found that different symptom inclusion and exclusion criteria were found to result in very large differences in the amount of potentially eligible participants, ranging from 8.1%-62.% of the pooled sample according to different prominent symptom criteria, and 10.2%-50.2% criteria for predominant criteria. These findings have been replicated using the CATIE trial data (Lieberman et al., 2005), which found inclusion rates between 6%-37% (Dunayevich et al., 2014). These findings suggest that restrictive symptom inclusion criteria can result in a substantially smaller potential pool from which potential participants can be recruited from. As Dunayevich and Rabinowitz highlight (Dunayevich et al., 2014, Rabinowitz et al., 2013), this is likely to result in additional challenges in trial recruitment, and prehaps more concerningly, may result in the study samples being less generalisable to the clinical population who such treatments may be appropriate for. Dunayevich and colleagues (Dunayevich et al., 2014) found that correlations between the changes in negative and positive symptoms were largely consistent regardless of the severity of exclusion criteria used, which suggests that more restrictive criteria may not reduce the association as much as originally presumed. However, Rabinowitz (Rabinowitz et al., 2013) found larger treatment effects for negative symptoms in samples that included participants with both prominent negative and positive symptoms, relative to samples with predominant negative symptoms. These findings merit further consideration as such work may pave the way to appropriately standardising how such criteria are adopted in clinical trials.

The aim of this study was to compare how different eligibility criteria impact the association between negative symptoms, and positive and depressive symptoms, whilst considering their effect on the potential sample pool. This work builds upon the work by Dunayevich (Dunayevich et al., 2014), but with a number of important differences. Firstly, given depressive symptoms have been found to both mimic and cause negative symptoms (see section 1.9.1.2. for a review), it is important to determine whether different eligibility criteria currently used influence the relationship between depressive and negative symptoms. Secondly, in the Dunayevich study the presence of blunted affect, avolition and impairments in conversation

flow were required in addition to the symptom levels specified by the different study criteria. While this may follow the EMA recommendations for negative symptom trials (EMA., 2012), implementing such criteria homogenises the subsamples and makes it difficult to evaluate the impact of the criteria which are being evaluated. Third, in the Dunayevich study the standard PANSS negative subscale was used, despite concerns that the latter formulation includes items which measure cognitive symptoms (Blanchard et al., 2011). In order to address this concern, in the present study the alternative negative subscale configuration proposed by Marder was adopted (Marder et al., 1997). Finally, Dunayevich adopted the last observation carried forward method of as a way of dealing with missing data, which may under-report the degree of change over time given the methods assumes responses remain constant (Hamer and Simpson, 2009). In this study, multi-level modelling was adopted.

4.2. Research Questions

- What is the change in negative symptoms over time in stable outpatients, after excluding participants with prominent positive and depressive symptoms?
- 2) What is the impact of adopting different inclusion and exclusion criteria on the proportion of eligible participants?
- 3) What is the impact of adopting different symptom inclusion and exclusion criteria on the association between negative symptoms, and positive and depressive symptoms?

4.3. Methods

4.3.1. Sample

The data used in this analysis was taken from the DIALOG Study, which is a cluster randomised controlled trial designed to evaluate the effectiveness of a computer-mediated intervention to structure patient-clinician dialogue (Priebe et al., 2007). The study took place in 6 European countries; Granada, Spain; Groningen, The Netherlands; London, United Kingdom; Lund, Sweden; Mannheim, Germany; and Zurich, Switzerland. All participants were community psychiatric outpatients, in receipt of at least 3 months of continuous care from the same service. All participants were aged 18-65, required to be capable of providing informed consent, had a sufficient understanding of the first language from the host country, and were

routinely meeting their keyworker every 2 months with the plan of continuing this for the next 12 months. Participants with either a severe organic psychiatric illness or primary substance abuse disorder were excluded. A more detailed description of the intervention is outlined elsewhere (Priebe et al., 2007). Various outcome measures, including symptoms were measured at baseline and at follow-up 12 months later. In DIALOG study the sample included patients diagnosed with schizophrenia-related disorders such as schizoaffective and schizophreniform disorders, however for the purposes of this study they were excluded from this investigation.

4.3.2. Scales

Socio-demographic details were obtained using the MANSA (Priebe et al., 1999). Symptoms of schizophrenia were measured using the PANSS (Kay et al., 1987), a full summary of which is presented in section 1.9.

Whilst the majority of the original studies which used the PANSS to define their symptom inclusion and exclusion criteria used the standard negative subscale formulation, in one case a modified version was adopted (Klingberg et al., 2011). One significant limitation of the standard PANSS negative subscale is that it includes items which are now recognised to relate to cognitive deficits, rather than what we currently understand negative symptoms to constitute (Blanchard et al., 2011). Consequently, the PANSS Marder reconfiguration of the negative symptom subscale was adopted (Marder et al., 1997). This alternative structure replaces the abstract thinking and stereotypical thinking items with motor retardation and active social avoidance. In cases where more than 1 item was missing from each subscale, the summary score from removed from the analysis.

4.3.3. Prominent and predominant symptom inclusion criteria under evaluation:

The minimum negative symptom and maximum positive symptom inclusion and exclusion criteria evaluated in this study were either identified in the final sample of the systematic review completed in chapter 3, or were used in the analysis completed by Dunayevich (Dunayevich et al., 2014). Prominent symptom criteria refer to utilising a minimum threshold of negative symptoms, whilst predominant symptom criteria refer to utilising both a minimum negative symptoms criterion, and a maximum level of positive symptoms. Within these criteria, minimum negative symptoms can either be specified by the negative subscale score
being above a minimum level threshold, or a particular number of individual negative symptoms being above certain level of severity. The positive symptom exclusion criteria can either be determined by implementing a maximum threshold for positive symptoms, or else by specifying that a measure of negative symptoms be above a measure of positive symptoms by a set amount.

In this study, a different interpretation of what constitutes prominent and predominant criteria was adopted, compared to that used in the Dunayevich investigation. In the Dunayevich study, symptom criteria were only defined as "predominant" if maximum PANSS positive subscale thresholds were specified below the minimum negative symptoms. In criteria where the maximum level of positive symptoms were specified on a single item level, such as Möllers (i.e. that no more than two PANSS positive items could be above 4) there would be defined as a "prominent" negative symptom criteria. However, while such criteria can result in positive symptoms being higher than negative symptoms in theory, in reality this was rare. For example, in the subsample of participants eligible according to the criteria defined by Moller, no participants were found to have higher positive than negative symptoms. Given it was found that such criteria typically resulted in negative symptoms being predominant, in this study they were defined as such.

In total, 16 different methods to determine prominent or predominant negative symptoms were uncovered, the details of which are presented in table 10. All those that used either the SANS, SAPS or the BPRS in the inclusion/exclusion criteria were excluded from this analysis, leaving a total of 11 different methods (3 prominent and 8 predominant).

Whilst no measure of EPS was recorded, data regarding the type and dose of medication prescribed was available. Using this data as a proxy-measure, the prescribed antipsychotic medication dose was transformed into the Defined Daily Dose measure in accordance with the WHO Anatomical Therapeutic Chemical Classification ATC/DDD index (WHO, 2013), which has been found to be a reliable tool for standardising antipsychotic dosages (Nose et al., 2008).

Table 10. Summary of unreferit symptom exclusion enterior adopted in negative symptom thats										
Stud	у		Year	Scales used	Symptom inclusion criterion					
<u>Crite</u>	eria fo	or prominent neg	ative sym	<u>ptoms</u>						
1	*	Kinon	2006	PANSS	At least three items of the PANSS negative subscale ≥ 4 , or at least two ≥ 5					
2		Alvarez	2006	SANS	Baseline SANS >10					
3	*	Cirici	2008	PANSS	At least three items of the PANSS negative subscale ≥ 4					
4	*	Kaphzan	2014	PANSS	≥19 PANSS negative subscale score.					
Criteria for predominant negative symptoms										
5		Bodkin	2005	SANS, BPRS	SANS >12, two subscales >3, ≥5 on the BPRS thinking disturbance subscale					
6	*	Kane	2012	PANSS	\geq 15 on the negative subscale, no positive items \geq 4					
7		Lasser	2013	SANS	>54 on SANS 18, <2 on <1 SANS global score,					
8		Lecrubier	2006	SANS & PANSS	negative symptoms: ≥10 on SANS summary, >4 on any PANSS positive symptom item					
9		Loo	1997	SANS & SAPS	Inclusion: score of \ge 60 on SANS and \le 50 on the SAPS					
10	*	Ollie	2006	PANSS	PANSS negative subscale ≥6 points over positive subscale					
11	*	Schoemaker	2014	PANSS	≥4 on at least 3 items, ≥21 overall on the negative subscale. No more than ≤19 on the PANSS positive					
12	*	Rabinowitz	2013	PANSS	PANSS negative subscale > positive subscale					
13	*	Moller	2004	PANSS	PANSS negative items "Blunted Affect" and "Conversation Flow" \geq 4, with at least one other \geq 4, and no more than two positive items \geq 4					
14	*	Riedal	2005	PANSS	PANSS negative subscale ≥21 and ≥1 over the positive subscale score					
15	*	Stauffer	2012	PANSS	At least 3 PANSS negative subscale items ≥4, or at least two items ≥5, and a positive subscale score ≤19					
16	*	Klingberg	2011	PANSS	1 PANSS negative item \geq 4, no positive item \geq 6					

Table 10: Summary of different symptom exclusion criterion adopted in negative symptom trials

* Included in the analysis

4.3.4. Analysis plan

The subsamples were produced by implementing the different prominent and predominant criteria specified in table 10, and removing all participants that presented with at least moderately severe depressive symptoms (>4 on the PANSS). In the first part of the analysis, the change in negative symptoms over time was examined in each of the subsamples using a Students repeated-measures t-test. In the second part of the analysis, the association between negative, and positive and depressive symptoms were evaluated by using the *xtreg* command in STATA. In the third part of the analysis longitudinal modelling was used to explore the association between negative symptoms, and positive and depressive symptoms. PANSS Marder negative symptoms scores were included as a dependent variable, nested within participants included as a random effect. In the null model only time point was included as a covariate. Positive and depressive symptoms where then added to the model, and the change in the variance explained was calculated.

Previous studies in this area suggest that symptom eligibility criteria can result in a substantial reduction in the eligible sample pool (Rabinowitz et al., 2013; Dunayevich et al., 2014). With regards to the minimum number of participants necessary to provide meaningful estimates in a regression model, a number of different thresholds have been proposed. Whilst a number of more complex models have been devised (Kelley and Maxwell, 2003), a minimum of 50 participants was suggested a basic rule of thumb, with more participants needed when additional independent variables are added to the model (Green, 1991, Harris, 2014). Based on estimates produced by Miles and Shevlin (Miles and Shevlin, 2001), Field (Field, 2009) suggested that 40 participants would be a sufficient number when a large effect size is predicted and 3 predictors are included in the model (which is the maximum number added in this investigation). Following these guidelines, samples with fewer than 40 participants were omitted from the final summary.

4.4. Results

4.4.1. Summary of sample

In total, 354 participants from the sample were eligible for this investigation, of which data were available for 351. The socio-demographic details are presented in table 11. On average, participants were 42.1 years old, predominantly male (67.8%), and reported a long history of

schizophrenia (mean=16.0 years). Participants had experienced a median of 3 psychiatric hospitalisations, and reported a mean defined daily dose of antipsychotic medication of 1.36 (SD=0.95). Recruitment was split fairly equally between the 6 European countries, with the fewest recruited in Zurich (n=41, 11.7%), and the most in London (n=73, 20.8%).

Variables	Baseline n=351		
Study centre (N, %)			
Granada	68	19.4%	
Groningen	57	16.2%	
London	73	20.8%	
Lund	50	14.3%	
Mannheim	62	17.7%	
Zurich	41	11.7%	
Socio-demographic details			
Age years (mean, SD)	42.1	11.5	
Female gender (n, %)	106	32.2%	
Illness Duration years (n, SD)	16.0	10.2	
Prev. number of admissions (median, IQR)	3	1-6	
Defined Daily Dose at baseline*	1.36	0.95	

Table 11: Demographic details of sample

*Equivalents calculated using WHO database estimates.

Of the 351 participants that were included in the study, 310 participants were assessed 12 months later, resulting in a study retention rate of 88.3%. Any differences in the baseline socio-demographic details, symptom scores and retention rates between countries are presented in table 12. A significant difference in study retention between research sites was detected (χ (5)= 16.3, P=.006). Groningen reported the highest study retention rate (98.2%), whilst Granada reported the lowest (77.9%). Significantly more females than males dropped out as a proportion of the sample (χ (1)=4.14, P=.042). No other significant differences in dropout were detected in either the baseline symptom scores, or the socio-demographic information.

	Non-dropouts			Drop	outs		D*
	N	Mean	SD / %	n	mean	SD / %	г
Study site (n, %)							.006
Granada	53		77.9%	15		22.1%	
Groningen	56		98.2%	1		1.8%	
London	63		86.3%	10		13.7%	
Lund	48		96.0%	2		4.0%	
Mannheim	53		85.5%	9		14.5%	
Zurich	37		90.2%	4		9.8%	
Socio-demographic information							
Age years (mean, SD)	309	42.1	11.4	41	42	12.2	.929
Female gender (n, %)	310	88	28.4%	41	18	43.9%	.042
Illness Duration years (n, SD)	307	16.1	10.0	41	15.6	12.0	.750
Prev. number of admissions (median, IQR)	303	3.0	1-6	38	2	1-5	.254
DDD at baseline (mean, SD)*	282	1.37	0.95	37	1.31	0.94	.713
Symptom scores (mean, SD)							
Negative symptom subscale	309	17.1	7.0	41	16	5.5	.326
Marder negative symptom subscale	309	17.0	7.4	41	16.4	5.7	.622
Positive symptom subscale	310	15.3	6.0	41	15.2	5.6	.899
General Psychopathology subscale	310	32.8	10.2	41	31	8.5	.272
Depression item	310	2.6	1.4	41	2.6	1.4	.976

Table 12 comparison centre, baseline sociodemographic, and symptom levels between those that did and did not attend the 12-month follow-up

*DDD=Defined daily dose, Determined through using the students-T, Mann Whitney U test or Chi square, depending upon whether the data was recorded as a mean, median or count value

4.4.2. Baseline and change in symptoms over time

Mean symptom levels are presented in table 13. Negative symptoms at baseline were in the low to moderate-to-moderate range (mean=16.9, SD=7.2), and were slightly lower 12 months later (mean difference -1.6, t(309)= 3.93, P<.001). When compared to the results in chapter 3, this reduction was found to be slightly smaller than that detected in the control study arms (ES change in current study= 0.22, 95% CI 0.13 to 0.32; chapter 3 control arm change= 0.33, 95% CI 0.16 to 0.50). Depressive (mean=2.6, SD=1.4) and positive (mean=15.3, SD=5.9) symptoms were also in the low-to-moderate range, which in both cases were significant lower at follow up (positive symptoms: mean difference -0.9, t(308)=2.31, P=.022; depressive symptoms: mean difference -0.9, t(308)=2.31, P=.022; depressive symptoms:

Table 13: symptom levels at baseline and 12 months follow-up whole sample

	T1			Т2			- +	P
	Ν	Mean	SD	N	Mean	SD	- L	F
Marder negative symptom subscale	350	16.9	7.2	308	15.3	6.4	5.19	<.001
Positive symptom subscale	351	15.3	5.9	306	14.6	6.2	2.24	0.025
General Psychopathology subscale	347	32.6	10.4	307	30.7	9.8	4.01	<.001
Depression item	351	2.6	1.4	310	2.4	1.3	2.58	0.010

Prior to implementing the positive and negative symptom criteria, all participants presenting with moderately severe depressive symptoms at baseline as measured by a score of at least 5 on the PANSS depression item were omitted. In this subsample the change in negative symptoms over time was marginally lower (mean difference -1.39, SE=0.33), with the majority of the sample remaining eligible (91.7%; n=321).

Having removed all cases presenting with at least moderately severe depressive symptoms at baseline, the different prominent and predominant negative symptom eligibility criteria were then applied. The mean level of negative symptoms and their change over time in the different subsamples are presented in table 14. As would be expected given participants with low symptoms were omitted, in the 3 prominent symptom criteria subsamples negative symptoms were much higher compared to the whole sample at the baseline stage. In addition, a significantly larger reduction in the level of negative symptoms over time was detected in all three samples, ranging from a mean difference of -4.29 in the Kaphzan subsample, to -4.94 in the Cirici subsample. Of the three different prominent symptom criteria, the method adopted by Kaphzan was found to be the least restrictive, with 30.3% (n=104) of the whole sample being eligible at baseline. The most restrictive was the criteria adopted by Cirici, where 23.4% (n=82) were found to be eligible.

In the 8 predominant symptom criteria subsamples, the mean level of negative symptoms at baseline increased as increasingly restrictive inclusion were adopted, as did the reduction in the proportion of eligible participants. In the broadest criteria (adopted by Rabinowitz et al., 2013), 50.0% (n=175) of the whole sample was eligible, whilst in the most restrictive criteria (Moller et al., 2004) only 6.3% (n=23) were eligible. At baseline, the mean negative symptoms in the Rabinowitz subsample was 19.88 (SD=6.64), substantially lower than the Moller subsample (mean=26.86, SD=5.01). The largest mean reduction in symptoms was present in criteria which specified a high minimum negative symptoms subscale score (Riedal et al., 2005; Schoemaker et al., 2014). The lowest reduction occurred in criteria which either specified a

minimum threshold on single items (Moller et al., 2004; Klingberg et al., 2011) or criteria which merely specify predominance, rather than requiring any a minimum degree of negative symptom severity (Rabinowitz et al., 2013).

	N at T1	Negative symptoms T1		Negative symptor T2	Negative symptoms T2		Change scores T2-T1		
		mean	SD	Mean	SD	mean	SE	Ρ*	
All eligible data	350	16.99	7.38	15.35	6.37	-1.64	0.32	<.001	
Removing ≥5 depression	321	16.35	7.10	14.96	6.27	-1.39	0.33	<.001	
Prominent symptoms criterion									
Kaphzan et al, 2014 criteria	106	24.48	4.72	19.19	6.25	-5.29	0.51	<.001	
Kinnon et al., 2006 criteria	84	25.43	4.79	20.12	6.23	-5.31	0.55	<.001	
Cirici et al, 2008 criteria	82	25.55	4.78	20.16	6.31	-5.38	0.56	<.001	
Predominant symptoms criterion									
Rabinowitz et al., 2013 criteria	175	19.88	6.64	16.65	6.41	-3.23	0.44	<.001	
Klingberg et al, 2011 criteria	149	20.82	5.86	17.57	6.21	-3.25	0.49	<.001	
Ollie et al., 2006 criteria	84	23.61	6.07	18.59	6.55	-5.01	0.59	<.001	
Kane et al, 2012 criteria	71	20.48	4.85	17.08	6.01	-3.40	0.62	<.001	
Riedal et al., 2005 criteria	70	26.17	4.54	20.02	6.38	-6.15	0.57	<.001	
Stauffer et al., 2004 criteria	58	24.92	4.65	19.33	6.41	-5.59	0.66	<.001	
Schoemaker et al, 2014 criteria	48	25.95	4.26	20.02	6.55	-5.93	0.73	<.001	
Moller et al., 2004 criteria	23	26.86	5.01	23.00	5.83	-3.86	1.03	.001	

Table 14: Change in Marder negative symptoms by the exclusion criteria

*Calculated by repeated measures t-test

4.4.3. Association between negative symptoms, and positive and depressive symptoms after adopting different eligibility criteria

The association between negative symptoms, and positive and depressive symptoms over time in the whole sample are presented in table 15. Both positive and depressive symptoms were found to be positively associated with negative symptoms, with the covariates explaining 19.2% of the variance (R^2 =.192). After removing participants with at least moderate depressive symptoms, the association between negative symptoms and depressive symptoms was slightly lower but, still remained significant (B=-1.03, 95% CI 0.65 to 1.41). In this subsample, positive and depressive symptoms explained 14.9% of the variance (see table 16). Even with the inclusion of positive and negative symptoms as covariates, negative symptoms were found to significantly reduce over time (B=-1.16, 95% -1.75 to -0.57).

n-250	Change ov	er time		Change ov	Change over time + Covariates			
11-350	В	B 95% CI		В	95% CI			
Cons	18.53	17.42	19.63	10.67	8.92	12.42		
Timepoint	-1.61	-2.22	-1.00	-1.16	-1.75	-0.57		
Positive symptoms				0.27	0.19	0.35		
Depressive symptoms				1.26	0.92	1.61		
Random Part								
Sigma_u	5.55	5.04	6.11	4.83	4.36	5.35		
Sigma_e	3.90	3.60	4.22	3.72	3.44	4.03		
Derived estimates								
R				.192				
Р	0.67			0.627				

 Table 15: Association between positive and depressive symptoms and negative symptoms over time in

 the whole sample

 Table 16: Association between positive and depressive symptoms and negative symptoms over time

 after removing participants with moderate depressive symptoms

	Change ove	er time		Change ov	Change over time + Covariates		
n=321	B 95% CI			В	95% CI		
Cons	17.6	16.48	18.73	10.93	9.15	12.72	
Timepoint	-1.35	-1.98	-0.72	-1.10	-1.71	-0.49	
Positive symptoms				0.27	0.18	0.35	
Depressive symptoms				1.03	0.65	1.41	
Random Part							
Sigma_u	5.28	4.77	5.85	4.75	4.27	5.29	
Sigma_e	3.87	3.57	4.20	3.73	3.44	4.05	
Derived estimates							
R				.149			
Р	0.651			0.619			

4.4.4. Prominent negative symptom inclusion criteria:

The association between negative symptoms, and positive and depressive symptoms over time in the subsamples which adopted prominent negative symptoms criteria are presented in tables 17 to 19. In all three cases, a substantial reduction in the association between negative symptoms and the covariates was detected, with only 8.1% (R^2 =0.81) of the variance explained in the Kaphzan subsample, and 11.7% (R^2 =0.81) in the Cirici subsample. This weaker association appears to be primarily attributable to a much weaker relationship between negative symptoms and depressive symptoms, which was no longer found to be significant in any of the three subsamples.

	Change over time			Change over time + Covariates		
n=106	В	95% CI		В	95% CI	
						
Cons	29.33	27.65	31.02	24.24	21.47	27.02
Timepoint	-5.16	-6.13	-4.20	-4.93	-5.86	-3.99
Positive symptoms				0.21	0.10	0.33
Depressive symptoms				0.46	-0.11	1.03
Random Part						
Sigma_u	4.15	3.43	5.01	4.00	3.31	4.84
Sigma_e	3.46	3.01	3.98	3.29	2.86	3.79
Derived estimates						
R				.081		
Р	0.590			0.596		

Table 17: Association between positive and depressive symptoms and negative symptoms over time after adopting the Kaphzan eligibility criteria

	Change ov	Change over time			Change over time + Covariates		
n=84	В	95% CI		В			
Cons	30.32	28.45	32.19	24.78	21.79	27.77	
Timepoint	-5.19	-6.26	-4.13	-4.99	-6.01	-3.98	
Positive symptoms				0.23	0.11	0.36	
Depressive symptoms				0.54	-0.09	1.17	
Derived estimates							
Sigma_u	4.22	3.44	5.20	4.01	3.24	4.97	
Sigma_e	3.39	2.9	3.97	3.20	2.72	3.76	
Derived estimates							
R				.098			
Р	0.608			0.612			

Table 18: Association between positive and depressive symptoms and negative symptoms over time after adopting the Kinnon eligibility criteria

Table 19: Association between positive and depressive symptoms and negative symptoms over time after adopting the Cirici eligibility criteria

	Change ove	Change over time			Change over time + Covariates			
n=82	В	95% CI		В	95% CI			
Cons	30.60	28.69	32.51	24.77	21.73	27.81		
Timepoint	-5.29	-6.37	-4.21	-5.05	-6.08	-4.01		
Positive symptoms				0.25	0.12	0.37		
Depressive symptoms				0.57	-0.07	1.2		
Random Part								
Sigma_u	4.26	3.46	5.25	4.00	3.22	4.96		
Sigma_e	3.40	2.90	3.98	3.20	2.72	3.76		
Derived estimates								
R				.117				
Р	0.611			0.609				

4.4.5. Predominant negative symptom exclusion criteria

The association between negative symptoms, and positive and depressive symptoms over time in the subsamples which adopted predominant negative symptoms criteria are presented in tables 20 to 27. Substantial differences in the association between the covariates and negative symptoms were found, depending upon the method employed. In subsamples obtained by using criteria which determined predominance as negative symptoms being higher than positive symptoms by a specified amount (Ollie et al., 2006; Rabinowitz et al., 2013), the association between the covariates was found to be much higher than in the other subsamples (Rabinowitz criteria R^2 =.339; Ollie criteria R^2 =.333). This stronger relationship between the covariates and negative symptoms appears to be driven primarily between a greater association between positive and negative symptoms (Rabinowitz subsample: Positive symptoms B=0.60; 95% Cl 0.48 to 0.71; Ollie subsample: positive symptoms B=0.54; 95% Cl 0.39 to 0.69).

	Change o	Change over time			Change over time + Covariates			
n=175	В	95% CI		В	95% CI			
Cons	22.83	21.32	24.34	13.48	11.42	15.54		
Timepoint	-3.16	-4.01	-2.31	-3.32	-4.08	-2.56		
Positive symptoms				0.60	0.48	0.71		
Depressive symptoms				0.76	0.30	1.21		
Random Part								
Sigma_u	5.07	4.40	5.83	3.87	3.31	4.52		
Sigma_e	3.87	3.46	4.32	3.45	3.09	3.86		
Derived estimates								
R				.339				
Р	0.632			0.557				

Table 20: Association between positive and depressive symptoms and negative symptoms over time after adopting the Rabinowitz eligibility criteria

	Change ove	er time		Change over time + Covariates		
n=84	В	95% CI		В	95% CI	
			••••			
Cons	27.96	25.93	29.98	20.25	17.55	22.95
Timepoint	-4.85	-5.98	-3.72	-5.13	-6.13	4.13
Positive symptoms				0.54	0.39	0.69
Depressive symptoms				0.62	0.00	1.23
Random Part						
Sigma_u	5.02	4.13	6.10	3.93	3.18	4.87
Sigma_e	3.49	2.97	4.10	3.08	2.61	3.63
Derived estimates						
R				.333		
Р	0.674			0.620		

Table 21: Association between positive and depressive symptoms and negative symptoms over time
after adopting the Ollie eligibility criteria

In subsamples which were determined by separate minimum negative and maximum positive thresholds the findings are presented in tables 22 to 26. Across all five criteria, the association between the covariates and negative symptoms were found to be broadly similar, comparable to those detected in the prominent criteria samples. The smallest association between was in the Schoemaker subsample (R^2 =.080), while the largest association was detected in the Riedal subsample (R^2 =.153). Consistent with the findings derived from the prominent negative symptom subsamples, including positive and depressive symptoms as covariates did not significantly reduce the change in negative symptoms over time in any of the five predominant subsamples.

	Change ov	er time		Change ov	ver time + Cov	ariates
n=149	В	95% CI		В	95% CI	
Cons	23.67	22.06	25.28	17.24	14.68	19.79
Timepoint	-3.13	-4.07	-2.19	-2.99	-3.87	-2.1
Positive symptoms				0.30	0.17	0.42
Depressive symptoms				0.73	0.18	1.27
Random Part						
Sigma_u	4.4	3.72	5.20	4.23	3.59	5.01
Sigma_e	3.94	3.49	4.44	3.65	3.24	4.13
Derived estimates						
R				.105		
Ρ	0.555			0.574		

Table 22: Association between positive and depressive symptoms and negative symptoms over time after adopting the Klingberg eligibility criteria

Table 23: Association between positive and depressive symptoms and negative symptoms over time after adopting the Kane eligibility criteria

	Change ov	er time		Change ov	ver time + Cov	variates
n=71	В	95% CI		В	95% CI	
Cons	23.6	21.56	25.63	18.82	15.92	21.73
Timepoint	-3.32	-4.49	-2.14	-3.71	-4.81	-2.62
Positive symptoms				0.32	0.14	0.51
Depressive symptoms				0.68	-0.03	1.39
Random Part						
Sigma_u	4.01	3.18	5.07	3.95	3.16	4.95
Sigma_e	3.41	2.87	4.05	3.11	2.62	3.7
Derived estimates						
R				.088		
Р	0.580			0.618		

	Change ove	er time		Change ov	ver time + Cov	ariates
n=70	В	95% CI		В	95% CI	
Cons	32.06	30.08	34.03	24.6	21.57	27.63
Timepoint	-6.08	-7.18	-4.99	-5.89	-6.82	-4.95
Positive symptoms				0.38	0.23	0.52
Depressive symptoms				0.51	-0.10	1.13
Random Part						
Sigma_u	4.36	3.51	5.41	4.17	3.39	5.14
Sigma_e	3.19	2.69	3.78	2.71	2.28	3.23
Derived estimates						
R				.153		
Р	0.651			0.702		

Table 24: Association between positive and depressive symptoms and negative symptoms over time after adopting the Riedal eligibility criteria

Table 25: Association between positive and depressive symptoms and negative symptoms over time after adopting the Stauffer eligibility criteria

	Change ov	er time		Change ov	ver time + Cov	ariates
n=58	В	95% CI		В	95% CI	
Cons	30.08	27.86	32.29	23.78	20.51	27.06
Timepoint	-5.47	-6.72	-4.22	-5.51	-6.58	-4.43
Positive symptoms				0.36	0.20	0.53
Depressive symptoms				0.60	-0.17	1.36
Random Part						
Sigma_u	4.30	3.37	5.48	4.35	3.45	5.48
Sigma_e	3.26	2.69	3.94	2.76	2.27	3.37
Derived estimates						
R				.089		
Р	0.635			0.712		

	Change ov	er time		Change over	time + Covaria	tes
n=48	В	95% CI		В	95% CI	
Cons	31.52	29.08	33.95	24.39	20.53	28.26
Timepoint	-5.83	-7.21	-4.44	-5.62	-6.81	-4.42
Positive symptoms				0.42	0.21	0.62
Depressive symptoms				0.50	-0.34	1.35
Random Part						
Sigma_u	4.17	3.18	5.48	4.27	3.31	5.51
Sigma_e	3.3	2.68	4.07	2.79	2.25	3.46
Derived estimates						
R				.080		
Р	0.614			.701		

Table 26: Association between positive and depressive symptoms and negative symptoms over time after adopting the Schoemaker eligibility criteria

In the final predominant criteria adopted by Moller the association between negative symptoms and the covariates was found to be very small (R^2 =0.48, see table 27). However, due to the very small sample size (n=23) this analysis was omitted from the final summary due to concerns that the estimates were not sufficiently stable. In an examination for possible outliers (see figure 8) two estimates in particular (denoted by the red circles) were found to have a disproportionate impact on the association between positive and negative symptoms. When these were removed, the association between negative symptoms and the covariates was found to be substantially larger (R^2 =.192, in comparison to R^2 =.048), which further supported the decision to omit these findings.

A summary of the associations between the covariates and negative symptoms over the 11 different subsamples are presented in table 28.

	Change over	er time		Change ov	er time + Covar	iates
n=23	В	95% CI		В	95% CI	
Cons	30.19	26.73	33.65	26.73	21.89	31.58
Timepoint	-3.71	-5.67	-1.76	-4.38	-6.25	-2.52
Positive symptoms				0.21	-0.01	0.42
Depressive symptoms				0.59	-0.59	1.77
Random Part						
Sigma_u	4.11	2.77	6.10	4.17	2.85	6.11
Sigma_e	3.26	2.41	4.40	2.97	2.19	4.02
Derived estimates						
R				.048		
Р	0.614			0.664		

Table 27: Association between positive and depressive symptoms and negative symptoms over time after adopting the Moller eligibility criteria

Figure 8: Scatterplot of positive and negative symptoms that participants experience in the Moller subsample.^a



^acircles denote outliers.

Eligibility criteria	N	R ²
All eligible data	350	.192
Removing ≥5 depression	321	.149
Prominent negative criteria		
Kaphzan et al, 2014 criteria	106	.081
Kinnon et al., 2006 criteria	84	.098
Cirici et al, 2008 criteria	82	.117
Predominant negative criteria		
Rabinowitz et al., 2013 criteria	175	.339
Klingberg et al, 2011 criteria	149	.105
Ollie et al., 2006 criteria	84	.333
Kane et al, 2012 criteria	71	.088
Riedal et al., 2005 criteria	70	.153
Stauffer et al., 2004 criteria	58	.089
Schoemaker et al, 2014 criteria	48	.081
Moller et al., criteria	23	_

Table 28: Association of depressive and positive symptoms to
negative symptoms, after controlling for covariates

4.5. Discussion

4.5.1. Main findings

Consistent with the findings presented in chapter 3, significant reductions in negative symptoms over time were detected even after different minimum negative and maximum depressive and positive symptom inclusion criteria were implemented on the overall sample. This reduction in negative symptoms over time was substantially larger in subsamples which resulted in higher mean negative symptoms at baseline.

Excluding participants who present with at least moderately severe depressive symptoms and low negative symptoms resulted in a weaker association between depressive and negative symptoms. However, implementing a maximum positive symptom criterion did not appear to reduce the association between positive and negative symptoms, and in some cases even increased it. Adopting increasingly restrictive eligibility criteria substantially reduced the amount of eligible participants in a standard outpatient sample pool, but appeared to have a minimal effect on the association between these variables. These findings suggest that whilst adopting such eligibility criteria may be effective in reducing the association between negative and depressive symptoms, it may not be an effective method to reduce the impact of positive symptoms on negative symptoms.

4.5.2. Strengths and limitations

One strength of this study is the relatively high number of different criteria evaluated (11), and the consistency of the findings between them which suggests the findings are relatively robust. Another strength of the study is the fact the sample evaluated had high follow up rates over a period of 12 months (88.3%), minimising a possible selection bias due to attrition. In addition, longitudinal modelling was adopted which allows for a more appropriate way to deal with any missing data, relative to other methods such as LOCF which was adopted by earlier studies (Dunayevich et al., 2014).

Another strength of the study is the fact that the sample was obtained from a nonpharmacological trial evaluating a treatment which was designed to improve quality of life, as opposed to clinical symptoms (Priebe et al., 2007). Consequently, participants were not randomised to receive a drug which may either significantly alter positive symptoms, or else result in side effects that can either mimic or exacerbate negative symptoms. In the Rabinowitz study (Rabinowitz et al., 2013), the sample was obtained from pooling a number of pharmacological studies all designed to treatment the symptoms of schizophrenia. In the CATIE study (Lieberman et al., 2005) which was the sample used by Dunayevich (Dunayevich et al., 2014), 5% of participants discontinued the randomised treatment due to side effects, whilst it is unclear what proportion experienced side effects but still continued with the medication. In both cases, changes in positive symptoms and extrapyramidal symptoms would be expected to be higher than in the sample used in this study which may present additional challenges in disentangling the antecedents of negative symptoms.

One significant limitation of the study is that despite the relatively large initial sample size (n=350), adopting increasingly restrictive inclusion criteria resulted in a substantial loss of participants. Consequently, in the smallest sample obtained by implementing the criteria proposed by Möller (Moller et al., 2004) there were an insufficient number of participants upon which to establish reliable estimates. In addition, the low number of participants in a number of the subsamples meant that comparisons could not be drawn between groups when the R^2 values were similar. Consequently, whilst this study concludes that adopting increasingly

restrictive inclusion criteria do not result in further changes in the association between negative symptoms and positive and depressive symptoms, small differences may become evident if the study was replicated in a far larger sample. However, this may not necessarily be informative in deciding what inclusion criteria to adopt if a small decrease in the association between negative and positive symptoms comes at the cost of a substantial reduction in the pool of potentially eligible participants, which is what this and earlier studies suggest (Dunayevich et al., 2014, Rabinowitz et al., 2013).

Another possible limitation is that in the sample used clinical stability was defined as being an outpatient under the care of the same clinical team for at least three months. In both the EMA guidelines (EMA., 2012) and the persistent negative symptoms criteria (Buchanan, 2007) it is recommended that participants should exhibit clinical stability for six months. As a result, the shorter period of clinical stability required could have potentially impacted the findings. However, in this study participants typically had a very long illness duration (mean 16 years, SD=10.2) and all reported being in the same outpatient service for at least one year. As a result, it is highly unlikely that adopting the recommended 6 month clinical stability criteria would have substantially altered the sample.

In this study, it is important to consider that all participants were considered stable outpatients, with the mean severity of symptoms at baseline low. As a result, it is not clear whether the eligibility criteria evaluated may have a greater impact in inpatients, or in samples which present with higher positive symptoms. However, there is a general consensus that patients in the acute phase of schizophrenia are not appropriate participants for negative symptom clinical trials (Marder et al., 2013). In addition, if the aim is to recruit such participants then it is likely that an even smaller proportion of participants would be eligible, further exacerbating issues regarding recruitment and sample generalisability.

The final limitation of the study is that EPS were not assessed, and so were not included in the analysis. As outlined in section 1.3.1.2., EPS such as bradykinesia and akinesia can mimic negative symptoms such as blunted affect, so it is possible that at least some of the symptoms reported may be a consequence of medication effects, rather than core psychopathology. In this study the defined daily dose of antipsychotic medication in the sample was relatively high (mean DDD=1.37, SD=0.95), which may be problematic given high antipsychotic dose have been found to be associated with higher negative symptoms (Perenyi et al., 1998). If a significant proportion of the negative symptoms reported in this study were associated to EPS, then depending upon whether these were addressed following the baseline assessment this may have either caused a greater or lesser change in negative symptoms than may have

otherwise been the case. In future studies, including EPS either as a covariate in the model, or as part of the exclusion criteria may help to further disentangle how different eligibility criteria may change the association between negative symptoms and causes of secondary symptoms.

4.5.4. Summary

Implementing minimum negative and maximum depressive symptom eligibility criteria were found to reduce the association between these symptoms. These findings suggest adopting such criteria may be a helpful tool in helping to minimise secondary negative symptoms caused by depressive symptomology. However, adopting minimum negative and maximum positive symptom eligibility criteria was not found to reduce the association between these symptoms, and in some cases even increased it. In highly restrictive inclusion criteria the association between these symptoms did not change, but did result in a large proportion of the sample being ineligible. In conjunction with previous findings (Dunayevich et al., 2014), these results suggest that restrictive symptom inclusion criteria may not be an effective method to limit the impact of positive symptoms on negative symptom change.

Chapter 5. Comparing the CAINS and the PANSS as a measure of negative symptoms

5.1. Introduction

In the MATRICs consensus statement on negative symptoms (Kirkpatrick et al., 2006) the limitations of existing assessment tools were identified as a significant barrier in the development of new treatments (see section 1.8. for a full summary). As a consequence, the panel recommended forming a working group with the aim of developing and testing new scales, both in terms of their psychometric properties and their sensitivity to symptom change. Following the MATRICs meeting, the Collaboration to Advance Negative Symptom Assessment in Schizophrenia (CANSAS) was established (Blanchard et al., 2011), which in turn led to the development of two new scales; the Clinical Assessment Interview for Negative Symptoms (CAINS) (Horan et al., 2011), and the Brief Negative Symptom Scale (BNSS) (Strauss et al., 2012). These scales have been designed to reflect our current conception of negative symptoms, which include alogia, blunted affect, asociality, anhedonia and avolition.

Conceptually, the CAINS is viewed as a significant advance on earlier instruments such as the PANSS (Kay et al., 1987) and the SANS (Andreasen, 1983) which are currently the most used scales in negative symptom assessment. Despite their extensive use in schizophrenia research, the PANSS and the SANS are recognised to have a number of limitations (Blanchard et al., 2011),. These include combining cognitive symptoms into negative symptom subscales, capturing multiple, conceptually distinct domains within a singular item, and not making an appropriate distinction between anticipatory and consummatory pleasure, which is important given it appears it is only the former which relates to negative symptoms (Gard et al., 2007). In addition, the PANSS and SANS largely rely on behavioural referents; be they either interviewer observations, or reports from the interviewee, a care giver, or clinicians, as evidence of experiential deficits. This is potentially problematic, given symptoms such as anhedonia and apathy relate primarily to experiential states, where what participants actually do can be affected a number of secondary factors.

As summarised in section 1.9.1.7., the CAINS has been to have good psychometric properties (Kring et al., 2013). The interrater reliability for the CAINS experiential and expressive subscale was high (ICC=.93 and ICC=.77 respectively), whilst the test-retest reliability of the two

subscales was found to be comparable to that found with the negative symptom subscale of the PANSS (Kay et al., 1987). The CAINS subscales significantly correlated with the corresponding items on the SANS (Andreasen, 1983) and the BPRS (Overall and Gorham, 1962), and was divergent from distinct concepts such as depression, IQ, positive symptoms and EPS. In addition, the CAINS was found to moderately correlate with indicators of functioning, relating specifically to what patients actually do, as opposed to what they physically capable of doing (Kring et al., 2013).

With the CAINS scale exhibiting promising conceptual and psychometric characteristics, the next stage of evaluation proposed by the MATRIC consensus statement would be to determine the sensitivity of the scale (Kirkpatrick et al., 2006). This is important, as earlier negative symptom assessment scales have been found to report very different effect sizes (Eckert et al., 1996). Further evidence for this is presented in chapter 3 of this investigation, where it was found that the SANS reported a significantly greater change in negative symptoms over time relative to the PANSS. If the CAINS is found to be a more sensitive instrument than existing assessment scales then this could represent a major step forward in the evaluation of new therapeutics given it would be more capable of detecting any treatment effects, and would also mean that smaller sample sizes would be required to detect meaningful changes.

In a related issue to determining the sensitivity of the scale, assessing whether the measure adds predictive power above what can be predicted by other sources has been considered an important but neglected area in most areas of applied psychology (Hunsley and Meyer, 2003). This concept, known as incremental validity, was originally proposed by Sechrest (Sechrest, 1963) who stated that in order for a new psychological test to be valid then it must result in an improvement in prediction over existing methods which are routinely adopted as part of the assessment process. Yates and Taub (Yates and Taub, 2003) went on to suggest that that the assessment of incremental validity needs to be considered against the associated costs (both time and financial) of using the measure over existing alternatives. Based on experiences from a recent full-scale trial which used both the CAINS and the PANSS as part of its full assessment battery (Priebe et al., 2013a), the CAINS scale at 13 items is slightly quicker to complete than the 30-item PANSS assessment, and requires an equivalent degree of training and experience, meaning the associated costs can be considered similar. Therefore, if the CAINS is found to provide greater predictive power of concepts related to the negative symptom construct, then it could be argued that this scale has greater incremental validity in this context.

As outlined in section 1.6., negative symptoms have been found to be associated with a reduced frequency of recreational activities, lower quality and frequency of social interactions, and reduced work performance (Hunter and Barry, 2012, Lysaker and Davis, 2004, Milev et al., 2005). In this study, the degree to which the CAINS and the PANSS negative subscale could predict such outcomes was compared. These outcomes include the participant's reported social network size, the number of activities they took part in over the previous week, and whether they could identify anyone as being close friend, which were collectively defined as indicators of social impoverishment. The term 'social impoverishment' is more typically used in social psychology, and refers to concepts such as community integration and participation (Gracia et al., 1995).The decision to adopt this particular label, rather than more common alternatives which include terms such as social or functional 'impairment', 'deficit' and/or 'disability' was due to concerns that this might imply that reported behaviours are considered pathological, when may be problematic given the factors associated with these behaviours were not fully explored.

In this part of the investigation three different aims will be addressed. First, the convergent validity between the CAINS and the PANSS negative subscale will be examined, given in the original study (Kring et al., 2013), the convergent validity was examined using the BPRS anergia scale. Whilst the PANSS negative and BPRS anergia symptom subscales have been found to be highly correlated (r=.82) (Bell et al., 1992), a number of negative symptoms are not assessed by the BPRS, such as apathy and asociality, and so it could be argued that the scale provides a poorer coverage of the negative symptom construct. It is notable that whilst the SANS and the PANSS were recognised as appropriate measures for use in clinical trials for negative symptoms in both the NIMH-MATRICs and the ISCTM consensus statements (Kirkpatrick et al., 2006; Marder et al., 2011), the BPRS was not mentioned. The PANSS is recognised as one of the most extensively used scales in the field of schizophrenia research (Mortimer, 2007) and, as identified in the systematic review presented in chapter 3, has been used in a number of large trials designed to evaluate the effectiveness of treatments for negative symptoms. Consequently, understanding the relationship between the PANSS negative subscale and the CAINS may give a better understanding both of the CAINS scale itself, and also place earlier findings which have used the PANSS in better context in light of developments in our understanding of what constitutes negative symptoms.

The second part of this investigation will assess the degree of agreement between the CAINS and the PANSS to examine whether there are any systematic differences in how they capture negative symptoms. This investigation will involve both comparing how the scales differentiate

between participants that present with high and low negative symptoms, and comparing how sensitive the two scales are in detecting changes in negative symptoms over time.

The third aim will be to examine the incremental validity of the CAINS by comparing its predictive ability of objective and subjective measures of social impoverishment with the PANSS negative subscale using Dominance Analysis (Azen and Budescu, 2003, Azen and Traxel, 2009). Negative symptoms relate specifically to deficits in motivation to take part in recreational activities and impairments in the quality and quantity of social networks, so the scale which better predicts these outcomes could be considered a better indicator of the negative symptom construct. Given the PANSS negative subscale in its standard form has been extensively used in negative symptom trials in the past, both this and the alternative Marder configuration was included in the analysis.

5.2. Research questions

- 1) What is the association between the CAINS and the PANSS negative subscale?
- 2) Does the CAINS provide a greater differentiation between participants that report high and low symptoms, relative to the PANSS negative subscale?
- 3) Is the CAINS more sensitive in detecting symptom change over time relative to the PANSS negative subscale?
- 4) Is the CAINS a better predictor of objective social outcomes relative to the PANSS negative subscale?

5.3. Method

5.3.1. Sample

The data for this study was taken from the NESS study (Priebe et al., 2013a), which was a multi-site randomised controlled trial designed to evaluate the effectiveness and cost effectiveness of body psychotherapy in the treatment of negative symptoms of schizophrenia. The inclusion criteria for the study included having an established diagnosis of schizophrenia according to ICD-10, displaying negative symptoms for a period of at least six months, not changing the type of antipsychotic medication six weeks prior the baseline assessment, and an ability and willingness to take part in a physically active group. In addition, participants were required to present with at least moderate levels of negative symptoms, indicated by a score of at least 18 on the negative subscale of the PANSS. Because this skewed the baseline

distribution of the PANSS negative subscale, in this study only the end of treatment and 6 month follow-up data were analysed. A full summary of the study has been reported elsewhere (Priebe et al., 2013a)

5.3.2. Scales

The psychometric properties of the Clinical Assessment Interview for Negative symptoms (CAINS) (Horan et al., 2011, Kring et al., 2013) have been summarised in detail in section 1.9.1.7. Whilst the authors have proposed reporting the experiential and expressive subscales separately given they appear to represent distinct constructs (Blanchard and Cohen, 2006), the current consensus regarding the evaluation of negative symptoms for clinical trials recommend evaluating this outcome as a singular construct (Kirkpatrick et al., 2006, Marder et al., 2011). Consequently, both the individual subscales and the overall summary score were evaluated.

In this study the CAINS was compared to the PANSS negative subscale (Kay et al., 1987). A full summary regarding the details of this scale have been summarised in section 1.9.1.3. Given it is important to understand the relationship between the CAINS and the PANSS both in the context of earlier studies and in line with current recommendations (Marder et al., 2011), the original PANSS negative subscale and the Marder negative symptom subscale (Marder et al., 1997) were included. In the standard version, the items which formulate the negative symptom construct include affective blunting, emotional withdrawal, apathetic social withdrawal, poor rapport, stereotypical thinking, difficulties in abstract thinking and impaired conversation flow. In the alternative Marder subscale, stereotypical thinking and difficulties in abstract thinking were removed, and replaced with the active social avoidance and motor retardation items.

In order to capture objective and self-report indicators of social impoverishment, individual items from the Manchester Short Assessment Scale (MANSA) (Priebe et al., 1999), an adapted version of the Social Network Scale (SNS) (Dunn et al., 1990), and the Time Use Survey (TUS) (Lader et al., 2006) were used. From the SNS, the question "Who did you see or speak to in the last week?" was asked, with the number of friend contacts added together. From the MANSA, item 4 was used which asked "Do you have anyone you would call a close friend?". With the TUS, participants were asked whether they had taken part in activities like going out to eat, shopping for anything over than food, going to any place of entertainment, or going out

on any form of outdoor trip, with all activities added together to provide a final summary score.

5.3.3. Analysis plan

5.3.3.1. Associations between the CAINS and the PANSS

In the first part of the analysis the correlations between the CAINS and the PANSS negative subscales were calculated. In the initial development phase of the CAINS (Horan et al., 2011) a significantly stronger correlation was detected between the BPRS anergia subscale and CAINS expressive subscale, relative to the experiential subscale. This fits with the limitations identified in the review of earlier scales (Blanchard et al., 2011) which state that scales such as the PANSS, BPRS and the SANS appear to insufficiently tap into experiential states. Whilst there is a substantial degree of overlap between the BPRS anergia subscale and the PANSS negative subscale, the PANSS includes extra items relating to social withdrawal and alogia, so at present it is unclear whether the relationship would be similar to that identified by Horan and colleagues (Horan et al., 2011).

5.3.3.2. A comparison of how the PANSS and CAINS differentiates between high and low negative symptoms

In the next part of the analysis, the relationship between the CAINS and the PANSS negative subscales were further explored. Tukey mean-difference plots were conducted to examine the degree of agreement between the two scales across the range of symptom severity. This form of plotting the data is extensively used in biomedical statistics in the form of Bland-Altman plots (Bland and Altman, 1986), where objective measures are taken (i.e. blood pressure) against which one can be compared to a recognised "gold standard". However, in this context where there is no gold standard against which either scale can be compared, the aim instead will be to explore systematic differences between the scales. Given the different scaling of the subscale scores (0-4 in the CAINS, versus 7-49 for the PANSS negative subscale) the scores plotted were a proportion of their possible range. The Bland Altman plots were modelled in STATA using the *batplot* command. Trends in the degree of change between the scales were explored in a regression analysis, using the STATA *regress* command. Both the standard PANSS negative subscale, in addition to the alternative Marder configuration of the negative

symptom construct (Marder et al., 1997) were evaluated, in addition to both the CAINS total, and experiential and expressive subscales.

5.3.3.3. Examination of the difference in negative symptom change scores between the CAINS and PANSS over time

In order to explore the sensitivity of the PANSS and CAINS in detecting negative symptom change, the analysis was extended to look at the relationship between the change scores in the PANSS and CAINS over time. To achieve this, a comparison of the rate of change in one scale relative to the other was examined in a subsample of the dataset which included all participants that reported a clinically meaningful change in negative symptoms, in the same direction, over both the CAINS and PANSS. However, determining what constitutes a clinically meaning change in negative symptoms using these rating scales is a complex issue, particularly so in the case of the CAINS given the scale has not yet been extensively tested.

In the PANSS total score, a 25% reduction in symptoms has been found to correspond to a 1 point reduction in the CGI (Guy, 1976, Leucht et al., 2006), denoting observable clinical improvement. However, it is not clear how this reported figure translates to the negative symptom subscale, while at lower symptom levels the degree of improvement required appears to be substantially less. Regarding negative symptom change specifically, in the NESS trial (Priebe et al., 2013a) a 3-point reduction in the PANSS negative subscale score was deemed clinically significant in the power analysis calculation, which was the figure adopted to denote clinically meaningful change in this analysis. However, given the uncertainty around this as an appropriate figure, in a sensitivity analysis an equivalent 2-point and 4-point change in the PANSS negative symptom subscale was also examined. Given the analysis was conducted as a change in the proportional score of the scale, as opposed to raw-change scores, these figures were transformed into percentage changes in the PANSS negative subscale has a range of 42 (7-49), meaning a 3-point change equates to a proportional change of at least $\pm 7.14\%$, a 2-point change to at least $\pm 4.76\%$, and a 4-point change to at least $\pm 9.52\%$.

Given all participants had previously been randomised to receive a treatment of either Pilates or Body Psychotherapy (see Priebe et al., 2013a for a full description of the original study) it is possible that the symptom trajectory may be different depending upon the treatment group they were randomised into. Consequently, trends were examined using a mixed effects models

fitted by maximum likelihood estimation. The mean percentage change scores were added as an independent variable; the difference in that change between the CAINS and the PANSS negative subscales were added as a dependent variable, and therapy group allocation was included as a random effect.

5.3.3.4. Assessment of the incremental validity of the CAINS over the PANSS

In the final part of the study the aim was to assess the incremental validity of the CAINS over the PANSS negative subscale as a predictor of social impoverishment, as defined by the number of friend contacts over the past, and the number of activities they have taken part in over the past week, and whether they have somebody they consider to be a close friend. In the first part of the analysis, these relationships were explored using logistic or Poisson univariate regression, depending upon whether the dependent variable was count or dichotomous in nature. Given there is likely to be substantial a correlation between the PANSS negative subscales and CAINS as they have been designed to measure the same construct (i.e. negative symptoms), multivariate regression analysis was not conducted given concerns regarding collinearity. As an alternative, dominance analysis was conducted, which has been used in the past to compare the predictive strength of two different scales designed to measure the same construct (Miller et al., 2012).

Whilst in a traditional multivariate regression model the aim is to build the strongest predictive model, in dominance analysis the question instead focuses on "given X number of predictors, which is more useful in predicting the criterion Y?" (Pedhazur, 1982). To answer this question, the contribution of one individual variable is compared with the other independent variables in all possible subset models, in addition to the full model where all predictors are included. In doing so, the superiority of a predictor can be specified at three levels; the first (and lowest) level is conditional dominance, which is specified by the variable with the higher mean contribution of the explained variance over all the models (L1). If one variable dominates another at each model size then it is said to generally dominate (L2). Lastly, if one variable provides greater explanatory power over another in every subset model then is recognised to completely dominate (L3). To test the reproducibility of complete dominance 50 bootstrapped samples were produced, which is a number considered adequate to calculate estimates of standard error (Mooney et al., 1993). Following the method proposed by Azen and Budescu (Azen and Budescu, 2003), in all bootstrapped replications the variable was given a score of 1 if it completely dominated the other variable, 0.5 if neither variable completely dominated, and

0 if it was completely dominated by the other variable. The figures reported denote the mean of these findings over all the bootstrapped samples. In this study the dependent variables were either binary or count outcomes, so a logistic or poisson statistic was adopted as appropriate, modelled within the STATA *domin* command (Luchman, 2014). Following the recommendations from Azen and Traxel (Azen and Traxel, 2009), McFadden's pseudo R^2 was adopted to provide an estimate of the measure of fit (McFadden). All analysis was conducted using STATA version 12.0 (STATA Corp, 2012).

5.4. Results

5.4.1. Description of sample

The participant characteristics are presented in table 29. The mean age of participants was 42.2 years, and were predominately male (74%), unemployed (96%), with a long history of schizophrenia (mean= 12.8 years, SD= 8.8). Participants presented with moderate levels of negative symptoms (PANSS negative=21.68, SD=5.04; CAINS total= 2.12, SD= 0.59) and low levels of positive symptoms (PANSS positive =13.21, SD=4.48).

The inter-rater reliability for both the CAINS and the PANSS between raters measured at study end was high (PANSS total ICC= .85; CAINS total ICC= .80). The study retention between the two time points included in this study was found to be very high (252/264; 95.5%).

Verieble	Total	
variable	N=265.	
Age (mean, SD)	41.93	10.5
Gender (n, %)		
Male	194	73.5%
Female	70	26.5%
Ethnicity (n, %)		
White	135	51.1%
Black	78	29.5%
Asian	30	11.4%
Other	21	8.0%
Employment (n, %)		
Work/training/education	7	2.6%
Other	258	97.4%
Living situation (n, %)		
Alone	149	56.4%
With others	115	43.6%
No. of Children (median, IQR)	0	0-6
PANSS total score (mean, SD)	64.76	12.82
PANSS negative	21.68	5.04
PANSS positive	13.21	4.48
PANSS Marder negative	20.51	5.42
CAINS total score (mean, SD)	2.12	0.59
Expression Subscale	1.85	0.98
Experience Subscale	2.34	0.64
# Activities completed (median, IQR)	3	1-7
Do you have a close friend (n, %)		
Yes	167	64.5%
No	92	35.5%
# friends contacts (median, IQR)	1	0-2

Table 29: Clinical and socio-demographic information of the sample at baseline

SD, standard deviation; IQR, interquartile range;

5.4.2. Association between variables

A correlation matrix of the outcome variables assessed is presented in table 30. The correlation between the PANSS negative subscale and the CAINS expression subscale (r=.754) and CAINS total score (r=.742) was high, whilst the correlation with the CAINS experience subscale was moderate (r=.480). The four different methods in which to assess negative symptoms (the PANSS negative subscale standard format, the PANSS negative Marder configuration, the CAINS experiential subscale, and the CAINS expressive subscale) were all

positively associated with each other, and negatively associated with the number of friend contacts participants reported in the past week, whether they had a close friend, and the number of activities taken part in over the past week.

	CAINS Total	CAINS Expression	CAINS Experience	PANSS Negative	PANSS Marder negative	# Activities	# Friend contacts	Close friend (Y/N)
CAINS Total	**	.685	.870	.742	.730	159	304	347
CAINS Expression		**	.237	.754	.777	091	081	208
CAINS Experience			**	.480	.448	151	350	322
PANSS Negative				**	.888	139	237	309
PANSS Marder negative					**	114	204	346
# Activities						**	.113	.046
# Friend contacts							**	.263
close friend (Y/N)								**

Table 30: correlation matrix of the variables at end of treatment stage	
---	--

5.4.3. Relationship between the PANSS negative and the CAINS total subscales

A mean of the proportional values of the CAINS total score and the PANSS negative subscale score, plotted against the difference between these values is presented in figure 9a. The CAINS was found to report negative symptoms at a proportionally higher level, relative to the PANSS negative subscale, as indicated by the dotted line. In addition, this proportionally larger score of the CAINS over the PANSS increased as the average of the CAINS total and PANSS negative subscale values increased (B=0.26, 95% CI 0.16 to 0.35, P<.001. R^2 =.101). This relationship, whilst still significant, was found to be slightly weaker in the comparison of the CAINS total and the PANSS Marder negative subscale (B=0.17, 95% CI 0.07 to 0.27, P=.001. R^2 =.047), with greater dispersion in the values evident (see figure 9b).

Figure 9a: Relationship between the CAINS total and the the PANSS Negative subscale.

Figure 9b: Relationship between the CAINS total and PANSS Marder Negative subscale.



The x axis denotes the mean score the PANSS negative and CAINS subscale for each participant. The y axis denotes the degree of difference between the CAINS subscale and the PANSS negative subscale. The Solid line denotes where the proportional scale scores are equivalent. Scores below this line indicate that the participant was rated proportionately higher on the PANSS negative subscale in comparison to the CAINS subscale, whilst those higher than the dotted indicate that they scored proportionately higher on the CAINS. The Dotted line denotes the mean difference between the subscales over the sample.

5.4.4. Degree of agreement between the PANSS negative subscale and the CAINS experiential and expressive subscales

In order to assess the level of agreement between each of the CAINS subscales and the PANSS negative subscale, two further Tukey-mean plots were plotted comparing an average of the subscales against the difference between the scores (see figures 10a and 10b). In a similar fashion to the relationship between the CAINS total subscale score and the PANSS negative subscale, the mean difference between the two scales over the whole sample indicate that the CAINS experiential subscale scored participants substantially higher as a proportion of its range, relative to the PANSS negative subscale. In addition, this difference increased as the average of the CAINS total and PANSS negative subscale values increased (B=0.28; 95% CI 0.18 to 0.37, P<.001. R^2 =.118).

In figure 10b, which depicts the level of disagreement between the PANSS negative scale and the CAINS expression subscale, large differences between the scales were also evident. However, in contrast with the previous plots, a much clearer systematic difference was evident over the range of severity. When the mean level of negative symptoms between the PANSS negative subscale and the CAINS Expression subscale was low, the CAINS Expression subscale appeared to rate the symptoms as lower. When the mean score from the two scales was high, the experience subscale proportional score was substantially larger. In the regression analysis a strong positive relationship between the difference in the subscale scores and the mean of the two subscales was evident (B=0.77; 95% CI 0.70 to 0.85, P<.001, R^2 =.606). Overall, this suggests that the CAINS expression subscale in particular provides a much greater differentiation between patients that report high and low symptoms, relative to the PANSS negative subscale.



The x axis denotes the mean score the PANSS negative and CAINS subscale for each participant. The y axis denotes the degree of difference between the CAINS subscale and the PANSS negative subscale. The Solid line denotes where the proportional scale scores are equivalent. Scores below this line indicate that the participant was rated proportionately higher on the PANSS negative subscale in comparison to the CAINS subscale, whilst those higher than the dotted indicate that they scored proportionately higher on the CAINS. The Dotted line is the mean difference between the subscales over the sample.

In order to examine whether the differences are attributable to the PANSS negative subscale including items not recognised to be part of the negative symptom construct, the analysis was replicated using the PANSS Marder negative factor configuration (Marder et al., 1997). The relationship between the CAINS expressive and experiential subscales, and the PANSS Marder negative subscale are presented in figures 11a and 11b. In both the expressive and experiential subscales, the relationships detected were found to be highly consistent with the results reported in the PANSS negative symptom subscale analysis. In a comparison of the CAINS expresive and the PANSS Marder negative subscale, the experiential subscale reported participants as having substantially higher negative symptoms as a proportion of its range, relative to the PANSS negative subscale. This difference increased as the average of the CAINS total and PANSS negative subscale values increased (B=0.30, 95% CI 0.16 to 0.45, P<.001. R^2 =.061). In the comparison between the CAINS expressive subscale and the PANSS

140

Marder negative subscale, when the two scales reported negative symptoms in the low range the expressive subscale reported symptoms as lower, and when a mean of the scales reported negative symptoms in the high range the expressive subscale score was again much higher. This relationship was found to explain a large proportion of the variance between the scoring of the two scales (B=0.69, 95% CI 0.61 to 0.77, P<.001, R^2 =.544).

Figure 11a: Relationship between the CAINS experiential subscale and the PANSS Marder negative subscale score



Figure 11b: Relationship between the CAINS expressive subscale and the PANSS Marder negative subscale score



The x axis denotes the mean score the PANSS negative and CAINS subscale for each participant. The y axis denotes the degree of difference between the CAINS subscale and the PANSS negative subscale. The Solid line denotes where the proportional scale scores are equivalent. Scores below this line indicate that the participant was rated proportionately higher on the PANSS negative subscale in comparison to the CAINS subscale, whilst those higher than the dotted indicate that they scored proportionately higher on the CAINS. The Dotted line is the mean difference between the subscales over the sample.

5.4.5. A comparison of the change scores in the PANSS negative and the CAINS total score over time

In the next part of the analysis, a comparison of the change over time assessed using the CAINS and the PANSS was compared. Meaningful change of negative symptoms was determined as a mean change of at least $\pm 7.14\%$ in the CAINS and PANSS subscale analysed, which corresponds to a change of 3 points in the PANSS negative subscale. In a sensitivity analysis, subsamples of participants who reported an equivalent mean change of at least 2 points (mean proportional change of $\geq \pm 4.76\%$) and 4 points (mean proportional change of $\geq \pm 9.52\%$) in the PANSS negative subscale over both scales were also examined. In addition to the CAINS total score and the PANSS negative subscale being compared, the analysis was extended to include the subscales of the CAINS, and the PANSS Marder negative symptom configuration.

5.4.5.1. A comparison of the PANSS negative subscale standard configuration and the CAINS total score

Of the 241 participants who had complete data for both the CAINS total and PANSS negative subscale, 109 reported a reported a mean change of at least ±7.14%. Of these, 106 (97.2%) reported the change as occurring in the same direction in both scales and so were therefore eligible for analysis.

A comparison of the degree of change over time in the PANSS negative subscale and the CAINS total score using a mean change of $\geq \pm 7.14\%$ as an indicator of clinically meaningful change is presented in figure 12. The scatterplot suggests that as the mean change in symptom scores in the PANSS negative and CAINS total score increased, a greater magnitude of change in the CAINS was reported relative to the PANSS negative subscale. However, in a regression analysis this relationship was not found to be statistically significant (B= 0.34, 95% CI -0.04 to 0.74, P=0.081), instead only suggesting evidence of a possible weak effect.



Figure 12. Relationship between the CAINS total and PANSS subscale change scores using a ±7.14% mean difference over time as an indicator of change

The 'x' axis denotes the mean % change in both the PANSS negative and CAINS total score over 6 months. The 'y' axis denotes the degree in which the % change was larger in the CAINS subscale (>.0), or the PANSS negative subscale (<.0).

In the sensitivity analysis, a comparison of the change over time in the PANSS negative and CAINS total scores was completed in participants that reported a $\geq \pm 4.76\%$ mean change, and a $\geq \pm 9.52\%$ mean change in negative symptom scores respectively. In both cases the findings were highly consistent with the original $\pm 7.14\%$ threshold, suggesting that as the magnitude of change measured in the PANSS negative subscale and the CAINS increases, the more the CAINS changes relative to the PANSS. In the sample obtained through adopting a change of at least $\pm 4.76\%$ to determine a clinically significant change, this relationship was found to be significant (B= 0.37, 95% CI 0.06 to 0.68, P=.020). In the sample obtained through adopting a change of at least $\pm 9.52\%$ to determine a clinically significant change, the relationship was not found to be statistically significant (B= 0.47, 95% CI -0.03 to 0.96, P=.063), instead suggesting evidence of a weak effect.

5.4.5.2. A comparison of the PANSS Marder negative subscale standard configuration and the CAINS total score

The analysis completed in section 5.4.5.1. was replicated using the PANSS Marder negative subscale in place of the standard PANSS negative factor configuration. Complete data was available for 241 participants, of which 108 reported a mean change of at least ±7.14% in the CAINS total and PANSS Marder negative subscale scores. Of this subsample, 105 participants (97.2%) reported a change in the same direction in the both the CAINS total and PANSS Marder negative subscale scores.

The comparison of the degree of change over time in the PANSS Marder negative subscale and the CAINS total score is presented in figure 13. The magnitude of change in the CAINS total score was found to significantly increase as mean change in both the PANSS Marder negative subscale and CAINS total score increased (B=0.46, 95% CI 0.06 to 0.85, P=.020). In the sensitivity analysis, a comparison of the change over time in the PANSS Marder negative subscale and CAINS was completed in participants that reported a mean change of least $\pm 4.76\%$, and $\pm 9.52\%$. In both cases the magnitude of change in the CAINS total score over the PANSS Marder negative subscale score was found to significantly increase as the mean change in both increased ($\geq \pm 4.76\%$ subsample: B= 0.32, 95% CI 0.03 to 0.63, P=.034; $\geq \pm 9.52\%$ subsample: B= 0.57, 95% CI 0.11 to 1.03, P=.016)
Figure 13. Relationship between the CAINS total and PANSS Marder negative subscale change scores using a ±7.14% mean difference over time threshold as an indicator of change



The 'x' axis denotes the mean % change in both the PANSS negative and CAINS total score over 6 months. The 'y' axis denotes the degree in which the % change was larger in the CAINS subscale (>.0), or the PANSS negative subscale (<.0).

5.4.6. A comparison of the PANSS negative subscales and the CAINS experiential and expressive subscales over time

Following the comparison of the PANSS negative and PANSS Marder negative subscales to the CAINS total scores, the relationship between the PANSS negative subscale and the CAINS expressive and experiential subscales were also examined.

5.3.6.1. Comparison of the CAINS Experiential and the PANSS negative subscales

Of the 242 participants who had complete data at both time points for the CAINS experiential and PANSS negative subscale, 103 reported a reported a mean change in symptoms of at least $\pm 7.14\%$. Of these, 95 (92.2%) were eligible for analysis. A comparison of the degree of change

over time in the CAINS experiential subscale and the PANSS negative subscale is presented in figure 14. The magnitude of change in the CAINS experiential subscale over the PANSS negative subscale was found to significantly increase as the overall mean change score increased (B= .71, 95% CI 0.32 to 1.11, P<.001). This significant increase was also found to be consistent when clinically meaningful change was set at \geq ±4.76% and \geq ±9.52% respectively (\geq ±4.76% criteria: B=0 .64, 95% CI 0.32 to 0.95, P<.001; \geq ±9.52% criteria B= 0.62, 95% CI 0.08 to 1.16, P=.024).

Figure 14. Relationship between the CAINS experiential and PANSS negative subscale change scores using a ±7.14% mean difference over time as an indicator of change



The 'x' axis denotes the mean % change in both the PANSS and CAINS subscales over 6 months. The 'y' axis denotes the degree in which the % change was larger in the CAINS subscale (>.0), or the PANSS negative subscale (<.0).

In a comparison between of the degree of change over time in the CAINS experiential subscale and the PANSS Marder negative subscale, data was available for 242 participants. Of 113 participants that reported a mean change of at least ±7.14% in the two subscales, 101 (89.4%) were eligible for analysis. A comparison of the degree of change over time in the CAINS experiential subscale and the PANSS Marder negative subscale is presented in figure 18. The magnitude change in the CAINS experiential subscale over the PANSS Marder negative subscale was found to significantly increase as the mean change in both subscales increased (B=0.98, 95% 0.59 to 1.37, P<.001). This was found to be consistent in the sensitivity analysis where clinically meaningful change was set at $\geq \pm 4.76\%$ and $\geq \pm 9.52\%$ ($\geq \pm 4.76\%$ criteria: B=0.75, 95\% 0.44 to 1.06, P<.001; $\geq \pm 9.52\%$ criteria: B=0.91, 95% 0.41 to 1.41, P<.001)

Figure 15. Relationship between the CAINS experiential and PANSS Marder negative subscale change scores using a $\pm 7.14\%$ mean difference over time as an indicator of change



The 'x' axis denotes the mean % change in both the PANSS and CAINS subscales over 6 months. The 'y' axis denotes the degree in which the % change was larger in the CAINS subscale (>.0), or the PANSS negative subscale (<.0).

5.3.6.2. Comparison of the CAINS Expressive and the PANSS negative subscales

Of the 247 participants who had complete data for both the CAINS expressive and PANSS negative subscale, 131 reported a reported a mean change of at least ±7.14%. Of these, 128 (97.7%) were eligible for analysis.

A comparison of the degree of change over time in the CAINS expressive subscale and the PANSS negative subscale is presented in figure 16. The magnitude of change in the CAINS expressive subscale over the PANSS negative subscale was found to significantly increase as the mean change in both scales increased (B= 0.94, 95% CI 0.66 to 1.21, P<.001). This increase was found to be consistent in the sensitivity analysis comparing the magnitude of change in subsamples which defined a clinically meaningful change at \geq ±4.76% and \geq ±9.52% respectively (\geq ±4.76% criteria: B=0 .88, 95% CI 0.66 to 1.11, P<.001; \geq ±9.52% criteria: B= 1.03, 95% CI 0.68 to 1.37, P<.001).

Figure 16. Relationship between the CAINS expressive and PANSS negative subscale change scores using a \pm 7.14% mean difference over time as an indicator of change





In a comparison between of the degree of change over time in the CAINS expressive subscale and the PANSS Marder negative subscale, complete data was available for 248 participants. One hundred and thirty one participants reported a mean change of symptoms of at least $\pm 7.14\%$, of which 129 (98.4%) were eligible.

A comparison of the degree of change over time in the CAINS expressive subscale and the PANSS Marder negative subscale is presented in figure 17. The change in the CAINS expressive subscale over the PANSS Marder negative subscale was found to significantly increase as the mean change in both subscales increased (B=0.83, 95% CI 0.57 to 1.09, P<.001). This was found to be consistent in the sensitivity analysis comparing the magnitude of change in subsamples which defined a clinically meaningful change at > \pm 4.76% and > \pm 9.52% (B=0.91, 95% 0.69 to 1.11, P<.001; B=0.96, 95% 0.64 to 1.29, P<.001).

Figure 17. Relationship between the CAINS expressive and PANSS Marder negative subscale change scores using a \pm 7.14% mean difference over time as an indicator of change



The 'x' axis denotes the mean % change in both the PANSS and CAINS subscales over 6 months. The 'y' axis denotes the degree in which the % change was larger in the CAINS subscale (>.0), or the PANSS negative subscale (<.0).

5.4.7. An assessment of the incremental validity of the CAINS over the PANSS negative subscales

In the final part of this investigation the incremental validity of the CAINS over the PANSS negative subscale in three indicators of social impoverishment was examined. These include the number of activities participants reported taking part in over the previous week, whether they could identify somebody they considered to be a close friend, and the number of friends they have been in contact with over the past week. Univariate regression analyses examining the associations between these indicators and the CAINS total and experiential and expressive subscales, the PANSS negative standard configuration, and alternative PANSS negative Marder factor loading are presented in table 31. In all 5 forms of negative symptom assessment a significant negative association was detected with all three outcomes. In the outcome "how many activities did you take part in over the past week" the CAINS expressive subscale was found to explain the smallest proportion of the variance (B=-0.18, 95% CI -0.23 to -0.12, P<.001; R²=.014), and CAINS total score the largest (B=-0.43, 95% CI -0.52 to -0.35, P<.001; R^2 =.032). Regarding whether participants report having a close friend, the CAINS expressive subscale again explained the lowest proportion of the variance (B=-0.47, 95% CI -0.74 to -0.20, P<.001; R²=.036) and the CAINS total scale the highest (B=-1.45, 95% CI -1.96 to -0.93, P<.001; R^2 =.108). In the last outcome, assessing the number of activities participants reported taking part in over the past week, the CAINS expressive subscale again explained the smallest degree of variance (B=-0.15, 95% Cl -0.27 to -0.04, P<.010; R^2 =.007), and the CAINS experiential subscale the highest (B=-0.90, 95% CI -1.07 to -0.74, P<.001; R²=.118).

	Univariate	e analysis			
Predictor variables	В	95%	% CI	Р	Ra
How many activities did you take part in last week?					
CAINS Expression	-0.18	-0.23	-0.12	<.001	.014
CAINS Experience	-0.35	-0.43	-0.28	<.001	.026
CAINS total score	-0.43	-0.52	-0.35	<.001	.032
PANSS Negative	-0.05	-0.06	-0.04	<.001	.027
PANSS Marder negative	-0.04	-0.05	-0.03	<.001	.019
Do you have a close Friend?					
CAINS Expression	-0.47	-0.74	-0.20	.001	.036
CAINS Experience	-1.28	-1.77	-0.80	<.001	.096
CAINS total score	-1.45	-1.96	-0.93	<.001	.108
PANSS Negative	-0.15	-0.20	-0.09	<.001	.083
PANSS Marder negative	-0.15	-0.21	-0.10	<.001	.103
Number of reported friend contacts in past week					
CAINS Expression	-0.15	-0.27	-0.04	.010	.007
CAINS Experience	-0.90	-1.07	-0.74	<.001	.118
CAINS total score	-0.90	-1.09	-0.71	<.001	.091
PANSS Negative	-0.09	-0.11	-0.06	<.001	.062
PANSS Marder negative	-0.07	-0.09	-0.04	<.001	.046

Table 31: Univariate analysis of the association between assessments of negative symptoms and indicators of social impoverishment

The pseudo R^2 was calculated using logistic or poisson regression depending upon whether the independent variable was count or dichotomous in nature.

5.4.7.1. Incremental validity using the whole CAINS score

The dominance analysis comparing the CAINS total score and the PANSS negative subscale is presented in table 32. Overall, the CAINS total subscale score was found to dominate the PANSS negative subscale in all three cases. Both scales were found to be a weak predictor of the number of activities participants took part in over the past week (R^2_M = .033). Of this association, only a negligible amount of unique variance was explained by the PANSS negative subscale (R^2 =.001), relative to the CAINS total score (R^2 =.009). The CAINS total subscale was found to completely dominate, albeit with a relatively low degree of reproducibility (.627).

Relative to the number of activities undertaken, the CAINS total and PANSS negative subscale was found to be a stronger predictor of whether they report having a close friend (R^2_M = .114). Whilst the majority of this association was shared (R^2 =.081), the CAINS total score explained a

larger degree of unique variance in comparison to the PANSS negative subscale (R^2 =.026, in comparison to R^2 =.007). The CAINS total subscale was found to completely dominate with a moderately degree of reproducibility (.740).

In the number of reported friend contacts, the CAINS total and PANSS negative subscale explained a moderate proportion of the variance ($R^2_M = .092$). Of this variance, a relatively large proportion was explained uniquely by the CAINS total score ($R^2 = .031$), whilst only a negligible amount was determined uniquely by the PANSS negative subscale ($R^2 = .001$). In the bootstrapped estimates the CAINS was found to dominate with a very high degree of consistency (.923).

Indic	cators of social impo	verishme	nt						
		No. additional variables in existing model		Dominance	Dominance Standardised weight weight		Bootstrapped reproducibility of complete dominance		
		0	1	weight	weight		CAINS total	PANSS Negative	
How	many activities did	l you take	e part in las	st week?	~~~				
λ ₁	CAINS LOLAI	.032	.009	.020	.615	1 ^a	-	.627	
X ₂	PANSS Negative	.024	.001	.013	.385	2	.373	-	
Over Do y	rall fit R ² _M = .033 rou have a close Frie	end?							
X ₁	CAINS total	.108	.026	.067	.586	1 ^a	-	.740	
X ₂	PANSS Negative	.088	.007	.047	.414	2	.260	-	
Over	rall fit R^2_M = .114								
Num	ber of reported frie	end conta	cts in past	week					
X_1	CAINS total	.091	.031	.061	.661	1 ^a	-	.923	
X_2	PANSS Negative	.061	.001	.031	.339	2	.077	-	
Over	rall fit R^2_M = .092								

Table 32: Dominance Analysis comparing the PANSS negative subscale and CAINS total score as predictors of indicators of social impoverishment

^a Variable completely dominated all other predictors in the full model

The dominance analysis of the PANSS Marder negative subscale and the CAINS total score is presented in table 33. In how many activities participants completed in the past week, the CAINS total score was found to explain a substantially larger proportion of unique variance relative to the PANSS Marder negative subscale (R^2 =.015, in comparison to R^2 =<.001), and be completely dominant to a moderately high degree of reproducibility (.808). In the number of

friends contacts reported over the past week, the proportion of unique variance explained by the CAINS total score was even higher (R^2 =.047, in comparison to R^2 =<.001), and was completely dominant in a large proportion of the bootstrapped samples (.961). In whether participants report having a close friend, the CAINS total and PANSS Marder negative subscale together explained a moderate proportion of the variance (R^2_M = .125), with both scales explaining an equal proportion of the unique variance (R^2 = .017). However, the PANSS Marder negative subscale was found to marginally dominate the CAINS total score (standardised weight=.502, to .498), which was found to reproduced in 62.7% of bootstrapped samples

pre			impovensin	ment				
		No. variables in existing model		Dominance	Dominance Standardised		Bootstrapped reproducibility of complete dominance	
		0	1	weight	weight	Kdlik	CAINS total	PANSS Marder
Ноч	v many activities di	d you tak	e part in las	st week?				
X_1	CAINS total	.032	.015	.023	.728	1ª	-	.808
X_2	PANSS Marder	.017	<.001	.009	.272	2	.192	-
Ove	erall fit $R^2_M = .032$							
Do	you have a close Fri	end?						
X_1	CAINS total	.108	.017	.062	.499	2	-	.627
X_2	PANSS Marder	.108	.017	.063	.502	1ª	.373	-
Ove	erall fit R^2_M = .125							
Nur	nber of reported fri	end cont	acts in past	week				
X_1	CAINS total	.091	.047	.069	.760	1 ^a	-	.961
X_2	PANSS Marder	.043	>.001	.022	.240	2	.039	-
Ove	erall fit R^2_M = .091							

Table 33: Dominance Analysis comparing the PANSS Marder negative subscale and CAINS total score as predictors of indicators of social impoverishment

^a Variable completely dominated all other predictors in the full model

5.4.7.2. Incremental validity using the CAINS experience and expression subscales

The dominance analysis of the separate CAINS subscales and the PANSS negative subscale is presented in table 34. In all three cases, the CAINS experiential subscale was found to dominate the other two subscales, with the CAINS expressive subscale explaining the smallest proportion of the variance.

In a measure of how many activities participants took part in over the past week, the three subscales together were only able to explain a relatively small proportion of the variance (R^2_M) = .034). The largest proportion of the variance was explained by the CAINS experiential subscale (standardised weight .492), with both the CAINS expression subscale ($R^2 < .001$), and the PANSS negative subscale (R² =.002) accounting for only a very small proportion of unique variance. In the bootstrapped estimates, the complete dominance of the CAINS experiential subscale completely dominated the CAINS expressive subscale in 76.5% of cases, and the PANSS negative subscale in 56.9% of cases. In whether participants report having a close friend, a moderate proportion of the total variance was explained by the three subscales $(R^2_M =$.126). The CAINS experiential subscale explained both the largest degree of variance (standardised weight .497), and largest proportion of unique variance (R^2 =.035). The CAINS experiential subscale was found to completely dominate the CAINS expressive subscale to a high degree of reproducibility (.980), and the PANSS negative subscale to a moderate degree (.657). In the number of friend contacts participants report having over the past week, the three subscales explained a moderate degree of variance ($R^2_M = .134$). The CAINS experiential subscale explained the largest degree of variance (standardised weight .630), including unique variance (R^2 =.049). The experiential subscale was found to dominate the CAINS expressive subscale to a high degree of reproducibility (.951), and the PANSS negative subscale to a moderate degree of reproducibility (.725).

		No. var	iables in model (K)	existing	Dom	Std.	Donk	Bootstrapp comp	oed reprodu llete domin	ucibility of ance	
		0	1	2	weight	weight	Kalik	CAINS Express.	CAINS Exper.	PANSS Neg.	
How many activities did you take part in last week? (n=248)											
X_1	C. Expression	.013	.003	<.001	.005	.154	3	-	.235	.216	
X_2	C. Experience	.026	.015	.010	.017	.492	1	.765	-	.569	
X_3	PANSS Negative	.024	.010	.002	.012	.353	2	.784	.431	-	
Ove	Overall fit R^2_M = .034										
Do	you have a close Fri	end? (n=2	254)								
X_1	C. Expression	.035	.009	<.001	.015	.117	3	-	.020	.108	
X_2	C. Experience	.096	.057	.035	.063	.497	1 ^a	.980	-	.657	
X_3	PANSS Negative	.088	.043	.015	.049	.386	2	.892	0.343	-	
Ove	erall fit R^2_M = .126										
Nur	nber of reported fri	end conta	acts in pa	st week (n	=255)						
X_1	C. Expression	.006	.012	.010	.009	.068	3	-	.049	.108	
X_2	C. Experience	.117	.088	.049	.085	.630	1ª	.951	-	.725	
X_3	PANSS Negative	.061	.044	.017	.041	.302	2	.892	.275	-	
Ove	erall fit R^2_M = .134										

Table 34: Dominance Analysis comparing the PANSS negative and CAINS subscales as predictors of indicators of social impoverishment

^a Variable completely dominated all other predictors in the full model

The dominance analysis of the CAINS subscales and the PANSS Marder negative subscale is presented in table 35. The CAINS experiential subscale was found to dominate as a predictor of how many activities the participant took part in, and the number of friend contacts they made over the past week, whilst the PANSS Marder subscale was found to dominate as a predictor of whether the participants report having somebody they consider to be a close friend.

	No. add	litional va	riables in				Bootstrapped reproducibility				
	Additional	exis	sting mod	el (K)	Dom.	Std.		of complete dominance			
C	ontribution of	0	1	2	weigh t	weight	Rank	CAINS Express.	CAINS Exper.	PANSS Neg.	
Hov	v many activities d	id you ta	ke part in	last week	? (n=248)						
X_1	C. Expression	.013	.003	.002	.006	.179	3	-	.245	.471	
X_2	C. Experience	.026	.016	.015	.019	.591	1 ª	.755	-	.657	
X_3	PANSS Marder	.017	.005	<.001	.007	.231	2	.529	.343	-	
Overall fit $R_M^2 = .032$											
Do	you have a close Fr	iend? (n=	=254)								
X_1	C. Expression	.035	.013	.005	.018	.120	3	-	.069	.000	
X_2	C. Experience	.096	.055	.028	.060	.406	2	.931	-	.324	
X_3	PANSS Marder	.108	.065	.037	.070	.475	1ª	1.000	.676	-	
Ove	rall fit $R^2_M = .147$										
Nur	nber of reported fr	riend con	tacts in pa	ast week (r	n=255)						
X_1	C. Expression	.006	.009	.006	.007	.054	3	-	.010	.115	
X_2	C. Experience	.117	.095	.066	.093	.729	1 ª	.990	-	.904	
X_3	PANSS Marder	.043	.030	.010	.028	.217	2	.885	.096	-	
Ove	rall fit $R^2_M = .127$										

Table 35: Dominance Analysis comparing the PANSS Marder negative and CAINS subscales as predictors of indicators of social impoverishment

^a Variable completely dominated all other predictors in the full model

Regarding how many activities the participants took part in over the past week, the three subscales together explained a relatively small proportion of the variance ($R^2_M = .032$). In a similar manner to the PANSS negative subscale, the PANSS Marder negative subscale explains only a minimal amount of unique variance (R^2 <.001). In the bootstrapped estimates the experiential subscale was found to dominate both the CAINS expressive subscale in 75.5% of cases, and the PANSS Marder negative subscale in 65.7% of cases. In whether participants report having a close friend, contrasting with the PANSS negative subscale the PANSS Marder negative subscale was found to dominate the CAINS expressive subscale with a moderate degree of reproducibility in the bootstrapped analysis (.676). Whilst the CAINS expressive subscale was only found to explain a relatively small proportion of unique variance (R^2 =.005), both the CAINS experiential and the PANSS Marder negative subscale explained a relatively large proportion of the unique variance (R^2 =.028 and R^2 =.037 respectively).

5.5. Discussion

5.5.1. Main findings

Overall, the association between the CAINS total score and the PANSS negative subscales was found to be high. When the CAINS experiential and expressive subscales were considered separately, the PANSS negative subscales were found to be strongly associated to the expressive subscale, but only moderately associated to the experiential subscale. This finding was consistent both with the original configuration of the PANSS negative subscale, and the alternative proposed by Marder (Marder et al., 1997) which is considered to be a closer representation of the negative symptom construct (Marder et al., 2011). These results appear to support earlier findings which suggest the PANSS insufficiently tap into the experiential features of negative symptoms (Blanchard et al., 2011).

In a comparison of the sensitivity of the two scales, the CAINS was found to report a greater distinction between participants that reported high and low symptoms, and a larger degree of symptom change as the mean change in both scales increased . This relationship was found to be particularly robust in the comparison between the CAINS expressive subscale and the PANSS negative subscales. In an assessment of incremental validity, the findings suggest that the CAINS total and the CAINS experience subscale in particular is a superior predictor to both the PANSS negative standard and Marder configuration in a range of indicators relating to social impoverishment. However, this relationship was not found to be consistent throughout, with the PANSS Marder negative subscale found to be a superior predictor of whether participants report having a close friend.

5.5.2. Strengths and limitations:

The trial from which this data was obtained was a rigorous, multisite trial with a high degree of inter-rater agreement on the instruments under investigation. The minimal dropout between the two stages assessed means the result is unlikely to have been influenced by a selection bias due to attrition. Despite the fact the methodologies undertaken here have been extensively used in other disciplines, this study appears to be the first to use either dominance analysis or the Bland-Altman plot method to evaluate the properties of this important new scale in relation to the PANSS negative subscales. In addition, whilst the convergent validity of the CAINS with the BPRS and the SANS has been evaluated (Kring et al., 2013), the scale has

not yet been compared to the PANSS despite the instruments extensive use in schizophrenia research (Mortimer, 2007). In the assessment of symptom change sensitivity, the direction and the effect sizes of the relationships detected were broadly consistent over the three different thresholds evaluated, suggesting the results are robust.

One important limitation of the study is the fact that the sample included only outpatients with at least a moderate level of negative symptoms, assessed ten weeks prior to first time point included in this analysis. As a result, it is unclear whether the difference detected between the CAINS and PANSS negative subscales detected here would be consistent during either a period of acute psychotic exacerbation, or in samples where the mean level of negative symptoms are very low. In addition, this study was completed with participants that were diagnosed with schizophrenia only, and so it is not clear whether these findings would extend to patients with other psychotic disorders. Evaluating these outcomes in different populations would be helpful in determining the generalisability of these findings.

Another issue to consider is the fact that the threshold used to determine meaningful negative symptom change at 7.14% is substantially lower than the 25% change proposed following a comparison of PANSS and the CGI (Leucht et al., 2006). As stated previously, this figure was not adopted given the figure corresponds to the PANSS total score, as opposed to the negative subscale specifically. In addition, this 25% change score figure was not linear, with smaller changes required with participants who reported lower symptoms at baseline. From a practical standpoint, adopting a 25% change threshold would have resulted in the vast majority of the sample not being eligible, meaning such an analysis would not be feasible in this dataset. As an example of this, only five participants experienced a >25% in the PANSS negative subscale. In the sensitivity analysis looking at a higher change score threshold (\geq ±9.5%) slightly larger coefficients were reported, even if the estimates were less stable due to the smaller sample overall. This suggests that the differences between the CAINS and the PANSS negative subscale may be even larger if more stringent change thresholds were adopted.

In the assessment of incremental validity, one limitation to consider is the fact that social impoverishment, determined in this study by social network size, frequency of activity, and the closeness of friendships, may be significantly influenced by environmental factors such as finances, geography, and the behaviour of friends and family. As a result, it is possible that while these findings appear to support the improved incremental validity of the CAINS, this improvement may be attributable to the fact that it is influenced to a greater extent by environmental factors, rather than a better representation of core negative symptom

psychopathology. This argument is supported by the finding that the incremental validity of the CAINS experiential subscale appears to be more prominent in determining social network size and the frequency of activities conducted, which would be more strongly influenced by environmental factors, relative to how close they feel to friends. However, while this is possible, it could be considered somewhat surprising given the development process of the scale. As outlined in the initial CANSAS review (Blanchard et al., 2011), one of the main limitations of earlier scales is the fact that they appear to rely predominantly on observational and behavioural indicators. In response to this, the CAINS was designed to focus more on the experiential features of negative symptoms, rather than relying on observational referents, with the aim of limiting possible environmental effects. This being the case, further work examining the incremental validity of the CAINS over the PANSS negative subscale in alternative areas, including those less influenced by environmental effects, would be informative.

This study suggests that the CAINS may represent an important advance in negative symptom assessment over the PANSS, both in terms of its sensitivity and incremental validity. This is important, given the PANSS is recognised to be one of the most established assessment tools used in schizophrenia research (Mortimer, 2007). However, in the NIMH-MATRICs consensus statement it has been suggested the SANS may be a more sensitive instrument than the PANSS, given multiple items are used to measure single constructs (Kirkpatrick et al., 2006), a conclusion supported by the findings reported in chapter 3. Therefore, while this study supports the use of the CAINS over the PANSS negative subscale in assessing negative symptoms, at present it is not clear whether this improvement in sensitivity and incremental validity also extends to the SANS. A replication of this study using the SANS in place of the PANSS would be important in determining whether the CAINS is a more sensitive instrument than the best of those previously available.

The final limitation is that given there was no effect of treatment in the original trial from which the data was obtained (Priebe et al., 2013a), it was not possible to assess whether the CAINS was a more sensitive instrument to evaluate treatment effects, relative to the PANSS negative subscale. The finding that the CAINS total and subscale scores appears to better differentiate between participants that report high and low symptoms, both cross-sectionally and in terms of clinically meaningful change over time, supports the argument that the CAINS may be a more sensitive instrument. However, it is not clear whether or how this may translate into larger effects sizes in clinical trials.

5.5.4. Summary

In comparison to earlier negative symptom assessment tools such as the PANSS negative subscale and the SANS, the CAINS assessment tool is recognised to represent a much closer conception of the negative symptom construct as it is currently understood (Blanchard et al., 2011). In addition, a recent evaluation of the scale suggest excellent psychometric properties, with high test-retest reliability, good convergent and divergent validity and, with appropriate training, excellent inter-rater reliability (Kring et al., 2013). The current study supports both the incremental validity and the sensitivity of the scale relative to one of the most extensively used scales in negative symptom assessment (the PANSS), adding further weight to the current body of evidence supporting the CAINS as an alternative measure of negative symptoms. In addition, the findings support the conclusion that reporting expressive and experiential subscales separately may yield further insights, given their heterogeneous association to outcomes relating to social impoverishment (Horan et al., 2011). Given the limitation of existing scales have been identified as a major barrier in the development of new treatments for these symptoms (Kirkpatrick et al., 2006), the CAINS may be considered an important advance in the field of schizophrenia research.

Chapter 6. Association between negative symptoms and subjective quality of life

6.1. Introduction

The findings in chapter 5 suggest that the CAINS (Kring et al., 2013) may be a more sensitive instrument to detect changes in negative symptoms, and that examining experiential and expressive deficits as separate constructs may yield new insights in the field. One such area that may benefit from examining negative symptoms in this manner is reviewing the relationship between negative symptoms and quality of life (QOL). For decades there has been a consensus that improving psychopathological symptoms alone should not be deemed a sufficient outcome in schizophrenia (Priebe, 2007). In addition to symptom remission, objective (OQOL) and subjective (SQOL) quality of life domains such as the ability to manage ones needs, maintain satisfying relationships, and participate in productive and enjoyable activities are seen as central to the concept of recovery (Liberman et al., 2002). In order to achieve these aims, understanding the relationship between quality of life and psychopathological symptoms is important.

Quality of life is typically defined to comprise of two separate constructs, including subjective components such as wellbeing and satisfaction with life, and objective components such as daily life functioning and external resources, both material and social (Katschnig, 2000, Lehman et al., 1982, Priebe, 2007). As highlighted in section 1.6, there is a wealth of evidence to suggest there is a strong association between negative symptoms and various aspects of objective quality of life (Ho et al., 1998, Hunter and Barry, 2012, Milev et al., 2005, Whitty et al., 2008), and either a weak or no association with subjective quality of life (Eack and Newhill, 2007, Fitzgerald et al., 2001, Priebe et al., 2011b)

The absence of a strong relationship between negative symptoms and subjective quality of life may appear somewhat surprising given the their impact on functional outcomes. However, only weak-to-moderate associations between objective and subjective indicators of quality of life have been detected in the past (r=.04–.57; Priebe and Fakhoury, 2007). This finding extends to patients who exhibit good insight their illness, suggesting this is not attributable to merely to a lack of awareness (Doyle et al., 1999). One reason for this weak association may be that subjective quality of life appears to be determined by multiple processes, including the comparison between expectations and aspirations, a comparison with others, and adaptation

over time (Priebe, 2007). Given schizophrenia is highly chronic in nature, with functional impairments usually present long before the first psychotic episode (Cornblatt et al., 2007), patients are likely to have had a long time to adapt to their current situation. Regarding comparison with others, patients with schizophrenia have been found to have small friendship networks, with the majority of members being other mental health service users (Harley et al., 2012), who may well have experienced similar disadvantages and impairments. Consequently, whilst the quality of life in these patients may be objectively low, it may not appear so when considered in the context of earlier experiences, or to those they are typically in contact with. However, it may also be possible that the relationship between subjective quality of life and negative symptoms has previously been under-reported due to the manner in which negative symptoms have previously been assessed.

As noted in a review by Blanchard (Blanchard et al., 2011), many of the current assessment tools used to measure negative symptoms such as the BPRS (Overall and Gorham, 1962), the SANS (Andreasen, 1983), and the PANSS (Kay et al., 1987) insufficiently tap into what the interviewees themselves experience, instead relying primarily on behavioural referents. As a result, there is evidence to suggest that these scales appear to relate much more to the expressive features of negative symptoms, rather than to deficits in experience (Horan et al., 2011), which is supported by the findings presented in chapter 5. This being the case, it is perhaps unsurprising that scales which primarily focus on external, behavioural referents of negative symptoms report a such stronger association with external, objective indicators of QOL, in comparison to an individuals subjective experiences. In order to examine the link between negative symptoms and SQOL properly, it is important to utilise an assessment tool which appropriately captures the experiential features of negative symptoms.

With this in mind, the aim of this study is to re-evaluate the link between negative symptoms of schizophrenia and SQOL using the the two subscales of the CAINS which measure expressive and experiential deficits in schizophrenia separately, in addition to the PANSS negative subscale. We tested the hypothesis that the association between negative symptoms and SQOL relate specifically to experiential deficits, as opposed to expressive deficits, both cross sectionally and over time. In addition, we hypothesised that this relationship would maintain after controlling for depressive symptoms. Controlling for depression is important, given the associations consistently detected between low mood and subjective quality of life (Eack and Newhill, 2007, Priebe et al., 2011b)

6.2. Research questions

1) What is the cross-sectional association between experiential and expressive deficits and subjective quality of life, and how does this compare to an assessment of negative symptoms as a single construct using the PANSS negative subscale?

2) What is the association in changes over time between experiential and expressive deficits and subjective quality of life, and how does this compare to an assessment of negative symptoms as a single construct using the PANSS subscale?

6.3. Method

6.3.1. Sample

The data used in this analysis was taken from the NESS Study (Priebe et al., 2013a), which is a multisite randomised controlled trial designed to evaluate the effectiveness and cost-effectiveness of body psychotherapy as a treatment for schizophrenia. All participants were community outpatients diagnosed with schizophrenia (F20.0-F20.9). The inclusion criteria for the study included displaying negative symptoms for a period of at least six months, not changing the type of antipsychotic medication six weeks prior the baseline assessment, and an ability and willingness to take part in a physically active group. In addition, all participants were required to present with at least moderate levels of negative symptoms at the baseline assessment stage, defined by a score of at least 18 on the PANSS negative subscale (Kay et al., 1987). Further details regarding the procedures, inclusion and exclusion criteria are described elsewhere (Priebe et al., 2013a).

6.3.2. Assessment tools

Assessments were conducted at three timepoints; baseline, end of treatment 10 weeks later, and then at six months follow-up. At each stage the CAINS (Horan et al., 2011), the PANSS (Kay et al., 1987), the Manchester Short Assessment of Quality of Life scale (MANSA) (Priebe et al., 1999), and the Calgary depression scale (Addington et al., 1993) were completed.

The CAINS (Horan et al., 2011) is a scale designed to address the conceptual and methodological limitations of earlier negative symptom assessment tools (Blanchard et al.,

2011). Although the scale can provide a single summary score, the authors recommend reporting the emotional experience and expression subscales seperately given the evidence that they are measure distinct constructs (Kring et al., 2013). The PANSS (Kay et al., 1987) is a 30-item semi-structured interview designed to provide an overall measure of the symptoms of schizophrenia. In this study, the Marder factor solution of the PANSS negative subscale was adopted (Marder et al., 1997). This subscale excludes the abstract thinking and stereotypical items from the negative subscale given they are understood to relate more to cognitive defecits (Bryson et al., 1999), and are replaced by the active social withdrawal and motor retardation items.

Subjective quality of life was measured by the MANSA (Priebe et al., 1999). The questionnaire consists of 16 items; 12 subjective and 4 objective. The 12 subjective items are rated on a 1-7 scale, and cover self-rated satisfaction regarding employment, finances, recreational activities, friendships, safety, housing, health, sex-life, family and overall life satisfaction. The 4 objective items are rated as yes or no, and cover whether they have been a victim of a crime, been accused of crime, have anyone they consider a close friend, or have seen a friend in the past 7 days. In this analysis, a mean total of the 12 subjective items were calculated to create a summary score. In cases where more than 1 item was not completed, the summary score was not included in the analysis.

Given the substantial overlap between low mood and a reduced subjective quality of life, it has been recommended that any investigation of subjective indicators of social outcomes be controlled for mood as a potential confounder (Priebe, 2007). This being the case, depressive symptoms were assessed using the Calgary depression scale (Addington et al., 1993), which a scale designed to measure such symptoms specifically in schizophrenia populations. The questionnaire comprises of 9 items, rated from 0-2, with a higher score indicating more severe depressive symptoms.

6.3.3. Analysis

In the first part of the analysis the association between SQOL and expressive deficits, experiential deficits, depressive symptoms, and a summary of the overall negative symptom construct was examined cross-sectionally using univariate regression analysis. Any predictors approaching significance (p<.10) were then included in a multivariate regression model. In order to test the robustness of any associations detected, this part of the analysis was

replicated using both the end of treatment and six month follow up data in a sensitivity analysis.

In the second part of the analysis, longitudinal modelling was used to explore the association between the change in negative symptoms and SQOL over time. Symptom change-scores were calculated by subtracting the follow-up stage assessment scores (T3) from the end of treatment scores (T2), and then the end of treatment scores (T2) from the baseline scores (T1), resulting in two panels of change-score data. The univariate associations between the change scores in SQOL and different measures of negative symptoms were then evaluated by way of multi-level modelling, with each change score (level-1) nested within participants (level-2) included as a random effect. Variables significant in univariate analysis were then analysed in a multivariate model, including the change scores of the Calgary depression scale to control for the association between depression and SQOL. All analysis was conducted using STATA version 11.0 (StataCorp, 2009).

6.4. Results

6.4.1. Description of sample

The participant characteristics are presented in table 36. Participants were mostly male (73.8%) with a relatively long history of illness (mean=13.6 years, SD=9.1). At study intake, participants reported depressive symptoms in the low range (Calgary scale mean=4.70, SD=4.37), and negative symptoms in the moderate range (PANSS Marder negative subscale total score=22.1, SD=4.8). A total of 275 participants completed the baseline assessment and 255 were successfully followed up, resulting in a retention rate of 92.7%. No significant differences were detected between completers and drop-outs at the baseline assessment in either the MANSA, CAINS, or PANSS negative standard or Marder subscale scores. The interrater reliability scores between the assessors in both the CAINS and the PANSS was high (PANSS ICC=.850; CAINS total ICC=.802). At all three timepoints the Calgary depression mean score distributions were found to be highly positively skewed, so a square root transformation of the data was performed.

Table 36: socio-demographic details of the sample								
Variable	Total							
	N=275							
Age (mean, SD)	42.2	10.65						
Gender (n, %)								
Male	203	73.8%						
Female	72	26.2%						
Ethnicity (n, %)								
White	142	51.6%						
Black	80	29.1%						
Asian	30	10.9%						
Other	23	8.4%						
Employment (n, %)								
Unemployed	263	95.6%						
Other	12	4.4%						
Living situation (n, %)								
Alone	156	56.7%						
With others	119	43.3%						
Duration of illness in years (mean, SD)	13.6	9.1						
Number hospitalisations (median, IQR)	3	1-5						

The mean values of the CAINS subscales, the PANSS Marder negative subscale, the Calgary and the MANSA are presented in table 37. A small significant reduction was detected in the CAINS experience subscale, and the PANSS Marder negative subscale. No significant difference was detected in the Calgary scale, the MANSA, or the CAINS expressive subscale.

	T1	T1		2	T3	3	C	D
Variable	Mean	SD	Mean	SD	Mean	SD	Г	r
MANSA SQOL	4.44	0.93	4.58	0.89	4.52	0.95	1.51	0.221
PANSS Negative (Marder)	22.07	4.83	20.51	5.43	20.13	5.61	10.07	<.001
CAINS Expression	1.94	0.92	1.85	0.98	1.78	1.04	1.76	0.174
CAINS Experience	2.42	0.62	2.24	0.64	2.29	0.71	4.92	0.008
Calgary Depression scale	4.69	4.36	3.92	4.30	4.11	4.15	2.37	0.094

Table 37: Mean scores and change over time.

6.4.2. The association between negative symptoms and subjective quality of life

The associations between negative symptoms and SQOL are reported in table 38. In the crosssectional analysis at baseline a significant negative association was found between SQOL and the CAINS experiental subscale (B=-0.43, 95% CI=-0.62 to -0.25, R^2 =.083). No association was detected between the SQOL and the PANSS Marder negative subscale (B=-.01, 95% CI=-0.04 to 0.01, R^2 =.004), or the CAINS expression subscale (B=-0.09, 95% CI=-0.03 to 0.22, R^2 =.009). As expected, a strong negative association between depressive symptoms and SQOL was detected (B=-0.10, 95% CI= -0.12 to -0.08, R^2 =.220). In the multivariate analysis, a significant negative association was detected between experiential symptoms and SQOL after controlling for depressive symptoms (adj. B= -0.25, 95% CI -0.42 to -0.08).

	Univaria	te mode					_	Multivariate model				
	В	SE B	95% CI		Р	R²		Adj. B	SE B	95% CI		Р
CAINS Experiential	-0.434	0.092	-0.616	-0.252	<.001	.083		-0.254	0.087	-0.425	-0.083	.004
CAINS Expressive	0.093	0.064	-0.033	0.218	.147	.009		-				
PANSS Negative	-0.013	0.012	-0.037	0.012	.306	.004		-				
Calgary Scale	-0.098	0.012	-0.122	-0.075	<.001	.220		-0.090	0.012	-0.114	-0.066	<.001

Table 38: Cross sectional associations between negative symptoms, depression and SQOL at baseline

In the sensitivity analysis, the association between negative symptoms and SQOL are presented in table 39. Overall, the findings were broadly consistent with those reported at the baseline stage. At both the end of treatment and 6 months follow-up, the Calgary depression

scale and the CAINS experiential subscale was found to be negatively associated with SQOL. After controlling for depressive symptoms, an inverse association between SQOL and the CAINS experiential subscale remained significant at 6-months follow-up (adj. B=-0.20, 95% CI=-0.36 to -0.04, P=.014). At the end of treatment stage the relationship between experiential symptoms and SQOL was not found be significant (adj. B=-0.15, 95% CI=-0.32 to 0.01, P=.061), however evidence of a weak effect was detected.

	Univaria	te model					Multivariate model					
	В	SE B	95% CI		Р	R²	Adj. B	SE B	95% CI		Р	
Associations a	at end of	treatmen	t									
Experiential	-0.314	0.087	-0.486	-0.143	<.001	.053	-0.154	0.082	-0.316	0.007	.061	
Expressive	0.049	0.059	-0.068	0.166	.414	.003	-					
Negative	-0.018	0.011	-0.040	0.003	.093	.012	-					
Scale	-0.095	0.012	-0.118	-0.071	<.001	.217	-0.089	0.012	-0.113	-0.065	<.001	
Associations a	at 6 mont	hs follow	-up									
CAINS Experiential	-0.373	0.086	-0.541	-0.204	<.001	.077	-0.200	0.081	-0.359	-0.042	.014	
CAINS Expressive	0.098	0.596	-0.019	0.215	.102	.012	-					
PANSS negative	-0.008	0.011	-0.031	0.014	.452	.002	-					
Calgary Scale	-0.114	0.013	-0.140	-0.087	<.001	.234	-0.103	0.014	-0.131	-0.075	<.001	

Table 39: Sensitivity analysis exploring the cross sectional associations at end of treatment and 6 months follow up

6.4.3. The association between the changes in negative symptoms and subjective quality of life over time

The association between the change in negative symptoms and SQOL over time are reported in table 40. A negative association was detected between SQOL and experiential symptoms (B=-0.21, 95% CI= -0.34 to -0.09, P=.001), depressive symptoms (B=-04, 95% CI= -0.05 to -0.02, P<.001), and the PANSS Marder negative subscale (B=-02, 95% CI= -0.03 to 0.00, P=.044). No significant association was detected between the changes in expressive symptoms and SQOL. In the multivariate analysis, experiential symptoms, depressive symptoms and the PANSS Marder negative subscale were all included in the model. Significant negative associations between SQOL and depressive symptoms (adj. B=-0.03, 95% CI= -0.05 to -.01, P=014), and

experiential symptoms (adj. B=-0.18, 95% CI= -0.31to -.04, P=.008) were detected, whilst the relationship between the PANSS Marder negative subscale and SQOL was no longer significant (adj. B= 0.00, 95% CI= -0.02 to 0.02, P=.882).

			0	0	, ,	, ,					
	Univariat	te model			Multivar	Multivariate model					
	В	SE B	95% CI		Р	R²	Adj. B	SE B	95% CI		Р
CAINS Experiential CAINS Expressive	-0.212 -0.048	0.063 0.052	-0.336 -0.150	-0.089 0533	.001 .352	.027 .002	-0.175	0.067	-0.306	-0.045	.008
PANSS negative	-0.016	0.008	-0.321	.000	.044	.009	-0.001	0.009	-0.019	0.016	.882
Calgary Scale	-0.035	0.010	-0.055	-0.015	<.001	.028	-0.026	0.010	-0.046	-0.005	.014

Table 40: Associations between the change in negative symptoms, depression and SQOL over time

6.5. Discussion

6.5.1. Main findings

These findings suggest that there is a stronger association between SQOL and negative symptoms than has been previously reported (Eack and Newhill, 2007), however this relationship appears to relate specifically to experiential deficits. In the cross-sectional analysis, a significant negative association was detected between subjective quality of life and the experiential deficits in negative symptoms of schizophrenia, which remained significant after controlling for depressive symptoms. In an analysis of the association between negative symptoms and SQOL over time, again only the experiential features of negative symptoms were found to be a significant predictor. These findings suggest that relationship between SQOL and negative symptoms may have previously been under-reported due to a number of assessment tools not adequately assessing experiential deficits, and the way in which negative symptoms have been typically examined as a singular construct.

6.5.2. Strengths and weaknesses

The data came from a large, multi-centre trial with excellent study retention rates (92.7% from baseline to the final follow-up assessment). The inter-rater reliability between the assessors on

both the PANSS and the CAINS was high (PANSS ICC=.85; CAINS ICC=.80). In the sensitivity analysis, the findings were broadly consistent at all three assessment stages. The specificity of the results (i.e. that only the relationship to experiential deficits, and not expressive deficits, was significant) suggest that the effect over time is not a consequence of regression to the mean. Lastly, a significant association was detected between the change scores in expressive symptoms and SQOL, despite only a small reduction in these variables over time, which suggests that the findings are robust.

The main limitation of the study relates to a possible selection bias. The data came from a study evaluating participants which all reported at least moderate levels of negative symptoms at baseline, and were all outpatients at study intake. Consequently, it is not clear whether similar estimates would be present in either acute samples, or in participants with lower negative symptoms at study intake. In addition, the sample is predominantly male, and chronic in nature with an average duration of illness of 13.6 years. This being the case, it is not clear whether the findings are generalisable to females, which may be significant given there is some evidence which suggests that different factors may influence subjective quality different between genders (Röder-Wanner and Priebe, 1998). In addition, evaluating the relationship in samples with a less chronic duration of illness may also be important given SQOL appears to be lower in newly treated patients (Priebe et al., 2000), with evidence to suggest that the impact of symptoms on SQOL appears to be more prominent in the early stages of the illness (Browne et al., 2000).

6.5.4. Conclusion

The link between SQOL and negative symptoms appears to be stronger than what was originally presumed, however this relationship appears to exist almost exclusively with experiential deficits. These findings support the model which suggests that an improvement in symptoms can result in improvements in SQOL, and so have important implications for treatment development. In the future, it may be important to focus on interventions which are effective at ameliorating experiential symptoms if the aim is to improve quality of life as part of a wider programme to support patient recovery. In addition, these findings lend further support for the use of the CAINS over other assessment tools for negative symptoms such as the PANSS, and highlight the importance of assessing the experiential and expressive subdomains of negative symptoms separately.

Chapter 7. The impact of negative symptoms on the subjective initial appraisal of psychiatric inpatient treatment

7.1. Introduction

Following the examination of the link between negative symptoms and subjective quality of life, in this investigation the aim is to extend this to another patient reported outcome, namely the appraisal of treatment. The evidence suggests that patients with schizophrenia who have a more positive initial appraisal of their treatment are likely to benefit more from that treatment eventually. This applies to pharmacological treatment (Awad et al., 1995, Bartkó et al., 1987, Van Putten et al., 1981, Van Putten and May, 1978) and complex interventions (Bröker et al., 1995, Priebe and Gruyters, 1994, Priebe and Gruyters, 1995, Priebe et al., 2010a). In inpatients, a more positive initial appraisal of treatment was associated with lower symptom levels after one month (Richardson et al., 2011a), at discharge (Bröker et al., 1995), lower social impairment at 3 and 12 months (Priebe et al., 2011a), and lower subsequent involuntary readmission rates (Priebe et al., 2009). Developing a greater understanding of the factors which influence how patients appraise their hospital treatment in the initial stages would help identify those at risk of being less satisfied with their treatment, and could assist in the development of new interventions to maximise patients' initial appraisal of treatment.

Factors which may significantly impact the initial appraisal of inpatient treatment are the symptoms that patients experience at the point of admission. Regarding how symptoms may impact the initial appraisal of treatment, a number of different theories could be considered. One argument is that subjective initial appraisal may be inversely related with mood, given the consistent association detected between depressive symptoms and other patient reported outcomes (Fakhoury et al., 2002, Priebe et al., 1998). However, an alternative argument could be that appraisal of inpatient treatment is closely related to insight, which has been found to be positively associated with positive and negative symptoms, and inversely associated with depressive symptoms (Mintz et al., 2003). This may be significant, as lower insight has been found to be associated with a less positive attitude to treatment such as the medication prescribed (Freudenreich et al., 2004). In an examination of the temporal ordering of symptoms and inpatient treatment satisfaction (Richardson et al., 2011a) inverse correlations were detected between initial appraisal of treatment and positive, manic, and depressive-anxiety symptoms. However, these relationships were not considered in a multivariate model,

so it is unclear whether this association reflects a specific action to that symptom domain, or represents a shared association with other symptom types.

Whilst any links between symptoms and subjective initial response (SIR) may not yet be clear, the association between the appraisal of treatment and the legal status of the hospital admission is a lot more established. In a review comparing involuntary and voluntary hospital treatments, involuntary patients were found to be more dissatisfied with treatment, and were more likely to report that the hospitalisation was not justified (Greenwood et al., 1999, Kallert et al., 2007). This difference does not appear to be attributable to more severe psychopathology, given only minimal differences in the presentation of symptoms have been found at the point of admission (Kallert et al., 2008). However, while the severity of symptoms may be similar between voluntary and involuntary symptoms, their impact on the appraisal of treatment may not be. Experiencing positive symptoms such as paranoid delusions and manic symptoms such as agitation may be more difficult when there is a greater restriction in autonomy. Patients with negative symptoms who experience severe apathy and social withdrawal may struggle with being placed in a busy ward against their will. Consequently, in the analysis the impact of symptoms will be compared between those who are admitted voluntarily and involuntarily.

During an acute episode when schizophrenia patients would typically be admitted, positive symptoms can often dominate the picture, masking other features of the disorder such as negative symptoms (Möller, 2007). In addition, the severity of symptoms experienced is likely to be temporary in nature as symptoms are controlled with drug treatment. Consequently, implementing the inclusion criteria to determine either persistent negative symptoms (Buchanan, 2007) or disentangling primary and secondary symptoms (Carpenter et al., 1985) is highly challenging in newly admitted patients. However, given the aim of this investigation is to examine the impact of symptoms over a very short period (i.e. the point of admission) determining antecedents of negative symptoms may not be of primary importance in this context, whilst conducting multivariate analysis can control for the impact of other symptoms. As a result, no attempt at a distinction between primary and secondary negative symptoms will be made.

In this study the aim is to identify what types of symptoms are associated with the initial appraisal of hospital treatment in patients with schizophrenia, after adjusting for the influence of socio-demographic and other clinical characteristics given their impact on patient reported outcomes.

7.2. Research questions

- 1) What is the association between negative symptoms and initial appraisal of inpatient treatment, and how does this compare to other symptoms of schizophrenia?
- 2) Is the association between symptoms and initial appraisal of treatment different in those that were voluntarily and involuntarily admitted?

7.3. Methods

7.3.1. Design

This investigation is an exploratory, cross-sectional study examining the association between symptoms and the initial appraisal of inpatient treatment, after controlling for other sociodemographic and clinical variables. All the assessments took place within the first week following the admission, and were conducted by a researcher not involved in the patients' treatment. The diagnosis was obtained from the participants' medical notes at the point of discharge.

7.3.2. Sample

The current sample was obtained from pooling data from three studies. The InvolvE study (Priebe et al., 2009) assessed characteristics and experiences of in-patients in 22 hospitals across England. The EUNOMIA study (Kallert et al., 2005, Priebe et al., 2010a) was a related study with a similar design conducted at sites in 11 European countries. Both studies had a focus on involuntarily admitted patients, but also included voluntary patients who felt coerced to treatment. The EUNOMIA data from the London site was included as part of the InvolvE study, and so these were omitted from the EUNOMIA sample to ensure that participants were not included twice. The third study, EDEN (Kallert et al., 2007), was a randomised controlled trial comparing outcomes of voluntary patients in day hospitals with those in conventional inpatient settings. Details of the rationale, methods and findings of each of the three studies have been described in in detail in the referenced papers.

All three studies included consecutively admitted patients aged between 18 and 65 who had the capacity to provide informed consent. The exclusion criteria consistent over all three studies included being admitted because of acute intoxication, being a forensic patient, and being transferred from another hospital. In the EDEN study patients requiring 1-to-1

supervision were excluded, whilst in the EUNOMIA and InvolvE studies voluntary patients who did not report a coercion level of at least 3 on the MacArthur Admission Experience Survey were excluded. For this study, only participants with a primary diagnosis of schizophrenia or a related disorder according to ICD-10 (F20-29) were included. Therefore, participants with other psychotic disorders such as schizoaffective disorder were included in this study, unlike with other investigations completed in this body of work. In the previous investigations completed in this study participants were omitted given concerns regarding both the diagnostic stability of schizoaffective disorder (Malhi et al., 2008), and the fact that the disorder appears to follow a different longitudinal course (Harrow et al., 1997). However, given symptoms in the current investigation were only considered cross-sectionally, the longitudinal course of symptoms was not deemed as relevant. With respect to the EDEN study, only patients treated in conventional inpatient hospital settings were included.

Whilst the substantially larger samples that can be derived from pooling multiple studies has benefits in terms of detecting very small effects and interpreting non-significant results, the technique is also considered to have a limitations. The most significant problem is that such a technique can ignore important subgroup characteristics (Bravata and Olkin, 2001), which can in turn result in either generate associations that do not exist, or hide associations that do (Pearson et al., 1899, Robinson, 1950, Yule, 1903). Therefore, in order to test the robustness of the estimates obtained from the pooled analyses, the analysis will also be conducted on the largest single study sample (EUNOMIA).

7.3.3. Measures

An assessment of the initial appraisal of hospital treatment was obtained using the Client Scale for Assessment of Treatment (CAT) (Priebe and Gruyters, 1995). The questionnaire has seven items and asks the patient whether they believe they are receiving the right care, whether their psychiatrist understands them and if other staff are pleasant to them, if they believe they are on the right medication, if they feel well respected and regarded, whether the care received has been helpful, and whether they feel other elements of their care are appropriate. Scores range from 0 ("not at all"), to 10 ("yes entirely"). Patients with missing data on four or more items were excluded from the analysis. Recent factorial analysis supports the use of the CAT as a meaningful unidimensional scale, stable over three different European countries (Richardson et al., 2011b). In previous research the CAT has been found to have excellent internal consistency (α =.90) (Priebe et al., 2009).

Symptom severity was measured on the 24-item version of the Brief Psychiatric Rating Scale (BPRS-E) (Overall and Gorham, 1962). While the scale is recognised to have a number of significant limitations in the measurement of negative symptoms (as summarised in section 1.9.1.2), the instrument was adopted given the need to capture a broader clinical picture, assessing manic, positive and depressive symptoms in addition to negative symptoms. Whilst the scale may not be recommended as a primary outcome for negative symptom trials in either the MATRICS or ISCTM consensus statements (Kirkpatrick et al., 2006, Marder et al., 2011), the scale has been successfully used as a measure of negative symptoms in other recent major investigations (Kring et al., 2013) suggesting the use of the scale is appropriate.

With the BPRS, various 4 and 5 factor models have been proposed, with a summary of these outlined in section 1.9.1.2. In this study, the model proposed by Velligan (Velligan et al., 2005) was adopted, given the study was one of the few to be completed with a sufficiently large sample size in which to conduct factor-analysis, and the fact that both confirmatory factor analysis and exploratory analysis was completed. The Velligan model includes 4 different factors, assessing negative, positive, manic/activation, and affective symptoms separately. The negative symptom subscale comprises of 3 items, including "blunted affect", "motor retardation" and "social withdrawal". The positive subscale comprises of 4 items, including "conceptual disorganisation", "hallucinations", "bizarre behaviour" and "unusual thought content". The mania subscale includes 3 items, assessing "excitement", "motor hyperactivity" and "tension". Finally, the depressive-anxiety subscale includes the "anxiety", "depression", "suicidality" and "guilt" items. In the BPRS a mean of the items in each subscale are calculated to determine a summary score, with scores ranging from 1 ("Not present") to 7 ("extremely severe"). In the original studies the inter-rater reliability between researchers using the BPRS was high, with researchers on the InvolvE project achieving a Cohen's Kappa score of .90, while researchers on the EDEN and EUNOMIA studies achieved an intra-class coefficient score of .78.

Socio-demographic details including age, gender, marital status, previous admission history and employment status were obtained using the MANSA (Priebe et al., 1999) in the InvolvE and EUNOMIA studies, and the Clinical History Schedule (Kallert et al., 2000) in the EDEN study. The categories used were identical in the three studies. The employment and marital status the variables were dichotomised i.e. "employed, in training or education" vs. "unemployed/other", and "married" vs. "unmarried" (which included those divorced, single and widowed). This was done for two reasons. Firstly, in the case of employment the aim was to differentiate between those who did and did not having a regular occupation, rather than focusing on whether participants received a payment for employment. Secondly, a number of

the groups contained only a small number of individuals (i.e. in training or education totalled less than 4% of participants, whilst housewife/househusband totalled less than 3%), meaning the groups would be insufficiently powered to detect small differences.

7.3.4. Analysis

In the first part of the analysis correlations between the different symptom factors were explored, given concerns that symptoms such as depression and negative symptoms can be difficult to distinguish in scales that have not been designed specifically for this purpose (Addington et al., 1993, Collins et al., 1996) (See section 1.3.1.3 for a review). While there is evidence to suggest that the measurement of depression can be conflated with negative symptoms in instruments such as the Hamilton Depression Scale (Addington et al., 1996), this has finding has been inconsistent, with other studies suggesting that these constructs can be measured independently by scales such as the BPRS (Newcomer et al., 1990).

In the next part of the analysis, the association between symptoms and the initial appraisal of impatient was assessed, after controlling for a number of clinical and sociodemographic variables. These included age, gender, employment status, marital status, and the legal status of admission. Any association between symptoms and SIR that were found to approach a level of significance (P<.10) were then added to a multivariate model in order to determine whether the association was specific to that symptom cluster. In the final part of the analysis, interaction effects between legal status and each BPRS subscale score were added to the multivariate model separately in order to examine whether the effect of symptom severity on the appraisal of treatment differed between those legally detained and those admitted voluntarily.

The impact of different studies was not controlled for, given one study (EDEN) included only voluntary patients. Since one aim was to assess the impact of legal status, entering this variable would have confounded this part of the analysis. In order to rule out the possibility of any bias in the results occurring due to the heterogeneity of the samples used, a subgroup analysis was conducted on the largest dataset (EUNOMIA).

To tackle any potential issues which could arise through a listwise deletion of incomplete cases (Little and Rubin, 1989) a multiple imputation using a chained equation of 10 cycles was conducted. All values included in the analysis were entered both as predictors and for imputing, other than the dependent variable which was included as a predictor only. Following

the recommendations by Spratt (Spratt et al., 2010), 25 imputations were conducted with the analysis conducted on the pooled data. R^2 scores were obtained from the imputed data in accordance with Rubin's rules by calculating the estimate from each imputed sample and then calculating an overall mean of these estimates (Rubin, 1987). All the analysis was conducted in SPSS version 20.

7.4. Results

7.4.1. Sample characteristics

In total, 2316 patients met the inclusion criteria. 211 patients were omitted due to missing or incomplete CAT scores, leaving 2105 patients in the final sample. The clinical and sociodemographic characteristics of this sample are presented in Table 41. The mean age of participants was 38.5 years (SD=11.24), with a fairly even split between males and females (56.9%-43.1%). Participants experienced a relatively long duration of illness (7.84 years, SD=9.11), most had experienced at least one previous inpatient admission (76.6%), and in the current hospital stay the majority were admitted involuntary (72.7%). The most severe symptoms experienced were positive symptoms (mean=2.92, SD=1.26), and the least severe were manic symptoms (mean=1.97, SD=1.11). Overall, participants reported being moderately satisfied with their treatment (CAT mean=6.00, SD=1.28).

Characteristics	<i>n</i> or mean	% or SD
Patients (N)	2105	
% detained involuntarily	1530	72.7%
Age (years)	38.49	11.24
Gender (% female)	906	43.1%
Number married (n, %)	454	22.0%
Employed/training/education	465	22.4%
Previous psychiatric admission (%Yes)	1517	76.7%
Illness Duration (mean, SD)	7.84	9.11
BPRS (mean, SD)		
Depression/anxiety subscale score	2.24	1.08
Mania subscale score	1.97	1.11
Negative subscale score	2.14	1.08
Positive subscale score	2.92	1.26
CAT mean score	6.00	1.28

Table 41: Socio-demographic and clinical characteristics of the patients

BPRS= Brief Psychiatric Rating Scale; CAT= Client Assessment of Treatment scale

As stated previously, the sample was pooled from three different studies, with the sociodemographic and clinical breakdown studies presented in table 42. In total, 1556 participants were recruited as part of the EUNOMIA project, 393 as part of the InvolvE study, and 156 as part of the EDEN study. As anticipated, given the nature of the three projects some differences were noted between the studies. In the InvolvE and EUNOMIA studies, a substantial proportion of participants were admitted on an involuntary basis, whilst 100% of the sample was admitted voluntarily in the EDEN study. Although the proportion of participants who had been previously experienced a psychiatric admission (71.4-77.5%), and age (35.83 years – 39.17 years) were similar between studies, a number of other differences were evident. In the EDEN study participants were more likely to be female, either employed and/or in training, married, and with a lower duration of illness. Regarding symptoms, in the EDEN study the sample was found to include experience more severe depressive symptoms, and less severe negative, positive and manic symptoms. In a comparison of the InvolvE and EUNOMIA datasets, manic and negative symptoms were found to be slightly higher in the EUNOMIA sample, whilst depressive/anxiety symptoms were slightly higher in the InvolvE sample, with positive symptoms being broadly similar. Participants reported being the least satisfied in the InvolvE sample (CAT mean=5.36, SD= 2.91), and most satisfied in the EDEN sample (CAT mean=7.69, SD=1.98)

A missing value analysis indicated that 2.8% of all predictor values (653 in total) were missing, with 21.1% of cases (445) missing at least one. Of the 653 values missing in total, 53.9% (352 in total) related to the duration of illness.

			Stud	ies		
Variables	Euno	mia	Invo	IvE	Ede	en
	N or mean	% or SD	N or mean	% or SD	N or mean	% or SD
n	1556		393		156	
% detained involuntarily	1169	75.1%	361	91.9%	156	0%
Age (years)	39.17	11.09	35.83	11.10	38.38	12.15
Gender (% female)	689	44.3%	116	29.5%	101	64.7%
Marital status (% married)	328	21.4%	67	17.5%	59	39.9%
Employed/in training	327	21.3%	71	18.4%	67	43.8%
Prev. admission (%Yes)	1179	77.5%	288	74.6%	50	71.4%
Illness Duration	8.41	9.38	6.91	8.45	4.68	7.05
BPRS						
Depression/anxiety	2.09	1.01	2.43	1.20	3.20	0.90
Mania	1.93	1.00	1.63	0.84	1.55	0.51
Negative	2.27	1.11	1.77	0.92	1.85	0.89
Positive	3.07	1.22	3.01	1.21	1.35	0.50
САТ	6.00	2.69	5.36	2.91	7.69	1.98

Table 42: demographics by study

BPRS= Brief Psychiatric Rating Scale; CAT= Client Assessment of Treatment scale

7.4.2. Symptoms and other patient characteristics associated with initial appraisal of treatment

Correlations between symptoms and the initial appraisal of treatment are presented in table 43. A moderate negative correlation was detected between the initial appraisal of treatment and positive and manic symptoms (r=-.228 and r=-.233 respectively). A weak negative correlation was found between negative symptoms and SIR (r=-.041), and a weak positive correlation was detected between depressive symptoms and SIR (r=.049). Although moderate correlations were detected between some symptoms, these were not of a sufficient level to suggest that collinearity may be an issue in the multivariate analysis. The strongest correlation was detected between positive and manic symptoms (r=.422), whilst the correlation between negative and depressive/anxiety symptoms was much lower than what was originally anticipated (r=.127).

	Subjective initial appraisal (CAT)	Depression/Anxiety symptoms	Manic symptoms	Negative symptoms	Positive symptoms
Subjective initial appraisal (CAT)	**	.049	233	041	228
Depression/Anxiety symptoms		**	028	.127	080
Manic symptoms			**	.104	.422
Negative symptoms				**	.250
Positive symptoms					**

 Table 43: Correlation table of symptoms subscales and appraisal of treatment

The associations between symptoms and the initial appraisal of treatment are presented in table 44. After controlling for age, gender, admission history, illness duration, legal status, employment situation and marital status, higher mania and positive symptoms were found to be associated with a lower initial appraisal of treatment (manic symptoms adj. B= -0.474, 95% CI -0.577 to -0.372, P<.001, R^2 =.036; positive symptoms adj. B= -0.367, 95% CI -0.460 to -0.274,
P<.001, R^2 =.027). A negative relationship between negative symptoms and initial appraisal of treatment was detected, however this was not significant, with the estimates found to be just outside the specified threshold to be included in the multivariate analysis (adjusted coefficient= -0.089, 95% CI -0.197 to -0.019, P=.106). No relationship between depressive/anxiety symptoms and initial appraisal of treatment was found.

In the multivariate analysis, both higher mania and positive symptoms were found to be significantly associated with a lower assessment of the appraisal of treatment (mania adj. B= - .358, -0.469 to -0.247, P<.001; positive symptoms adj. B =-0.250, 95% CI -0.350 to -0.151, P<.001). However, including both symptoms into the model appeared to only explain a relatively small proportion of additional variance (R^2 =.045).

.

Table 44: Associations bet	ween symptoms and su	bjective initial response

Predictor variables ^a	Univariate analysis			Multivariate analysis ^B			
	В	95 %CI	Р	R ²	В	95 % CI	Р
Depressive/anxiety	0.028	-0.082 to 0.138	.619	<.001			
Mania	-0.474	-0.577 to -0.372	<.001	.036	-0.358	-0.469 to -0.247	<.001
Negative	-0.089	-0.197 to 0.019	.106	.001			
Positive	-0.367	-0.460 to -0.274	<.001	.027	-0.250	-0.350 to -0.151	<.001

^aAssociation between SIR and symptoms, after adjusting for age, gender, admission history, illness duration, legal status, employment status, and marital status.

^B*R*² change after adding symptoms= .045

In order to ensure the findings are not attributable to subgroup differences between studies (Bravata and Olkin, 2001) a subgroup analysis was completed on just the largest dataset, namely the EUNOMIA study. The findings are presented in table 45. Overall, the findings were broadly consistent with those reported in the pooled analysis. When symptoms were considered separately, a significant negative association was detected between SIR, and mania and positive symptoms (mania adj. B= -0.521, 95% CI -0.633 to -0.408, P<.001, R^2 =.050; positive adj. B= -0.416, 95% CI -0.524 to -0.308, P<.001), whilst no relationship between depressive/anxiety symptoms and SIR was detected (adj. B= 0.023, 95% CI -0.111 to 0.158, P=.735). In a slight change to the pooled analysis, the negative symptom subscale was found to be significantly associated with SIR, albeit only explaining a very small proportion of the variance (adj. B= -0.157, 95% CI -0.278 to -0.035, P=.012, R^2 =.002). When symptoms were included in the model together, the findings were consistent with the original findings, with significant negative associations detected between mania an positive symptoms (mania symptoms adj. B= -0.401, 95% CI -0.526 to -0.277, P<.001; positive symptoms adj. B= -0.239,

95% CI -0.360 to -0.117, P<.001), whilst the relationship to negative symptoms was nonsignificant (negative symptoms adj. B= -0.102, 95% CI -0.225 to 0.021, P=.104). In the multivariate model, the symptoms again explained only a relatively small proportion of the total variance in appraisal of treatment (R^2 =.061).

Table 45: Associations between symptoms and subjective initial response in the EUNOMIA dataset

Predictor variables ^a -	Univariate analysis			Multivari	Multivariate analysis ^B		
	В	95 %CI	Р	R ²	В	95 % CI	Р
Depressive/anxiety	0.023	-0.111 to 0.158	.735	<.001			
Mania	-0.521	-0.633 to -0.408	<.001	.050	-0.401	-0.526 to -0.277	<.001
Negative	-0.157	-0.278 to -0.035	.012	.004	-0.102	-0.225 to 0.021	.104
Positive	-0.416	-0.524 to -0.308	<.001	.036	-0.239	-0.360 to -0.117	<.001

^aAssociation between SIR and symptoms, after adjusting for age, gender, admission history, illness duration, legal status, employment status, and marital status.

 ${}^{\rm B}R^2$ change after adding symptoms in the multivariate model= .061

7.4.4 Legal status of admission, symptom severity and initial treatment evaluation

The next stage of the analysis explored whether symptom levels had a different predictive value in voluntary and involuntary patients. In order to achieve this, an interaction effect between legal status and each symptom subscale score was separately entered into the model. Amongst the four symptom subtypes, a significant interaction effect was found for manic symptoms (adj. B=0.407, 95% Cl 0.138 to 0.676, P=.003), with the results presented in figure 18. The graph suggests that whilst SIR is broadly similar between voluntary and involuntary patients when manic symptoms were very low; as they increase the appraisal of treatment reduces far more in involuntarily admitted patients, relative to those admitted voluntarily. No difference in the impact of symptoms between those that were and were not admitted involuntarily were detected in positive symptoms (adj. B= 0.116, 95% Cl -0.092 to 0.325, P=.274), negative symptoms (adj. B= -0.123, 95% Cl -0.343 to 0.120, P=.345).

In a post-hoc analysis separating those admitted involuntarily from those admitted on a voluntary basis, a significant negative association between manic symptoms and SIR was detected in those admitted involuntarily (adj. B=-0.548, 95% CI -0.666 to -0.430, P<.001, R^2 change=.053), whilst the relationship was non-significant in the voluntary sample (adj. B=-0.124, 95% CI -0.340 to 0.093, P=.263. R^2 change=.002).

Figure 18: Association between manic symptoms and SIR between those that have been voluntarily and involuntarily admitted.



7.5. Discussion

7.5.1. Main findings

The negative symptoms of schizophrenia were not found to be associated with the subjective initial appraisal of impatient treatment. Instead, manic and positive symptoms were found to be negatively associated how hospital treatment is perceived in the initial stages, after controlling for other clinical and sociodemographic variables. The impact of manic symptoms appeared to be particularly strong in those patients who were detained involuntarily, as opposed to those who are voluntarily admitted. Somewhat surprisingly, no association between depressive symptoms and appraisal of treatment was detected, despite the consistent negative association between patient reported outcomes and mood which has previously been found (Fakhoury et al., 2002, Priebe et al., 1998).

7.5.2. Strengths and limitations

Regarding the strengths of the study, the large sample size (2105 patients) provides sufficient statistical power to detect even small effect sizes, thus allowing the interpretation of negative findings. In this study observer-ratings of symptoms were used to predict patient-rated appraisals of the appraisal of treatment, meaning the associations detected are not a consequence of a generalised tendency to provide more or less positive ratings in different self-reporting measures (Fakhoury et al., 2002, Hansson et al., 2007). In addition, results were consistent in the sensitivity analysis evaluating the EUNOMIA dataset in isolation, suggesting the findings are robust. Finally, the Interrater reliability was high in all three studies. In the InvolvE study all researchers achieved an interrater reliability of Cohen's Kappa =.90 in the BPRS, whilst in both the EDEN and EUNOMIA an interrater reliability score of ICC=.78 was reported.

One significant limitation of the study is the fact that there appears to be a number of differences between the study samples pooled, both in terms of sociodemographic and clinical characteristics. In two of the three studies (InvolvE and EUNOMIA), voluntary patients were included only if they expressed a level of being coerced. In the third study (EDEN) no involuntary patients were recruited. In the EDEN study participants appeared to report lower symptoms in all domains, other than in depression/anxiety, and reported a more positive appraisal of treatment, whilst a number of sociodemographic differences were evident. As highlighted previously, one significant issue with the pooling multiple studies is the fact that it can ignore important subgroup characteristics, which can in turn influence any associations between variables (Bravata and Olkin, 2001, Pearson et al., 1899, Robinson, 1950, Yule, 1903). However, given the findings of the pooled sample were found to be highly consistent with those in an analysis of the single largest dataset (EUNOMIA). it suggests that this may not be as significant an issue as may first have been assumed.

Another significant limitation of the study relates to the context in which participants were recruited. Given the aim of all three studies was to recruit and assess all participants within the first week of an inpatient admission, it is likely that there may have been a small subsample of patients who were too acutely unwell to be eligible for recruitment due to a lack of capacity. Whilst problematic, this particular issue is common to all studies which aim to recruit participants in this phase of treatment. In a related issue, it is also possible that patients who were highly dissatisfied with treatment may be less likely to co-operate in research being conducted in this setting. However, his may not be as significant an issue as first assumed

given all assessors were independent to the clinical team. This being the case, it is quite possible that this sub-group of patients may be more keen to be involved in the research as it provides an independent platform in which to discuss their dissatisfaction with aspects of their treatment.

Another limitation is due to the lack of available data, it was not possible to consider what treatment participants received during the first days of hospital treatment, and whether specific treatments were linked a more positive or negative appraisal of treatment. For example it has been recognised that patients with high positive symptoms are significantly more likely to be prescribed a high dose of antipsychotic medication (Sim et al., 2004), while involuntary admissions have been found to be a significant predictor of antipsychotic polypharmacy and high-dose (Lelliott et al., 2002). High-dose and polypharmacy have been both been recognised to increase adverse side effects (Centorrino et al., 2004), which in turn are considered to be a significant source of discomfort and treatment appraisal. Future work assessing the nature of treatments that participants receive during the initial hospitalisation period may help determine possible mediating factors in associations between symptoms and the appraisal of hospital treatment.

7.5.3. Summary

The findings suggest that patients with different symptom profiles experience and respond to psychiatric hospital treatment in different ways and, as a result, express different appraisals of treatment within a few days after admission. Patients with schizophrenia experiencing more severe manic and positive symptoms, and manic symptoms in particular if detained involuntarily, are more likely to report a more negative initial evaluation of treatment. Higher negative symptoms do not appear to impact the appraisal of inpatient treatment.

The findings underline the importance of assessing patients' initial appraisal of treatment as a relevant process variable in both research and routine practice. Further studies may explore environmental factors, processes of interaction between service users and staff, treatment components and mediating processes as an explanation into the association of manic and positive symptoms with more negative appraisals of hospital treatment after only a few days. In a recent review Mullen (Mullen, 2009) has suggested that acute inpatient settings are at present too custodial and over reliant on medication, and suggests that providing more

psychosocial interventions as an alternative may be a way in which to try and address these issues. If these are problems which disproportionally affect those experiencing higher positive and manic symptoms, then this possible solution may go some way to addressing the more negative experiences these particular patients report.

Chapter 8. Discussion

8.1. Thesis aims

The main aim of this thesis was to explore areas which could either inform the design of negative symptom clinical trials, or build upon the current developments in symptom assessment. This included examining how negative symptoms change over time, the impact of different symptom inclusion criteria adopted in clinical trials, a comparison of different scales designed to measure negative symptoms, and the nature of their association to other constructs such as subjective quality of life and the initial appraisal of treatment.

In the first part of this investigation the aim was to explore the longitudinal course of negative symptoms in stable schizophrenia outpatients over a period of up to 3 years, pooling a broad array of studies by way of meta-analysis. In a meta-regression, study-level variables such as the assessment scale used and the eligiblity criteria adopted were examined as predictors of differing longitudinal course. Despite symptom inclusion criteriafor negative symptom trials being consistently recommended in the literature (Kirkpatrick et al., 2006, Marder et al., 2013, Marder et al., 2011), there has been substantial variation in the manner in which they have been adopted (Dunayevich et al., 2014, Rabinowitz et al., 2013). Therefore, in Chapter 4 the aim was to further explore the impact of adopting differen symptom inclusion criteria on the associations between these negative, and positive and depressive symptoms.

As stated previously, one of the major recent advances in the field has been the development of new instruments to measure negative symptoms, such as the CAINS (Kring et al., 2013) and the BNSS (Kirkpatrick et al., 2011). Whilst it is recognised that CAINS has highly promising psychometric properties (Kring et al., 2013) and represents an important advance on earlier scales such as the PANSS and the SANS from a conceptual point of view (Blanchard et al., 2011), to date little work has been conducted examining the sensitivity of the scale. In chapter 5, a comparison of how the PANSS and the CAINS differentiate between participants that report high and low negative symptoms, and differences in the degree to which both scales report symptom change over time was conducted. In addition, the incrimental validity of the CAINS was compared to the PANSS negative subscales in various indicators of social improvishment. The main aim of this investigation was to determine whether the CAINS is a more sensitive instument than the PANSS negative subscale, and whether the CAINS is a stronger pedictor of outcomes known to be related to the negative symptom construct.

One important feature of the CAINS is the fact that the scale is designed to measure expressive and experiential features of negative symptoms as distinct constructs. Measuring these two features of negative symptoms seperately allows us to examine whether associations between negative symptoms and other constructs may relate specifically expressive or experiential deficits, or both. In chapter 6 the association between subective quality of life and the expressive and experiential components of negative symptom was explored, which has previously been found to be relatively weak (Eack and Newhill, 2007, Priebe et al., 2011b). The aim of this investigation was determine whether the association between SQOL and negative symptoms have previously been underreported due to limitations in existing scales which do not sufficiently tap into the experiential features of negative symptoms (Blanchard et al., 2011).

In the final part of the investigation (chapter 7), the impact of negative symptoms was extended to another patient report outcome, namely the initial appraisal of inpatient psychiatric treatment. Subjective initial appraisal is an important outcome given it has been found to have significant implications on later treatment response (Bartkó et al., 1987, Bröker et al., 1995, Priebe and Gruyters, 1994, Priebe et al., 2009). The aim of this investigation was to determine whether negative, positive, manic or depressive symptoms are associated with SIR.

8.2. Summary of findings and comparisons to the literature

8.2.1. The change in negative symptoms over time in outpatients

The meta-analysis presented in chapter 3 suggests that there is a significant reduction in the severity of negative symptoms over time in all treatment contexts. This finding was supported in the analysis completed in chapter 4, where a reduction in negative symptoms over time was present regardless of the eligibility criteria adopted. In chapter 6, a significant reduction over time was detected in the PANSS negative subscale and the CAINS experiential subscale, but not in the expressive subscale. Collectively, the findings from chapters 3, 4 and 6 all point to the conclusion that there is a significant reduction in negative symptoms over time in stable outpatients diagnosed with schizophrenia, including in samples which are selected to minimise the impact of secondary negative symptoms. However, due to the inconsistent evidence it is unclear whether this reduction is consistent across all types of negative symptoms, or whether expressive and experiential symptoms follow a different trajectory.

The findings that negative symptoms significantly reduce over time are contrary both to the earliest conceptions of schizophrenia, which suggested that negative symptoms follow a path of progressive deterioration (Bleuler, 1950, Kraepelin, 1971), and our current understanding of negative symptoms which suggest that they are highly stable in the non-acute phase (Möller, 2007). Instead, these results appear to mirror the findings of a study by Quinlan (Quinlan et al., 1995), who evaluated the longitudinal course of symptoms in chronic schizophrenic outpatients. Over two years significant reductions in negative symptoms were detected, leading Quinlan to propose that the notion of a progressive downward course of schizophrenia may need to be reconsidered.

In the assessment of individual negative symptoms in chapter 3, the reduction in alogia, anhedonia, blunted affect and avolition all appeared very similar. These findings are in contrast with the findings in chapter 6 where significant reductions in experiential, but not expressive symptoms were evident. The reasons for the differences between these findings are not clear. One possible theory could be that the non-significant reduction in expressive symptoms detected in chapter 6 could be attributable to floor effects, given the level of expressive symptoms were relatively low at baseline (mean= 1.94, SD= 0.92). However, it also possible that the lack of any differences between the types of symptoms uncovered in chapter 3 may be attributable the limitations of the assessment scales used. As stated previously, the SANS and PANSS is recognised to merge conceptually distinct experiential symptom domains into single subscales such as anhedonia-asociality, in addition to including features now not recognised to be part of the negative symptoms construct (Blanchard et al., 2011). In addition, these scales insufficiently tap into experiential symptoms of the disorder (Blanchard et al., 2011, Kring et al., 2013), a theory further supported by the findings presented in chapter 5 which only found a moderate association between the PANSS negative and CAINS experiential subscales (r=.480). This being the case, it is quite possible that a reduction in the experiential symptoms may be under-reported when these scales are used. The third possibility it that expressive symptoms such blunted affect and alogia are more stable than experiential symptoms such as anhedonia, avolition and asociality. Greater stability in expressive symptoms have been reported in the past (Johnstone et al., 1987; Meuser et al., 1991; Dollfus and Petit, 1995; Kelley et al., 2008), possibly because they are less affected by causes of secondary negative symptoms such as positive symptoms and medication side effects (Angrist et al., 1980; Kelley et al., 1999).

Participants who were allocated to receive SGA's or adjunctive medications reported a greater reduction in negative symptoms, relative to those allocated to other conditions. At first glance,

these findings appear to suggest that these medications may be more effective in the treatment of negative symptoms, contrasting with earlier studies which have not found a consistent difference in the efficacy of SGA's over other treatments (i.e. Leucht et al., 2009; see section 1.7 for a review). However, this finding is complicated by the fact that in the placebo/control/treatment as usual samples a greater reduction in negative symptoms were detected in those that were part of a second generation antipsychotic or adjunctive medication drug trial. This suggests that there may be undetermined features of these types of studies which result in a greater change in negative symptoms being detected, rather than the effectiveness of the treatments themselves. Given a number of the SGA and adjunctive drug studies that were completed more recently, it is possible that this finding relates to an increasing placebo effect which has been found in psychiatric pharmacological trials (Kinon et al., 2011, Rutherford and Roose, 2013, Walsh et al., 2002).

8.2.2. The impact of adopting different symptom inclusion criteria in negative symptom trials

In the meta-regression completed in chapter 3, studies that adopted a minimum negative symptom inclusion criterion reported a small but significant increase in the reduction of negative symptoms over time. However, this difference was not significant in the multivariate model. In chapter 4, a substantial reduction in negative symptoms over time was evident as the minimum negative symptom eligibility criteria increased. These reductions were present despite maximum depressive and positive symptom eligibility criteria also being adopted, which should theoretically minimise secondary symptoms and so increase the stability of negative symptoms over time (Möller, 2007). The reason why the minimum negative symptom inclusion criterion appeared to have a much greater impact in the analysis completed in chapter 4, relative to the meta-regression completed in chapter 3 is not clear. One possibility is that in chapter 3 the substantial variation in the manner in which these criteria are implemented, in addition to the large degree of heterogeneity between the pooled studies themselves may have resulted in a number of extraneous factors and a substantial loss of power, masking any differences which may exist. If this is the case, then the findings of chapter 4 may be a closer representation of the impact of adopting such criteria.

Over the different inclusion criteria examined in chapter 4, the association between positive and negative symptoms appeared to be largely consistent regardless of how restrictive the inclusion criteria were. Of the 11 criteria examined, the only difference that appeared to exist

was when predominant criteria were determined by negative symptoms being over positive symptoms by a specified amount. In these cases (Olie et al., 2006, Rabinowitz et al., 2013) the association between positive and negative symptoms substantially increased. These findings are broadly consistent with those reported by Dunayevich (Dunayevich et al., 2014), who found that adopting different inclusion criteria did not minimise the association between positive and negative symptoms . In other work, Stauffer (Stauffer et al., 2012) found that negative symptom change was not different between participant samples that reported prominent or predominant negative symptoms. Whilst these findings merit further exploration, they point to the conclusion that most symptom inclusion criteria adopted in clinical trials may not be effective in reducing the association between positive in negative symptoms and, based on the findings of the current study, may even increase it in some cases.

Regarding the association between depressive and negative symptoms, it is interesting to note that this was reduced both by implementing a minimum negative, and maximum depressive symptom criterion. In the whole sample the association between negative and depressive symptoms was found to be relatively large. This association was found to decrease slightly when all participants with at least moderate depressive symptoms were omitted, which in turn resulted in a small reduction in the severity of negative symptoms. This finding supports earlier work which suggests that high levels of depressive symptoms may either mimic or exacerbate negative symptoms (Carpenter et al., 1985). However, a larger decrease in the association between these symptoms occurred when a minimum negative symptom criterion was implemented, the reason for which is not entirely clear. One theory is that implementing a minimum negative symptom threshold means that participants would be required to present with a broader range of negative symptoms, some of which have not found to be significantly elevated in depressed verses non-depressed schizophrenia patients such as affective flattening (Barnes et al., 1989). A second possibility is that participants may need to present with more severe individual symptoms, making it easier to distinguish between depressive and negative symptoms. One example of this may be in anhedonia, where an inability to experience pleasure is a feature of both negative symptoms and depression. However, in depression this manifests itself as an overall anhedonic deficit, whilst in negative symptoms this is limited specifically to anticipatory pleasure, with the experience of sensory pleasure remaining largely intact (Gard et al., 2007, Strauss et al., 2011). Given this distinction is somewhat nuanced, it is likely to be more challenging to determine when the symptoms experienced are relatively mild. A reduction in the association between negative symptoms and depression was similar when a number of individual negative symptoms were required to be above a mild level of severity, and in criteria that specified an overall negative symptom subscale score be over a

certain threshold, making it difficult to determine whether either (or if indeed both) mechanisms may be involved.

The findings in chapter 4 appear to support the conclusions outlined in the NEWMEDS consensus statement (Marder et al., 2013). In this report it was recommended that participants who present with depressive symptoms that do not overlap with negative symptoms should be excluded from negative symptom clinical trials, whilst no consensus was reached regarding excluding participants with high positive symptoms. This conclusion is in contrast to the NIMH-MATRICS consensus statement (Kirkpatrick et al., 2006), the criteria of persistent negative symptoms outlined by Buchanan (Buchanan, 2007), and the current guidelines of the European Medicines Agency (EMA., 2012) which all recommend excluding participants with at least moderate levels of positive symptoms.

8.2.3. A comparison of the different negative symptom assessment scales.

In chapter 3 participants that were assessed using the SANS were found to report a significantly greater reduction in negative symptoms, relative to those assessed by either the PANSS negative subscale or the BPRS anergia subscale. This difference was relatively large, even after controlling for a range of other study-level factors (ES=0.43). In chapter 5, the CAINS was found to provide a greater differentiation between participants that report high and low symptoms, and reported a greater change in negative symptoms relative to the PANSS as the mean change in both scales increased. In addition, the CAINS was found to be a stronger predictor in a range of outcomes relating to social impoverishment.

The findings in chapter 3 and 5 suggest that different assessment tools can result in a large variation in the severity and change in negative symptoms reported. This supports the conclusions of the recent MATRICS consensus report which suggested that inappropriate scales may act as a significant barrier to the development of new treatments (Kirkpatrick et al., 2006). They concluded that new scales should be developed that adhere to our current conception of the negative symptom construct, and be a more sensitive instrument to detect symptom changes. Prior to developing the CAINS and the BNSS, a review of the negative symptom construct outlined the conceptual requirements of the scales (Blanchard et al., 2011). In follow up work, the CAINS was developed, evaluated and refined, with the final scale being found to have excellent psychometric properties (Forbes et al., 2010, Horan et al., 2011, Kring et al., 2013). In the current study, the findings suggest that the CAINS may be a more

sensitive instrument to detect changes in negative symptoms relative to the PANSS, thus fulfilling the final parameter outlined in the consensus report (Kirkpatrick et al., 2006).

The findings from chapter 3 which suggest that the SANS is a more sensitive instrument to detect negative symptom change over the PANSS and the BPRS is in consistent with recommendations outlined in the MATRICS consensus statement (Kirkpatrick et al., 2006). In the report, they suggest that the SANS may be preferable given multiple items are used to measure separate constructs which should improve the psychometric properties of the scale. However in the same report it is also recognises that the scale has a number of limitations, including items which are not considered part of the negative symptom construct such as attention deficits, inappropriate affect, and poverty of speech content (Buchanan and Carpenter, 1994), and not specifying between anticipatory and consummatory hedonic deficits (Gard et al., 2007). If the advantages of the SANS are attributable to the fact that multiple items are used (as proposed by Kirkpatrick and colleagues) (Kirkpatrick et al., 2006), then it follows that this same advantage would also extend to other scales with adopt multiple items such as the CAINS.

8.2.4. The Impact of adopting different eligibility criteria on the sample pool size

Consistent with the findings from Rabinowitz and Dunayevich (Dunayevich et al., 2014, Rabinowitz et al., 2013), the results of chapter 4 suggest that adopting more restrictive inclusion criteria can result in a substantial reduction in the eligible sample pool. In the current study, adopting the most restrictive criteria (Moller et al., 2004) resulted in only 6.6% of the total sample being eligible. This figure is comparable to similar studies by Dunayevich and Rabinowitz which found that only 6.0% and 8.1% were eligible in the most stringent criteria adopted. This is in stark contrast to the broadest criteria examined in the current investigation, which resulted in 50% of the sample still being eligible. With so many participants deemed ineligible in the most restrictive criteria, this could have two major implications. Firstly, excluding so many potential participants can make it substantially harder to recruit the required number of participants, which is already recognised to be a challenging and complex issue for clinical trials (McDonald et al., 2006). Secondly, it could potentially result in trial samples not being sufficiently generalisable to populations that these treatments would ideally be designed to treat. Over 60% of patients have been found to present with at least one moderate negative symptom in routine care (Bobes et al., 2010), and between 20-30% are thought to present with deficit syndrome (Kirkpatrick et al., 2006). In the current study, the

participants were older (mean age 42.1 years), where the prevalence of deficit syndrome in clinical populations is thought to be even higher (37% in those over 45) (Harris et al., 1991). Therefore, it is likely that implementing such criteria is likely to result in a substantial proportion of participants who either present with persistent negative symptoms or deficit syndrome not being eligible for negative symptom trials that adopt these criteria, despite the fact that the treatments under evaluation would be designed precisely for this patient group. Utilising more broad criteria may aid ease of trial recruitment, whilst also producing study samples which are more generalisable to the treatment target population. It is possible that this lack of generalisability may in part explain why such a difference was found between pragmatic trials which utilised broader criteria such as the CUtLASS (Jones et al., 2006) and the CATIE (Lieberman et al., 2005) studies, compared to the early industry studies.

8.2.5. What is nature of the relationship between the CAINS and PANSS negative subscale?

In chapter 5, the CAINS total and the PANSS negative subscale was found to be highly correlated (r=.742). Consistent with earlier studies which examined the relationship between the BPRS and the CAINS (Horan et al., 2011), the PANSS negative subscale was found to relate primarily to the expressive features of negative symptoms (r=.754), with only a moderate correlation between experiential subscale and the PANSS negative subscale present (r=.480). These findings are consistent with the conclusions outlined in a review by Blanchard and colleagues (Blanchard et al., 2011), which suggested that older scales such as the PANSS insufficiently tap into experiential features of negative symptoms, instead relying primarily on behavioural referents. These referents can include the frequency of activities, interviewer observations, and reports from family members or carers, and mean the experiential features of symptoms such as anhedonia and avolition are not directly addressed. This is important, as it means that link between outcomes with negative symptoms that one would expect to relate more to experiential than expressive deficits may have been under-reported in studies that have used older scales. This conclusion is supported by the findings presented in chapter 6, where significant associations were detected between experiential deficits and subjective quality of life using the CAINS, whilst no relationship was detected using the PANSS negative subscale.

8.2.6. The association between the negative symptom construct and other variables when measured by the CAINS, PANSS and BPRS

In chapter 7, negative symptoms were not found to be associated with subjective initial response. In chapter 5, significant associations between negative symptoms and indicators of social impoverishment such as the number of activities participants took part in over the past week, the number of friend contacts made over the past week, and whether they had anyone they considered a close friend were detected. In the majority of cases, the CAINS was found to be a better predictor of these outcomes than the PANSS negative subscale score. In chapter 6, no relationship was detected between negative symptoms and subjective quality of life when it was measured as a singular construct using the PANSS, however significant associations between experiential deficits and SQOL were present.

In chapter 5, the CAINS total subscale was found to be a better predictor of the number of friends they have made contact with, the number of activities they have taken part in over the past week, and whether participants reported having a close friend. Given the PANSS has been criticised for focusing primarily on behavioural referents (Blanchard et al., 2011), it is somewhat surprising that while the scale explained only a minimal degree of unique variance in the two behavioural outcomes (both =.001), at least some unique variance was explained in the subjective outcome which related to whether participants had somebody they would consider as being a close friend (=.007). This was more apparent in the comparison between the PANSS Marder negative subscale and the CAINS, where similarly little unique variance was explained by the PANSS subscale in the two behavioural outcomes (both <.001), but explained a relatively large proportion in reporting whether they had a close friend (=.017). In this item, the degree of shared and unique variance explain by the PANSS Marder negative subscale was slightly higher than the CAINS total score, with the Marder subscale being dominant in 62.7% of cases.

These findings point to a number of conclusions. Firstly, the PANSS Marder negative subscale was found to be a superior predictor of whether participants have a close friend whilst the PANSS negative subscale was not, lending further support to recommendations which suggest that the Marder subscale may be a more appropriate method to adopt in clinical trials evaluating negative symptoms (Marder et al., 2011, Marder et al., 1997). Secondly, despite the PANSS Marder negative subscale being marginally better than the CAINS at predicting whether participants report have a close friend, in the other 5 comparisons made the CAINS total score was found to be dominant with the PANSS typically explaining little to no additional variance.

This suggests that the CAINS should be considered as superior at predicting functional impairments related to the negative symptom construct. Somewhat contrary to expectations, this advantage appears to be particularly strong in predicting more concrete, behavioural outcomes related to the negative symptom construct. This is important, as determining the incremental validity of new psychometric scales should be considered a crucial step in the scale development process (Hunsley and Meyer, 2003), despite it often being neglected (Haynes and Lench, 2003).

In chapter 7, higher negative symptoms were not found to result in a lower subjective initial appraisal of inpatient psychiatric treatment. Instead, it was found that positive and manic symptoms were associated with a more negative assessment. Given the impact of subjective initial appraisals on various social and psychopathological outcomes (Bröker et al., 1995, Priebe et al., 2010a, Richardson et al., 2011a) this study suggests that a greater consideration of these symptoms during the admission stage may help longer term outcomes. These findings are partly consistent with the results of Richardson and colleagues (Richardson et al., 2011a) who reported a less positive initial treatment appraisal for patients with higher levels of mania, positive and depression/anxiety symptoms. The findings from the current study are in contrast to a body of literature in psychiatry which has typically found that mood symptoms have the greatest impact on patient reported outcomes (Priebe, 2007). It has been suggested that this reflects a negative rating bias of patients with high levels of depression, rather than a specific experience of treatment (Fakhoury et al., 2002, Hansson et al., 2007, Priebe et al., 1998).

One possibility for the finding that positive and manic symptoms lead to a more negative assessment of treatment may relate to how these symptoms might impact upon being placed in the confined environment of a hospital ward. In a qualitative review of involuntary inpatient experiences of treatment (Katsakou and Priebe, 2007), a number of features of treatment were recognised to negatively impact upon a patient's experience of admission. Of these experiences, a number may particularly resonate with those suffering more intense manic and positive psychotic symptoms. Patients experiencing greater levels of positive symptoms such as paranoia or hallucinations may struggle more in an unfamiliar setting which may feel frightening or insecure. Patients with severe manic symptoms are likely to be more agitated, and struggle more with being contained in an enclosed space with rigid rules and limited opportunities for activities. The finding that manic symptoms influence patients' appraisal even more in those involuntarily admitted, who therefore would experience an even greater restriction to their autonomy, lends further support to this argument. Patients who exhibit 'disturbed behaviour' or experience acute positive symptom exacerbation are also more likely

to receive combined antipsychotics and high doses of antipsychotic medication (Paton et al., 2008), which can lead to a higher side effect burden (Centorrino et al., 2004), potentially impacting the appraisal of the treatment they receive further.

8.2.7. The association between negative symptoms and other constructs when expressive and experiential constructs are considered separately

Whilst the relationship between negative symptoms of schizophrenia and subjective quality of life (SQOL) has been examined in the past (Eack and Newhill, 2007, Priebe et al., 2011b) this is the first time the link between SQOL and expressive and experiential deficits as separate constructs have been considered. As anticipated, the association between experiential deficits and SQOL detected in this investigation appears to be significantly stronger than in previous studies which have examined the relationship to negative symptoms as a singular construct. This stronger association was found both cross-sectionally and in changes over time, and in contrast to earlier investigations, remained present after controlling for depressive symptoms. These findings support the current developments in negative symptom assessment which advocate measuring experiential and expressive symptoms as distinct negative symptom constructs (Blanchard and Cohen, 2006, Horan et al., 2011, Kring et al., 2013), suggesting that new insights in the field of negative symptoms may be uncovered by utilising the new CAINS assessment instrument. In addition, this finding suggests that if the primary aim of treatment is to improve the patients' quality of life, then the focus on treatment should be on the experiential, as opposed expressive features of the illness.

In the comparison of the PANSS negative and CAINS subscales, the experiential scale was found to be the strongest predictor of the number of friends contacts and social activities completed in the past week, in addition to whether participants report having somebody they consider to be a close friend. The finding that the expressive subscale explains little variance in these outcomes is not surprising, given the outcomes evaluated would be considered to relate more to experiential deficits, such as asociality and avolition. One interesting finding was that in the outcome concerning the number of reported friend contacts, the experiential subscale was found to explain a slightly greater proportion of variance than the whole CAINS subscale $(R^2=.118$ in comparison to $R^2=.091$), further supporting the argument that in some situations it may be beneficial to utilise the subscales separately.

Overall, these findings support an emerging evidence base which suggests that experiential and expressive symptoms represent distinct constructs (Blanchard and Cohen, 2006), and as such should be measured separately (Horan et al., 2011). Not only does assessing these symptoms in this manner better resemble the negative symptom construct as it is currently understood, the findings here suggests that assessing symptoms in this manner may yield new insights which may have previously been missed when negative symptoms were measured as a singular concept.

8.3. Project Strengths and Limitations

One strength of this study is that the investigation comes at a time when there are significant developments in the field of negative symptoms of schizophrenia. Recent consensus statements have placed increasing focus on refining methodological and conceptual issues around trials designed to evaluate treatments for negative symptoms (Marder et al., 2013, Marder et al., 2011). The work of this project builds upon these findings, providing further insight on how negative symptoms may change over time in the context of a research trial, and the impact of different symptom inclusion criteria on the associations between negative symptoms and factors known to exacerbate these symptoms. In addition, the current investigation uses date from one of the first large-scale studies to use the CAINS, which have been considered an important advance in the field. Whilst earlier validation studies suggest that the scale as excellent psychometric properties (Horan et al., 2011, Kring et al., 2013), this investigation suggests the scale be a more sensitive instrument, be a stronger predictor of social and functional impairment, and by disentangling expressive and experiential deficits, may lead to further important insights in the field.

One significant limitation of the investigation as a whole is that all of the analysis has been completed using data from existing studies, either through pooling them by way of metaanalysis, or conducting analysis on datasets obtained from previous studies. Consequently, the scope of the analysis was limited by the nature of the data available. In chapter 3, an insufficient number of studies used a maximum level of EPS as an inclusion criterion for eligibility, meaning the impact of these criteria on negative symptom change could not be included in the meta-regression. In addition, a substantial degree of heterogeneity between the pooled studies was unexplained, which resulted in considerable imprecision in the estimates. This may account for the differences in the findings between chapters 3 and 4, where minimum negative and depressive symptom criteria were evaluated. In chapter 4, whilst the original dataset was relatively large (N=507), there was insufficient power to obtain

meaningful estimates from the most restrictive criteria evaluated due to the fact that the trial had not been powered for this purpose. In addition, the lack of any assessment of EPS meant these could not be considered as a third factor associated with secondary negative symptoms. In this sample the mean DDD was found to be relatively high, meaning it is possible that some participants may have experienced notable EPS. As a result, it is possible that parkinsonian symptoms such as akinesia may have mimicked expressive symptoms, resulting in a higher rating of negative symptoms that may not represent core psychopathology. Finally, using an assessment scale for depressive symptoms designed specifically for schizophrenia populations, such as the Calgary (Addington et al., 1992), would have been preferable given the issues in disentangling depressive and negative symptoms with the PANSS (Collins et al., 1996).

Given the meta-regression from chapter 3 had found that the SANS may be a more sensitive instrument to detect negative symptom change, relative to the PANSS, in chapter 5 it may have been preferable to compare the SANS to the CAINS in the analysis. However, given the extensive use of the PANSS in negative symptom treatment trials, as evidenced in the meta-analysis in chapter 3, determining the relationship between the two scales should still be regarded as an important step forward. In chapter 7, given the BPRS provides a relatively poor coverage of the negative symptom construct, the PANSS may have been a preferable assessment tool, given factor analytic studies have suggested the scale can provide relatively good coverage of various symptom clusters (see section 1.9.1.3. for a review).

Whilst prospective studies designed to address the stated aims would have meant that many of these limitations could have been addressed from the outset, this would have required significant resources far in excess of what would be available in an investigation of this scale.

8.4. Implications

8.4.1. Implications on research

The findings from chapters 3, 4 and 6 all suggest that negative symptoms appear to reduce over time to a greater extent than originally presumed, with the meta-analysis suggesting that there is a substantial degree of heterogeneity in the rate of change that exists between studies. These results could be considered to have a number of important implications to future research. Firstly, these findings underline the importance of comparing outcomes to an appropriate control, and the difficulty in determining whether within-group reductions found

in negative symptoms are attributable to treatment. This is important, given the assumption that negative symptoms are highly stable and treatment-resistant have previously led some researchers to conclude that within-group pre-post reductions suggest a clinically significant improvement as an effect of treatment (Sensky et al., 2000). Secondly, if there is more variability and a greater reduction in negative symptoms over time, and/or a larger placebo effect than previously assumed, then this suggests it may be harder to detect treatment effects than was previously thought, suggesting larger samples sizes may be required in some cases.

Building upon recent developments in our understanding of the negative symptom construct (Blanchard and Cohen, 2006, Blanchard et al., 2011), the CAINS is considered to represent a significant advance on older scales used to measure these symptoms (Kring et al., 2013). In an assessment of incremental validity, the CAINS was found to be more strongly associated to functional impairments related to negative symptoms, relative to the PANSS negative subscale. In chapter 6, the CAINS experiential subscale was negatively associated with subjective quality of life, whilst no relationship to the PANSS negative subscale was detected. In chapter 5, the CAINS was found to be a more sensitive than the PANSS, both in differentiating between participants with high and low negative symptoms, and in measuring changes in these symptoms over time. Given the limitations of earlier negative symptom assessment scales have been identified as a major barrier in the development of new treatments (Kirkpatrick et al., 2006), these findings suggest the CAINS may be a superior scale to adopt in clinical trials for negative symptom treatments. If so, then this may represent an important advance in treatment evaluation in an area which been identified as an unmet therapeutic need (Kirkpatrick et al., 2006). The findings from chapters 5 and 6 also suggest that the ability of the CAINS to evaluate expressive and experiential deficits as separate constructs may open up new insights and opportunities in the research of negative symptoms.

The results of chapter 4 suggest that adopting minimum negative and maximum depressive symptom study inclusion criteria reduces the association between negative and depressive symptoms. This supports the recommendations outlined in both the MATRICS and NEWMEDS reports (Kirkpatrick et al., 2006, Marder et al., 2013), and highlight the importance and effectiveness of adopting such criteria to minimise secondary negative symptoms attributable to depressive symptomology in future clinical trials. By contrast, adopting predominant negative symptom criteria was not found to reduce the association between negative and positive symptoms, and in most cases resulted in a substantial reduction in the sample pool size. If these inclusion criteria are not effective, then it suggests that alternative methodologies

may need to be considered to disentangle co-occurring changes in positive symptoms from negative symptoms in clinical trials for negative symptom treatments. Once such method may be to adopt a path analysis design, which has been adopted in a number of earlier pharmacological studies evaluating SGA's for negative symptoms (Alvarez et al., 2006, Moller et al., 1995, Tollefson and Sanger, 1997). However, such a method comes with the drawback that much larger samples may be required (MacCallum et al., 1996).

The findings in chapter 6 suggest the association between negative symptoms and SQOL may be significantly stronger than previously assumed, however the relationship appears to relate specifically to experiential subdomain of these symptoms. Such findings appear to support a model proposed by Priebe (Priebe, 2007) which suggests that quality of life may be improved by treatment indirectly via an improvement in symptoms. Given improvements in quality of life are considered central to the concept of recovery (Liberman et al., 2002) these findings suggest it is important to develop and evaluate new treatments which focus on the experiential symptoms of the negative symptom construct in particular. Given the initiation and development of social relationships are seen as a key feature of all resource-orientated models of psychiatric interventions (Priebe et al., 2014), such treatments may represent a promising area which could address experiential deficits such as amotivation and asociality, and merit further consideration.

8.4.2. Implications on policy

At present, both the FDA and EMA policy guidelines state that for negative symptom treatments to be considered eligible for approval then their effectiveness needs to be determined in clinical trials which evaluate negative symptoms as a singular construct (Marder et al., 2011). However, there is increasing evidence to suggest that negative symptoms comprise of at least two distinct domains, namely expressive and experiential symptoms (Blanchard and Cohen, 2006, Horan et al., 2011, Kring et al., 2013). In the current investigation completed in chapters 5 and 6, experiential deficits were found to be uniquely associated to objective indicators of social impoverishment, such as the participation of activities and friend contacts made over the past week, in addition to other important outcomes such as impaired subjective quality of life. Such links between experiential deficits and functional disability have been noted elsewhere, and have in turn led to efforts towards developing treatments specifically for experiential symptoms (Reddy et al., 2015). If such treatments prove to be effective at reducing experiential symptoms, and any functional disability which derive from

these symptoms, then the current policy guidelines may merit reconsidering. As outlined in the MATRICS consensus statement (Kirkpatrick et al., 2006), if the subdomains of negative symptoms respond differently to treatment, then it is possible that a combined negative symptom assessment may underreport clinically meaningful improvements in an individual symptom.

The current EMA guidelines require clinical trials evaluating adjunctive medications for negative symptoms to recruit participants that present with predominant, as opposed to prominent negative symptoms (EMA., 2012). The rationale for implementing such a policy is to determine that changes in negative symptoms are not attributable to co-occurring changes in positive symptoms. However, in chapter 4 it was found that adopting various different maximum positive symptom exclusion criteria did not reduce the association between positive and negative symptoms, and in some cases actually increased it. Furthermore, the increasingly restrictive criteria adopted appeared to result in a substantial reduction in the available sample pool which may have implications on trial recruitment and the generalisability of findings. Given that adopting maximum positive symptom criteria was found to be ineffective, these findings suggest that if the aim is to reduce the association between the variables then alternative methods may need to be proposed in new policy guidelines.

8.4.3. Implications on practice

The findings from chapters 3, 4, and 6 all suggest that negative symptoms may significantly reduce over time in stable outpatients, even after adopting criteria which aims to reduce the impact of secondary negative symptoms. Whilst acknowledging that the changes in negative symptoms were relatively small, and the fact it is difficult to determine what the nature of this measured change actually is, this improvement over time appears to lend support to the recovery model of schizophrenia (Warner, 2009), given these symptoms links to functional outcomes (Hunter and Barry, 2012, Lysaker and Davis, 2004, Whitty et al., 2008). Such findings may provide hope to those that experience these symptoms, in addition to those who work with patients to help try and alleviate the impact of this devastating disorder.

The findings in both chapters 5 and 6 suggest that the experiential and expressive subdomains of negative symptoms appear to be related to outcomes differently. In support of these findings, the link between functional disability and experiential symptoms in particular has been noted in the literature (Reddy et al., 2015). In chapter 5, experiential symptoms were

found to have a stronger inverse relationship to how socially engaged the participants reports to be, relative to experiential symptoms. In chapter 6, experiential symptoms were found to be significantly associated to subjective quality of life, whilst no relationship to expressive symptoms was apparent. Given the the ability to maintain satisfying relationships and participate in productive and enjoyable activities are seen as central to the concept of recovery (Liberman et al., 2002), these findings suggest that extra consideration should be given to addressing experiential symptoms in routine practice.

The findings in chapter 7 emphasise the importance of measuring symptoms at the point of inpatient admission, and considering the impact of manic and positive symptoms in particular during the initial phase of acute treatment. Whilst no association between negative symptoms and initial appraisal of impatient was detected, the results underline the importance of assessing patients' initial appraisal of treatment as a relevant process variable in routine practice. In a recent review Mullen (Mullen, 2009) has suggested that acute inpatient settings are at present too custodial and over reliant on medication, and suggests that providing more psychosocial interventions as an alternative may be a way in which to try and address these issues. If these are problems which disproportionally affect those experiencing higher positive and manic symptoms, then this possible solution may go some way to addressing the experiences these particular patients report.

8.5. Future directions

The results in this investigation point to a number of possible future directions for research. The findings in chapter 3, 4 and 6 all suggest that negative symptoms improve during the study period. However, understanding the nature of the improvement in negative symptoms over time requires further attention. The duration between time points was not found to be a significant predictor, but with the considerable variability in treatment duration, post treatment follow-up duration, and post treatment provision between studies, this is perhaps not surprising. In longitudinal studies assessing persistent negative symptoms over very long periods of time, there has been little evidence of a gradual linear improvement towards symptom remission (Strauss et al., 2010), whilst the rate of recovery in schizophrenia remains low (Jääskeläinen et al., 2013). One explanation of the improvement in negative symptoms noted throughout this investigation could be the non-specific effects of being involved in the research, possible related to the "trial effect" (Braunholtz et al., 2001). However, the mechanism that drives change in this context is unclear. One possibility is that a reduction in

negative symptoms found between the first and subsequent assessments may be a consequence of increased familiarity with the assessor and the assessment process itself. If a good rapport was established in the first assessment, then it is possible that this may result in the participant being less anxious in subsequent assessments. This is significant, given symptoms of anxiety in schizophrenia are positively associated with passive withdrawal and poorer psychosocial functioning (Lysaker and Salyers, 2007). If this is the case, then it suggests that the improvement detected between assessments may be secondary rather than primary, in nature. Studies exploring the possibility that secondary negative symptoms may be more prominent in initial assessments may be merited, given the importance attached to minimising transient secondary negative symptoms in clinical trials of treatments for these symptoms (Kirkpatrick et al., 2006).

Another possible reason for the reduction in negative symptoms typically found in research settings, regardless of the type of intervention administered, may be related to the type of social interactions which occur within these studies. If the improvement in negative symptoms is related to increased social contact during studies, then treatments that promote social interactions such as befriending schemes and group psychotherapies may be a possible step forward. Recent findings suggesting a non-specific effect of group therapies on negative symptoms (Orfanos et al., 2015) appear to support this possibility. As a result, additional studies which examine the impact of these more socially-oriented treatments may be informative.

Whilst the meta-analysis did consider studies that implemented criteria which aimed to limit the impact of secondary negative symptoms, none used criteria which would qualify as determining a diagnosis of deficit syndrome (Carpenter et al., 1988, Kirkpatrick et al., 1989). The principle reason for this is that studies that did adopt these criteria often included participants with schizoaffective disorder or inpatients (Buchanan et al., 2007). Amending the inclusion criteria to focus on these particular studies would allow for a greater understanding of any change in primary negative symptoms. Due a lack of available data the impact of extrapyramidal side effects on the variability of negative symptom change could not be explored, which would be informative in further determining longitudinal course. In addition, as more studies are completed using the newer BNSS and CAINS scales, an evaluation of these tools would be helpful in determining their sensitivity to other assessment tools.

The findings in chapter 4 suggest that adopting a maximum positive symptom exclusion criterion does not appear to result in a reduction in the association between positive and

negative symptoms. However, given much of the criteria evaluated resulted in a substantial number of participants being excluded, the samples on which many of the estimates were based on were small. Therefore, replicating this study in a far larger sample would be informative, particularly given larger sample sizes would allow for a greater comparison between the different R^2 values through calculating narrower confidence intervals. Larger sample sizes would also allow for the possibility of conducting path analysis, looking at the proportion of change in negative symptoms over time associated with co-occurring changes in positive, depressive, and extra-pyramidal symptoms in subsamples obtained using the different eligibility criteria.

The results in chapter 5 suggest that the CAINS may be a more sensitive instrument than the PANSS in measuring changes in negative symptoms. However, given the results from the meta-regression in chapter 3 suggest that the SANS may also be more sensitive than the PANSS, a comparison of the CAINS and the SANS would be informative in determining whether the new scale represents a significant advance over all previously available scales. Another step forwards would be a comparison of the results obtained from different scales within a clinical trial where a treatment effect is present. If the CAINS produces larger effect sizes than other scales used, then this may go some way improving the evaluation of treatments designed to treatment these symptoms.

In chapter 6, the finding that experiential symptoms were primarily linked to subjective quality of life suggest that it may be beneficial to revisit the link between the different subdomains of negative symptoms and other functional outcomes. In addition, it emphasises the pressing need to develop treatments that are particularly effective at treating experiential symptoms, namely asociality, anhedonia and avolition. Following the findings in chapter 7, further studies may explore environmental factors, processes of interaction between service users and staff, treatment components, and mediating processes as an explanation into the association of manic and positive symptoms with more negative appraisals of hospital treatment after only a few days.

8.6. Concluding Statement

Over the years the concept of negative symptoms have undergone a number of important revisions, from the "avolitional syndrome" proposed by Kraepelin (Kraepelin, 1971), to the current model of negative symptoms which include alogia, blunted affect, avolition, anhedonia

and asociality (Blanchard et al., 2011). The findings of this thesis suggest that the CAINS, which was developed to better represent our current interpretation of the negative symptom construct (Kring et al., 2013), may represent a significant advance in older assessment tools. This is important, as the limitations inherent to earlier assessment scales have been identified as a significant barrier in the development of new treatments for negative symptoms (Kirkpatrick et al., 2006).

In this study CAINS was found to be a more sensitive instrument in detecting negative symptom change, reported a greater differentiation between participants who reported high and low symptoms, and reported stronger association to functional outcomes and SQOL relative to the PANSS negative subscale. The fact that the CAINS is designed to measure the expressive and experiential features of negative symptoms as distinct constructs may be particularly important, given the evidence presented here which suggests the subdomains may be independently associated to functional outcomes. The finding that it is only the experiential deficits of the negative symptom construct that are associated to SQOL suggest that more targeted treatments which focus on these particular symptoms may represent a new way forward in supporting the recovery of patients with schizophrenia.

Overall, many of the findings in this investigation could be considered to offer renewed optimism to those who experience the negative symptoms of schizophrenia. Collectively the results suggest that negative symptoms may improve more than was originally assumed, challenging earlier models of schizophrenia which have either suggested negative symptoms are either highly stable, or get progressively more severe. This improvement over time was found in stable outpatient samples with relatively low positive and depressive symptoms, suggesting that improvements in these symptoms may be achievable at all phases of the disorder.

References

- Adams, DH, Kinon, BJ, Baygani, S, Millen, BA, Velona, I, Kollack-Walker, S Walling, DP 2013. A long-term, phase 2, multicenter, randomized, open-label, comparative safety study of pomaglumetad methionil (LY2140023 monohydrate) versus atypical antipsychotic standard of care in patients with schizophrenia. *BMC Psychiatry*, **13**, 143.
- Addington, D, Addington, J Atkinson, MG 1996. A psychometric comparison of the Calgary Depression Scale for Schizophrenia and the Hamilton Depression Rating Scale. *Schizophrenia Research*, **19**, 205-212.
- Addington, D, Addington, J Maticka-Tyndale, E 1993. Assessing depression in schizophrenia: the Calgary Depression Scale. *The British Journal of Psychiatry*, **22**, 39–44.
- Addington, D, Addington, J, Maticka-Tyndale, E Joyce, J 1992. Reliability and validity of a depression rating scale for schizophrenics. *Schizophrenia Research*, **6**, 201-208.
- Addington, J Addington, D 1991. Positive and negative symptoms of schizophrenia: their course and relationship over time. *Schizophrenia Research*, **5**, 51-59.
- Addington, J Addington, DG 2000. Neurocognitive and social functioning in schizophrenia: A 2.5 year follow-up study. *Schizophrenia Research*, **44**, 47-56.
- Aguglia, E, De Vanna, M, Onor, ML Ferrara, DG 2002. Insight in persons with schizophrenia effects of switching from conventional neuroleptics to atypical antipsychotics. *Progress* in Neuro-Psychopharmacology & Biological Psychiatry, **26**, 1229-1233.
- Aleman, A, Kahn, RS Selten, J-P 2003. Sex differences in the risk of schizophrenia: evidence from meta-analysis. *Archives of General Psychiatry*, **60**, 565-571.
- Allan, ER, Sison, CE, Alpert, M, Connolly, B Crichton, J 1998. The relationship between negative symptoms of schizophrenia with extrapyramidal side effects with halperidol and olanzapine. *Psychopharmacology Bulletin*, **34**, 71-74.
- Alpert, M, Clark, A Pouget, ER 1994. The syntactic role of pauses in the speech of schizophrenic patients with alogia. *Journal of Abnormal Psychology*, **103**, 750-757.
- Alphs, L, Morlock, R, Coon, C, Cazorla, P, Szegedi, A Panagides, J 2011. Validation of a 4-item Negative Symptom Assessment (NSA-4): a short, practical clinical tool for the assessment of negative symptoms in schizophrenia. *International Journal of Methods in Psychiatric Research*, **20**, e31-e37.
- Alphs, L, Summerfelt, A, Lann, H Muller, R 1988. The negative symptom assessment: a new instrument to assess negative symptoms of schizophrenia. *Psychopharmacology Bulletin*, **25**, 159-163.
- Alptekin, K, Erkoc, S, Gogus, AK, Kultur, S, Mete, L, Ucok, A Yazici, KMG 2005. Disability in schizophrenia: Clinical correlates and prediction over 1-year follow-up. *Psychiatry Research*, **135**, 103-111.
- Alvarez, E, Ciudad, A, Olivares, JM, Bousono, M Gomez, JCG 2006. A randomized, 1-year followup study of olanzapine and risperidone in the treatment of negative symptoms in outpatients with schizophrenia. *Journal of Clinical Psychopharmacology*, **26**, 238-249.
- Amador, XF, Kirkpatrick, B, Buchanan, RW, Carpenter, WT, Marcinko, L Yale, SA 1999. Stability of the diagnosis of deficit syndrome in schizophrenia. *American Journal of Psychiatry*, **156**, 637-639.

- Andreasen, NC 1979. Thought, language, and communication disorders: I. Clinical assessment, definition of terms, and evaluation of their reliability. *Archives of General Psychiatry*, 36, 1315-1321.
- Andreasen, NC 1982. Negative symptoms in schizophrenia: definition and reliability. *Archives* of General Psychiatry, **39**, 784-788.
- Andreasen, NC 1983. Scale for the assessment of negative symptoms. Iowa City: University of Iowa.
- Andreasen, NC 1984. *Scale for the assessment of positive symptoms.* Iowa City: University of Iowa.
- Andreasen, NC, Arndt, S, Alliger, R, Miller, D Flaum, M 1995. Symptoms of schizophrenia: methods, meanings, and mechanisms. *Archives of General Psychiatry*, **52**, 341-351.
- Andreasen, NC Flaum, M 1991. Schizophrenia: the characteristic symptoms. *Schizophrenia Bulletin*, **17**, 27-49.
- Andreasen, NC, Flaum, M, Swayze, VW, Tyrrell, G Arndt, S 1990. Positive and negative symptoms in schizophrenia: a critical reappraisal. *Archives of General Psychiatry*, **47**, 615-621.
- Andreasen, NC Olsen, S 1982. Negative v positive schizophrenia: definition and validation. Archives of General Psychiatry, **39**, 789-794.
- Andreasen, NC, Olsen, SA, Dennert, JW Smith, MR 1982. Ventricular enlargement in schizophrenia: Relationship to positive and negative symptoms. *American Journal of Psychiatry*, **139**, 297-302.
- Andrew, A, Knapp, M, Mccrone, PR, Parsonage, M Trachtenberg, M 2012. *Effective interventions in schizophrenia: the economic case*, Personal Social Services Research Unit, London School of Economics and Political Science, London, UK.
- Angrist, B, Rotrosen, J Gershon, SJ 1980. Differential effects of amphetamine and neuroleptics on negative vs. positive symptoms in schizophrenia. *Psychopharmacology*, **72**, 17-19.
- Apa 2013. *Diagnostic and Statistical Manual of Mental Disorders (DSM-5®)*, American Psychiatric Pub.
- Arango, C, Garibaldi, G Marder, SR 2013. Pharmacological approaches to treating negative symptoms: A review of clinical trials. *Schizophrenia Research*, **150**, 346-352.
- Arndt, S, Andreasen, NC, Flaum, M, Miller, D Nopoulos, PJ 1995. A longitudinal study of symptom dimensions in schizophrenia. Prediction and patterns of change. Archives of General Psychiatry, 52, 352-360.
- Arns, PG Linney, JA 1993. Work, self, and life satisfaction for persons with severe and persistent mental disorders. *Psychosocial Rehabilitation Journal*, **17**, 63-79.
- Arseneault, L, Cannon, M, Poulton, R, Murray, R, Caspi, A Moffitt, TE 2002. Cannabis use in adolescence and risk for adult psychosis: longitudinal prospective study. *BMJ*, **325**, 1212-1213.
- Ascher-Svanum, H, Zhu, B, Faries, DE, Salkever, D, Slade, EP, Peng, X Conley, RR 2010. The cost of relapse and the predictors of relapse in the treatment of schizophrenia. *BMC Psychiatry*, **10**, 2.
- Awad, A, Hogan, T, Voruganti, L Heslegrave, R 1995. Patients' subjective experiences on antipsychotic medications: implications for outcome and quality of life. *International Clinical Psychopharmacology*, **10**, 123-132.

- Axelrod, BN Alphs, LD 1993. Training novice raters on the Negative Symptom Assessment scale. *Schizophrenia Research*, **9**, 25-28.
- Axelrod, BN, Goldman, RS Alphs, LD 1993. Validation of the 16-item negative symptom assessment. *Journal of Psychiatric Research*, **27**, 253-258.
- Azen, R Budescu, DV 2003. The dominance analysis approach for comparing predictors in multiple regression. *Psychological Methods*, **8**, 129.
- Azen, R Traxel, N 2009. Using dominance analysis to determine predictor importance in logistic regression. *Journal of Educational and Behavioral Statistics*, **34**, 319-347.
- Bales, A, Peterson, MJ, Ojha, S, Upadhaya, K, Adhikari, B Barrett, BG 2009. Associations between betel nut (Areca catechu) and symptoms of schizophrenia among patients in Nepal: A longitudinal study. *Psychiatry Research*, **169**, 203-211.
- Barch, D Berenbaum, H 1994. The relationship between information processing and language production. *Journal of Abnormal Psychology*, **103**, 241-250.
- Barch, DM, Sheline, YI, Csernansky, JG Snyder, AZ 2003. Working memory and prefrontal cortex dysfunction: specificity to schizophrenia compared with major depression. *Biological Psychiatry*, 53, 376-384.
- Barnes, TR 1989. A rating scale for drug-induced akathisia. *The British Journal of Psychiatry*, **154**, 672-676.
- Barnes, TR, Liddle, PF, Curson, DA Patel, M 1989. Negative symptoms, tardive dyskinesia and depression in chronic schizophrenia. *The British Journal of Psychiatry*, **7**, 99-103.
- Bartkó, G, Herczeg, I Békésy, M 1987. Predicting outcome of neuroleptic treatment on the basis of subjective response and early clinical improvement. *Journal of Clinical Psychiatry*, **48**, 363-365.
- Barton, R 1959. Institutional Neurosis. John Wright and Sons: Bristol. England.
- Beck, AT, Steer, RA Carbin, MG 1988. Psychometric properties of the Beck Depression Inventory: Twenty-five years of evaluation. *Clinical Psychology Review*, **8**, 77-100.
- Becker, BJ 1988. Synthesizing standardized mean-change measures. *British Journal of Mathematical and Statistical Psychology*, **41**, 257-278.
- Behere, RV, Arasappa, R, Jagannathan, A, Varambally, S, Venkatasubramanian, G, Thirthalli, J, Subbakrishna, DK, Nagendra, HR Gangadhar, BNG 2011. Effect of yoga therapy on facial emotion recognition deficits, symptoms and functioning in patients with schizophrenia. Acta Psychiatrica Scandinavica, **123**, 147-153.
- Bell, M, Milstein, R, Beam-Goulet, J, Lysaker, P Cicchetti, D 1992. The Positive and Negative Syndrome Scale and the Brief Psychiatric Rating Scale: Reliability, Comparability, and Predictive Validity. *The Journal of Nervous and Mental Disease*, **180**, 723-728.
- Bell, MD, Lysaker, PH, Beam-Goulet, JL, Milstein, RM Lindenmayer, J-P 1994. Five-component model of schizophrenia: assessing the factorial invariance of the positive and negative syndrome scale. *Psychiatry Research*, **52**, 295-303.
- Bellack, AS, Morrison, RL, Wixted, JT Mueser, KT 1990. An analysis of social competence in schizophrenia. *The British Journal of Psychiatry*, **156**, 809-818.
- Bellack, AS, Sayers, M, Mueser, KT Bennett, M 1994. Evaluation of social problem solving in schizophrenia. *Journal of Abnormal Psychology*, **103**, 371-378.
- Beneke, M Rasmus, W 1992. Clinical Global Impressions (ECDEU): some critical comments. *Pharmacopsychiatry*, **25**, 171-176.

- Bentall, RP 1990. The illusion of reality: a review and integration of psychological research on hallucinations. *Psychological Bulletin*, **107**, 82-95.
- Berardelli, A, Rothwell, J, Hallett, M, Thompson, P, Manfredi, M Marsden, C 1998. The pathophysiology of primary dystonia. *Brain*, **121**, 1195-1212.
- Berenbaum, H, Kerns, JG, Vernon, LL Gomez, JJ 2008. Cognitive correlates of schizophrenia signs and symptoms: I. Verbal communication disturbances. *Psychiatry Research*, **159**, 147-156.
- Berenbaum, H Oltmanns, TF 1992. Emotional experience and expression in schizophrenia and depression. *Journal of Abnormal Psychology*, **101**, 37-44.
- Berman, I, Viegner, B, Merson, A, Allan, E, Pappas, D Green, Al 1997. Differential relationships between positive and negative symptoms and neuropsychological deficits in schizophrenia. *Schizophrenia Research*, **25**, 1-10.
- Berner, P. 1983. *Diagnostic criteria for schizophrenic and affective psychoses*. World Psychiatric Association.
- Berridge, KC Robinson, TE 1998. What is the role of dopamine in reward: hedonic impact, reward learning, or incentive salience? *Brain Research Reviews*, **28**, 309-369.
- Berridge, KC Robinson, TE 2003. Parsing reward. Trends in neurosciences, 26, 507-513.
- Bhowmick, S, Hazra, A Ghosh, MG 2010. Amisulpride versus olanzapine in the treatment of schizophrenia in Indian patients: Randomized controlled trial. *Australian & New Zealand Journal of Psychiatry*, **44**, 237-242.
- Bilder, RM, Mukherjee, S, Rieder, RO Pandurangi, AK 1985. Symptomatic and neuropsychological components of defect states. *Schizophrenia Bulletin*, **11**, 409-419.
- Bio, DS Gattaz, WFG 2011. Vocational rehabilitation improves cognition and negative symptoms in schizophrenia. *Schizophrenia Research*, **126**, 265-269.
- Blanchard, JJ, Bellack, AS Mueser, KT 1994. Affective and social-behavioral correlates of physical and social anhedonia in schizophrenia. *Journal of Abnormal Psychology*, **103**, 719-728.
- Blanchard, JJ Cohen, AS 2006. The structure of negative symptoms within schizophrenia: implications for assessment. *Schizophrenia Bulletin*, **32**, 238–245.
- Blanchard, JJ, Kring, AM, Horan, WP Gur, R 2011. Toward the next generation of negative symptom assessments: the collaboration to advance negative symptom assessment in schizophrenia. *Schizophrenia Bulletin*, **37**, 291-299.
- Blanchard, JJ, Mueser, KT Bellack, AS 1998. Anhedonia, positive and negative affect, and social functioning in schizophrenia. *Schizophrenia Bulletin*, **24**, 413-424.
- Bland, JM Altman, D 1986. Statistical methods for assessing agreement between two methods of clinical measurement. *The Lancet*, **327**, 307-310.
- Bleuler, E 1950. *Dementia praecox or the group of schizophrenias*, New York, NY: International Universities Press.
- Bobes, J, Arango, C, Garcia-Garcia, M Rejas, J 2010. Prevalence of negative symptoms in outpatients with schizophrenia spectrum disorders treated with antipsychotics in routine clinical practice: findings from the CLAMORS study. *Journal of Clinical Psychiatry*, **71**, 280-286.

- Bobes, J, Ciudad, A, Alvarez, E, San, L, Polavieja, P Gilaberte, I 2009. Recovery from schizophrenia: results from a 1-year follow-up observational study of patients in symptomatic remission. *Schizophrenia Research*, **115**, 58-66.
- Bodkin, J, Siris, SG, Bermanzohn, PC, Hennen, J Cole, JOG 2005. Double-Blind, Placebo-Controlled, Multicenter Trial of Selegiline Augmentation of Antipsychotic Medication to Treat Negative Symptoms in Outpatients With Schizophrenia. *American Journal of Psychiatry*, **162**, 388-390.
- Bourque, F, Van Der Ven, E Malla, A 2011. A meta-analysis of the risk for psychotic disorders among first-and second-generation immigrants. *Psychological Medicine*, **41**, 897-910.
- Boydell, J, Van Os, J, Mckenzie, K, Allardyce, J, Goel, R, Mccreadie, RG Murray, RM 2001. Incidence of schizophrenia in ethnic minorities in London: ecological study into interactions with environment. *BMJ*, **323**, 1336.
- Braff, DL, Ryan, J, Rissling, AJ Carpenter, WT 2013. Lack of use in the literature from the last 20 years supports dropping traditional schizophrenia subtypes from DSM-5 and ICD-11. *Schizophrenia Bulletin*, **34**, 751-753.
- Braunholtz, DA, Edwards, SJ Lilford, RJ 2001. Are randomized clinical trials good for us (in the short term)? Evidence for a "trial effect". *Journal of Clinical Epidemiology*, **54**, 217-224.
- Bravata, DM Olkin, I 2001. Simple pooling versus combining in meta-analysis. *Evaluation & the health professions*, **24**, 218-230.
- Breier, A, Wolkowitz, OM, Doran, AR, Roy, A, Boronow, J, Hommer, DW Pickar, D 1987. Neuroleptic responsivity of negative and positive symptoms in schizophrenia. *American Journal of Psychiatry*, **144**, 1549-1555.
- Brekke, J, Kay, DD, Lee, KS Green, MF 2005. Biosocial pathways to functional outcome in schizophrenia. *Schizophrenia Research*, **80**, 213-225.
- Brekke, JS, Debonis, JA Graham, JW 1994. A latent structure analysis of the positive and negative symptoms in schizophrenia. *Comprehensive Psychiatry*, **35**, 252-259.
- Bröker, M, Röhricht, F Priebe, S 1995. Initial assessment of hospital treatment by patients with paranoid schizophrenia: a predictor of outcome. *Psychiatry Research*, **58**, 77-81.
- Broome, MR, Woolley, JB, Tabraham, P, Johns, LC, Bramon, E, Murray, GK, Pariante, C, Mcguire, PK Murray, RM 2005. What causes the onset of psychosis? *Schizophrenia Research*, **79**, 23-34.
- Brown, AS, Schaefer, CA, Wyatt, RJ, Begg, MD, Goetz, R, Bresnahan, MA, Harkavy-Friedman, J, Gorman, JM, Malaspina, D Susser, ES 2014. Paternal age and risk of schizophrenia in adult offspring. *American Journal of Psychiatry*, **159**, 1528-1533.
- Browne, S, Clarke, M, Gervin, M, Waddington, JL, Larkin, C O'callaghan, E 2000. Determinants of quality of life at first presentation with schizophrenia. *The British Journal of Psychiatry*, **176**, 173-176.
- Bryson, G, Bell, M, Greig, T Kaplan, E 1999. Internal consistency, temporal stability and neuropsychological correlates of three cognitive components of the Positive and Negative Syndrome Scale (PANSS). *Schizophrenia Research*, **38**, 27-35.
- Buchanan, R, Javitt, D, Marder, S, Schooler, N, Gold, J, Mcmahon, R, Heresco-Levy, U
 Carpenter, W 2007. The Cognitive and Negative Symptoms in Schizophrenia Trial (CONSIST): the efficacy of glutamatergic agents for negative symptoms and cognitive impairments. *American Journal of Psychiatry*, **164**, 1593-1602.

- Buchanan, RW 2007. Persistent negative symptoms in schizophrenia: an overview. *Schizophrenia Bulletin*, **33**, 1013-1022.
- Buchanan, RW, Breier, A, Kirkpatrick, B, Elkashef, A, Munson, RC, Gellad, F Carpenter Jr, W 1993. Structural abnormalities in deficit and nondeficit schizophrenia. *American Journal of Psychiatry*, **150**, 59-65.
- Buchanan, RW Carpenter, WT 1994. Domains of psychopathology: an approach to the reduction of heterogeneity in schizophrenia. *The Journal of Nervous and Mental Disease*, **182**, 193-204.
- Buckley, PF, Miller, BJ, Lehrer, DS Castle, DJ 2009. Psychiatric comorbidities and schizophrenia. *Schizophrenia Bulletin*, **35**, 383-402.
- Burger, GK, Calsyn, RJ, Morse, GA, Klinkenberg, WD Trusty, ML 1997. Factor structure of the expanded brief psychiatric rating scale. *Journal of Clinical Psychology*, **53**, 451-454.
- Cabeza, IGA, Amador, MS, López, CA De Chavez, MG 2000. Subjective response to antipsychotics in schizophrenic patients: clinical implications and related factors. *Schizophrenia Research*, **41**, 349-355.
- Cannon, M, Jones, PB Murray, RM 2002. Obstetric complications and schizophrenia: historical and meta-analytic review. *American Journal of Psychiatry*, **159**, 1080-1092.
- Cannon, TD, Cadenhead, K, Cornblatt, B, Woods, SW, Addington, J, Walker, E, Seidman, LJ, Perkins, D, Tsuang, M Mcglashan, T 2008. Prediction of psychosis in youth at high clinical risk: a multisite longitudinal study in North America. *Archives of General Psychiatry*, **65**, 28-37.
- Cardno, AG, Jones, LA, Murphy, KC, Asherson, P, Scott, LC, Williams, J, Owen, MJ Mcguffin, P 1996. Factor analysis of schizophrenic symptoms using the OPCRIT checklist. *Schizophrenia Research*, **22**, 233-239.
- Cardno, AG, Marshall, EJ, Coid, B, Macdonald, AM, Ribchester, TR, Davies, NJ, Venturi, P, Jones, LA, Lewis, SW Sham, PC 1999. Heritability estimates for psychotic disorders: the Maudsley twin psychosis series. *Archives of General Psychiatry*, **56**, 162-168.
- Caroff, SN 1980. The neuroleptic malignant syndrome. *The Journal of clinical psychiatry*, **41**, 79-83.
- Caroff, SN, Davis, VG, Miller, DD, Davis, SM, Rosenheck, RA, Mcevoy, JP, Campbell, EC, Saltz, BL, Riggio, S, Chakos, MH, Swartz, MS, Keefe, RS, Stroup, TS, Lieberman, JA Catie, IJ 2011. Treatment outcomes of patients with tardive dyskinesia and chronic schizophrenia. *The Journal of clinical psychiatry*, **72**, 295-303.
- Carpenter, W Kirkpatrick, B 2015. Concepts and methods when considering negative symptom course. *Psychological Medicine*, **45**, 2135-2136.
- Carpenter, WJ Strauss, JS 1974. Cross-cultural evaluation of Schneider's first-rank symptoms of schizophrenia: A report from the International Pilot Study of Schizophrenia. *The American Journal of Psychiatry*, 682-687.
- Carpenter, WT, Jr., Heinrichs, DW Alphs, LD 1985. Treatment of negative symptoms. *Schizophrenia Bulletin*, **11**, 440-452.
- Carpenter, WT, Jr., Heinrichs, DW Wagman, AM 1988. Deficit and nondeficit forms of schizophrenia: the concept. *American Journal of Psychiatry*, **145**, 578-583.
- Centorrino, F, Goren, JL, Hennen, J, Salvatore, P, Kelleher, JP Baldessarini, RJ 2004. Multiple versus single antipsychotic agents for hospitalized psychiatric patients: case-control study of risks versus benefits. *American Journal of Psychiatry*, **161**, 700-706.

- Chapman, LJ, Chapman, JP Raulin, ML 1976. Scales for physical and social anhedonia. *Journal of Abnormal Psychology*, **85**, 374-382.
- Cheniaux, E, Landeira-Fernandez, J, Telles, LL, Lessa, JLM, Dias, A, Duncan, T Versiani, M 2008. Does schizoaffective disorder really exist? A systematic review of the studies that compared schizoaffective disorder with schizophrenia or mood disorders. *Journal of Affective Disorders*, **106**, 209-217.
- Chiolero, A, Paradis, G, Rich, B Hanley, JA 2013. Assessing the relationship between the baseline value of a continuous variable and subsequent change over time. *Frontiers in Public Health*, **1**, 29.
- Chouinard, G, Annable, L, Serrano, M, Albert, JM Charette, R 1975. Amitriptyline-perphenazine interaction in ambulatory schizophrenic patients. A controled study of drug interaction. *Archives of General Psychiatry*, **32**, 1295-1307.
- Cirici, AR Obiols, LJ 2008a. Validity of a social skills training program for schizophrenic patients. Actas Espanolas de Psiquiatria, **36**, 123-132.
- Cirici, AR Obiols, LJJ 2008b. Validity of a social skills training program for schizophrenic patients. Actas Espanolas de Psiquiatria, **36**, 123-132.
- Clark, LA Watson, D 1995. Constructing validity: Basic issues in objective scale development. *Psychological Assessment*, **7**, 309-319.
- Cohen, AS Minor, KS 2010. Emotional experience in patients with schizophrenia revisited: meta-analysis of laboratory studies. *Schizophrenia Bulletin*, **36**, 143-150.
- Cohen, AS, Mitchell, KR Elvevåg, B 2014. What do we really know about blunted vocal affect and alogia? A meta-analysis of objective assessments. *Schizophrenia Research*, **159**, 533-538.
- Cohen, AS, Najolia, GM, Brown, LA Minor, KS 2011. The state-trait disjunction of anhedonia in schizophrenia: potential affective, cognitive and social-based mechanisms. *Clinical Psychology Review*, **31**, 440-448.
- Cohen, JD, Barch, DM, Carter, C Servan-Schreiber, D 1999. Context-processing deficits in schizophrenia: converging evidence from three theoretically motivated cognitive tasks. *Journal of Abnormal Psychology*, **108**, 120-133.
- Collins, AA, Remington, G, Coulter, K Birkett, K 1996. Depression in schizophrenia: a comparison of three measures. *Schizophrenia Research*, **20**, 205-209.
- Comrey, AL Lee, HB 2013. A first course in factor analysis, Psychology Press.
- Corcoran, R, Mercer, G Frith, CD 1995. Schizophrenia, symptomatology and social inference: investigating "theory of mind" in people with schizophrenia. *Schizophrenia Research*, **17**, 5-13.
- Cornblatt, BA, Auther, AM, Niendam, T, Smith, CW, Zinberg, J, Bearden, CE Cannon, TD 2007. Preliminary findings for two new measures of social and role functioning in the prodromal phase of schizophrenia. *Schizophrenia Bulletin*, **33**, 688-702.
- Corrigan, P 2004. How stigma interferes with mental health care. *American psychologist*, **59**, 614-625.
- Cougnard, A, Marcelis, M, Myin-Germeys, I, De Graaf, R, Vollebergh, W, Krabbendam, L, Lieb, R, Wittchen, H-U, Henquet, C Spauwen, J 2007. Does normal developmental expression of psychosis combine with environmental risk to cause persistence of psychosis? A psychosis proneness–persistence model. *Psychological Medicine*, **37**, 513-527.

- Cramer, P, Bowen, J O'neill, M 1992. Schizophrenics and social judgement. Why do schizophrenics get it wrong? *The British Journal of Psychiatry*, **160**, 481-487.
- Crawford, MJ, Killaspy, H, Barnes, TR, Barrett, B, Byford, S, Clayton, K, Dinsmore, J, Floyd, S, Hoadley, A, Johnson, T, Kalaitzaki, E, King, M, Leurent, B, Maratos, A, O'neill, FA, Osborn, D, Patterson, S, Soteriou, T, Tyrer, P Waller, D 2012a. Group art therapy as an adjunctive treatment for people with schizophrenia: a randomised controlled trial (MATISSE). *Health technology assessment (Winchester, England)*, 1-76.
- Crawford, MJ, Killaspy, H, Barnes, TR, Barrett, B, Byford, S, Clayton, K, Dinsmore, J, Floyd, S, Hoadley, A, Johnson, T, Kalaitzaki, E, King, M, Leurent, B, Maratos, A, O'neill, FA, Osborn, DP, Patterson, S, Soteriou, T, Tyrer, P Waller, D 2012b. Group art therapy as an adjunctive treatment for people with schizophrenia: multicentre pragmatic randomised trial. *BMJ*, **344**, e846.
- Crichton, P 1996. First-rank symptoms or rank-and-file symptoms? *The British Journal of Psychiatry*, **169**, 537-540.
- Crow, TJ 1980. Molecular pathology of schizophrenia: more than one disease process? *BMJ*, **280**, 66-68.
- Crow, TJ 1985. The two-syndrome concept: origins and current status. *Schizophrenia Bulletin*, **11**, 471-486.
- Crow, TJ, Frith, CD, Johnstone, EC Owens, DG 1980. Schizophrenia and cerebral atrophy. *Lancet*, **1**, 1129-1130.
- Curson, DA, Pantelis, C, Ward, J Barnes, TR 1992. Institutionalism and schizophrenia 30 years on. Clinical poverty and the social environment in three British mental hospitals in 1960 compared with a fourth in 1990. *The British Journal of Psychiatry*, **160**, 230-241.
- Dalkey, N Helmer, O 1963. An experimental application of the Delphi method to the use of experts. *Management science*, **9**, 458-467.
- Davidson, L Stayner, D 1997. Loss, loneliness, and the desire for love: Perspectives on the social lives of people with schizophrenia. *Psychiatric Rehabilitation Journal*, **20**, 3-12.
- Davies, G, Welham, J, Chant, D, Torrey, EF Mcgrath, J 2003. A systematic review and metaanalysis of Northern Hemisphere season of birth studies in schizophrenia. *Schizophrenia Bulletin*, **29**, 587-593.
- Davies, P 1974. Conditioned after-images: I. British Journal of Psychology, 65, 191-204.
- Deister, A Marneros, A 1993. Long-term stability of subtypes in schizophrenic disorders: a comparison of four diagnostic systems. *European Archives of Psychiatry and Clinical Neuroscience*, **242**, 184-190.
- Dersimonian, R Laird, N 1986. Meta-analysis in clinical trials. *Controlled clinical trials*, **7**, 177-188.
- Dingemans, P, Linszen, D, Lenior, M Smeets, R 1995. Component structure of the expanded brief psychiatric rating scale (BPRS-E). *Psychopharmacology*, **122**, 263-267.
- Dollfus, S Petit, M 1995a. Principal-component analyses of PANSS and SANS-SAPS in schizophrenia: their stability in an acute phase. *European Psychiatry*, **10**, 97-106.
- Dollfus, S Petit, MG 1995b. Stability of positive and negative symptoms in schizophrenic patients: A 3-year follow-up study. *European Psychiatry* **10**, 228-236.
- Dossenbach, M, Erol, A, Kessaci, MEM, Shaheen, MO, Sunbol, MM, Boland, J, Hodge, A, O'halloran, RA Bitter, IG 2004. Effectiveness of Antipsychotic Treatments for Schizophrenia: Interim 6-Month Analysis From a Prospective Observational Study (IC-

SOHO) Comparing Olanzapine, Quetiapine, Risperidone, and Haloperidol. *Journal of Clinical Psychiatry*, **65**, 312-321.

- Doyle, M, Flanagan, S, Browne, S, Clarke, M, Lydon, D, Larkin, C O'callaghan, E 1999. Subjective and external assessments of quality of life in schizophrenia: relationship to insight. *Acta Psychiatrica Scandinavica*, **99**, 466-472.
- Dunayevich, E, Chen, CY, Marder, SR Rabinowitz, J 2014. Restrictive symptomatic inclusion criteria create barriers to clinical research in schizophrenia negative symptoms: An analysis of the CATIE dataset. *European Neuropsychopharmacology*, **24**, 1615-1621.
- Dunn, M, O'driscoll, C, Dayson, D, Wills, W Leff, J 1990. The TAPS Project. 4: An observational study of the social life of long-stay patients. *The British Journal of Psychiatry*, **157**, 842-848.
- Duval, S Tweedie, R 2000. Trim and Fill: A Simple Funnel-Plot–Based Method of Testing and Adjusting for Publication Bias in Meta-Analysis. *Biometrics*, **56**, 455-463.
- Dworkin, RH, Oster, H, Clark, SC White, SR 1998. *Affective expression and affective experience in schizophrenia, in Origins and and Development of schizophrenia,* American Psychological Association.
- Eack, SM Newhill, CE 2007. Psychiatric symptoms and quality of life in schizophrenia: a metaanalysis. *Schizophrenia Bulletin*, **33**, 1225–1237.
- Eadie, M 2007. The neurological legacy of John Russell Reynolds (1828–1896). *Journal of Clinical Neuroscience*, **14**, 309-316.
- Eaton, WW, Thara, R, Federman, B, Melton, B Liang, K-Y 1995. Structure and course of positive and negative symptoms in schizophrenia. *Archives of General Psychiatry*, **52**, 127-134.
- Eckert, SL, Diamond, PM, Miller, AL, Velligan, DI, Funderburg, LG True, JE 1996. A comparison of instrument sensitivity to negative symptom change. *Psychiatry Research*, **63**, 67-75.
- European Medicines Agency. 2012. *Guideline on clinical investigation of medicinal products, including depot preparations in the treatment of schizophrenia* EMA/CHMP/40072/2010.
- Emsley, R, Rabinowitz, J, Torreman, M Group, R-I-EPGW 2003. The factor structure for the Positive and Negative Syndrome Scale (PANSS) in recent-onset psychosis. *Schizophrenia Research*, **61**, 47-57.
- Eysenck, HJ 1960. *The psychology of politics*, Transaction publishers.
- Fakhoury, WK, Kaiser, W, Roeder-Wanner, U-U Priebe, S 2002. Subjective Evaluation: Is there more than one criterion? *Schizophrenia Bulletin*, **28**, 319-327.
- Faris, REL Dunham, HW 1939. *Mental disorders in urban areas: an ecological study of schizophrenia and other psychoses.* Oxford, England.
- Fearon, P, Kirkbride, JB, Morgan, C, Dazzan, P, Morgan, K, Lloyd, T, Hutchinson, G, Tarrant, J, Lun Alan Fung, W Holloway, J 2006. Incidence of schizophrenia and other psychoses in ethnic minority groups: results from the MRC AESOP Study. *Psychological Medicine*, **36**, 1541-1550.
- Fenton, WS, Blyler, CR Heinssen, RK 1997. Determinants of medication compliance in schizophrenia: empirical and clinical findings. *Schizophrenia Bulletin*, **23**, 637-651.
- Fenton, WS Mcglashan, TH 1994. Antecedents, symptom progression, and long-term outcome of the deficit syndrome in schizophrenia. *The American Journal of Psychiatry*, **151**, 351-356.

- Fenton, WS Mcglashan, THJ 1991. Natural history of schizophrenia subtypes. II. Positive and negative symptoms and long-term course. *Archives of General Psychiatry*, **48**, 978-986.
- Field, A 2009. Discovering statistics using SPSS, Sage publications.
- Fitzgerald, PB, Williams, C, Corteling, N, Filia, S, Brewer, K, Adams, A, De Castella, A, Rolfe, T, Davey, P Kulkarni, J 2001. Subject and observer-rated quality of life in schizophrenia. *Acta Psychiatrica Scandinavica*, **103**, 387-392.
- Fleischhacker, WW, Eerdekens, M, Karcher, K, Remington, G, Llorca, PM, Chrzanowski, W, Martin, S Gefvert, OJ 2003. Treatment of schizophrenia with long-acting injectable risperidone: a 12-month open-label trial of the first long-acting second-generation antipsychotic. *The Journal of clinical psychiatry*, **64**, 1250-1257.
- Forbes, C, Blanchard, JJ, Bennett, M, Horan, WP, Kring, A Gur, R 2010. Initial development and preliminary validation of a new negative symptom measure: the Clinical Assessment Interview for Negative Symptoms (CAINS). *Schizophrenia Research*, **124**, 36-42.
- Foussias, G Remington, G 2010. Negative symptoms in schizophrenia: avolition and Occam's razor. *Schizophrenia Bulletin*, **36**, 359-369.
- Franke, P, Maier, W, Hardt, J, Hain, C Cornblatt, BA 1994. Attentional abilities and measures of schizotypy: their variation and covariation in schizophrenic patients, their siblings, and normal control subjects. *Psychiatry Research*, 54, 259-272.
- Freeman, D Garety, PA 2000. Comments on the content of persecutory delusions: Does the definition need clarification? *British Journal of Clinical Psychology*, **39**, 407-414.
- Freudenreich, O, Cather, C, Evins, AE, Henderson, DC Goff, DC 2004. Attitudes of schizophrenia outpatients toward psychiatric medications: relationship to clinical variables and insight. *Journal of Clinical Psychiatry*, **65**, 1372-1376.
- Frith, CD 1979. Consciousness, information processing and schizophrenia. The British Journal of Psychiatry, 134, 225-235.
- Frith, CD Corcoran, R 1996. Exploring 'theory of mind'in people with schizophrenia. *Psychological Medicine*, **26**, 521-530.
- Fusar-Poli, P, Deste, G, Smieskova, R, Barlati, S, Yung, AR, Howes, O, Stieglitz, R-D, Vita, A, Mcguire, P Borgwardt, S 2012. Cognitive functioning in prodromal psychosis: a metaanalysis. Archives of General Psychiatry, 69, 562-571.
- Fusar-Poli, P, Papanastasiou, E, Stahl, D, Rocchetti, M, Carpenter, W, Shergill, S Mcguire, P 2015. Treatments of negative symptoms in schizophrenia: meta-analysis of 168 randomized placebo-controlled trials. *Schizophrenia Bulletin*, **41**, 892-899.
- Gaebel, W, Riesbeck, M, Wolwer, W, Klimke, A, Eickhoff, M, Von Wilmsdorff, M, Jockers-Scherubl, MC, Kuhn, KU, Lemke, M, Bechdolf, A, Bender, S, Degner, D, Schlosser, R, Schmidt, LG, Schmitt, A, Jager, M, Buchkremer, G, Falkai, P, Klingberg, S, Kopcke, W, Maier, W, Hafner, H, Ohmann, C, Salize, HJ, Schneider, F Moller, HJG 2007.
 Maintenance treatment with risperidone or low-dose haloperidol in first-episode schizophrenia: 1-year results of a randomized controlled trial within the German research network on schizophrenia. *Journal of Clinical Psychiatry*, 68, 1763-1774.
- Galderisi, S, Maj, M, Mucci, A, Cassano, GB, Invernizzi, G, Rossi, A, Vita, A, Dell'osso, L, Daneluzzo, E Pini, S 2002. Historical, psychopathological, neurological, and neuropsychological aspects of deficit schizophrenia: a multicenter study. *American Journal of Psychiatry*, **159**, 983-990.
- Gard, DE, Gard, MG, Kring, AM John, OP 2006. Anticipatory and consummatory components of the experience of pleasure: a scale development study. *Journal of Research in Personality*, **40**, 1086-1102.
- Gard, DE, Kring, AM, Gard, MG, Horan, WP Green, MF 2007. Anhedonia in schizophrenia: distinctions between anticipatory and consummatory pleasure. *Schizophrenia Research*, **93**, 253-260.
- Gard, DE, Sanchez, AH, Cooper, K, Fisher, M, Garrett, C Vinogradov, S 2014. Do People With Schizophrenia Have Difficulty Anticipating Pleasure, Engaging in Effortful Behavior, or Both? *Journal of Abnormal Psychology*, **123**, 771-782.
- Geddes, JR, Verdoux, H, Takei, N, Lawrie, SM, Bovet, P, Eagles, JM, Heun, R, Mccreadie, RG, Mcneil, TF O'callaghan, E 1999. Schizophrenia and complications of pregnancy and labor: an individual patient data meta-analysis. *Schizophrenia Bulletin*, **25**, 413-423.
- Gelenberg, AJ Mandel, MR 1977. Catatonic reactions to high-potency neuroleptic drugs. Archives of General Psychiatry, **34**, 947-950.
- Gilbert, PL, Harris, MJ, Mcadams, LA Jeste, DV 1995. Neuroleptic withdrawal in schizophrenic patients: a review of the literature. *Archives of General Psychiatry*, **52**, 173-188.
- Goff, DC, Herz, L, Posever, T, Shih, V, Tsai, G, Henderson, DC, Freudenreich, O, Evins, A, Yovel, I, Zhang, H Schoenfeld, DG 2005. A six-month, placebo-controlled trial of D-cycloserine co-administered with conventional antipsychotics in schizophrenia patients. *Psychopharmacology*, **179**, 144-150.
- Gold, JM, Randolph, C, Carpenter, CJ, Goldberg, TE Weinberger, DR 1992. Forms of memory failure in schizophrenia. *Journal of Abnormal Psychology*, **101**, 487-494.
- Gold, JM, Waltz, JA, Prentice, KJ, Morris, SE Heerey, EA 2008. Reward processing in schizophrenia: a deficit in the representation of value. *Schizophrenia Bulletin*, **34**, 835-847.
- Goldberg, SC 1985. Negative and deficit symptoms in schizophrenia do respond to neuroleptics. *Schizophrenia Bulletin*, **11**, 453-456.
- Goldman, R, Tandon, R, Liberzon, I Greden, JF 1992. Measurement of depression and negative symptoms in schizophrenia. *Psychopathology*, **25**, 49-56.
- Goldman, RS, Tandon, R, Liberzon, I, Goodson, J Greden, JF 1991. Stability of positive and negative symptom constructs during neuroleptic treatment in schizophrenia. *Psychopathology*, 24, 247-252.
- Gorna, K, Jaracz, K, Rybakowski, F Rybakowski, J 2008. Determinants of objective and subjective quality of life in first-time-admission schizophrenic patients in Poland: a longitudinal study. *Quality of Life Research*, **17**, 237-247.
- Gracia, E, García, F Musitu, G 1995. Macrosocial determinants of social integration: Social class and area effect. *Journal of Community & Applied Social Psychology*, **5**, 105-119.
- Grant, P Beck, A 2009. Evaluation sensitivity as a moderator of communication disorder in schizophrenia. *Psychological Medicine*, **39**, 1211-1219.
- Grant, PM Beck, AT 2010. Asocial beliefs as predictors of asocial behavior in schizophrenia. *Psychiatry Research*, **177**, 65-70.
- Green, MF, Kern, RS, Braff, DL Mintz, J 2000. Neurocognitive deficits and functional outcome in schizophrenia: are we measuring the right stuff? *Schizophrenia Bulletin*, **26**, 119.

- Green, MF, Kern, RS Heaton, RK 2004a. Longitudinal studies of cognition and functional outcome in schizophrenia: implications for MATRICS. Schizophrenia Research, 72, 41-51.
- Green, MF, Nuechterlein, KH, Gold, JM, Barch, DM, Cohen, J, Essock, S, Fenton, WS, Frese, F, Goldberg, TE Heaton, RK 2004b. Approaching a consensus cognitive battery for clinical trials in schizophrenia: the NIMH-MATRICS conference to select cognitive domains and test criteria. *Biological Psychiatry*, 56, 301-307.
- Green, MF, Penn, DL, Bentall, R, Carpenter, WT, Gaebel, W, Gur, RC, Kring, AM, Park, S, Silverstein, SM Heinssen, R 2008. Social cognition in schizophrenia: an NIMH workshop on definitions, assessment, and research opportunities. *Schizophrenia Bulletin*, **34**, 1211-1220.
- Green, SB 1991. How many subjects does it take to do a regression analysis. *Multivariate behavioral research*, **26**, 499-510.
- Greenwood, N, Key, A, Burns, T, Bristow, M Sedgwick, P 1999. Satisfaction with in-patient psychiatric services. Relationship to patient and treatment factors. *The British Journal of Psychiatry*, **174**, 159-163.
- Gross, JJ 1998. The emerging field of emotion regulation: an integrative review. *Review of general psychology*, **2**, 271-299.
- Grove, WM, Lebow, BS, Clementz, BA, Cerri, A, Medus, C Iacono, WG 1991. Familial prevalence and coaggregation of schizotypy indicators: a multitrait family study. *Journal of Abnormal Psychology*, **100**, 115-121.
- Gruenberg, EM, Brandon, S Kasius, RV 1966. Identifying cases of the social breakdown syndrome. *The Milbank Memorial Fund Quarterly*, 150-155.
- Gur, RE, Kohler, CG, Ragland, JD, Siegel, SJ, Lesko, K, Bilker, WB Gur, RC 2006. Flat affect in schizophrenia: relation to emotion processing and neurocognitive measures. *Schizophrenia Bulletin*, **32**, 279-287.
- Gureje, O, Aderibigbe, Y Obikoya, O 1995. Three syndromes in schizophrenia: validity in young patients with recent onset of illness. *Psychological Medicine*, **25**, 715-725.
- Guy, W 1976. Clinical global impression scale. *The ECDEU Assessment Manual for Psychopharmacology-Revised.*: National Institute of Mental Health, Rockville, MD.
- Häfner, H, Löffler, W, Maurer, K, Hambrecht, M Heiden, W 1999. Depression, negative symptoms, social stagnation and social decline in the early course of schizophrenia. *Acta Psychiatrica Scandinavica*, **100**, 105-118.
- Häfner, H, Maurer, K, Trendler, G, An Der Heiden, W, Schmidt, M Könnecke, R 2005. Schizophrenia and depression: challenging the paradigm of two separate diseases—a controlled study of schizophrenia, depression and healthy controls. *Schizophrenia Research*, **77**, 11-24.
- Hamer, RM Simpson, PM 2009. Last observation carried forward versus mixed models in the analysis of psychiatric clinical trials. *The American Journal of Psychiatry*, **166**, 639-641.
- Hamilton, M 1960. A rating scale for depression. *Journal of Neurology Neurosurgery and Psychiatry*, **23**, 56-62.
- Handest, P Parnas, J 2005. Clinical characteristics of first-admitted patients with ICD–10 schizotypal disorder. *The British Journal of Psychiatry*, **187**, s49-s54.

- Hansson, L, Björkman, T Priebe, S 2007. Are important patient-rated outcomes in community mental health care explained by only one factor? *Acta Psychiatrica Scandinavica*, **116**, 113-118.
- Hardy-Baylé, M-C, Sarfati, Y Passerieux, C 2003. The cognitive basis of disorganization symptomatology in schizophrenia and its clinical correlates: toward a pathogenetic approach to disorganization. *Schizophrenia Bulletin*, **29**, 459-471.
- Harley, EW-Y, Boardman, J Craig, T 2012. Friendship in people with schizophrenia: a survey. *Social Psychiatry and Psychiatric Epidemiology*, **47**, 1291-1299.
- Haro, J, Kamath, S, Ochoa, SO, Novick, D, Rele, K, Fargas, A, Rodriguez, M, Rele, R, Orta, J Kharbeng, A 2003. The Clinical Global Impression–Schizophrenia scale: a simple instrument to measure the diversity of symptoms present in schizophrenia. Acta Psychiatrica Scandinavica, **107**, 16-23.
- Harris, MJ, Jeste, DV, Krull, A, Montague, J Heaton, RK 1991. Deficit syndrome in older schizophrenic patients. *Psychiatry Research*, **39**, 285-292.
- Harris, RJ 2014. A primer of multivariate statistics, Psychology Press.
- Harrow, M Jobe, TH 2007. Factors involved in outcome and recovery in schizophrenia patients not on antipsychotic medications: a 15-year multifollow-up study. *The Journal of Nervous and Mental Disease*, **195**, 406-414.
- Harrow, M, Sands, JR, Silverstein, ML Goldberg, JF 1997. Course and outcome for schizophrenia versus other psychotic patients: a longitudinal study. *Schizophrenia Bulletin*, **23**, 287-303.
- Harvey, C, Curson, D, Pantelis, C, Taylor, J Barnes, T 1996. Four behavioural syndromes of schizophrenia. *The British Journal of Psychiatry*, **168**, 562-570.
- Harvey, PD Keefe, RS 2001. Studies of cognitive change in patients with schizophrenia following novel antipsychotic treatment. *American Journal of Psychiatry*, **158**, 176-184.
- Harvey, PD, Koren, D, Reichenberg, A Bowie, CR 2006. Negative symptoms and cognitive deficits: what is the nature of their relationship? *Schizophrenia Bulletin*, **32**, 250-258.
- Hayes, RL, Halford, WK Varghese, FTJ 1995. Social skills training with chronic schizophrenic patients: Effects on negative symptoms and community functioning. *Behavior Therapy*, 26, 433-449.
- Haynes, SN Lench, HC 2003. Incremental validity of new clinical assessment measures. *Psychological Assessment*, **15**, 456.
- Hecht, EM Landy, C 2012. Alpha-2 receptor antagonist add-on therapy in the treatment of schizophrenia; a meta-analysis. *Schizophrenia Research*, **134**, 202-206.
- Heckers, S, Weiss, AP, Deckersbach, T, Goff, DC, Morecraft, RJ Bush, G 2004. Anterior cingulate cortex activation during cognitive interference in schizophrenia. *American Journal of Psychiatry*, **161**, 707-715.
- Heerey, EA Gold, JM 2007. Patients with schizophrenia demonstrate dissociation between affective experience and motivated behavior. *Journal of Abnormal Psychology*, **116**, 268-278.
- Heerey, EA, Robinson, BM, Mcmahon, RP Gold, JM 2007. Delay discounting in schizophrenia. *Cognitive Neuropsychiatry*, **12**, 213-221.
- Henquet, C, Murray, R, Linszen, D Van Os, J 2005. The environment and schizophrenia: the role of cannabis use. *Schizophrenia Bulletin*, **31**, 608-612.

- Henry, JD, Green, MJ, De Lucia, A, Restuccia, C, Mcdonald, S O'donnell, M 2007. Emotion dysregulation in schizophrenia: reduced amplification of emotional expression is associated with emotional blunting. *Schizophrenia Research*, **95**, 197-204.
- Herbener, ES Harrow, M 2002. The course of anhedonia during 10 years of schizophrenic illness. *Journal of Abnormal Psychology*, **111**, 237-248.
- Higgins, J Green, S 2009. Cochrane handbook for systematic reviews of interventions. Version 5.0. 2. Updated September 2009. The Cochrane Collaboration.
- Hirsch, SR, Kissling, W, B,,Uml, J, Power, A O'connor, RJ 2002. A 28-week comparison of ziprasidone and haloperidol in outpatients with stable schizophrenia. *The Journal of clinical psychiatry*, **63**, 516-523.
- Ho, B-C, Nopoulos, P, Flaum, M, Arndt, S Andreasen, NC 1998. Two-year outcome in firstepisode schizophrenia: predictive value of symptoms for quality of life. *American Journal of Psychiatry*, **155**, 1196–1201.
- Hoekert, M, Kahn, RS, Pijnenborg, M Aleman, A 2007. Impaired recognition and expression of emotional prosody in schizophrenia: review and meta-analysis. *Schizophrenia Research*, 96, 135-145.
- Hong, LE, Avila, MT, Adami, H, Elliot, A Thaker, GK 2003. Components of the smooth pursuit function in deficit and nondeficit schizophrenia. *Schizophrenia Research*, **63**, 39-48.
- Hor, K Taylor, M 2010. Review: Suicide and schizophrenia: a systematic review of rates and risk factors. *Journal of Psychopharmacology*, **24**, 81-90.
- Horan, WP, Kring, AM Blanchard, JJ 2006. Anhedonia in schizophrenia: a review of assessment strategies. *Schizophrenia Bulletin*, **32**, 259-273.
- Horan, WP, Kring, AM, Gur, RE, Reise, SP Blanchard, JJ 2011. Development and psychometric validation of the Clinical Assessment Interview for Negative Symptoms (CAINS). *Schizophrenia Research*, **132**, 140–145.
- Hori, A, Tsunashima, K, Watanabe, K, Takekawa, Y, Ishihara, I, Terada, T Uno, M 1999. Symptom classification of schizophrenia changes with the duration of illness. *Acta Psychiatrica Scandinavica*, **99**, 447-452.
- Howes, OD Kapur, S 2009. The dopamine hypothesis of schizophrenia: version III—the final common pathway. *Schizophrenia Bulletin*, **35**, 549-562.
- Hsu, C-C Sandford, BA 2007. The Delphi technique: making sense of consensus. *Practical Assessment, Research & Evaluation*, **12**, 1-8.
- Hunsley, J Meyer, GJ 2003. The incremental validity of psychological testing and assessment: conceptual, methodological, and statistical issues. *Psychological Assessment*, **15**, 446-455.
- Hunter, R Barry, S 2012. Negative symptoms and psychosocial functioning in schizophrenia: neglected but important targets for treatment. *European psychiatry*, **27**, 432–436.
- Ihara, K, Morgan, C, Fearon, P, Dazzan, P, Demjaha, A, Lloyd, T, Kirkbride, JB, Hayhurst, H, Murray, RM Jones, PB 2009. The prevalence, diagnostic significance and demographic characteristics of Schneiderian first-rank symptoms in an epidemiological sample of first-episode psychoses. *Psychopathology*, **42**, 81-91.
- Jääskeläinen, E, Juola, P, Hirvonen, N, Mcgrath, JJ, Saha, S, Isohanni, M, Veijola, J Miettunen, J 2013. A systematic review and meta-analysis of recovery in schizophrenia. Schizophrenia Bulletin, **39**, 1296-1306.

- Jablensky, A 2000. Epidemiology of schizophrenia: the global burden of disease and disability. European Archives of Psychiatry and Clinical Neuroscience, **250**, 274-285.
- Jackson, HJ, Burgess, PM, Minas, I Joshua, S 1990. *Psychometric properties of the Manchester Scale*, New York: Hafner.
- Jackson, JH 1958. Selected writings of John Hughlings Jackson, Staples Press.
- Jaeger, J, Bitter, I, Czobor, P Volavka, J 1990. The measurement of subjective experience in schizophrenia: the Subjective Deficit Syndrome Scale. *Comprehensive Psychiatry*, **31**, 216-226.
- Jäger, M, Haack, S, Becker, T Frasch, K 2011. Schizoaffective disorder–an ongoing challenge for psychiatric nosology. *European psychiatry*, **26**, 159-165.
- Jaspers, K, Hoenig, J Hamilton, MW 1997. General psychopathology, JHU Press.
- Jauhar, S, Mckenna, P, Radua, J, Fung, E, Salvador, R Laws, K 2014. Cognitive-behavioural therapy for the symptoms of schizophrenia: Systematic review and meta-analysis with examination of potential bias. *British Journal of Psychiatry*, **204**, 20-29.
- Johnstone, EC Frith, CD 1996. Validation of three dimensions of schizophrenic symptoms in a large unselected sample of patients. *Psychological Medicine*, **26**, 669-679.
- Johnstone, EC, Owens, DG, Frith, CD Crow, TJ 1987. The relative stability of positive and negative features in chronic schizophrenia. *The British Journal of Psychiatry*, **150**, 60-64.
- Jones, PB, Barnes, TR, Davies, L, Dunn, G, Lloyd, H, Hayhurst, KP, Murray, RM, Markwick, A Lewis, SW 2006. Randomized controlled trial of the effect on Quality of Life of secondvs first-generation antipsychotic drugs in schizophrenia: Cost Utility of the Latest Antipsychotic Drugs in Schizophrenia Study (CUtLASS 1). Archives of General Psychiatry, 63, 1079-1087.
- Kahlbaum, KL 1874. Klinische Abhandlungen über psychische Krankheiten: Die Katatonie oder das Spannungsirresein: eine klinische Form psychischer Krankheit, Hirschwald.
- Kallert, T, Schützwohl, M, Matthes, C Group, ES 2000. *The client socio-demographic and clinical history inventory*. Medizinische Fakultät TU Dresden.
- Kallert, TW, Glöckner, M, Onchev, G, Raboch, J, Karastergiou, A, Solomon, Z, Magliano, L,
 Dembinskas, A, Kiejna, A Nawka, P 2005. The EUNOMIA project on coercion in
 psychiatry: study design and preliminary data. *World Psychiatry*, 4, 168-172.
- Kallert, TW, Glöckner, M Schützwohl, M 2008. Involuntary vs. voluntary hospital admission. *European Archives of Psychiatry and Clinical Neuroscience*, **258**, 195-209.
- Kallert, TW, Priebe, S, Mccabe, R, Kiejna, A, Rymaszewska, J, Nawka, P, Ocvar, L, Raboch, J, Starkova-Kalisova, L, Koch, R Schutzwohl, M 2007. Are day hospitals effective for acutely ill psychiatric patients? A European multicenter randomized controlled trial. J Clin Psychiatry, 68, 278-287.
- Kane, J, Honigfeld, G, Singer, J Meltzer, H 1988. Clozapine for the treatment-resistant schizophrenic: a double-blind comparison with chlorpromazine. Archives of General Psychiatry, 45, 789-796.
- Kane, J, Safferman, A, Pollack, S, Johns, C, Szymanski, S, Kronig, M Lieberman, J 1994. Clozapine, negative symptoms, and extrapyramidal side effects. *The Journal of clinical psychiatry*, 55, 74-77.

- Kane, JM, Fleischhacker, WW, Hansen, L, Perlis, R, Pikalov 3rd, A Assuncao-Talbott, S 2009. Akathisia: an updated review focusing on second-generation antipsychotics. *The Journal of clinical psychiatry*, **70**, 627-643.
- Kane, JM, Mackle, M, Snow-Adami, L, Zhao, J, Szegedi, A Panagides, JJ 2011. A randomized placebo-controlled trial of asenapine for the prevention of relapse of schizophrenia after long-term treatment. *The Journal of clinical psychiatry*, **72**, 349-355.
- Kane, JM, Yang, R Youakim, JM 2012. Adjunctive armodafinil for negative symptoms in adults with schizophrenia: A double-blind, placebo-controlled study. *Schizophrenia Research*, **135**, 116-122.
- Kaphzan, H, Ben-Shachar, D Klein, E. 2014. Entacapone augmentation of antipsychotic treatment in schizophrenic patients with negative symptoms; A double-blind placebo-controlled study. *International Journal of Neuropsychopharmacology*, **17**, 337-340.
- Kasckow, JW, Twamley, E, Mulchahey, JJ, Carroll, B, Sabai, M, Strakowski, SM, Patterson, T Jeste, DV 2001. Health-related quality of well-being in chronically hospitalized patients with schizophrenia: comparison with matched outpatients. *Psychiatry Research*, **103**, 69-78.
- Katsakou, C Priebe, S 2007. Patient's experiences of involuntary hospital admission and treatment: a review of qualitative studies. *Epidemiologia e Psichiatria Sociale*, **16**, 172-178.
- Katsanis, J, Iacono, WG Beiser, M 1990. Anhedonia and perceptual aberration in first-episode psychotic patients and their relatives. *Journal of Abnormal Psychology*, **99**, 202-206.
- Katschnig, H 2000. Schizophrenia and quality of life. *Acta Psychiatrica Scandinavica*, **102**, 33–37.
- Katschnig, H, Freeman, H Sartorius, N 2006. *Quality of life in mental disorders*, John Wiley & Sons Incorporated.
- Kay, SR, Flszbein, A Opfer, LA 1987. The positive and negative syndrome scale (PANSS) for schizophrenia. *Schizophrenia Bulletin*, **13**, 261–276.
- Kay, SR, Opler, LA Lindenmayer, J-P 1988. Reliability and validity of the positive and negative syndrome scale for schizophrenics. *Psychiatry Research*, **23**, 99-110.
- Kay, SR Sevy, S 1990. Pyramidical model of schizophrenia. *Schizophrenia Bulletin*, **16**, 537-545.
- Kelley, K Maxwell, SE 2003. Sample size for multiple regression: obtaining regression coefficients that are accurate, not simply significant. *Psychological Methods*, **8**, 305.
- Kelley, ME, Haas, GL Van Kammen, DP 2008. Longitudinal progression of negative symptoms in schizophrenia: A new look at an old problem. *Schizophrenia Research*, **105**, 188-196.
- Kelley, ME, Van Kammen, DP Allen, DN 1999. Empirical validation of primary negative symptoms: independence from effects of medication and psychosis. *American Journal* of Psychiatry, **156**, 406-411.
- Kendler, KS Diehl, SR 1993. The genetics of schizophrenia: a current, genetic-epidemiologic perspective. *Schizophrenia Bulletin*, **19**, 261-285.
- Kern, RS, Nuechterlein, KH, Green, MF, Baade, LE, Fenton, WS, Gold, JM, Keefe, RS, Mesholam-Gately, R, Mintz, J Seidman, LJ 2008. The MATRICS Consensus Cognitive Battery, part 2: co-norming and standardization. *American Journal of Psychiatry*, **165**, 214-220.
- Khashan, AS, Abel, KM, Mcnamee, R, Pedersen, MG, Webb, RT, Baker, PN, Kenny, LC Mortensen, PB 2008. Higher risk of offspring schizophrenia following antenatal

maternal exposure to severe adverse life events. *Archives of General Psychiatry*, **65**, 146-152.

- Kiang, M, Christensen, BK, Remington, G Kapur, S 2003. Apathy in schizophrenia: clinical correlates and association with functional outcome. *Schizophrenia Research*, 63, 79-88.
- Kinon, BJ, Noordsy, DL, Liu-Seifert, H, Gulliver, AH, Ascher-Svanum, H Kollack-Walker, S 2006. Randomized, double-blind 6-month comparison of olanzapine and quetiapine in patients with schizophrenia or schizoaffective disorder with prominent negative symptoms and poor functioning. *Journal of Clinical Psychopharmacology*, **26**, 453-461.
- Kinon, BJ, Potts, AJ Watson, SB 2011. Placebo response in clinical trials with schizophrenia patients. *Current Opinion in Psychiatry*, **24**, 107-113.
- Kirkpatrick, B, Amador, XF, Flaum, M, Yale, SA, Gorman, JM, Carpenter, WT, Tohen, M Mcglashan, T 1996a. The deficit syndrome in the DSM-IV Field Trial: I. Alcohol and other drug abuse. *Schizophrenia Research*, **20**, 69-77.
- Kirkpatrick, B, Buchanan, RW, Breier, A Carpenter, WTJJ 1994. Depressive symptoms and the deficit syndrome of schizophrenia. *Journal of Nervous & Mental Disease*, **182**, 452-455.
- Kirkpatrick, B, Buchanan, RW, Mckenny, PD, Alphs, LD Carpenter, WT 1989. The schedule for the deficit syndrome: an instrument for research in schizophrenia. *Psychiatry Research*, **30**, 119-123.
- Kirkpatrick, B, Buchanan, RW, Ross, DE Carpenter, WT 2001. A separate disease within the syndrome of schizophrenia. *Archives of General Psychiatry*, **58**, 165-171.
- Kirkpatrick, B, Castle, D, Murray, RM Carpenter Jr, WT 2000a. Risk factors for the deficit syndrome of schizophrenia. *Schizophrenia Bulletin*, **26**, 233-242.
- Kirkpatrick, B, Fenton, WS, Carpenter, WT, Jr. Marder, SR 2006. The NIMH-MATRICS consensus statement on negative symptoms. *Schizophrenia Bulletin*, **32**, 214-219.
- Kirkpatrick, B Galderisi, S 2008. Deficit schizophrenia: an update. World Psychiatry, 7, 143-147.
- Kirkpatrick, B, Ram, R Bromet, E 1996b. The deficit syndrome in the Suffolk County mental health project. *Schizophrenia Research*, **22**, 119-126.
- Kirkpatrick, B, Ross, DE, Walsh, D, Karkowski, L Kendler, KS 2000b. Family characteristics of deficit and nondeficit schizophrenia in the Roscommon Family Study. *Schizophrenia Research*, **45**, 57-64.
- Kirkpatrick, B, Strauss, GP, Nguyen, L, Fischer, BA, Daniel, DG, Cienfuegos, A Marder, SR 2011. The brief negative symptom scale: psychometric properties. *Schizophrenia Bulletin*, **37**, 300-305.
- Kitamura, T, Okazaki, Y, Fujinawa, A, Yoshino, M Kasahara, Y 1995. Symptoms of psychoses. A factor-analytic study. *The British Journal of Psychiatry*, **166**, 236-240.
- Klingberg, S, Wolwer, W, Engel, C, Wittorf, A, Herrlich, J, Meisner, C, Buchkremer, G Wiedemann, GJ 2011. Negative symptoms of schizophrenia as primary target of cognitive behavioral therapy: Results of the randomized clinical TONES study. *Schizophrenia Bulletin*, **37**, 98-110.
- Knutson, B, Fong, GW, Adams, CM, Varner, JL Hommer, D 2001. Dissociation of reward anticipation and outcome with event-related fMRI. *Neuroreport*, **12**, 3683-3687.
- Koehler, K, Guth, W Grimm, G 1977. First-rank symptoms of schizophrenia in Schneideroriented German centers. *Archives of General Psychiatry*, **34**, 810-813.

- Krabbendam, L Van Os, J 2005. Schizophrenia and urbanicity: a major environmental influence—conditional on genetic risk. *Schizophrenia Bulletin*, **31**, 795-799.
- Kraepelin, E 1971. *Dementia praecox and paraphrenia (1919). Translated by Barclay RM; edited by Robertson* GM. New York, Robert E Krieger.
- Kravariti, E, Morris, RG, Rabe-Hesketh, S, Murray, RM Frangou, S 2003. The Maudsley Early-Onset Schizophrenia Study: cognitive function in adolescent-onset schizophrenia. *Schizophrenia Research*, **65**, 95-103.
- Krawiecka, M, Goldberg, D Vaughan, M 1977. A standardized psychiatric assessment scale for rating chronic psychotic patients. *Acta Psychiatrica Scandinavica*, **55**, 299-308.
- Kring, AM Earnst, KS 1999. Stability of emotional responding in schizophrenia. *Behavior Therapy*, **30**, 373-388.
- Kring, AM, Gur, RE, Blanchard, JJ, Horan, WP Reise, SP 2013. The clinical assessment interview for negative symptoms (CAINS): final development and validation. *American Journal of Psychiatry*, **170**, 165–172.
- Kring, AM, Kerr, SL, Smith, DA Neale, JM 1993. Flat affect in schizophrenia does not reflect diminished subjective experience of emotion. *Journal of Abnormal Psychology*, **102**, 507-517.
- Kring, AM Neale, JM 1996. Do schizophrenic patients show a disjunctive relationship among expressive, experiential, and psychophysiological components of emotion? *Journal of Abnormal Psychology*, **105**, 249-257.
- Kross, E, Davidson, M, Weber, J Ochsner, K 2009. Coping with emotions past: the neural bases of regulating affect associated with negative autobiographical memories. *Biological Psychiatry*, 65, 361-366.
- Kulhara, P Chandiramani, K 1990. Positive and negative subtypes of schizophrenia: a follow-up study from India. *Schizophrenia Research*, **3**, 107-116.
- Kupper, Z, Ramseyer, F, Hoffmann, H, Kalbermatten, S Tschacher, W 2010. Video-based quantification of body movement during social interaction indicates the severity of negative symptoms in patients with schizophrenia. *Schizophrenia Research*, **121**, 90-100.
- Kurtz, MM Mueser, KT 2008. A meta-analysis of controlled research on social skills training for schizophrenia. *Journal of Consulting Clinical Psychology*, **76**, 491-504.
- Kurtz, MM Richardson, CL 2012. Social cognitive training for schizophrenia: A meta-analytic investigation of controlled research. *Schizophrenia Bulletin*, **38**, 1092-1104.
- Kwapil, TR 1998. Social anhedonia as a predictor of the development of schizophreniaspectrum disorders. *Journal of Abnormal Psychology*, **107**, 558.
- Lader, D, Short, S Gershuny, J. 2006. *The time use survey, 2005* [Online]. *URL: http://www.timeuse.org/information/publications/docs/TimeUse2005.pdf*.
- Lancon, C, Auquier, P, Nayt, G Reine, G 2000. Stability of the five-factor structure of the Positive and Negative Syndrome Scale (PANSS). *Schizophrenia Research*, **42**, 231-239.
- Lancon, C, Reine, G, Llorca, P Auquier, P 1999. Validity and reliability of the French-language version of the Positive and Negative Syndrome Scale (PANSS). *Acta Psychiatrica Scandinavica*, **100**, 237-243.
- Larsson, S, Andreassen, OA, Aas, M, Røssberg, JI, Mork, E, Steen, NE, Barrett, EA, Lagerberg, TV, Peleikis, D Agartz, I 2013. High prevalence of childhood trauma in patients with

schizophrenia spectrum and affective disorder. *Comprehensive Psychiatry*, **54**, 123-127.

- Lasser, RA, Dirks, B, Nasrallah, H, Kirsch, C, Gao, J, Pucci, ML, Knesevich, MA Lindenmayer, J-P 2013. Adjunctive lisdexamfetamine dimesylate therapy in adult outpatients with predominant negative symptoms of schizophrenia: open-label and randomizedwithdrawal phases. *Neuropsychopharmacology*, **38**, 2140-2149.
- Lecrubier, Y, Quintin, P, Bouhassira, M, Perrin, E Lancrenon, SG 2006. The treatment of negative symptoms and deficit states of chronic schizophrenia: Olanzapine compared to amisulpride and placebo in a 6-month double-blind controlled clinical trial. *Acta Psychiatrica Scandinavica*, **114**, 319-327.
- Lehman, AF, Ward, NC Linn, LS 1982. Chronic mental patients: the quality of life issue. *The American Journal of Psychiatry*, **139**, 1271–1276.
- Lelliott, P, Paton, C, Harrington, M, Konsolaki, M, Sensky, T Okocha, C 2002. The influence of patient variables on polypharmacy and combined high dose of antipsychotic drugs prescribed for in-patients. *The Psychiatrist*, **26**, 411-414.
- Lencz, T, Smith, CW, Mclaughlin, D, Auther, A, Nakayama, E, Hovey, L Cornblatt, BA 2006. Generalized and specific neurocognitive deficits in prodromal schizophrenia. *Biological Psychiatry*, **59**, 863-871.
- Lenzenweger, MF Dworkin, RH 1996. The dimensions of schizophrenia phenomenology. Not one or two, at least three, perhaps four. *The British Journal of Psychiatry*, **168**, 432-440.
- Leucht, S, Arbter, D, Engel, RR, Kissling, W Davis, JM 2009a. How effective are secondgeneration antipsychotic drugs? A meta-analysis of placebo-controlled trials. *Molecular Psychiatry*, **14**, 429-447.
- Leucht, S, Corves, C, Arbter, D, Engel, RR, Li, C Davis, JM 2009b. Second-generation versus firstgeneration antipsychotic drugs for schizophrenia: a meta-analysis. *Lancet*, **373**, 31-41.
- Leucht, S Engel, RR 2006. The relative sensitivity of the Clinical Global Impressions Scale and the Brief Psychiatric Rating Scale in antipsychotic drug trials. *Neuropsychopharmacology*, **31**, 406-412.
- Leucht, S, Kane, JM, Etschel, E, Kissling, W, Hamann, J Engel, RR 2006. Linking the PANSS, BPRS, and CGI: clinical implications. *Neuropsychopharmacology*, **31**, 2318-2325.
- Leucht, S, Kane, JM, Kissling, W, Hamann, J, Etschel, E Engel, RR 2005. What does the PANSS mean? *Schizophrenia Research*, **79**, 231-238.
- Levine, SZ Leucht, S 2013. Identifying clinically meaningful symptom response cut-off values on the SANS in predominant negative symptoms. *Schizophrenia Research*, **145**, 125-127.
- Levine, SZ, Rabinowitz, J, Engel, R, Etschel, E Leucht, S 2008. Extrapolation between measures of symptom severity and change: an examination of the PANSS and CGI. *Schizophr Res*, **98**, 318-322.
- Lewine, RR, Fogg, L Meltzer, HY 1983. Assessment of negative and positive symptoms in schizophrenia. *Schizophrenia Bulletin*, **9**, 368-376.
- Lewis, G, David, A, Andréassson, S Allebeck, P 1992. Schizophrenia and city life. *The Lancet,* **340**, 137-140.
- Liberati, A, Altman, DG, Tetzlaff, J, Mulrow, C, Gotzsche, PC, Ioannidis, JP, Clarke, M, Devereaux, PJ, Kleijnen, J Moher, D 2009. The PRISMA statement for reporting

systematic reviews and meta-analyses of studies that evaluate healthcare interventions: explanation and elaboration. *BMJ*, **339**, b2700.

- Liberman, RP, Kopelowicz, A, Ventura, J Gutkind, D 2002. Operational criteria and factors related to recovery from schizophrenia. *Int Rev Psychiatry*, **14**, 256–272.
- Liberman, RP, Wallace, CJ, Blackwell, G, Kopelowicz, A, Vaccaro, JV Mintz, J 1998. Skills training versus psychosocial occupational therapy for persons with persistent schizophrenia. *Am J Psychiatry*, **155**, 1087-1091.
- Liddle, PF 1987. The symptoms of chronic schizophrenia. A re-examination of the positivenegative dichotomy. *The British Journal of Psychiatry*, **151**, 145-151.
- Liddle, PF Barnes, TR 1988. The subjective experience of deficits in schizophrenia. *Comprehensive Psychiatry*, **29**, 157-164.
- Lieberman, JA, Dunbar, G, Segreti, AC, Girgis, RR, Seoane, F, Beaver, JS, Duan, N Hosford, DA 2013. A randomized exploratory trial of an alpha-7 nicotinic receptor agonist (TC-5619) for cognitive enhancement in schizophrenia. *Neuropsychopharmacology*, **38**, 968-975.
- Lieberman, JA, Stroup, TS, Mcevoy, JP, Swartz, MS, Rosenheck, RA, Perkins, DO, Keefe, RS, Davis, SM, Davis, CE, Lebowitz, BD, Severe, J, Hsiao, JK Clinical Antipsychotic Trials of Intervention Effectiveness, I 2005. Effectiveness of antipsychotic drugs in patients with chronic schizophrenia. *New England Journal of Medicine*, **353**, 1209-1223.
- Lindenmayer, J-P, Bernstein-Hyman, R Grochowski, S 1994. Five-Factor Model of Schizophrenia Initial Validation. *The Journal of Nervous and Mental Disease*, **182**, 631-638.
- Lindenmayer, J-P, Grochowski, S Kay, SR 1991. Schizophrenic patients with depression: psychopathological profiles and relationship with negative symptoms. *Comprehensive Psychiatry*, **32**, 528-533.
- Linscott, RJ, Allardyce, J Van Os, J 2010. Seeking verisimilitude in a class: a systematic review of evidence that the criterial clinical symptoms of schizophrenia are taxonic. *Schizophrenia Bulletin*, **36**, 811-829.
- Little, RJ Rubin, DB 1989. The analysis of social science data with missing values. *Sociological Methods & Research*, **18**, 292-326.
- Liu, F, Guo, X, Wu, R, Ou, J, Zheng, Y, Zhang, B, Xie, L, Zhang, L, Yang, L, Yang, S, Yang, J, Ruan, Y, Zeng, Y, Xu, X Zhao, J 2014. Minocycline supplementation for treatment of negative symptoms in early-phase schizophrenia: A double blind, randomized, controlled trial. *Schizophrenia Research Feb*, **153**, 169-176.
- Loebel, AD, Khanna, S, Rajadhyaksha, S, Siu, CO, Giller, E Potkin, SGJ 2007. Ziprasidone in treatment-resistant schizophrenia: a 52-week, open-label continuation study. *The Journal of Clinical Psychiatry*, **68**, 1333-1338.
- Long, JD Brekke, JS 1999. Longitudinal factor structure of the Brief Psychiatric Rating Scale in schizophrenia. *Psychological Assessment*, **11**, 498-506.
- Loo, H, Poirier-Littre, MF, Theron, M, Rein, W Fleurot, OG 1997. Amisulpride versus placebo in the medium-term treatment of the negative symptoms of schizophrenia. *British Journal of Psychiatry*, **170**, 18-22.
- Lorr, M, Mcnair, D, Klett, C Lasky, J 1962. Evidence of ten psychotic syndromes. *Journal of Consulting Psychology*, **26**, 185-189.
- Luchman, JN 2014. DOMIN: Stata module to conduct dominance analysis. Statistical Software Components.

- Lukoff, D, Nuechterlein, K Ventura, J 1986. Manual for the expanded brief psychiatric rating scale. *Schizophr Bull*, **12**, 594-602.
- Lykouras, L, Oulis, P, Psarros, K, Daskalopoulou, E, Botsis, A, Christodoulou, GN Stefanis, C 2000. Five-factor model of schizophrenic psychopathology: how valid is it? *European Archives of Psychiatry and Clinical Neuroscience*, **250**, 93-100.
- Lyne, JP, Kinsella, A O'donoghue, B 2012. Can we combine symptom scales for collaborative research projects? *Journal of Psychiatric Research*, **46**, 233-238.
- Lysaker, P Salyers, M 2007. Anxiety symptoms in schizophrenia spectrum disorders: associations with social function, positive and negative symptoms, hope and trauma history. *Acta Psychiatrica Scandinavica*, **116**, 290-298.
- Lysaker, PH Davis, LW 2004. Social function in schizophrenia and schizoaffective disorder: associations with personality, symptoms and neurocognition. *Health and Quality of Life Outcomes*, **2**, 1-6.
- Maccallum, RC, Browne, MW Sugawara, HM 1996. Power analysis and determination of sample size for covariance structure modeling. *Psychological Methods*, **1**, 130-149.
- Malaspina, D, Harlap, S, Fennig, S, Heiman, D, Nahon, D, Feldman, D Susser, ES 2001.
 Advancing paternal age and the risk of schizophrenia. *Archives of General Psychiatry*, 58, 361-367.
- Malhi, GS, Green, M, Fagiolini, A, Peselow, ED Kumari, V 2008. Schizoaffective disorder: diagnostic issues and future recommendations. *Bipolar Disorders*, **10**, 215-230.
- Malla, AK, Norman, RM, Williamson, P, Cortese, L Diaz, F 1993. Three syndrome concept of schizophrenia: a factor analytic study. *Schizophrenia Research*, **10**, 143-150.
- Marcelis, M, Navarro-Mateu, F, Murray, R, Selten, J-P Van Os, J 1998. Urbanization and psychosis: a study of 1942–1978 birth cohorts in The Netherlands. *Psychological Medicine*, **28**, 871-879.
- Marder, SR, Alphs, L, Anghelescu, I-G, Arango, C, Barnes, TR, Caers, I, Daniel, DG, Dunayevich, E, Fleischhacker, W, Garibaldi, G, Green, MF, Harvey, PD, Kahn, RS, Kane, JM, Keefe, RS, Kinon, B, Leucht, S, Lindenmayer, J-P, Malhotra, AK, Stauffer, V, Umbricht, D, Wesnes, K, Kapur, S Rabinowitz, J 2013. Issues and perspectives in designing clinical trials for negative symptoms in schizophrenia. *Schizophrenia Research*, **150**, 328-333.
- Marder, SR, Daniel, DG, Alphs, L, Awad, AG Keefe, RS 2011. Methodological issues in negative symptom trials. *Schizophrenia Bulletin*, **37**, 250-254.
- Marder, SR, Davis, JM Chouinard, G 1997. The effects of risperidone on the five dimensions of schizophrenia derived by factor analysis: combined results of the North American trials. *Journal of Clinical Psychiatry*, **58**, 538–546.
- Marder, SR, Essock, SM, Miller, AL, Buchanan, RW, Casey, DE, Davis, JM, Kane, JM, Lieberman, JA, Schooler, NR Covell, N 2014. Physical health monitoring of patients with schizophrenia. *American Journal of Psychiatry*, **161**, 1334-1339.
- Marder, SR Fenton, W 2004. Measurement and Treatment Research to Improve Cognition in Schizophrenia: NIMH MATRICS initiative to support the development of agents for improving cognition in schizophrenia. *Schizophrenia Research*, **72**, 5-9.
- Marder, SR, Glynn, SM, Wirshing, WC, Wirshing, DA, Ross, D, Widmark, C, Mintz, J, Liberman, RP Blair, KEJ 2003. Maintenance treatment of schizophrenia with risperidone or haloperidol: 2-year outcomes. *American Journal of Psychiatry*, **160**, 1405-1412.

- Marder, SR Meibach, RC 1994. Risperidone in the treatment of schizophrenia. *American Journal of Psychiatry*, **151**, 825-835.
- Marneros, A, Deister, A Rohde, A 1992. Validity of the negative/positive dichotomy for schizophrenic disorders under long-term conditions. *Schizophrenia Research*, **7**, 117-123.
- Mass, R, Schoemig, T, Hitschfeld, K, Wall, E Haasen, C 2000. Psychopathological syndromes of schizophrenia: evaluation of the dimensional structure of the positive and negative syndrome scale. *Schizophrenia Bulletin*, **26**, 167-177.
- Maziade, M, Roy, M-A, Martinez, M Cliche, D 1995. Negative, psychoticism, and disorganized dimensions in patients with familial schizophrenia or bipolar disorder: continuity and discontinuity between the major psychoses. *The American Journal of Psychiatry*, **152**, 1458-1463.
- Mcdonald, AM, Knight, RC, Campbell, MK, Entwistle, VA, Grant, AM, Cook, JA, Elbourne, DR, Francis, D, Garcia, J Roberts, I 2006. What influences recruitment to randomised controlled trials? A review of trials funded by two UK funding agencies. *Trials*, **7**, 9.
- Mcdonell, MG, Short, RA, Berry, CM Dyck, DG 2003. Burden in schizophrenia caregivers: impact of family psychoeducation and awareness of patient suicidality. *Family Process*, **42**, 91-103.
- Mcfadden, D 1974. Conditional logit analysis of qualitative choice behavior, Zarembka P., Frontiers in Econometrics, , Academic Press, New York.
- Mcghie, A Chapman, J 1961. Disorders of attention and perception in early schizophrenia. *British Journal of Medical Psychology*, **34**, 103-116.
- Mcglashan, TH Fenton, WS 1992. The positive-negative distinction in schizophrenia: review of natural history validators. *Archives of General Psychiatry*, **49**, 63-72.
- Mcglashan, TH Fenton, WS 1993. Subtype progression and pathophysiologic deterioration in early schizophrenia. *Schizophrenia Bulletin*, **19**, 71.
- Mcgorry, P, Bell, R, Dudgeon, P Jackson, H 1998. The dimensional structure of first episode psychosis: an exploratory factor analysis. *Psychological Medicine*, **28**, 935-947.
- Mcgrath, J 1999. Hypothesis: is low prenatal vitamin D a risk-modifying factor for schizophrenia? *Schizophrenia Research*, **40**, 173-177.
- Mcgrath, J, Saha, S, Chant, D Welham, J 2008. Schizophrenia: a concise overview of incidence, prevalence, and mortality. *Epidemiologic reviews*, **30**, 67-76.
- Mcgrath, JJ 2006. Variations in the incidence of schizophrenia: data versus dogma. *Schizophrenia Bulletin*, **32**, 195-197.
- Mckinney, WT, Moran, E, Kraemer, G Prange Jr, A 1980. Long-term chlorpromazine in rhesus monkeys: production of dyskinesias and changes in social behavior. *Psychopharmacology*, **72**, 35-39.
- Mckinney, WT Moran, EC 1981. Animal models of schizophrenia. *American Journal of Psychiatry*, **138**, 478-483.
- Meehl, PE 1962. Schizotaxia, schizotypy, schizophrenia. American psychologist, 17, 827-838.
- Mellers, J, Sham, P, Jones, P, Toone, B Murray, R 1996. A factor analytic study of symptoms in acute schizophrenia. Acta Psychiatrica Scandinavica, **93**, 92-98.

- Meltzer, HY, Bobo, WV, Lee, MA, Cola, P Jayathilake, KG 2010. A randomized trial comparing clozapine and typical neuroleptic drugs in non-treatment-resistant schizophrenia. *Psychiatry Research*, **177**, 286-293.
- Messias, E, Kirkpatrick, B, Bromet, E, Ross, D, Buchanan, RW, Carpenter, WT, Tek, C, Kendler, KS, Walsh, D Dollfus, S 2004. Summer birth and deficit schizophrenia: a pooled analysis from 6 countries. *Archives of General Psychiatry*, **61**, 985-989.
- Miles, J Shevlin, M 2001. Applying regression and correlation: A guide for students and researchers, Sage.
- Milev, P, Ho, B-C, Arndt, S Andreasen, NC 2005. Predictive values of neurocognition and negative symptoms on functional outcome in schizophrenia: a longitudinal first-episode study with 7-year follow-up. *American Journal of Psychiatry*, **162**, 495–506.
- Miller, BK, Konopaske, R Byrne, ZS 2012. Dominance analysis of two measures of organizational justice. *Journal of Managerial Psychology*, **27**, 264-282.
- Minas, IH, Klimidis, S, Stuart, GW, Copolov, DL Singh, BS 1994. Positive and negative symptoms in the psychoses: principal components analysis of items from the Scale for the Assessment of Positive Symptoms and the Scale for the Assessment of Negative Symptoms. *Comprehensive Psychiatry*, **35**, 135-144.
- Minas, IH, Stuart, G, Klimidis, S, Jackson, H, Singh, B Copolov, D 1992. Positive and negative symptoms in the psychoses: multidimensional scaling of SAPS and SANS items. *Schizophrenia Research*, **8**, 143-156.
- Mino, Y, Kodera, R Bebbington, P 1990. A comparative study of psychiatric services in Japan and England. *The British Journal of Psychiatry*, **157**, 416-420.
- Minozzi, S, Davoli, M, Bargagli, AM, Amato, L, Vecchi, S Perucci, CA 2010. An overview of systematic reviews on cannabis and psychosis: discussing apparently conflicting results. *Drug and Alcohol Review*, **29**, 304-317.
- Mintz, AR, Dobson, KS Romney, DM 2003. Insight in schizophrenia: a meta-analysis. *Schizophrenia Research*, **61**, 75-88.
- Mintz, S Alpert, M 1972. Imagery vividness, reality testing, and schizophrenic hallucinations. Journal of Abnormal Psychology, **79**, 310-316.
- Möller, H-J 2007. Clinical evaluation of negative symptoms in schizophrenia. *European psychiatry*, **22**, 380-386.
- Moller, HJ, Muller, H, Borison, RL, Schooler, NR Chouinard, GJ 1995. A path-analytical approach to differentiate between direct and indirect drug effects on negative symptoms in schizophrenic patients. A re-evaluation of the North American risperidone study. *European Archives of Psychiatry & Clinical Neuroscience*, **245**, 45-49.
- Moller, HJ, Riedel, M, Muller, N, Fischer, W Kohnen, RG 2004. Zotepine Versus Placebo in the Treatment of Schizophrenic Patients with Stable Primary Negative Symptoms: A Randomized Double-Blind Multicenter Trial. *Pharmacopsychiatry*, **37**, 270-278.
- Mooney, CZ, Duval, RD Duval, R 1993. *Bootstrapping: A nonparametric approach to statistical inference*, Sage.
- Morgan, C Fisher, H 2007. Environment and schizophrenia: environmental factors in schizophrenia: childhood trauma—a critical review. *Schizophrenia Bulletin*, **33**, 3-10.
- Morgan, C, Lappin, J, Heslin, M, Donoghue, K, Lomas, B, Reininghaus, U, Onyejiaka, A, Croudace, T, Jones, PB Murray, RM 2014. Reappraising the long-term course and

outcome of psychotic disorders: the AESOP-10 study. *Psychological Medicine*, **44**, 2713-2726.

- Morgan, V, Castle, D, Page, A, Fazio, S, Gurrin, L, Burton, P, Montgomery, P Jablensky, A 1997. Influenza epidemics and incidence of schizophrenia, affective disorders and mental retardation in Western Australia: no evidence of a major effect. *Schizophrenia Research*, **26**, 25-39.
- Morris, R, Rushe, T, Woodruffe, P Murray, R 1995. Problem solving in schizophrenia: a specific deficit in planning ability. *Schizophrenia Research*, **14**, 235-246.
- Morris, SB 2000. Distribution of the standardized mean change effect size for meta-analysis on repeated measures. *British Journal of Mathematical and Statistical Psychology*, **53**, 17-29.
- Morrison, AP, French, P, Walford, L, Lewis, SW, Kilcommons, A, Green, J, Parker, S Bentall, RP 2004. Cognitive therapy for the prevention of psychosis in people at ultra-high risk Randomised controlled trial. *The British Journal of Psychiatry*, **185**, 291-297.
- Mortimer, AM 2007. Symptom rating scales and outcome in schizophrenia. *The British Journal* of Psychiatry, **191**, s7-s14.
- Mucci, A, Galderisi, S, Kirkpatrick, B, Bucci, P, Volpe, U, Merlotti, E, Centanaro, F, Catapano, F Maj, M 2007. Double dissociation of N1 and P3 abnormalities in deficit and nondeficit schizophrenia. *Schizophrenia Research*, **92**, 252-261.
- Mueser, KT, Curran, PJ Mchugo, GJ 1997. Factor structure of the Brief Psychiatric Rating Scale in schizophrenia. *Psychological Assessment*, **9**, 196-204.
- Mueser, KT, Doonan, R, Penn, DL, Blanchard, JJ, Bellack, AS, Nishith, P Deleon, J 1996. Emotion recognition and social competence in chronic schizophrenia. *Journal of Abnormal Psychology*, **105**, 271-275
- Mulholland, C Cooper, S 2000. The symptom of depression in schizophrenia and its management. *Advances in Psychiatric Treatment*, **6**, 169-177.
- Mullen, A 2009. Mental health nurses establishing psychosocial interventions within acute inpatient settings. *International Journal of Mental Health Nursing*, **18**, 83-90.
- Murphy, BM, Burke, J, Bray, J, Walsh, D Kendler, K 1994. An analysis of the clinical features of familial schizophrenia. *Acta Psychiatrica Scandinavica*, **89**, 421-427.
- Nakaya, M, Suwa, H Ohmori, K 1999. Latent structures underlying schizophrenic symptoms: a five-dimensional model. *Schizophrenia Research*, **39**, 39-50.
- NCCfMH 2010. Schizophrenia: The NICE Guideline on Core Interventions in the Treatment and Management of Schizophrenia in Adults in Primary and Secondary Care (Updated edition). *London: The British Psychological Society and the Royal College of Psychiatrists*.
- Newcomer, JW, Faustman, WO, Yeh, W Csernansky, JG 1990. Distinguishing depression and negative symptoms in unmedicated patients with schizophrenia. *Psychiatry Research*, **31**, 243-250.
- Nierenberg, AA Dececco, LM 2000. Definitions of antidepressant treatment response, remission, nonresponse, partial response, and other relevant outcomes: a focus on treatment-resistant depression. *The Journal of Clinical Psychiatry*, **62**, 5-9.
- Nopoulos, P, Flashman, L, Flaum, M, Arndt, S Andreasen, N 1994. Stability of cognitive functioning early in the course of schizophrenia. *Schizophrenia Research*, **14**, 29-37.

- Nordgaard, J, Arnfred, SM, Handest, P Parnas, J 2008. The diagnostic status of first-rank symptoms. *Schizophrenia Bulletin*, **34**, 137-154.
- Nosarti, C, Reichenberg, A, Murray, RM, Cnattingius, S, Lambe, MP, Yin, L, Maccabe, J, Rifkin, L Hultman, CM 2012. Preterm birth and psychiatric disorders in young adult life. *Archives of General Psychiatry*, **69**, 610-617.
- Nose, M, Tansella, M, Thornicroft, G, Schene, A, Becker, T, Veronese, A, Leese, M, Koeter, M, Angermeyer, M Barbui, C 2008. Is the Defined Daily Dose system a reliable tool for standardizing antipsychotic dosages? *International Clinical Psychopharmacology*, 23, 287-290.
- Nuechterlein, KH, Barch, DM, Gold, JM, Goldberg, TE, Green, MF Heaton, RK 2004. Identification of separable cognitive factors in schizophrenia. *Schizophrenia Research*, **72**, 29-39.
- Nuechterlein, KH, Green, MF, Kern, RS, Baade, LE, Barch, DM, Cohen, JD, Essock, S, Fenton, WS, Frese Iii, FJ Gold, JM 2008. The MATRICS Consensus Cognitive Battery, part 1: test selection, reliability, and validity. *The American Journal of Psychiatry*, **165**, 203-213.
- Obermeier, M, Mayr, A, Schennach-Wolff, R, Seemüller, F, Möller, H-J Riedel, M 2010. Should the PANSS be rescaled? *Schizophrenia Bulletin*, **36**, 455-460.
- Ogino, S, Miyamoto, S, Miyake, N Yamaguchi, N 2014. Benefits and limits of anticholinergic use in schizophrenia: focusing on its effect on cognitive function. *Psychiatry and Clinical Neurosciences*, **68**, 37-49.
- Olie, JP, Spina, E, Murray, S Yang, RG 2006. Ziprasidone and amisulpride effectively treat negative symptoms of schizophrenia: Results of a 12-week, double-blind study. *International Clinical Psychopharmacology*, **21**, 143-151.
- Oltmanns, TF 1988. Approaches to the definition and study of delusions. New York: Wiley.
- Orfanos, S, Banks, C Priebe, S 2015. Are Group Psychotherapeutic Treatments Effective for Patients with Schizophrenia? A Systematic Review and Meta-Analysis. *Psychotherapy and Psychosomatics*, **84**, 241-249.
- Ösby, U, Correia, N, Brandt, L, Ekbom, A Sparén, P 2000. Mortality and causes of death in schizophrenia in Stockholm county, Sweden. *Schizophrenia Research*, **45**, 21-28.
- Oshima, I, Mino, Y Inomata, Y 2003. Institutionalisation and schizophrenia in Japan: social environments and negative symptoms: Nationwide survey of in-patients. *The British Journal of Psychiatry*, **183**, 50-56.
- Overall, JE Beller, SA 1984. The Brief Psychiatric Rating Scale (BPRS) in geropsychiatric research: I. Factor structure on an inpatient unit. *Journal of Gerontology*, **39**, 187-193.
- Overall, JE Gorham, DR 1962. The brief psychiatric rating scale. *Psychological Rep*, **10**, 799-812.
- Overall, JE Klett, J 1972. Applied multivariate analysis. New York McGraw-Hill
- Overall, JE Rhoades, M 1982. Refinement of phenomenological classification in clinical psychopharmacology research. *Psychopharmacology*, **77**, 24-30.
- Overall, JE Woodward, JA 1975. Conceptual validity of a phenomenological classification of psychiatric patients. *Journal of Psychiatric Research*, **12**, 215-230.
- Pach, J, Finkbeiner, T, Glaser, T, Haug, J, Osterheider, M Tegeler, J 1998. Positive and negative symptoms in chronic schizophrenic patients under maintenance therapy with flupenthixol decanoate for a twelve month perioid. *Fortschritte der Neurologie Psychiatrie*, 66, 442-449.

- Paton, C, Barnes, TR, Cavanagh, M-R, Taylor, D Lelliott, P 2008. High-dose and combination antipsychotic prescribing in acute adult wards in the UK: the challenges posed by prn prescribing. *The British Journal of Psychiatry*, **192**, 435-439.
- Pearson, K, Lee, A Bramley-Moore, L 1899. Mathematical contributions to the theory of evolution. VI. Genetic (reproductive) selection: Inheritance of fertility in man, and of fecundity in thoroughbred racehorses. *Philosophical Transactions of the Royal Society* of London. Series A, Containing Papers of a Mathematical or Physical Character, 257-330.
- Pedersen, CB Mortensen, PB 2006. Urbanization and traffic related exposures as risk factors for schizophrenia. *BMC Psychiatry*, **6**, 2.
- Pedhazur, EJ 1982. *Multiple regression in behavioral research: Explanation and prediction*, New York: Holt, Rinehart and Winston Inc.
- Peet, M Horrobin, DFG 2002. A dose-ranging exploratory study of the effects of ethyleicosapentaenoate in patients with persistent schizophrenic symptoms. *Journal of Psychiatric Research*, **36**, 7-18.
- Peralta, V Cuesta, MJ 1994. Psychometric properties of the positive and negative syndrome scale (PANSS) in schizophrenia. *Psychiatry Research*, **53**, 31-40.
- Peralta, V Cuesta, MJ 1999a. Diagnostic significance of Schneider's first-rank symptoms in schizophrenia. Comparative study between schizophrenic and non-schizophrenic psychotic disorders. *The British Journal of Psychiatry*, **174**, 243-248.
- Peralta, V Cuesta, MJ 1999b. Dimensional structure of psychotic symptoms: an item-level analysis of SAPS and SANS symptoms in psychotic disorders. *Schizophrenia Research*, **38**, 13-26.
- Peralta, V Cuesta, MJ 1999c. Negative, parkinsonian, depressive and catatonic symptoms in schizophrenia: a conflict of paradigms revisited. *Schizophrenia Research*, **40**, 245-253.
- Peralta, V Cuesta, MJ 2001. How many and which are the psychopathological dimensions in schizophrenia? Issues influencing their ascertainment. *Schizophrenia Research*, **49**, 269-285.
- Peralta, V Cuesta, MJ 2004. The deficit syndrome of the psychotic illness. *European Archives of Psychiatry and Clinical Neuroscience*, **254**, 165-171.
- Peralta, V, Cuesta, MJ De Leon, J 1992a. Formal thought disorder in schizophrenia: a factor analytic study. *Comprehensive Psychiatry*, **33**, 105-110.
- Peralta, V, Cuesta, MJ De Leon, J 1994. An empirical analysis of latent structures underlying schizophrenic symptoms: a four-syndrome model. *Biological Psychiatry*, **36**, 726-736.
- Peralta, V, Cuesta, MJ Farre, C 1997. Factor structure of symptoms in functional psychoses. *Biological Psychiatry*, **42**, 806-815.
- Peralta, V, De Leon, J Cuesta, MJ 1992b. Are there more than two syndromes in schizophrenia? A critique of the positive-negative dichotomy. *The British Journal of Psychiatry*, **161**, 335-343.
- Perenyi, A, Norman, T, Hopwood, M Burrows, G 1998. Negative symptoms, depression and parkinsonian symptoms in chronic, hospitalised schizophrenic patients. *Journal of affective disorders*, **48**, 163-169.
- Pfohl, B Winokur, G 1983. The micropsychopathology of hebephrenic/catatonic schizophrenia. *The Journal of nervous and mental disease*, **171**, 296-300.
- Pinel, P 1806. A treatise on insanity. London, United Kingdom: Messers Cadell & Davies, Strand

- Pinto, R, Bannerjee, A Ghosh, N 1979. A double-blind comparison of flupenthixol decanoate and fluphenazine decanoate in the treatment of chronic schizophrenia. *Acta Psychiatrica Scandinavica*, **60**, 313-322.
- Pogue-Geile, MF Harrow, M 1985. Negative Symptoms in Schizophrenia: Their Longitudinal Course and Prognostic Importance. *Schizophrenia Bulletin*, **11**, 427-439.
- Priebe, S 2007. Social outcomes in schizophrenia. British Journal of Psychiatry, 50, 15–20.
- Priebe, S Fakhoury, WK 2007. *Quality of life. In The clinical Handbook of schizophrenia* Guildford Press.
- Priebe, S Gruyters, T 1994. Patients' and caregivers' initial assessments of day-hospital treatment and course of symptoms. *Comprehensive Psychiatry*, **35**, 234-238.
- Priebe, S Gruyters, T 1995. Patients' assessment of treatment predicting outcome. *Schizophrenia Bulletin*, **21**, 87-94.
- Priebe, S, Huxley, P, Knight, S Evans, S 1999. Application and results of the Manchester Short Assessment of Quality of Life (MANSA). *International Journal of Social Psychiatry*, **45**, 7–12.
- Priebe, S, Kaiser, W, Huxley, PJ, Röder-Wanner, U-U Rudolf, H 1998. Do different subjective evaluation criteria reflect distinct constructs? *The Journal of Nervous and Mental Disease*, **186**, 385-392.
- Priebe, S, Katsakou, C, Amos, T, Leese, M, Morriss, R, Rose, D, Wykes, T Yeeles, K 2009. Patients' views and readmissions 1 year after involuntary hospitalisation. *The British Journal of Psychiatry*, **194**, 49-54.
- Priebe, S, Katsakou, C, Glöckner, M, Dembinskas, A, Fiorillo, A, Karastergiou, A, Kiejna, A, Kjellin, L, Nawka, P Onchev, G 2010a. Patients' views of involuntary hospital admission after 1 and 3 months: prospective study in 11 European countries. *The British Journal* of Psychiatry, **196**, 179-185.
- Priebe, S, Katsakou, C, Yeeles, K, Amos, T, Morriss, R, Wang, D Wykes, T 2011a. Predictors of clinical and social outcomes following involuntary hospital admission: a prospective observational study. *European Archives of Psychiatry and Clinical Neuroscience*, **261**, 377-386.
- Priebe, S, Mccabe, R, Bullenkamp, J, Hansson, L, Lauber, C, Martinez-Leal, R, Rossler, W, Salize, H, Svensson, B, Torres-Gonzales, F, Van Den Brink, R, Wiersma, D Wright, DJ 2007.
 Structured patient-clinician communication and 1-year outcome in community mental healthcare: cluster randomised controlled trial. *British Journal of Psychiatry*, **191**, 420-426.
- Priebe, S, Mccabe, R, Junghan, U, Kallert, T, Ruggeri, M, Slade, M Reininghaus, U 2011b. Association between symptoms and quality of life in patients with schizophrenia: a pooled analysis of changes over time. *Schizophrenia Research*, **133**, 17–21.
- Priebe, S, Omer, S, Giacco, D Slade, M 2014. Resource-oriented therapeutic models in psychiatry: conceptual review. *The British Journal of Psychiatry*, **204**, 256-261.
- Priebe, S, Reininghaus, U, Mccabe, R, Burns, T, Eklund, M, Hansson, L, Junghan, U, Kallert, T, Van Nieuwenhuizen, C Ruggeri, M 2010b. Factors influencing subjective quality of life in patients with schizophrenia and other mental disorders: a pooled analysis. *Schizophrenia Research*, **121**, 251-258.
- Priebe, S, Roeder-Wanner, U-U Kaiser, W 2000. Quality of life in first-admitted schizophrenia patients: a follow-up study. *Psychological Medicine*, **30**, 225–230.

- Priebe, S, Savill, M, Reininghaus, U, Wykes, T, Bentall, R, Lauber, C, Mccrone, P, Röhricht, F Eldridge, S 2013a. Effectiveness and cost-effectiveness of body psychotherapy in the treatment of negative symptoms of schizophrenia - A multi-centre randomised controlled trial. *BMC Psychiatry*, **13**, 26.
- Priebe, S, Yeeles, K, Bremner, S, Lauber, C, Eldridge, S, Ashby, D, David, AS, O'connell, N, Forrest, A Burns, T 2013b. Effectiveness of financial incentives to improve adherence to maintenance treatment with antipsychotics: cluster randomised controlled trial. *BMJ*, **347**, f5847.
- Prosser, ES, Csernansky, JG, Kaplan, J, Thiemann, S, Becker, TJ Hollister, LE 1987. Depression, parkinsonian symptoms, and negative symptoms in schizophrenics treated with neuroleptics. *The Journal of Nervous and Mental Disease*, **175**, 100-105.
- Purdon, SE, Jones, BD, Stip, E, Labelle, A, Addington, D, David, SR, Breier, A Tollefson, GD 2000. Neuropsychological change in early phase schizophrenia during 12 months of treatment with olanzapine, risperidone, or haloperidol. The Canadian Collaborative Group for research in schizophrenia. *Archives of General Psychiatry*, **57**, 249-258.
- Quinlan, DM, Schuldberg, D, Morgenstern, H Glazer, W 1995. Positive and negative symptom course in chronic community-based patients. A two-year prospective study. *British Journal of Psychiatry*, **166**, 634-641.
- Rabany, L, Weiser, M, Werbeloff, N Levkovitz, Y 2011. Assessment of negative symptoms and depression in schizophrenia: revision of the SANS and how it relates to the PANSS and CDSS. *Schizophrenia Research*, **126**, 226-230.
- Rabinowitz, J, Levine, Z, Garibaldi, G, Bugarski-Kirola, D, Berardo, CG Kapur, S 2012. Negative symptoms have greater impact on functioning than positive symptoms in schizophrenia: Analysis of CATIE data. *Schizophrenia Research*, **137**, 147-150.
- Rabinowitz, J, Werbeloff, N, Caers, I, Mandel, FS, Stauffer, V, Menard, F, Kinon, BJ Kapur, S
 2013. Negative symptoms in schizophrenia the remarkable impact of inclusion
 definitions in clinical trials and their consequences. *Schizophrenia Research*, 150, 334-338.
- Rado, S 1953. Dynamics and classification of disordered behavior. *American Journal of Psychiatry*, **110**, 406-416.
- Rado, S 1956. Psychoanalysis of behavior; collected papers, New York: Grune & Stratton.
- Ratakonda, S, Gorman, JM, Yale, SA Amador, XF 1998. Characterization of psychotic conditions: use of the domains of psychopathology model. *Archives of General Psychiatry*, **55**, 75-81.
- Ravanic, DB, Dejanovic, SMD, Janjic, V, Jovic, SD, Milovanovic, DR, Jakovljevic, V, Pantovic, V, Ravanic, B, Pantovic, M Pantovic, MMG 2009. Effectiveness of clozapine, haloperidol and chlorpromazine in schizophrenia during a five-year period. *Arquivos de Neuro-Psiquiatria*, **67**, 195-202.
- Read, J, Os, JV, Morrison, A Ross, CA 2005. Childhood trauma, psychosis and schizophrenia: a literature review with theoretical and clinical implications. *Acta Psychiatrica Scandinavica*, **112**, 330-350.
- Rector, NA 2004. Dysfunctional attitudes and symptom expression in schizophrenia: differential associations with paranoid delusions and negative symptoms. *Journal of Cognitive Psychotherapy*, **18**, 163-173.

- Reddy, LF, Horan, WP Green, MF 2015. Motivational Deficits and Negative Symptoms in Schizophrenia: Concepts and Assessments. *Current Topics in Behavioral Neurosciences*, 1–17.
- Reine, G, Lancon, C, Di Tucci, S, Sapin, C Auquier, P 2003. Depression and subjective quality of life in chronic phase schizophrenic patients. *Acta Psychiatrica Scandinavica*, **108**, 297-303.
- Reininghaus, U, Dutta, R, Dazzan, P, Doody, GA, Fearon, P, Lappin, J, Heslin, M, Onyejiaka, A,
 Donoghue, K Lomas, B 2015. Mortality in Schizophrenia and Other Psychoses: A 10 Year Follow-up of the ÆSOP First-Episode Cohort. Schizophrenia Bulletin, 41, 664-673.
- Rey, E, Bailer, J, Bräuer, W, Händel, M, Laubenstein, D Stein, A 1994. Stability trends and longitudinal correlations of negative and positive syndromes within a three-year follow-up of initially hospitalized schizophrenics. *Acta Psychiatrica Scandinavica*, **90**, 405-412.
- Richardson, M, Katsakou, C Priebe, S 2011a. Association of treatment satisfaction and psychopathological sub-syndromes among involuntary patients with psychotic disorders. *Social Psychiatry and Psychiatric Epidemiology*, **46**, 695-702.
- Richardson, M, Katsakou, C, Torres-González, F, Onchev, G, Kallert, T Priebe, S 2011b. Factorial validity and measurement equivalence of the Client Assessment of Treatment Scale for psychiatric inpatient care—A study in three European countries. *Psychiatry Research*, 188, 156-160.
- Richardson, P, Jones, K, Evans, C, Stevens, P Rowe, AG 2007. Exploratory RCT of art therapy as an adjunctive treatment in schizophrenia. *Journal of Mental Health*, **16**, 483-491.
- Riedel, M, Muller, N, Strassnig, M, Spellmann, I, Engel, RR, Musil, R, Dehning, S, Douhet, A, Schwarz, MJ Moller, HJG 2005. Quetiapine has equivalent efficacy and superior tolerability to risperidone in the treatment of schizophrenia with predominantly negative symptoms. *European Archives of Psychiatry and Clinical Neurosciences*, 255, 432-437.
- Rifkin, A 1987. Extrapyramidal side effects: A historical perspective. *Journal of Clinical Psychiatry*, **48**, 3-6.
- Ritsner, MS, Arbitman, M Lisker, A 2011. Anhedonia is an important factor of health-related quality-of-life deficit in schizophrenia and schizoaffective disorder. *The Journal of Nervous and Mental Disease*, **199**, 845-853.
- Robinson, D, Woerner, MG, Alvir, JMJ, Bilder, R, Goldman, R, Geisler, S, Koreen, A, Sheitman, B, Chakos, M Mayerhoff, D 1999. Predictors of relapse following response from a first episode of schizophrenia or schizoaffective disorder. *Archives of General Psychiatry*, 56, 241-247.
- Robinson, DG, Woerner, MG, Mcmeniman, M, Mendelowitz, A Bilder, RM 2014. Symptomatic and functional recovery from a first episode of schizophrenia or schizoaffective disorder. *American Journal of Psychiatry*, **161**, 473-479.
- Robinson, W 1950. Ecological correlations and the behavior of individuals. *American Sociological Review*, **15**, 351-357.
- Röder-Wanner, U-U Priebe, S 1998. Objective and subjective quality of life of first-admitted women and men with schizophrenia. *European Archives of Psychiatry and Clinical Neuroscience*, **248**, 250–258.
- Ross, DE, Thaker, GK, Buchanan, RW, Kirkpatrick, B, Lahti, AC, Medoff, D, Bartko, JJ, Goodman, J Tien, A 1997. Eye tracking disorder in schizophrenia is characterized by specific ocular

motor defects and is associated with the deficit syndrome. *Biological Psychiatry*, **42**, 781-796.

- Ross, DE, Thaker, GK, Buchanan, RW Lahti, AC 1996. Association of abnormal smooth pursuit eye movements with the deficit syndrome in schizophrenic patients. *The American Journal of Psychiatry*, **153**, 1158.
- Rubin, DB 1987. Multiple Imputation for Nonresponse in Surveys, John Wiley & Sons.
- Ruggeri, M, Koeter, M, Schene, A, Bonetto, C, Vàzquez-Barquero, JL, Becker, T, Knapp, M, Knudsen, HC, Tansella, M Thornicroft, G 2005. Factor solution of the BPRS-expanded version in schizophrenic outpatients living in five European countries. *Schizophrenia Research*, **75**, 107-117.
- Rule, A 2005. Ordered thoughts on thought disorder. The Psychiatrist, 29, 462-464.
- Rund, BR 1998. A review of longitudinal studies of cognitive function in schizophrenia patients. *Schizophrenia Bulletin*, **24**, 425.
- Rutherford, BR Roose, SP 2013. A model of placebo response in antidepressant clinical trials. *American Journal of Psychiatry*, **170**, 723-733.
- Ryan, RM Deci, EL 2000. Self-determination theory and the facilitation of intrinsic motivation, social development, and well-being. *American Psychologist*, **55**, 68-78
- Safferman, A, Lieberman, JA, Kane, JM, Szymanski, S Kinon, B 1991. Update on the clinical efficacy and side effects of clozapine. *Schizophrenia Bulletin*, **17**, 247-261.
- Saha, S, Chant, D, Welham, J Mcgrath, J 2005. A systematic review of the prevalence of schizophrenia. *PLoS Medicine*, **2**, e141.
- Saha, S, Chant, D, Welham, J Mcgrath, J 2006. The incidence and prevalence of schizophrenia varies with latitude. *Acta Psychiatrica Scandinavica*, **114**, 36-39.
- Salem, JE Kring, AM 1999. Flat affect and social skills in schizophrenia: evidence for their independence. *Psychiatry Research*, **87**, 159-167.
- Salokangas, RK 1997. Structure of schizophrenic symptomatology and its changes over time: prospective factor-analytical study. *Acta Psychiatrica Scandinavica*, **95**, 32-39.
- Sartorius, N, Shapiro, R Jablensky, A 1974. The international pilot study of schizophrenia. *Schizophrenia Bulletin*, **1**, 21-34.
- Sayers, SL, Curran, PJ Mueser, KT 1996. Factor structure and construct validity of the Scale for the Assessment of Negative Symptoms. *Psychological Assessment*, **8**, 269-280.
- Schneider, K 1959. *Clinical psychopathology.(Trans. by MW Hamilton)*, New York: Grune & Stratton.
- Schoemaker, JH, Jansen, WT, Schipper, J Szegedi, A 2014. The selective glycine uptake inhibitor org 25935 as an adjunctive treatment to atypical antipsychotics in predominant persistent negative symptoms of schizophrenia: Results from the GIANT trial. *Journal* of Clinical Psychopharmacology, **34**, 190-198.
- Schuldberg, D, Quinlan, DM, Morgenstern, H Glazer, W 1990. Positive and negative symptoms in chronic psychiatric outpatients: Reliability, stability, and factor structure.
 Psychological Assessment: A Journal of Consulting and Clinical Psychology, 2, 262-268.
- Schultz, W 2002. Getting formal with dopamine and reward. Neuron, 36, 241-263.
- Sechrest, L 1963. Incremental validity: A recommendation. *Educational and Psychological Measurement*, **12**, 153-158.

- Semiz, UB, Cetin, M, Basoglu, C, Ebrinc, S, Uzun, O, Herken, H, Balibey, H, Algul, A Ates, AG 2007. Clinical predictors of therapeutic response to clozapine in a sample of Turkish patients with treatment-resistant schizophrenia. *Progress in Neuro-Psychopharmacology & Biological Psychiatry*, **31**, 1330-1336.
- Sensky, T, Turkington, D, Kingdon, D, Scott, JL, Scott, J, Siddle, R, O'carroll, M Barnes, TRJ 2000. A randomized controlled trial of cognitive-behavioral therapy for persistent symptoms in schizophrenia resistant to medication. *Archives of General Psychiatry*, 57, 165-172.
- Serban, G, Siegel, S Gaffney, M 1992. Response of negative symptoms of schizophrenia to neuroleptic treatment. *Journal of Clinical Psychiatry*, **57**, 229-234.
- Sergi, MJ, Rassovsky, Y, Widmark, C, Reist, C, Erhart, S, Braff, DL, Marder, SR Green, MF 2007. Social cognition in schizophrenia: relationships with neurocognition and negative symptoms. Schizophrenia Research, 90, 316-324.
- Serretti, A, Macciardi, F Smeraldi, E 1996. Identification of symptomatologic patterns common to major psychoses: proposal for a phenotype definition. *American Journal of Medical Genetics*, **67**, 393-400.
- Sharpley, M, Hutchinson, G, Mckenzie, K Murray, R 2001. Understanding the excess of psychosis among the African-Caribbean population in England. Review of current hypotheses. *British Journal of Psychiatry*, **40**, s60-68.
- Shea, MT, Stout, R, Gunderson, J, Morey, LC, Grilo, CM, Mcglashan, T, Skodol, AE, Dolan-Sewell, R, Dyck, I Zanarini, MC 2002. Short-term diagnostic stability of schizotypal, borderline, avoidant, and obsessive-compulsive personality disorders. *American Journal of Psychiatry*, **159**, 2036-2041.
- Shtasel, DL, Gur, RE, Gallacher, F, Heimberg, C, Cannon, T Gur, RC 1992. Phenomenology and functioning in first-episode schizophrenia. *Schizophrenia Bulletin*, **18**, 449-462.
- Sim, K, Su, A, Leong, J, Yip, K, Chong, M, Fujii, S, Yang, S, Ungvari, G, Si, T Chung, E 2004. High dose antipsychotic use in schizophrenia: findings of the REAP (Research on East Asia psychotropic Prescriptions) study. *Pharmacopsychiatry*, **37**, 175-179.
- Simpson, G Angus, J 1970. A rating scale for extrapyramidal side effects. *Acta Psychiatrica Scandinavica*, **45**, 11-19.
- Singh, MM Kay, SR 1975. A comparative study of haloperidol and chlorpromazine in terms of clinical effects and therapeutic reversal with benztropine in schizophrenia. Theoretical implications for potency differences among neuroleptics. *Psychopharmacologia*, **43**, 103-113.
- Singh, SP, Singh, V, Kar, N Chan, K 2010. Efficacy of antidepressants in treating the negative symptoms of chronic schizophrenia: meta-analysis. *The British Journal of Psychiatry*, **197**, 174-179.
- Sipos, A, Rasmussen, F, Harrison, G, Tynelius, P, Lewis, G, Leon, DA Gunnell, D 2004. Paternal age and schizophrenia: a population based cohort study. *BMJ*, **329**, 1070.
- Siris, SG 1991. Diagnosis of Secondary Depression in Schizophrenia: Implications for DSM-IV. Schizophrenia Bulletin, **17**, 75-98.
- Slade, P 1976. Towards a Theory of Auditory Hallucinations: Outline of an Hypothetical Four-Factor Model. *British Journal of Social and Clinical Psychology*, **15**, 415-423.
- Smith, DA, Mar, CM Turoff, BK 1998. The structure of schizophrenic symptoms: a meta-analytic confirmatory factor analysis. *Schizophrenia Research*, **31**, 57-70.

- Snaith, P 1993. Anhedonia: a neglected symptom of psychopathology. *Psychological Medicine*, **23**, 957-966.
- Spratt, M, Carpenter, J, Sterne, JA, Carlin, JB, Heron, J, Henderson, J Tilling, K 2010. Strategies for multiple imputation in longitudinal studies. *American Journal of Epidemiology*, **172**, 478-487.
- Stata Corp, L 2012. Stata 12 statistical software. StataCorp LP.
- Statacorp, L 2009. Stata: release 11, StataCorp LP.
- Stauffer, VL, Song, G, Kinon, BJ, Ascher-Svanum, H, Chen, L, Feldman, PD Conley, RR 2012. Responses to antipsychotic therapy among patients with schizophrenia or schizoaffective disorder and either predominant or prominent negative symptoms. Schizophrenia Research, **134**, 195-201.
- Stolar, N, Berenbaum, H, Banich, MT Barch, D 1994. Neuropsychological correlates of alogia and affective flattening in schizophrenia. *Biological Psychiatry*, **35**, 164-172.
- Strauss, GP, Harrow, M, Grossman, LS Rosen, CG 2010. Periods of recovery in deficit syndrome schizophrenia: A 20-year multi-follow-up longitudinal study. *Schizophrenia Bulletin*, **36**, 788-799.
- Strauss, GP, Wilbur, RC, Warren, KR, August, SM Gold, JM 2011. Anticipatory vs. consummatory pleasure: what is the nature of hedonic deficits in schizophrenia? *Psychiatry Research*, **187**, 36-41.
- Strauss, JS Carpenter, WT 1977. Prediction of outcome in schizophrenia: III. Five-year outcome and its predictors. *Archives of General Psychiatry*, **34**, 159-163.
- Strauss, JS, Carpenter, WT Bartko, JJ 1974. Speculations on the processes that underlie schizophrenic symptoms and signs: III. *Schizophrenia Bulletin*, **1**, 61-69.
- Strauss, P, Keller, R, Buchanan, W, Gold, M, Fischer, A, Mcmahon, P, Catalano, T, Culbreth, J, Carpenter, T Kirkpatrick, B 2012. Next-generation negative symptom assessment for clinical trials: Validation of the Brief Negative Symptom Scale. *Schizophrenia Research*, 142, 88-92.
- Sumiyoshi, T, Park, S, Jayathilake, K, Roy, A, Ertugrul, A Meltzer, HY 2007. Effect of buspirone, a serotonin1A partial agonist, on cognitive function in schizophrenia: a randomized, double-blind, placebo-controlled study. *Schizophr Res*, **95**, 158-168.
- Susser, ES Lin, SP 1992. Schizophrenia after prenatal exposure to the Dutch Hunger Winter of 1944-1945. Archives of General Psychiatry, **49**, 983-988.
- Taiminen, TJ, Syvalahti, E, Saarijarvi, S, Niemi, H, Lehto, H, Ahola, V Salokangas, RKRG 1997.
 Citalopram as an adjuvant in schizophrenia: Further evidence for a serotonergic dimension in schizophrenia. *International Clinical Psychopharmacology*, **12**, 31-35.
- Tandon, R, Gaebel, W, Barch, DM, Bustillo, J, Gur, RE, Heckers, S, Malaspina, D, Owen, MJ, Schultz, S Tsuang, M 2013. Definition and description of schizophrenia in the DSM-5. *Schizophrenia Research*, **150**, 3-10.
- Tandon, R, Goldman, RS, Goodson, J Greden, JF 1990. Mutability and relationship between positive and negative symptoms during neuroleptic treatment in schizophrenia. *Biological Psychiatry*, 27, 1323-1326.
- Tandon, R, Ribeiro, SC, Dequardo, JR, Goldman, RS, Goodson, J Greden, JF 1993. Covariance of positive and negative symptoms during neuroleptic treatment in schizophrenia: a replication. *Biological Psychiatry*, **34**, 495-497.

- Tanenberg-Karant, M, Fennig, S, Ram, R, Krishna, J, Jandorf, L Bromet, EJ 1995. Bizarre delusions and first-rank symptoms in a first-admission sample: a preliminary analysis of prevalence and correlates. *Comprehensive Psychiatry*, **36**, 428-434.
- Tek, C, Kirkpatrick, B Buchanan, RWJ 2001. A five-year followup study of deficit and nondeficit schizophrenia. *Schizophrenia Research*, **49**, 253-260.
- Thompson, PA Meltzer, HY 1993. Positive, negative, and disorganisation factors from the Schedule for Affective Disorders and Schizophrenia and the Present State Examination. A three-factor solution. *The British Journal of Psychiatry*, **163**, 344-351.
- Tollefson, GD Sanger, TM 1997. Negative symptoms: A path analytic approach to a doubleblind, placebo- and haloperidol-controlled clinical trial with olanzapine. *American Journal of Psychiatry*, **154**, 466-474.
- Toomey, R, Kremen, WS, Simpson, JC, Samson, JA, Seidman, LJ, Lyons, MJ, Faraone, SV Tsuang, MT 1997. Revisiting the factor structure for positive and negative symptoms: evidence from a large heterogeneous group of psychiatric patients. *American Journal of Psychiatry*, **154**, 371-377.
- Tuominen, HJ, Tiihonen, J Wahlbeck, K 2005. Glutamatergic drugs for schizophrenia: a systematic review and meta-analysis. *Schizophrenia Research*, **72**, 225-234.
- Turetsky, BI, Colbath, EA Gur, RE 1998. P300 subcomponent abnormalities in schizophrenia: I. Physiological evidence for gender and subtype specific differences in regional pathology. *Biological Psychiatry*, **43**, 84-96.
- Turkington, D, Sensky, T, Scott, J, Barnes, TRE, Nur, U, Siddle, R, Hammond, K, Samarasekara, N Kingdon, DG 2008. A randomized controlled trial of cognitive-behavior therapy for persistent symptoms in schizophrenia: A five-year follow-up. *Schizophrenia Research*, 98, 1-7.
- Ucok, A, Serbest, S Kandemir, PEG 2011. Remission after first-episode schizophrenia: Results of a long-term follow-up. *Psychiatry Research*, **189**, 33-37.
- Van Der Does, A, Dingemans, P, Linszen, D, Nugter, M Scholte, W 1993. Symptom dimensions and cognitive and social functioning in recent-onset schizophrenia. *Psychological Medicine*, **23**, 745-753.
- Van Der Does, AJW, Dingemans, PMaJ, Linszen, DH, Nugter, MA Scholte, WF 1995. Dimensions and subtypes of recent-onset schizophrenia: A longitudinal analysis. *The Journal of Nervous and Mental Disease*, **183**, 681-687.
- Van Kammen, DP, Hommer, DW Malas, KLJ 1987. Effect of pimozide on positive and negative symptoms in schizophrenic patients: are negative symptoms state dependent? *Neuropsychobiology*, **18**, 113-117.
- Van Os, J, Fahy, T, Jones, P, Harvey, I, Sham, P, Lewis, S, Bebbington, P, Toone, B, Williams, M Murray, R 1996. Psychopathological syndromes in the functional psychoses: associations with course and outcome. *Psychological Medicine*, **26**, 161-176.
- Van Os, J, Gilvarry, C, Bale, R, Van Horn, E, Tattan, T White, I 1999. A comparison of the utility of dimensional and categorical representations of psychosis. *Psychological Medicine*, 29, 595-606.
- Van Praag, H 1976. About the impossible concept of schizophrenia. *Comprehensive Psychiatry*, **17**, 481-497.
- Van Putten, T, May, PR, Marder, SR Wittmann, LA 1981. Subjective response to antipsychotic drugs. *Archives of General Psychiatry*, **38**, 187-190.

- Van Putten, T May, R 1978. Subjective response as a predictor of outcome in pharmacotherapy: the consumer has a point. Archives of General Psychiatry, 35, 477-480.
- Vazquez-Barquero, JL, Lastra, I, Nunez, MC, Castanedo, SH Dunn, G 1996. Patterns of positive and negative symptoms in first episode schizophrenia. *The British Journal of Psychiatry*, **168**, 693-701.
- Velligan, D, Prihoda, T, Dennehy, E, Biggs, M, Shores-Wilson, K, Crismon, ML, Rush, AJ, Miller, A, Suppes, T Trivedi, M 2005. Brief psychiatric rating scale expanded version: How do new items affect factor structure? *Psychiatry Research*, **135**, 217-228.
- Ventura, J, Green, MF, Shaner, A Liberman, RP 1993. Training and quality assurance with the Brief Psychiatric Rating Scale:" the drift busters.". *International Journal of Methods in Psychiatric Research*, **3**, 221-244.
- Ventura, J, Nuechterlein, KH, Green, MF, Horan, WP, Subotnik, KL Mintz, J 2004. The timing of negative symptom exacerbations in relationship to positive symptom exacerbations in the early course of schizophrenia. *Schizophrenia Research*, **69**, 333-342.
- Ventura, J, Nuechterlein, KH, Subotnik, KL, Gutkind, D Gilbert, EA 2000. Symptom dimensions in recent-onset schizophrenia and mania: a principal components analysis of the 24item Brief Psychiatric Rating Scale. *Psychiatry Research*, **97**, 129-135.
- Voruganti, LP, Awad, AG, Parker, G, Forrest, C, Usmani, Y, Fernando, ML Senthilal, S 2007. Cognition, functioning and quality of life in schizophrenia treatment: results of a oneyear randomized controlled trial of olanzapine and quetiapine. *Schizophrenia Research*, **96**, 146-155.
- Walsh, BT, Seidman, SN, Sysko, R Gould, M 2002. Placebo response in studies of major depression: variable, substantial, and growing. *JAMA*, **287**, 1840-1847.
- Warner, R 2009. Recovery from schizophrenia and the recovery model. *Current Opinion in Psychiatry*, **22**, 374-380.
- Wessely, S, Buchanan, A, Reed, A, Cutting, J, Everitt, B, Garety, P Taylor, P 1993. Acting on delusions. I: Prevalence. *The British Journal of Psychiatry*, **163**, 69-76.
- West, LJ 1975. A clinical and theoretical overview of hallucinatory phenomena. In RK Siegal & LJ West: *Hallucinations: Behaviour Experience and Theory*, New York: Wiley
- Westen, D, Muderrisoglu, S, Fowler, C, Shedler, J Koren, D 1997. Affect regulation and affective experience: individual differences, group differences, and measurement using a Q-sort procedure. *Journal of Consulting and Clinical Psychology*, **65**, 429-439
- White, L, Harvey, PD, Opler, L Lindenmayer, J 1997. Empirical assessment of the factorial structure of clinical symptoms in schizophrenia. *Psychopathology*, **30**, 263-274.
- Whitty, P, Clarke, M, Mctigue, O, Browne, S, Kamali, M, Kinsella, A, Larkin, C O'callaghan, EG 2008. Predictors of outcome in first-episode schizophrenia over the first 4 years of illness. *Psychological Medicine*, **38**, 1141–1146.
- Who. 2013. ATC/DDD Index 2013 [Online]. Available: <u>http://www.whocc.no/atc_ddd_index/</u> [Accessed 06 January 2015 2014].
- Widlöcher, D Hardy-Bayle, M-C 1989. Cognition and control of action in psychopathology. Cahiers de Psychologie Cognitive/Current Psychology of Cognition, **9**, 583-615.
- Wiersma, D, Nienhuis, FJ, Slooff, CJ Giel, R 1998. Natural course of schizophrenic disorders: a 15-year followup of a Dutch incidence cohort. *Schizophrenia Bulletin*, **24**, 75-85.

- Wiersma, D, Wanderling, J, Dragomirecka, E, Ganev, K, Harrison, G, An Der Heiden, W, Nienhuis, F Walsh, D 2000. Social disability in schizophrenia: its development and prediction over 15 years in incidence cohorts in six European centres. *Psychological Medicine*, **30**, 1155-1167.
- Wing, JK Brown, GW 1970. Institutionalism and Schizophrenia: A Comparative Study of Three Mental Hospitals, 1960-1968, by JW King and GW Brown. With a Chapter by the Physician Superintendents of the Three Hospitals, University Press.
- Wise, RA 2002. Brain reward circuitry: insights from unsensed incentives. Neuron, 36, 229-240.
- Wohlberg, GW Kornetsky, C 1973. Sustained attention in remitted schizophrenics. *Archives of General Psychiatry*, **28**, 533-537.
- Wykes, T, Huddy, V, Cellard, C, Mcgurk, SR Czobor, P 2011. A meta-analysis of cognitive remediation for schizophrenia: methodology and effect sizes. *American Journal of Psychiatry*, **168**, 472-485.
- Xiang, Y, Weng, Y, Li, W, Gao, L, Chen, G, Xie, L, Chang, Y, Tang, WK Ungvari, GSJ 2006. Training patients with schizophrenia with the community re-entry module: a controlled study. Social Psychiatry & Psychiatric Epidemiology, 41, 464-469.
- Yates, BT Taub, J 2003. Assessing the costs, benefits, cost-effectiveness, and cost-benefit of psychological assessment: we should, we can, and here's how. *Psychological Assessment*, **15**, 478.
- York, GK Steinberg, DA 2011. Hughlings Jackson's neurological ideas. Brain, 134, 3106-3113.
- Yule, GU 1903. Notes on the theory of association of attributes in statistics. *Biometrika*, **2**, 121-134.
- Zammit, S, Allebeck, P, Andreasson, S, Lundberg, I Lewis, G 2002. Self reported cannabis use as a risk factor for schizophrenia in Swedish conscripts of 1969: historical cohort study. *BMJ*, **325**, 1199.
- Zammit, S, Allebeck, P, Dalman, C, Lundberg, I, Hemmingson, T, Owen, M Lewis, G 2003. Paternal age and risk for schizophrenia. *The British journal of psychiatry*, **183**, 405-408.
- Zhang, Z, Xu, X Ni, H 2013. Small studies may overestimate the effect sizes in critical care metaanalyses: a meta-epidemiological study. *Critical Care*, **17**, R2.
- Zoccali, R, Muscatello, MR, Bruno, A, Cambria, R, Mico, U, Spina, E Meduri, MG 2007. The effect of lamotrigine augmentation of clozapine in a sample of treatment-resistant schizophrenic patients: A double-blind, placebo-controlled study. *Schizophrenia Research*, **93**, 109-116.
- Zornberg, GL, Buka, SL Tsuang, MT 2000. Hypoxic-ischemia-related fetal/neonatal complications and risk of schizophrenia and other nonaffective psychoses: a 19-year longitudinal study. *American Journal of Psychiatry*, **157**, 196-202.

Appendix I. Meta-analysis examining the course of negative symptoms

Do negative symptoms of schizophrenia change over time? A meta-analysis of longitudinal data

M. Savill*, C. Banks, H. Khanom and S. Priebe

Unit for Social and Community Psychiatry, WHO collaborating Centre for Mental Health Service Development, Queen Mary University of London, London, UK

Background. Negative symptoms are a core component of schizophrenia which can severely impact quality of life and functional outcomes. These symptoms are understood to be highly stable but this has not been tested in a meta-analysis, despite the wealth of longitudinal data available.

Method. A systematic review of the literature was conducted, with eligible studies pooled into a random-effects metaanalysis. Planned meta-regressions were conducted to evaluate the impact of factors known to induce secondary negative symptoms, in addition to other possible sources of heterogeneity.

Results. The main analysis included 89 samples from 41 studies, totalling 5944 participants. Negative symptoms were found to significantly reduce in all treatment interventions, including in placebo and treatment as usual conditions, with a medium effect size (ES) present across all study conditions (ES = 0.66, 95% confidence interval 0.56–0.77, I^2 = 94.0%). In a multivariate meta-regression, only the type of scale used was found to significantly influence negative symptom change. No difference in outcome was found between studies that excluded patients with a high level of positive or depressive symptoms, compared to those that did not.

Conclusions. Negative symptoms were found to reduce in almost all schizophrenia outpatient samples. A reduction was found across all conditions, with effect sizes ranging from small to large depending upon the condition type. These findings challenge the convention that negative symptoms are highly stable and suggest that they may improve to a greater extent than what has previously been assumed.

Received 29 July 2014; Revised 20 October 2014; Accepted 22 October 2014

Key words: Longitudinal course, negative symptoms, schizophrenia, symptom assessment.

Introduction

Since Bleuler coined the term schizophrenia in the early 1900s negative symptoms have been recognized as a core feature of the disorder (Bleuler, 1950). The symptoms include alogia, asociality, blunted affect, anhedonia and amotivation (Blanchard *et al.* 2011), and have been found to severely impact both quality of life and social functioning (Norman *et al.* 2000).

Historically, negative symptoms were believed to increase over time as patients experience a progressive deterioration in functioning (Kraepelin, 1971). However, in observational studies which evaluated the progressive course of these symptoms the evidence initially suggested that these are largely stable over time (Pogue-Geile & Harrow, 1985; Fenton & McGlashan, 1991; Dollfus & Petit, 1995; Eaton *et al.*

(Email: m.savill@qmul.ac.uk)

1995). Later work recognized that the course was highly heterogenous, with some negative symptoms improving, often in tandem with improvement in positive symptoms (Addington & Addington, 1991). In an attempt to explain this heterogeneity Carpenter and colleagues proposed a distinction between those attributable to factors such as hospitalization, medication side-effects, depression, and elevated positive symptoms (known as secondary negative symptoms), from primary symptoms which were regarded as a core feature of the disorder itself (Carpenter *et al.* 1985). While secondary symptoms tend to improve relatively quickly once the causes are addressed, primary negative symptoms are thought to be largely persistent (Möller, 2007).

Broadly defined, primary negative symptoms refer to negative symptoms which are present both within and during periods of positive symptom exacerbation. However, distinguishing between primary and secondary negative symptoms can be a complex undertaking given the challenges in obtaining sufficient historical information and the level of clinical expertise required by the assessors. In light of this, Buchanan

^{*} Address for correspondence: Mr M. Savill, Unit for Social and Community Psychiatry, WHO collaborating Centre for Mental Health Service Development, Queen Mary University of London, Newham Centre for Mental Health, London E13 8SP, UK.

(2007) suggested the alternative, broader distinction of 'persistent negative symptoms', which include negative symptoms which remain present after usual treatments for secondary negative symptoms have failed. In the consensus statement for negative symptoms, it was proposed that distinguishing between primary and secondary negative symptoms was not essential for the purposes of testing therapeutics, as long as studies select participants with persistent symptoms and control for secondary sources of negative symptoms (Kirkpatrick *et al.* 2006).

To date, any advances in the treatment of schizophrenia have been found to provide only limited benefit to negative symptoms. In a meta-analysis which examined the efficacy of different secondgeneration antipsychotics, most were found not to provide a significant benefit over and above first-generation drugs, and in those that did the effect sizes were small (Leucht et al. 2009). Meta-analyses into the efficacy of adjunctive medications such as α_2 receptor antagonists (Hecht & Landy, 2012) and glutamatergic compounds (Tuominen et al. 2005) show some promise, while there is some evidence to suggest that adjunctive antidepressant medication may have some limited benefit (Singh et al. 2010). In a broader review evaluating the different pharmacological approaches in treating negative symptoms (Arango et al. 2013), new drugs that act on the NMDA and α_7 nicotinic receptors are highlighted as promising, but again more research is needed. In a series of meta-analyses on psychotherapeutic interventions, CBT was reported to have a small effect (Jauhar et al. 2014), social skills training a moderate effect (Kurtz & Mueser, 2008), while no effect was found for social cognitive training (Kurtz & Richardson, 2012). In a meta-analysis of cognitive remediation therapy which evaluated symptoms overall, a small effect was detected (Wykes et al. 2011). In the UK, NICE have previously recommended Arts therapies (NCCMH, 2010); however, this has since been challenged by the non-significant result of the MATISSE trial (Crawford et al. 2012). Overall, a lack of treatment efficacy has led to negative symptoms to be recognized as an unmet therapeutic need, and an important target for new interventions (Kirkpatrick et al. 2006).

Given the current focus on developing new interventions for negative symptoms, understanding their longitudinal course is important for future study design. Many of the earlier observational studies included inpatients, which is problematic given this population would typically receive far higher doses of antipsychotic medication and experience higher positive symptoms (Kasckow *et al.* 2001), and may reside in an under-stimulating environment (Oshima *et al.* 2003), which may induce negative symptoms secondary to the disorder itself. In addition, a number of the earlier studies included other illnesses such as schizoaffective disorder, which follows a different longitudinal course and can have poorer diagnostic stability, which again may influence symptom change over time (Malhi *et al.* 2008).

The objective of this study was to examine how negative symptoms change over time in schizophrenia outpatients, while exploring the impact of factors known to induce secondary negative symptoms. By pooling a wide variety of studies by way of meta-analysis, the aim was uncover broader trends in how these symptoms may change, as opposed to attempting to identify an estimate of effect size for a particular type of treatment. Following a systematic search, we conducted a meta-analysis of the within-group mean changes in negative symptoms. Only samples comprising exclusively of schizophrenia patients from the first assessment point were considered. Due to the expected heterogeneity between different interventions, separate effect size estimates were calculated for each treatment type. Finally, a series of planned meta-regressions were conducted to explore any impact of factors which may lead to secondary negative symptoms (Carpenter et al. 1985), and possible sources of methodological bias.

Method

Research in context

The systematic review was conducted following PRISMA statement guidelines (Liberati *et al.* 2009). An electronic search using the Medline, PsycINFO, EMBASE and CENTRAL databases was conducted dating back to 1962, which was when the Brief Psychiatric Rating Scale (BPRS) was first published (Overall & Gorham, 1962). The search was conducted on 26 April 2014 and contained three parameters. The first related to diagnosis, the second to negative symptoms, and the third an indicator that the study took place over at least two time points.

A hand-search of the American Journal of Psychiatry, Acta Scandinavica Psychiatrica, British Journal of Psychiatry, Schizophrenia Bulletin, JAMA Psychiatry, The Lancet, and Schizophrenia Research was conducted, either from 1962 or the date of first issue, and reference lists from all selected papers were hand-searched. During extraction, all assessments of negative symptoms, study inclusion/exclusion criteria, demographic details, industry sponsorship, and study methodology details were recorded. When necessary, corresponding authors were contacted for further information. In the case of missing standard deviations, a mean from the existing sample was imputed when possible. M.S. conducted the abstract screening, 20% of which was duplicated by C.B. with minimal discrepancies in selection detected. In the full paper screening phase M.S. conducted 100% of the screening, duplicated by H.K. and C.B. screening 50% of the sample each. All discrepancies were resolved without the need for S.P. to adjudicate as planned. At the full screening phase, all data were independently extracted onto a piloted extraction sheet.

Eligibility criteria

During the screening phase studies were excluded if they were clearly not relevant, did not have repeated assessments of negative symptoms at set time points, included no usable data on an exclusively schizophrenic sample, children or older adults, or were either under 10 weeks in length or over 3 years in length from the first follow-up assessment. Studies which included inpatients were considered, as long as the study included one time-point where the sample was exclusively outpatients, and then followed up from a standardized time-point from this assessment. Symptoms were required to be measured on a validated scale. Qualitative studies, case reports, letters to the editor, conference abstracts and book chapters were excluded. All articles were required to be published in a language which used Latin-based characters. Due to the analytical strategy adopted and the risk of small samples leading to biased estimates (Morris, 2000), studies with fewer than 50 participants were excluded.

Analysis plan

In the pooled analysis, the measure of effect size for each study was calculated using the standardized mean change (SMC) (Becker, 1988; Morris, 2000). The estimation of the variance was calculated using the large-sample approximation method recommended by Becker (1988), which can provide accurate estimates provided the sample sizes are adequately sized (Morris, 2000). The estimate of the correlation between the baseline and end of study scores was set at 0.633, based upon datasets held at our research group (Priebe *et al.* 2007) and a subsequent sensitivity analysis.

In deciding the appropriate effects model to adopt, the decision was complicated by the likelihood that multiple arms of the same study would be separately eligible for inclusion. One method of addressing this which was recognized in the *Cochrane Handbook* (Higgins & Green, 2011), is to conduct a two-level, fixed-effects meta-analysis across arms within studies, followed by a random-effects meta-analysis across studies, as a way to account for the mix in fixed and random effects that are likely to be present. However, this model adds considerable complexity to the analysis, while the handbook itself acknowledges that 'in practice the difference between different analyses is likely to be trivial' (section 16.5.5). This being the case, the method was not used, and the DerSimonion and Laird random-effects model was adopted (DerSimonian & Laird, 1986). All analysis was completed using Stata version 11 (StataCorp, 2009).

In cases where multiple scales were used to measure negative symptoms, the primary outcome measure was selected. When negative symptoms were measured over more than two time points, only the baseline and the study endpoint data were selected.

In the first stage of the analysis samples were grouped according to whether the intervention involved testing second-generation antipsychotics, first-generation antipsychotics, adjunctive medications, non-drug interventions, or placebo/treatment as usual (TAU) arms. In the next stage a series of planned univariate meta-regressions were conducted, with those found to approach significance (p < 0.10) entered into a multivariate model. First, we examined the impact of length of treatment in order to assess whether there was any trend over time. Next, we tested whether there was any difference between studies which incorporated a maximum threshold for positive and depressive symptoms, compared to those that did not, in order to assess whether the degree of change in negative symptoms varied dependent upon how studies dealt with factors which can cause secondary negative symptoms. We also tested the impact of blinding the assessors, a minimum negative symptom inclusion criterion, and whether the study received industry sponsorship.

In any examination of the change in a continuous variable over time which only includes two time points the issue of regression to the mean should be considered (Chiolero *et al.* 2013). This being the case, the mean negative symptoms at baseline were added to the final multivariate model to determine the degree of additional variance that may be explained by a greater reduction in negative symptoms being caused by a higher baseline symptom levels.

Results

A flow diagram depicting the search strategy for studies is included in Fig. 1. Of the 9480 articles screened, 49 articles were found and 41 were included in the final analysis (see Table 1). From these, a total of 89 separate samples were obtained. Of the 41 studies, five came from the USA; four each from Canada, Germany and the UK; three each from India, Spain and Turkey; two each from China, France and Italy;

4 M. Savill et al.



Fig. 1. Flow diagram outlining study selection procedure.

and one each from Brazil, Finland, Israel, Nepal, Poland and Serbia. Four studies were conducted in multiple countries, with sites in Northern America, Europe and Asia. Based on 51 samples, the median of study mean illness duration was 12.4 years (range 0.6–27.5 years). Twenty-three studies measured negative symptoms using the Positive and Negative Syndrome Scale (PANSS; Kay *et al.* 1987), 14 used the Scale to Assess Negative Symptoms (SANS; Andreasen, 1983), and four used the BPRS (Overall & Gorham, 1962). While studies which used alternative scales were screened, none met eligibility criteria. After pooling all 89 samples, a final total of 5944 participants were included in the meta-analysis.

As indicated in the forest plot (see Fig. 2), in all five intervention types a significant reduction in negative symptoms was found between the baseline and the follow-up assessment stage. Large effect sizes (ES) were detected in second-generation antipsychotics [ES = 1.09, 95% confidence interval (CI) 0.86–1.32, I^2 = 95.5%] and the adjunctive medication (ES = 0.97, 95% CI 0.68–1.26, I^2 = 91.7%) arms, while a small effect

size was noted in the placebo/TAU group (ES = 0.33, 95% CI 0.17–0.49, l^2 = 91.8%).

Next, a series of meta-regressions were conducted. In the univariate analyses the scale used, intervention type, study duration, and a minimum negative symptoms inclusion criterion were all associated with negative symptom change heterogeneity (see Table 2). A maximum level of positive symptoms and previous non-response to treatment as exclusion criteria were found to approach significance (p < 0.10), while other variables were non-significant. In the multivariate model, only the type of scale used and the type of intervention received remained significant. Studies which used the SANS found a significantly greater reduction in negative symptoms relative to those that used the PANSS (SANS: ES = 1.02, 95% CI 0.77-1.28; PANSS: ES = 0.66, 95% CI 0.56-0.77). Collectively, the scale used and the intervention type accounted for 43.65% of the variance. In a sensitivity analysis, the sample-level baseline negative symptoms were added to the model which was found to be a significant predictor (*B* = 0.01, s.e. = 0.00, 95% CI 0.00–0.02). However,

Authors	Year	Country	Study duration (weeks)	Outcome measure ^a	Individual symptoms reported	Intervention type ^b	п
Addington &	2000	Canada	130	PANSS	No	TAU	65
Aguglia <i>et al.</i> (2002)	2007	Italy	52	SANS	Yes	Non-drug intervention: psychoeducation	69
						TAU	66
Alptekin et al. (2005)	2005	Turkey	52	BPRS	No	TAU	382
Alvarez et al. (2006)	2006	Spain	48	SANS	Yes	SGA: olanzapine	120
Amell & Llandrich (2008)	2008	Spain	46	PANSS	Yes	SGA: risperidone Non-drug intervention: skills training group	115 35
						TAU	22
Bales et al. (2009)	2009	Nepal	18	PANSS	No	TAU	30
						TAU + betel nuts	30
Behere <i>et al.</i> (2011)	2011	India	16	PANSS	No	Non-drug intervention: yoga group	34
						Non-drug intervention: exercise group	31
\mathbf{P} : 1 (1 (2010)	0010	т 1.	10	CANC	NT		26
Bnowmick <i>et al.</i> (2010)	2010	India	12	SANS	INO	SGA: amisulpride	40
Big ℓ_{T} (2011)	2011	Brazil	26	PANSS	No		40 57
Bobes et al. (2009)	2011	Spain	34	BPRS	No	SCA: risperidone	362
Bodkin <i>et al.</i> (2005)	2005	USA	12	SANS	Ves	Adjunctive: selegiline	33
Doukin <i>et ut</i> . (2000)	2000	0011	12	0/11/0	105	Placebo	34
Crawford et al. (2012)	2012	UK	52	PANSS	No	TAU	137
C.u	_01_		0-			Non-drug intervention: activity group Non-drug intervention: art therapy group	140 140
Fleischhacker <i>et al.</i> (2003)	2003	Multi	52	PANSS	No	SGA: risperidone	120
						SGA: risperidone	228
						SGA: risperidone	267
Gaebel <i>et al.</i> (2007)	2007	Germany	52	PANSS	No	SGA: risperidone	77
						FGA: haloperidol	74
Gorna <i>et al.</i> (2008)	2008	Poland	52	PANSS	No	TAU	88
Hirsch <i>et al.</i> (2002)	2002	Germany	28	PANSS	No	SGA: ziprasidone	110
						FGA: haloperidol	117
Kane <i>et al.</i> (2011)	2011	USA	26	PANSS	No	TAU: remained on same drug	194
T((0010)	0010			DANGO		Placebo: switched to placebo	192
Kane <i>et al.</i> (2012)	2012	USA	24	PANSS	No	Adjunctive: armodafinil Adjunctive: armodafinil Adjunctive: armodafinil	70 69 71
						Placebo	70
Kaphzan <i>et al.</i> (2014)	2014	Israel	12	PANSS	No	Placebo	22
1 ()						Adjunctive: entacapone	23
Klingberg <i>et al.</i> (2011)	2011	Germany	52	PANSS	Yes	Non-drug intervention: CBT	99
000		,				Non-drug intervention: CRT	99
Lasser <i>et al.</i> (2013)	2013	USA	10	SANS	No	Adjunctive: lisdexamfetamine dimesylate	92
Lecrubier et al. (2006)	2006	France	26	SANS	No	Placebo	34
. ,						SGA: olanzapine SGA: olanzapine	70 70

Table 1. Characteristics of eligible studies

6 M. Savill et al.

Table 1 (cont.)

Authors	Year	Country	Study duration (weeks)	Outcome measure ^a	Individual symptoms reported	Intervention type ^b	n
						SGA: amisulpride	70
Liu et al. (2014)	2014	China	16	PANSS	No	Placebo	40
						Adjunctive: minocycline	39
Loebel <i>et al.</i> (2007)	2007	India	64	PANSS	No	SGA: ziprasidone	32
T (1007)	4005		24	0.4.3.10		SGA: ziprasidone	30
Loo et al. (1997)	1997	France	26	SANS	Yes	Placebo	72
M 1 (2010)	0010	LICA	52	DDDC	NT	SGA: amisulpride	69
Meitzer <i>et al.</i> (2010)	2010	USA	52	BPKS	INO	FGA: clozapine FGA: various first-generation drugs	40 45
Olie et al. (2006)	2006	Multi	12	PANSS	No	SGA: ziprasidone	59
	-000	man		111100	110	SGA: amisulpride	63
Pach <i>et al.</i> (1998)	1998	Germany	52	SANS	Yes	FGA: flupenthixol decanoate	63
Peet & Horrobin (2002)	2002	UK	12	PANSS	No	Placebo	31
						Adjunctive: eicosapentaenoic acid	32
						Adjunctive: eicosapentaenoic acid	32
						Adjunctive: eicosapentaenoic acid	27
Purdon <i>et al.</i> (2000)	2000	Canada	54	PANSS	No	SGA: olanzapine	21
						FGA: haloperidol	23
						SGA: risperidone	21
Ravanic et al. (2009)	2009	Serbia	52	PANSS	No	FGA: haloperidol	70
						FGA: haloperidol	35
						FGA: chlorpromazine	65
						FGA: chlorpromazine	40
						SGA: clozapine	65
						SGA: clozapine	50
Richardson <i>et al.</i> (2007)	2007	UK	38	SANS	No	TAU	46
						Non-drug intervention: art therapy group	43
Semiz et al. (2007)	2007	Turkey	12	SANS	No	SGA: clozapine	97
Schoemaker <i>et al.</i> (2014)	2014	Multi	12	SANS	Yes	Adjunctive: Org25935 low dose	71
						Adjunctive: Org25935 high dose	73
						Placebo	70
Sumiyoshi et al. (2007)	2007	USA	26	BPRS	No	Adjunctive: buspirone	30
						TAU	29
Taiminen et al. (1997)	1997	Finland	12	PANSS	No	TAU	39
						Adjunctive: citalopram	36
Turkington <i>et al.</i> (2008)	2008	UK	78	SANS	No	Non-drug intervention: CBT	46
						Non-drug intervention: befriending	44
Ucok <i>et al.</i> (2011)	2011	Turkey	52	SANS	No	TAU	52
						TAU	41
Voruganti et al. (2007)	2007	Canada	52	PANSS	No	SGA: olanzapine SGA: quetiapine	42 43

Table 1	(cont.)
---------	---------

Authors	Year	Country	Study duration (weeks)	Outcome measure ^a	Individual symptoms reported	Intervention type ^b	п
Xiang <i>et al.</i> (2006)	2006	China	34	PANSS	No	Non-drug intervention: community re-entry	48
						Non-drug intervention: counselling	48
Zoccali et al. (2007)	2007	Italy	24	SANS	Yes	Adjunctive: lamotrigine Placebo	26 25
List of studies not inclu	ided in	the main a	nalysis due to insu	fficient data			
Adams et al. (2013)	2013	Multi	24	NSA-16	No	SGA: multiple types SGA: LY2140023	130 131
Chouinard <i>et al.</i> (1975)	1975	Canada	12	BPRS	No	Placebo	24
						FGA: amitriptyline hydrochloride	24
						FGA: perphenazine	24
						FGA: amitriptyline perphenazine	24
Goff <i>et al.</i> (2005)	2005	USA	26	SANS	Yes	Adjunctive: D-cycloserine Placebo	26 25
Hayes <i>et al.</i> (1995)	1995	Australia	44	SANS	No	Non-drug intervention: skills training group	n/s
						Non-drug intervention: discussion group	n/s
Liberman et al. (1998)	1988	USA	156	BPRS	No	Non-drug intervention: occupational therapy	n/s
						Non-drug intervention: skills training group	n/s
Lieberman et al. (2013)	2013	Multi	12	SANS	No	Adjunctive: TC-5619	94 01
$M_{\rm em} = 1 (2002)$	2002	LIC A	104	CANC	N		91
Marder <i>et al.</i> (2003)	2003	USA	104	SAINS	res	training	33
						FGA: haloperidol + skills training	30
Pinto <i>et al.</i> (1979)	1979	UK	78	BPRS	No	FGA: flupenthixol decanoate	34
						FGA: fluphenazine decanoate	30

^a PANSS, Positive and Negative Syndrome Scale; BPRS, Brief Psychiatric Rating Scale; SANS, Scale to Assess Negative Symptoms; NSA-16, Negative Symptom Assessment – 16.

^b TAU, treatment as usual; SGA, second-generation antipsychotic; Adjunctive, adjunctive medication in addition to antipsychotic medication received; FGA, First-generation antipsychotic; CBT, cognitive behaviour therapy; CRT, cognitive remediation therapy.

the additional variance explained was relatively small (4.10%).

Although many of the studies evaluated extrapyramidal symptoms (EPS) as part of their analysis (53.7%), only three studies specified an EPS maximum threshold as an exclusion criterion (Klingberg *et al.* 2011; Lasser *et al.* 2013; Schoemaker *et al.* 2014). Given the lack of data, this was not included in the meta-regression analysis. Of those studies that did report EPS, they were generally considered to be in the low range at study intake, suggesting that the impact of EPS on negative symptoms was likely to be minimal.

Given the finding that second-generation antipsychotics and adjunctive medication arms resulted in much larger effect sizes than other treatment types, contrary to our expectations based on the existing literature (i.e. Leucht *et al.* 2009; Arango *et al.* 2013), the TAU and placebo control arms that they were compared to were explored in more depth. A substantially larger effect size was detected in TAU/placebo control arms which were part of the drugs trials, in



Fig. 2. Forest plot of the change in negative symptoms, by intervention type.

	Univariate a	nalysis	Multivariate analysis			
Predictor of negative symptom change	Coefficient	95% CI	р	Coefficient	95% CI	р
Study duration	-0.01 (0.00)	-0.01 to -0.00	0.035	-0.00 (0.00)	-0.01 to 0.00	0.388
Scale used (compared to PANSS)			0.002			
SANS	0.49 (0.14)	0.21 to 0.77		0.43 (0.13)	0.16 to 0.70	0.002
BPRS	-0.12 (0.26)	-0.63 to 0.40		-0.07 (0.23)	-0.54 to 0.39	0.760
Intervention type (compared to SGA)			< 0.001			
Non-drug intervention	-0.75 (0.18)	-1.11 to -0.38		-0.67 (0.18)	-1.04 to -0.30	0.001
TAU/placebo	-0.76 (0.16)	-1.07 to -0.45		-0.72 (0.16)	-1.04 to -0.40	< 0.001
Drug: FGA	-0.68 (0.18)	-1.10 to -0.25		-0.53 (0.20)	-0.93 to -0.13	0.010
Drug: augmentation	-0.12 (0.18)	-0.48 to 0.23		-0.17 (0.18)	-0.54 to 0.19	0.336
Min negative symptoms	0.37 (0.13)	0.10 to 0.63	0.007	0.11 (0.12)	-0.13 to 0.36	0.356
Max positive symptoms	0.26 (0.14)	-0.02 to 0.54	0.071	0.05 (0.13)	-0.21 to 0.32	0.685
Study supported by industry sponsorship	0.15 (0.14)	-0.14 to 0.43	0.309			
Exclusion: previous non-response	0.42 (0.18)	0.07 to 0.78	0.019	0.05 (0.17)	-0.29 to 0.40	0.752
Raters blinded to allocation ^a	-0.18 (0.15)	-0.47 to 0.12	0.234			
Exclusion: moderate levels of depression	0.13 (0.17)	-0.21 to 0.46	0.452			

Table 2. Univariate and multivariate meta-regressions examining the heterogeneity of negative symptom change

CI, Confidence interval; PANSS, Positive and Negative Syndrome Scale; SANS, Scale to Assess Negative Symptoms; BPRS, Brief Psychiatric Rating Scale; SGA, second-generation antipsychotic; TAU, treatment as usual; FGA, first-generation antipsychotic.

Values within parentheses are standard errors.

^a One study not included due to lack of data.

comparison to those that were not (ES = 0.67,95% CI 0.41-0.93, in comparison to ES = 0.15,95% CI 0.04-0.25). In a subsequent meta-regression this difference was found not to be attributable to either higher negative symptoms at baseline, or the type of assessment tool used, which were significant predictors in the full model.

Examination of individual negative symptoms

In 18 samples over nine studies the change in individual negative symptoms were also reported (see Table 1). Seven studies used the SANS as the rating tool, while two used the PANSS. Scores from different scales were combined using the method proposed by Lyne and colleagues (2012). A significant reduction was found in all four of the symptoms measured (affective blunting, alogia, avolition-apathy, anhedonia-asociality). Of the four, alogia appeared to reduce the least (ES = 0.64, 95% CI 0.45–0.83) and avolition-apathy the most (ES = 0.77, 95% CI 0.53–1.01); however, the difference between the items appeared minimal.

Eligible studies not pooled into the main analysis

Eight studies were found to be eligible, but could not be included in the main analysis (see Table 1). In line with the main results, 11 samples found some form of reduction in negative symptoms from baseline to end of study, five saw no change, and in two the change was not specified.

Discussion

Main results

The meta-analysis provided a clear result; negative symptoms of schizophrenia tend to improve significantly in an outpatient setting. A reduction in negative symptoms found across all intervention types, with the effect sizes ranging from small to large. A significant reduction was found in all four of the separate negative symptoms examined, covering both experiential and expressive features of the disorder. While substantial heterogeneity was present in the sample, a series of planned meta-regressions indicated that there was no difference in the reduction between studies which did and did not exclude participants with higher levels of positive or depressive symptoms. In addition, study-level methodological differences such as whether assessors were blinded, the symptom eligibility criteria, or whether the study received industry sponsorship did also not appear to influence the result.

Strengths and limitations

One of the main strengths of the study is that, despite the broad range of study interventions considered, the

findings are consistent. Of the 89 study arms included, only one found a clear significant increase in symptoms. In this case, the sample was part of a continuation study where patients who had previously responded well to their SGA medication were then switched to a placebo (Kane et al. 2011). In addition, when testing for the effect of regression to the mean, adding baseline negative symptoms to the multivariate model appeared to add relatively little additional explanatory power of the variance (4.1%), suggesting the findings are relatively robust. A further strength of this study is that despite the broad study inclusion criteria, removing samples which included inpatients at baseline, and other psychotic diagnoses, meant the participant inclusion criteria were relatively stringent in comparison to other observational studies that have looked at how negative symptoms change over time (i.e. Pogue-Geile & Harrow, 1985; Fenton & McGlashan, 1991; Dollfus & Petit, 1995; Eaton et al. 1995). When testing for heterogeneity, no difference in the effect size was detected between studies which excluded participants with elevated positive and depressive symptoms, which suggests the change unlikely to be attributable to a reduction in these factors which can induce secondary negative symptoms.

One limitation of the study is that because of the variance estimation method adopted a number of studies were excluded due being too small. However, given there is evidence to suggest that smaller studies can often present larger effect sizes (i.e. Zhang *et al.* 2013), our findings may have led to a more conservative estimate of the effect size. Another limitation is that, despite the number of studies included in the analysis (n = 41), the final sample of 5944 patients was smaller than what was anticipated. This was due to a number of the larger studies either containing inpatients (Lieberman *et al.* 2005), not using a validated negative symptoms scale (Dossenbach *et al.* 2004), or including patients with other psychotic disorders.

Another important issue to consider is that it is difficult to assess to what extent the reduction in severity is attributable to improvements in primary or secondary negative symptoms. However, difficulties making this distinction in research trials is not new (Buchanan, 2007), and the consensus statement suggests that such a distinction is not essential in trials as long as the symptoms are persistent and causes of secondary negative symptoms are adequately controlled for (Kirkpatrick et al. 2006). In this analysis, the eligible studies typically reported participants as being highly chronic in nature, reflected in the large median duration of illness (12.4 years), and many defined their sample as treatment-resistant, stable, non-acute, or in a maintenance period. Regarding whether secondary negative symptoms can be adequately controlled for

using study-level inclusion/exclusion criteria in a meta-regression of the heterogeneity present in a meta-analysis, this is also up for debate. However, despite these issues the high consistency of the directional change in negative symptoms in an outpatient sample, and the fact that there was no difference in this change between studies which did and did not control for factors which induce secondary negative symptoms does suggest that the improvement appears to occur to a greater extent to what was previously assumed. Further work examining the longitudinal course of negative symptoms in a study with clearly defined inclusion criteria relating to the persistence of negative symptoms, with appropriate controls for secondary negative symptoms, would provide stronger evidence for whether primary negative symptoms of the disorder are less stable than previously assumed.

Another limitation is that due to the substantial heterogeneity of the study designs, the fact that multiple arms of single studies were included which would naturally cluster together, and possible issues relating to the regression of the mean complicating the interpretation further, it was recognized that conducting an examination of publication bias important in typical meta-analytical studies (Higgins & Green, 2011) would have limited utility in this context. This being the case, such analysis was omitted so we cannot be certain as to whether publication bias influenced the results significantly. However, given a number of the studies were non-inferiority trials, dose-response studies, observational studies, and that control arms were used in this study as equivalent to experimental conditions, it would be unlikely that any publication bias would systematically inflate the overall effect sizes in the same manner as would typically be expected in a normal meta-analysis.

Finally, due to the lack of data, no-medication as a therapeutic option could not be evaluated, meaning it is not clear whether a reduction in negative symptoms would also occur in non-medicated patients. It is possible, however, that such an improvement could occur given there is some evidence to suggest that patients who do not immediately relapse upon termination of their antipsychotic regimen may experience improved global functioning over time (Harrow & Jobe, 2007).

Interpretation

These findings are contrary both to the earliest conceptions of schizophrenia, which suggested that negative symptoms follow a path of progressive deterioration (Bleuler, 1951; Kraepelin, 1971), and our current understanding of negative symptoms which suggest that they are highly stable in the non-acute phase (Möller, 2007). While acknowledging that the improvement in
negative symptoms were relatively small in the TAU, non-drug intervention arms and typical antipsychotic study arms, the improvement of negative symptoms over time appear to lend support to the recovery model of schizophrenia, particularly given the relationship of these symptoms to psychosocial functioning (Norman *et al.* 2000; Warner, 2009).

Given the limitations of the within-group design, the effect sizes presented cannot be used as an assessment on the effectiveness of any one treatment. As highlighted earlier, a series of meta-analyses have been conducted to evaluate treatments for schizophrenia using more appropriate designs (i.e. Kurtz & Mueser, 2008; Leucht et al. 2009; Jauhar et al. 2014). Overall, these reviews have detected relatively limited treatment benefits for negative symptoms, contrasting with the large within-group effect sizes noted here in the example of second-generation antipsychotic and adjunctive drug medication trials. Further investigation into the TAU and placebo study arms indicate that the effect sizes of drug study control arms are substantially larger than non-drug study controls, which suggest there is something inherent in the methodologies employed which makes these drug studies more likely to detect and report symptoms improvements. Many drug studies used placebo, as opposed to TAU, so a placebo effect may account for at least part of this difference. However, while it has been noted that the placebo effect is an increasing issue in schizophrenia drug trials (Kinon et al. 2011), given the effect size differences between drug and non-drug studies are so large (ES = 0.67, in comparison to ES = 0.15) it suggests that other factors inherent to the design and assessment may also be important. Regardless, the highly varied nature and outcomes study arms which fall under the heading of TAU and placebos merits further investigation.

Disentangling how regression to the mean issue relates to negative symptoms in an exclusively outpatient sample is complex issue worthy of further consideration. Higher mean levels of negative symptoms at baseline did predict a greater reduction. However, the additional proportion of the variance explained over and above the intervention type and assessment scale used was fairly small (4.1%), suggesting that the regression to the mean may not be as large as one might typically expect. This could be due to a number of factors. First, primary negative symptoms are thought to be highly stable (Möller et al. 2007) so it is perhaps unlikely that a substantial fluctuation around the mean level of symptoms over time would be expected, presuming secondary factors are appropriately considered. Second, by omitting samples which contained inpatients at baseline (but not necessarily at study end), the patients were a lot less likely to have been recruited during their most severe phase of their disorder, further minimizing the regression to the mean effect (Morton & Torgerson, 2003).

When testing for sources of heterogeneity in the course of negative symptoms, only the impact of assessment scale type remained significant, after controlling for intervention type. In comparison to studies which used the PANSS or the BPRS, a significantly greater change in negative symptoms was detected in studies that used the SANS. The finding that the SANS is a more sensitive instrument to detect change is in line with recommendations outlined in the MATRICS consensus statement (Kirkpatrick et al. 2006) and is perhaps unsurprising given the scales focus on negative symptoms, despite the conceptual and methodological issues that the scale is recognized to have (Blanchard et al. 2011). No eligible studies used either the Clinical Assessment Interview for Negative Symptoms (Horan et al. 2011) or the Brief Negative Symptom Scale (Strauss et al. 2012), therefore it is unknown how these new scales compare.

Understanding how negative symptoms change over time requires further attention given the duration between time points was not found to be a significant predictor. However, with the considerable variability in treatment duration, post-treatment follow-up duration, and post-treatment provision between studies, this is perhaps not surprising. In longitudinal studies assessing negative symptoms over very long periods of time, there has been little evidence of a linear improvement towards symptom remission (Strauss et al. 2010), while the rate of recovery in schizophrenia remains low (Jääskeläinen et al. 2013). Overall, this suggests that the trajectory of this improvement may be complex. One possible explanation of the improvement uncovered could be the non-specific effects of increased attention derived from being involved in research. Patients with prominent negative symptoms are typically very socially isolated, so increased contact time with researchers in itself may provide some therapeutic benefit.

Conclusions

Based on the available data of almost 6000 outpatients, negative symptoms of schizophrenia do not tend to be stable or deteriorate, but are instead likely to improve over time. This finding offers a further critique of the historical argument which suggests schizophrenia is a disorder of continual decline (Bleuler, 1950; Kraepelin, 1971) and instead provides further support to the recovery model of schizophrenia (Warner, 2009). Overall, these findings suggest that negative symptoms may not be as resistant to change as what has previously been assumed, and perhaps offer new hope to those who may experience such symptoms.

Acknowledgements

The authors thank Dr Stephen Bremer for statistical advice. This research received no specific grant from any funding agency, commercial or not-for-profit sectors.

Declaration of Interest

None.

References

Adams DH, Kinon BJ, Baygani S, Millen BA, Velona I, Kollack-Walker S, Walling DP (2013). A long- term, phase 2, multicenter, randomized, open-label, comparative safety study of pomaglumetad methionil (LY2140023 monohydrate) *versus* atypical antipsychotic standard of care in patients with schizophrenia. *BMC Psychiatry* **13**, 143.

Addington J, Addington DG (2000). Neurocognitive and social functioning in schizophrenia: a 2.5 year follow-up study. *Schizophrenia Research* **41**, 47–56.

Addington J, Addington DJ (1991). Positive and negative symptoms of schizophrenia. Their course and relationship over time. *Schizophrenia Research* 5, 51–59.

Aguglia E, De Vanna M, Onor ML, Ferrara DG (2002). Insight in persons with schizophrenia effects of switching from conventional neuroleptics to atypical antipsychotics. *Progress in Neuro-Psychopharmacology and Biological Psychiatry* **26**, 1229–1233.

Alptekin K, Erkoc S, Gogus AK, Kultur S, Mete L, Ucok A, Yazici KMG (2005). Disability in schizophrenia: clinical correlates and prediction over 1-year follow-up. *Psychiatry Research* **135**, 103–111.

Alvarez E, Ciudad A, Olivares JM, Bousono M, Gomez JCG (2006). A randomized, 1-year follow-up study of olanzapine and risperidone in the treatment of negative symptoms in outpatients with schizophrenia. *Journal of Clinical Psychopharmacology* **26**, 238–249.

Amell R, Llandrich JG (2008). Validity of a social skills training program for schizophrenic patients. *Actas Espanolas de Psiquiatria* 36, 123–132.

Andreasen NC (1983). Scale for the Assessment of Negative Symptoms. University of Iowa: Iowa City.

Arango C, Garibaldi G, Marder R (2013). Pharmacological approaches to treating negative symptoms: a review of clinical trials. *Schizophrenia Research* **150**, 346–352.

Bales A, Peterson MJ, Ojha S, Upadhaya K, Adhikari B, Barrett BG (2009). Associations between betel nut (Areca catechu) and symptoms of schizophrenia among patients in Nepal: a longitudinal study. *Psychiatry Research* **169**, 203–211.

Becker BJ (1988). Synthesizing standardized mean-change measures. *British Journal of Mathematical and Statistical Psychology* **41**, 257–278.

Behere RV, Arasappa R, Jagannathan A, Varambally S, Venkatasubramanian G, Thirthalli J, Subbakrishna DK, Nagendra HR, Gangadhar BNG (2011). Effect of yoga therapy on facial emotion recognition deficits, symptoms and functioning in patients with schizophrenia. *Acta Psychiatrica Scandinavica* **123**, 147–153. Bhowmick S, Hazra A, Ghosh MG (2010). Amisulpride *versus* olanzapine in the treatment of schizophrenia in Indian patients: randomized controlled trial. *Australian and New Zealand Journal of Psychiatry* **44**, 237–242.

Bio DS, Gattaz WFG (2011). Vocational rehabilitation improves cognition and negative symptoms in schizophrenia. *Schizophrenia Research* **126**, 265–269.

Blanchard JJ, Kring AM, Horan WP, Gur R (2011). Toward the next generation of negative symptom assessments: the collaboration to advance negative symptom assessment in schizophrenia. *Schizophrenia Bulletin* **37**, 291–299.

Bleuler E (1950). Dementia Praecox and the Group of Schizophrenias. New York: International Universities Press.

Bobes J, Ciudad A, Alvarez E, San L, Polavieja P, Gilaberte I (2009). Recovery from schizophrenia: results from a 1-year follow-up observational study of patients in symptomatic remission. *Schizophr Research* **115**, 58–66.

Bodkin J, Siris SG, Bermanzohn PC, Hennen J, Cole JOG (2005). Double-blind, placebo-controlled, multicenter trial of selegiline augmentation of antipsychotic medication to treat negative symptoms in outpatients with schizophrenia. *American Journal of Psychiatry* **162**, 388–390.

Buchanan RW (2007). Persistent negative symptoms in schizophrenia: an overview. *Schizophrenia Bulletin* **33**, 1013–1022.

Carpenter WT, Heinrichs DW, Alphs LD (1985). Treatment of negative symptoms. *Schizophrenia Bulletin* **11**, 440–452.

Chiolero A, Paradis G, Rich B, Hanley JA (2013). Assessing the relationship between the baseline value of a continuous variable and subsequent change over time. *Frontiers in Pulblic Health* **1** (http://www.ncbi.nlm.nih.gov/pmc/articles/ PMC3854983/). Accessed 11 November 2014.

Chouinard G, Annable L, Serrano M, Albert JM, Charette R (1975). Amitriptyline-perphenazine interaction in ambulatory schizophrenic patients. A controled study of drug interaction. *Archives of General Psychiatry* **32**, 1295–1307.

Crawford MJ, Killaspy H, Barnes TR, Barrett B, Byford S, Clayton K, Dinsmore J, Floyd S, Hoadley A, Johnson T, Kalaitzaki E, King M, Leurent B, Maratos A, O'neill FA, Osborn D, Patterson S, Soteriou T, Tyrer P, Waller D (2012). Group art therapy as an adjunctive treatment for people with schizophrenia: a randomised controlled trial (MATISSE). *British Medical Journal* 344 (http://www.bmj. com/content/344/bmj.e846).

Dersimonian R, Laird N (1986). Meta-analysis in clinical trials. *Controlled Clinical Trials* 7, 177–188.

Dollfus S, Petit MG (1995). Stability of positive and negative symptoms in schizophrenic patients: a 3-year follow-up study. *European Psychiatry* **10**, 228–236.

Dossenbach M, Erol A, Kessaci MEM, Shaheen MO, Sunbol MM, Boland J, Hodge A, O'Halloran RA, Bitter IG (2004). Effectiveness of antipsychotic treatments for schizophrenia: interim 6-month analysis from a prospective observational study (IC-SOHO) comparing olanzapine, quetiapine, risperidone, and haloperidol. *Journal of Clinical Psychiatry* 63, 312–321.

Eaton WW, Thara R, Federman B, Melton B, Liang KYJ (1995). Structure and course of positive and negative

symptoms in schizophrenia. *Archives of General Psychiatry* **52**, 127–134.

Fenton WS, McGlashan THJ (1991). Natural history of schizophrenia subtypes. II. Positive and negative symptoms and long-term course. *Archives of General Psychiatry* **48**, 978–986.

Fleischhacker WW, Eerdekens M, Karcher K, Remington G, Llorca PM, Chrzanowski W, Martin S, Gefvert OJ (2003). Treatment of schizophrenia with long-acting injectable risperidone: a 12-month open-label trial of the first long-acting second-generation antipsychotic. *Journal of Clinical Psychiatry* 64, 1250–1257.

Gaebel W, Riesbeck M, Wolwer W, Klimke A, Eickhoff M, Von Wilmsdorff M, Jockers-Scherubl MC, Kuhn KU, Lemke M, Bechdolf A, Bender S, Degner D, Schlosser R, Schmidt LG, Schmitt A, Jager M, Buchkremer G, Falkai P, Klingberg S, Kopcke W, Maier W, Hafner H, Ohmann C, Salize HJ, Schneider F, Moller HJG (2007). Maintenance treatment with risperidone or low-dose haloperidol in first-episode schizophrenia: 1-year results of a randomized controlled trial within the German research network on schizophrenia. *Journal of Clinical Psychiatry* **68**, 1763–1774.

Goff DC, Herz L, Posever T, Shih V, Tsai G, Henderson DC, Freudenreich O, Evins A, Yovel I, Zhang H, Schoenfeld DG (2005). A six-month, placebo-controlled trial of D-cycloserine co-administered with conventional antipsychotics in schizophrenia patients. *Psychopharmacology* **179**, 144–150.

Gorna K, Jaracz K, Rybakowski F, Rybakowski J (2008). Determinants of objective and subjective quality of life in first-time-admission schizophrenic patients in Poland: a longitudinal study. *Quality of Life Research* **17**, 237–247.

Harrow M, Jobe TH (2007). Factors involved in outcome and recovery in schizophrenia patients not on antipsychotic medications: a 15-year multi follow-up study. *Journal of Nervous and Mental Disease* 195, 406–414.

Hayes RL, Halford WK, Varghese FTJ (1995). Social skills training with chronic schizophrenic patients: effects on negative symptoms and community functioning. *Behavior Therapy* **26**, 433–449.

Hecht EM, Landy C (2012). Alpha-2 receptor antagonist add-on therapy in the treatment of schizophrenia; a meta-analysis. *Schizophrenia Research* **134**, 202–206.

Higgins JPT, Green S (2011). Cochrane Handbook for Systematic Reviews of Interventions, version 5.1.0. The Cochrane Collaboration (www.cochrane-handbook.org). Accessed 11 November 2014.

Hirsch SR, Kissling WB, Uml J, Power A, O'connor RJ (2002). A 28-week comparison of ziprasidone and haloperidol in outpatients with stable schizophrenia. *Journal of Clinical Psychiatry* **63**, 516–523.

Horan WP, Kring AM, Gur RE, Reise SP, Blanchard JJ (2011). Development and psychometric validation of the Clinical Assessment Interview for Negative Symptoms (CAINS). *Schizophrenia Research* **132**, 140–145.

Jääskeläinen E, Juola P, Hirvonen N, McGrath JJ, Saha S, Isohanni M, Veijola J, Miettunen J (2013). A systematic review and meta-analysis of recovery in schizophrenia. *Schizophrenia Bulletin* **39**, 1296–1306. Jauhar S, Mckenna PJ, Radua J, Fung E, Salvador R, Laws KR (2014). Cognitive-behavioural therapy for the symptoms of schizophrenia: systematic review and meta-analysis with examination of potential bias. *British Journal of Psychiatry* **204**, 20–29.

Kane JM, Mackle M, Snow-Adami L, Zhao J, Szegedi A, Panagides JJ (2011). A randomized placebo-controlled trial of asenapine for the prevention of relapse of schizophrenia after long-term treatment. *Journal of Clinical Psychiatry* 72, 349–355.

Kane JM, Yang R, Youakim JM (2012). Adjunctive armodafinil for negative symptoms in adults with schizophrenia: a double-blind, placebo-controlled study. *Schizophrenia Research* **135**, 116–122.

Kaphzan H, Ben-Shachar D, Klein E (2014). Entacapone augmentation of antipsychotic treatment in schizophrenic patients with negative symptoms; a double-blind placebo-controlled study. *International Journal of Neuropsychopharmacology* **17**, 337–340.

 Kasckow JW, Twamley E, Mulchahey JJ, Carroll B, Sabai M, Strakowski SM, Patterson T, Jeste DV (2001).
Health-related quality of well-being in chronically hospitalized patients with schizophrenia: comparison with matched outpatients. *Psychiatry Research* 103, 69–78.

Kay SR, Flszbein A, Opfer LA (1987). The positive and negative syndrome scale (PANSS) for schizophrenia. *Schizophrenia Bulletin* 13, 261.

Kinon BJ, Potts AJ, Watson SB (2011). Placebo response in clinical trials with schizophrenia patients. *Current Opinion* in Psychiatry 24, 107–113.

Kirkpatrick B, Fenton WS, Carpenter WT, Marder SR (2006). The NIMH-MATRICS consensus statement on negative symptoms. *Schizophrenia Bulletin* 32, 214–219.

Klingberg S, Wolwer W, Engel C, Wittorf A, Herrlich J, Meisner C, Buchkremer G, Wiedemann GJ (2011). Negative symptoms of schizophrenia as primary target of cognitive behavioral therapy: results of the randomized clinical TONES study. *Schizophrenia Bulletin* **37**, 98–110.

Kraepelin E (1971). *Dementia Praecox and Paraphrenia* (1919). Translated by Barclay RM; edited by Robertson GM. Robert E Krieger: New York.

Kurtz MM, Mueser KT (2008). A meta-analysis of controlled research on social skills training for schizophrenia. *Journal of Consulting and Clinical Psychology* **76**, 491–504.

Kurtz MM, Richardson CL (2012). Social cognitive training for schizophrenia: a meta-analytic investigation of controlled research. *Schizophrenia Bulletin* 38, 1092–1104.

Lasser A, Dirks B, Nasrallah H, Kirsch C, Gao J, Pucci L, Knesevich A, Lindenmayer JP (2013). Adjunctive lisdexamfetamine dimesylate therapy in adult outpatients with predominant negative symptoms of schizophrenia: open-label and randomized-withdrawal phases. *Neuropsychopharmacology* **38**, 2140–2149.

Lecrubier Y, Quintin P, Bouhassira M, Perrin E, Lancrenon SG (2006). The treatment of negative symptoms and deficit states of chronic schizophrenia: olanzapine compared to amisulpride and placebo in a 6-month double-blind controlled clinical trial. *Acta Psychiatrica Scandinavica* **114**, 319–327.

Leucht S, Arbter D, Engel RR, Kissling W, Davis JM (2009). How effective are second-generation antipsychotic drugs? A meta-analysis of placebo-controlled trials. *Molecular Psychiatry* **14**, 429–447.

Liberati A, Altman DG, Tetzlaff J, Mulrow C, Gotzsche PC, Ioannidis JP, Clarke M, Devereaux PJ, Kleijnen J, Moher D (2009). The PRISMA statement for reporting systematic reviews and meta-analyses of studies that evaluate health care interventions: explanation and elaboration. *Journal of Clinical Epidemiology* **62**, e1–e34.

Liberman RP, Wallace CJ, Blackwell G, Kopelowicz A, Vaccaro JV, Mintz J (1998). Skills training versus psychosocial occupational therapy for persons with persistent schizophrenia. *American Journal of Psychiatry* **155**, 1087–1091.

Lieberman A, Dunbar G, Segreti C, Girgis R, Seoane F, Beaver S, Duan N, Hosford A (2013). A randomized exploratory trial of an alpha-7 nicotinic receptor agonist (TC-5619) for cognitive enhancement in schizophrenia. *Neuropsychopharmacology* **38**, 968–975.

Lieberman JA, Stroup TS, McEvoy JP, Swartz MS, Rosenheck RA, Perkins DO, Keefe RS, Davis SM, Davis CE, Lebowitz BD, Severe J, Hsiao JK (2005) Effectiveness of antipsychotic drugs in patients with chronic schizophrenia. *New England Journal of Medicine* **353**, 1209–1223.

Liu F, Guo X, Wu R, Ou J, Zheng Y, Zhang B, Xie L, Zhang L, Yang L, Yang S, Yang J, Ruan Y, Zeng Y, Xu X, Zhao J (2014). Minocycline supplementation for treatment of negative symptoms in early-phase schizophrenia: a double blind, randomized, controlled trial. *Schizophrenia Research* 153, 169–176.

Loebel AD, Khanna S, Rajadhyaksha S, Siu CO, Giller E, Potkin SGJ (2007). Ziprasidone in treatment-resistant schizophrenia: a 52-week, open-label continuation study. *Journal of Clinical Psychiatry* 68, 1333–1338.

Loo H, Poirier-Littre MF, Theron M, Rein W, Fleurot OG (1997). Amisulpride *versus* placebo in the medium-term treatment of the negative symptoms of schizophrenia. *British Journal of Psychiatry* **170**, 18–22.

Lyne JP, Kinsella A, O'Donoghue B (2012). Can we combine symptom scales for collaborative research projects? *Journal* of Psychiatric Research 46, 233–238.

Malhi GS, Green M, Fagiolini A, Peselow ED, Kumari V (2008). Schizoaffective disorder: diagnostic issues and future recommendations. *Bipolar Disorders* **10**, 215–230.

Marder SR, Glynn SM, Wirshing WC, Wirshing DA, Ross D, Widmark C, Mintz J, Liberman RP, Blair KEJ (2003). Maintenance treatment of schizophrenia with risperidone or haloperidol: 2-year outcomes. *American Journal of Psychiatry* 160, 1405–1412.

Meltzer HY, Bobo WV, Lee MA, Cola P, Jayathilake KG (2010). A randomized trial comparing clozapine and typical neuroleptic drugs in non-treatment-resistant schizophrenia. *Psychiatry Research* **177**, 286–293.

Möller HJ (2007). Clinical evaluation of negative symptoms in schizophrenia. *European Psychiatry* 22, 380–386.

Morris SB (2000). Distribution of the standardized mean change effect size for meta-analysis on repeated measures.

British Journal of Mathamatical and Statistical Psychology 53, 17–29.

Morton V, Torgerson DJ (2003). Effect of regression to the mean on decision making in health care. *British Medical Journal* **326**, 1083.

NCCMH (2010). Schizophrenia: The NICE Guideline on Core Interventions in the Treatment and Management of Schizophrenia in Adults in Primary and Secondary Care, updated edition. The British Psychological Society and the Royal College of Psychiatrists: London.

Norman RM, Malla AK, Mclean T, Voruganti LP, Cortese L, Mcintosh E, Cheng S, Rickwood A (2000). The relationship of symptoms and level of functioning in schizophrenia to general wellbeing and the Quality of Life Scale. *Acta Psychiatrica Scandinavica* **102**, 303–309.

Olie JP, Spina E, Murray S, Yang RG (2006). Ziprasidone and amisulpride effectively treat negative symptoms of schizophrenia: results of a 12-week, double-blind study. *International Clinical Psychopharmacology* **21**, 143–151.

Oshima I, Yoshio M, Yoshimasa I (2003). Institutionalisation and schizophrenia in Japan: social environments and negative symptoms Nationwide survey of in-patients. *British Journal of Psychiatry* **183**, 50–56.

Overall JE, Gorham DR (1962). The brief psychiatric rating scale. *Psychological Reports* **10**, 799–812.

Pach J, Finkbeiner T, Glaser T, Haug J, Osterheider M, Tegeler JJ (1998). Positive and negative symptoms in chronic schizophrenic patients under maintenance therapy with flupenthixol decanoate for a twelve month perioid. *Fortschritte der Neurologie Psychiatrie* 66, 442–449.

Peet M, Horrobin DFG (2002). A dose-ranging exploratory study of the effects of ethyl-eicosapentaenoate in patients with persistent schizophrenic symptoms. *Journal of Psychiatric Research* **36**, 7–18.

Pinto R, Bannerjee A, Ghosh N (1979). A double-blind comparison of flupenthixol decanoate and fluphenazine decanoate in the treatment of chronic schizophrenia. *Acta Psychiatrica Scandinavica* **60**, 313–322.

Pogue-Geile MF, Harrow MG (1985). Negative symptoms in Schizophrenia: their longitudinal course and prognostic importance. *Schizophrenia Bulletin* **11**, 427–439.

Priebe S, Mccabe R, Bullenkamp J, Hansson L, Lauber C, Martinez-Leal R, Rossler W, Salize H, Svensson B, Torres-Gonzales F, Van Den Brink R, Wiersma D, Wright DJ (2007). Structured patient-clinician communication and 1-year outcome in community mental healthcare: cluster randomised controlled trial. *British Journal of Psychiatry* 191, 420–426.

Purdon SE, Jones BD, Stip E, Labelle A, Addington D, David SR, Breier A, Tollefson GD (2000). Neuropsychological change in early phase schizophrenia during 12 months of treatment with olanzapine, risperidone, or haloperidol. The Canadian Collaborative Group for research in schizophrenia. Archives of General Psychiatry 57, 249–258.

Ravanic DB, Dejanovic SMD, Janjic V, Jovic SD, Milovanovic DR, Jakovljevic V, Pantovic V, Ravanic B, Pantovic M, Pantovic MMG (2009). Effectiveness of clozapine, haloperidol and chlorpromazine in schizophrenia during a five-year period. *Arquivos de Neuro-Psiquiatria* **67**, 195–202.

Richardson P, Jones K, Evans C, Stevens P, Rowe AG (2007). Exploratory RCT of art therapy as an adjunctive treatment in schizophrenia. *Journal of Mental Health* **16**, 483–491.

Schoemaker JH, Jansen WT, Schipper J, Szegedi A (2014). The selective glycine uptake inhibitor org 25935 as an adjunctive treatment to atypical antipsychotics in predominant persistent negative symptoms of schizophrenia: results from the GIANT trial. *Journal of Clinical Psychopharmacology* **34**, 190–198.

Semiz UB, Cetin M, Basoglu C, Ebrinc S, Uzun O, Herken H, Balibey H, Algul A, Ates AG (2007). Clinical predictors of therapeutic response to clozapine in a sample of Turkish patients with treatment-resistant schizophrenia. *Progress in Neuro-Psychopharmacology and Biological Psychiatry* 31, 1330–1336.

Singh SP, Singh V, Kar N, Chan K (2010). Efficacy of antidepressants in treating the negative symptoms of chronic schizophrenia: meta-analysis. *British Journal of Psychiatry* 197, 174–179.

Statacorp L (2009). Stata: release 11, StataCorp LP.

Strauss GP, Harrow M, Grossman LS, Rosen CG (2010). Periods of recovery in deficit syndrome schizophrenia: a 20-year multi-follow-up longitudinal study. *Schizophrenia Bulletin* 36, 788–799.

Strauss GP, Keller WR, Buchanan RW, Gold JM, Fischer BA, Mcmahon RP, Catalano LT, Culbreth AJ, Carpenter WT, Kirkpatrick B (2012). Next-generation negative symptom assessment for clinical trials: validation of the brief negative symptom scale. *Schizophrenia Research* 142, 88–92.

Sumiyoshi T, Park S, Jayathilake K, Roy A, Ertugrul A, Meltzer HY (2007). Effect of buspirone, a serotonin1A partial agonist, on cognitive function in schizophrenia: a randomized, double-blind, placebo-controlled study. *Schizophrenia Research* 95, 158–168.

Taiminen TJ, Syvalahti E, Saarijarvi S, Niemi H, Lehto H, Ahola V, Salokangas RKRG (1997). Citalopram as an adjuvant in schizophrenia: further evidence for a serotonergic dimension in schizophrenia. *International Clinical Psychopharmacology* **12**, 31–35.

Tuominen HJ, Tiihonen J, Wahlbeck K (2005). Glutamatergic drugs for schizophrenia: a systematic review and meta-analysis. *Schizophrenia Research* 72, 225–234.

Turkington D, Sensky T, Scott J, Barnes TRE, Nur U, Siddle R, Hammond K, Samarasekara N, Kingdon DG (2008). A randomized controlled trial of cognitive-behavior therapy for persistent symptoms in schizophrenia: a five-year follow-up. *Schizophrenia Research* **98**, 1–7.

Ucok A, Serbest S, Kandemir PEG (2011). Remission after first-episode schizophrenia: results of a long-term follow-up. *Psychiatry Research* **189**, 33–37.

Voruganti LP, Awad AG, Parker G, Forrest C, Usmani Y, Fernando ML, Senthilal S (2007). Cognition, functioning and quality of life in schizophrenia treatment: results of a one-year randomized controlled trial of olanzapine and quetiapine. *Schizophrenia Research* **96**, 146–155.

Wykes T, Huddy V, Cellard C, Mcgurk SR, Czobor P (2011). A meta-analysis of cognitive remediation for schizophrenia: methodology and effect sizes. *American Journal of Psychiatry* 168, 472–485.

Xiang Y, Weng Y, Li W, Gao L, Chen G, Xie L, Chang Y, Tang WK, Ungvari GSJ (2006). Training patients with schizophrenia with the community re-entry module: a controlled study. *Social Psychiatry and Psychiatric Epidemiology* **41**, 464–469.

Zhang Z, Xu X, Ni H (2013). Small studies may overestimate the effect sizes in critical care meta-analyses: a metaepidemiological study. *Critical Care* 17, R2 (http://ccforum. com/content/17/1/R2). Accessed 11 November 2014.

Zoccali R, Muscatello MR, Bruno A, Cambria R, Mico U, Spina E, Meduri MG (2007). The effect of lamotrigine augmentation of clozapine in a sample of treatmentresistant schizophrenic patients: a double-blind, placebo-controlled study. *Schizophrenia Research* **93**, 109–116.

Appendix II. Symptom levels and initial appraisal of inpatient treatment

ARTICLE IN PRESS

Psychiatry Research I (IIII) III-III

Contents lists available at SciVerse ScienceDirect





Psychiatry Research

Symptom levels and initial appraisal of hospital treatment in patients with schizophrenia

Mark Savill^a, Jelena Jankovic^b, Christina Katsakou^a, Thomas Kallert^c, Stefan Priebe^{a,*}

^a Unit for Social and Community Psychiatry, Barts and the London School of Medicine and Dentistry, Queen Mary University of London, London, United Kingdom ^b Dudley and Walsall Mental Health Partnership Trust, Trafalgar House, 47 - 49 King Street, Dudley, West Midlands DY2 8PS, United Kingdom

^c Department of Psychiatry, Psychosomatics and Psychotherapy, Park Hospital Leipzig, Leipzig, Germany

ARTICLE INFO

Article history: Received 19 December 2011 Received in revised form 16 March 2012 Accepted 12 April 2012

Keywords: Initial assessment of treatment Inpatient environment Symptoms of schizophrenia Legal status of admission

ABSTRACT

The initial appraisal of treatment by inpatients with schizophrenia has been found to be a significant predictor of clinical outcomes. The study aim was to examine whether specific types of symptoms are associated with the initial appraisal of treatment after controlling other patient characteristics. Data of 2105 inpatients with schizophrenia (ICD-10 F20-9) were pooled from three national and international multi-centre studies. Patients were interviewed within the first week of their inpatient admission. Higher levels of manic and positive symptoms were significantly associated with a less favourable initial appraisal of treatment, whilst no association was found with depression/anxiety and negative symptoms. Detained patients had more negative initial treatment appraisals, and the association with manic symptoms was significantly stronger in detained patients compared to those admitted voluntarily. Whilst patient reported outcomes in psychiatry are usually associated with mood symptoms, this appears not to be the case for the initial appraisal by inpatients with schizophrenia. The association with manic and positive symptoms may be explained by the influence of such symptoms on the hospital experience. Focusing on the initial management of mania and positive symptoms might improve patients' appraisal of treatment in the inpatient environment.

© 2012 Elsevier Ireland Ltd. All rights reserved.

1. Introduction

Evidence indicates that patients with schizophrenia who have a more positive initial appraisal of their treatment are likely to benefit more from that treatment eventually. This applies to pharmacological treatment (Van Putten and May, 1978; Van Putten et al., 1978; Bartkó et al., 1987; Awad et al., 1995) and complex interventions (Priebe and Gruyters, 1994, 1995a; Bröker et al., 1995; Priebe et al., 2009, 2010c). In inpatients, a more positive initial appraisal of treatment was associated with lower symptom levels after one month (Richardson et al., 2010), at discharge (Bröker et al., 1995), lower social disability at 3 and 12 months (Priebe et al., 2010a), and lower subsequent involuntary readmission rates (Priebe et al., 2009). Developing a greater understanding of the factors which influence how patients initially appraise their hospital treatment would help identify those at risk of being less satisfied with their treatment, and could assist in the development of new interventions to maximise patients' initial appraisal of treatment.

E-mail address: s.priebe@gmul.ac.uk (S. Priebe).

Whilst previous studies have controlled the predictive association of initial treatment assessments with later clinical outcomes for socio-demographic and clinical characteristics, at present it is unclear whether specific types of symptoms are associated with a more or less favourable initial assessment of treatment. The importance of analysing the influence of symptom types, rather than global levels of psychopathology, has been emphasised in the literature (Shafer, 2005; Richardson et al., 2010). Previous findings suggest other patient reported outcomes, including patients' overall appraisal of psychiatric treatment, are negatively associated with mood, and that this could be due to a negative rating bias in people with higher levels of depressive symptoms, as opposed to specific experiences of treatment (Priebe et al., 1998; Fakhoury et al., 2002; Hansson et al., 2007). In this study we aimed to identify what types of symptoms are associated with the initial appraisal of hospital treatment in patients with schizophrenia, after adjusting the influence of socio-demographic and other clinical characteristics.

2. Methods

An exploratory cross-sectional study examining the association between patients' initial appraisal of inpatient treatment and various socio-demographic and symptom severity measures, analysing a pooled data set specifically collated

^{*} Correspondence to: Unit for Social and Community Psychiatry, Newham Centre for Mental Health, London E13 8SP, United Kingdom. Tel.: +44 20 7540 4210.

^{0165-1781/}\$-see front matter © 2012 Elsevier Ireland Ltd. All rights reserved. http://dx.doi.org/10.1016/j.psychres.2012.04.011

2

for this purpose. All the data was obtained within the first week of treatment apart from the diagnosis, which was recorded from patient notes at the point of discharge.

2.1. Sample

The current sample was obtained from pooling data from three studies. The InvolvE study (Priebe et al., 2009) assessed characteristics and experiences of inpatients in 22 hospitals across England. The EUNOMIA study (Kallert et al., 2005; Priebe et al., 2010b) was a related study with a similar design conducted at sites in 11 European countries. Both studies had a focus on detained patients but also included voluntary patients who felt coerced to treatment. The EUNOMIA study's data from London was included as part of the InvolvE study, and so these were omitted from the EUNOMIA sample to ensure that participants were not included twice. The third study, EDEN (Kallert et al., 2007), was a randomised controlled trial comparing outcomes of voluntary patients in day hospitals with those in conventional in-patient settings. Details of the rationale, methods and findings of each of the three studies have been described elsewhere.

All three studies included consecutively admitted patients who had the capacity to provide informed consent, aged between 18 and 65. Consistent exclusion criteria were as follows: admitted because of acute intoxication, being a forensic patient, and being transferred from another hospital. In the EDEN study patients requiring 1-to-1 supervision were excluded, whilst in the EUNOMIA and InvolvE study voluntary patients who did not report a coercion level of at least three on the MacArthur Admission Experience Survey were excluded. For this study, only patients with a main diagnosis of schizophrenia or a related disorder according to ICD-10 (F20-29, WHO, 1998) were included. With respect to the EDEN study, only patients treated in conventional hospital settings were included.

2.2. Measures

Initial appraisal of treatment was obtained using the Client Scale for Assessment of Treatment (CAT; Priebe and Gruyters, 1995b; Richardson et al., 2011). The questionnaire has seven items and asks the patient whether they believe they are receiving the right care, whether their psychiatrist understands them and if other staff are pleasant to them, if they believe they are on the right medication, if they feel well respected and regarded, whether the care received has been helpful, and whether they feel other elements of their care are appropriate. Scores range from 0 ("not at all"), to 10 ("yes entirely"). Patients with missing data on four or more items were excluded from the analysis. Recent factorial analysis supports the use of the CAT as a meaningful unidimensional scale, stable over three different European countries (Richardson et al., 2011), and in previous research the CAT has been found to have excellent internal consistency (α =0.90, Priebe et al., 2009).

Symptom severity was measured on the 24-item version of the Brief Psychiatric Rating Scale (BPRS; Ventura et al., 1993). Scores range from 1 ("not present") to 7 ("extremely severe"). In examining the severity of symptoms at a sub-syndrome level with the BPRS a number of different 4 and 5 factor models have been postulated (Van der Does et al., 1995; Berger et al., 1997; Ventura et al., 2000; Velligan et al., 2005). Of those, Velligans 4-factor model was selected. Interrater reliability was high in all the three studies, with researchers on the EDEN and EUNOMIA studies achieved an intra-class coefficient score of 0.78.

Socio-demographic details including age, gender, marital status, previous admission history and employment status were obtained using the MANSA (Priebe et al., 1999) in the InvolvE and EUNOMIA studies and the Clinical History Schedule (Kallert et al., 2000) in the EDEN studies. The categories used were identical in the three studies.

All the assessments took place within the first week following the admission (in the EDEN study always within the first two days), and were conducted by a researcher not involved in the patients' treatment.

2.3. Analysis

In univariate analyses linear regressions were conducted with the initial appraisal of treatment as the dependant variable. In addition to the BPRS subscale scores, the following variables were tested as potential predictors: age, gender, employment status, marital status, and the legal status of admission. For the employed" (students were treated as employed for this analysis), and "married" vs. "unemployed" (students were treated as employed for this analysis), and "married" us, "unmarried" (which included those divorced, single and widowed). The patient was defined as involuntarily held if they were initially admitted on a voluntary basis but later detained within the first week, prior to the research assessment taking place. We did not test whether differences in the initial appraisal of treatment were explained by the study from which the data were taken, because in one of the three studies (EDEN) there were only voluntary patients. Since we aimed to analyse the impact of legal status, we did not enter study as a predictor as it would have confounded part of the analysis.

In the next step a multiple linear regression was conducted with all the considered predictor variables entered into the model simultaneously. In the final part of the analysis, interaction effects between legal status and each BPRS subscale score were added to the multivariate model separately in order to examine whether the effect of symptom severity on the appraisal of treatment differed between those legally detained and those admitted voluntarily. In order to rule out the possibility of any bias in the results occurring due to the heterogeneity of the samples used, a subgroup analysis was conducted on the largest dataset (EUNOMIA).

To tackle any potential issues which could arise through a listwise deletion of incomplete cases (Little and Rubin, 1987) a multiple imputation using a chained equation of 10 cycles was conducted. All values included in the analysis were entered both as predictors and for imputing. Twenty-five imputations were conducted (Spratt et al., 2010), with the analysis conducted on the pooled data. R^2 scores were obtained from the imputed data using the method outlined by Harel (2009), which involves conducting a Fisher's *r*-to-*z* transformation for each imputed *R* score, combining them in accordance with Rubin's rules (Rubin, 1987), before transforming them back to R^2 . All the analysis was conducted on SPSS version 18.

3. Results

3.1. Sample characteristics

Around 2316 patients met the inclusion criteria. Of which about 211 patients were omitted due to missing or incomplete CAT scores, leaving 2105 patients in the final sample. The clinical and socio-demographic characteristics of this sample are presented in Table 1. Of the final total, 1556 were recruited as part of the EUNOMIA project, 393 as part of the InvolvE study, and 156 as part of the EDEN study. A missing value analysis indicated that 2.8% of all predictor values (653 in total) was missing, with 21.1% of cases (445) missing at least one. Of the 653 values missing in total, 53.9% (352 in total) related to the duration of illness.

3.2. Symptoms and other patient characteristics associated with initial appraisal of treatment

The univariable and multivariable associations of symptoms and other patient characteristics with the initial appraisal of treatment are shown in Table 2.

Higher levels of mania and positive symptoms, lower levels of depression/anxiety symptoms, being involuntarily admitted, unemployed and unmarried were all significantly associated with a less favourable initial appraisal of treatment in univariable analyses. After simultaneously adjusting for potential confounding by other variables; older age, being male, not being detained,

Table 1

Socio-demographic and clinical characteristics of the patients.

Characteristics	n or Mean	% or SD
Patients (<i>N</i>)	2105	
% Detained involuntarily	1530	72.7%
Age (years)	38.49	11.24
Gender (% female)	906	43.1%
Marital status (% married)	454	22.0%
Employment		
Standard paid employment	385	18.5%
Unemployed (incl. disability benefits)	1554	74.9%
Other (e.g. student, home-maker)	138	6.7%
Previous psychiatric admission (%yes)	1517	76.7%
Illness duration	7.84	9.11
BPRS		
Depression/anxiety subscale score	2.24	1.08
Mania subscale score	1.84	0.96
Negative subscale score	2.14	1.08
Positive subscale score	2.92	1.26
CAT mean score	6.00	1.28

ARTICLE IN PRESS

M. Savill et al. / Psychiatry Research & (****) ***-***

Table 2

Univariate and multivariate analysis of variables associated with patients' initial appraisal of treatment.

Predictor variables	Univariate ar	nalysis		Multivariate	analysis	
	В	95% CI	р	В	95% CI	р
Socio-demographic data						
Age	0.010	0.000 to 0.020	0.061	0.021	0.009 to 0.033	0.001
Female vs. male	-0.214	-0.450 to 0.022	0.076	-0.346	-0.581 to -0.111	0.004
Past admission (Yes/No)	-0.178	0.235 to -0.473	0.235	-0.022	-0.338 to 0.293	0.889
Illness duration	-0.014	-0.028 to 0.000	0.056	-0.012	-0.030 to 0.005	0.172
Detained legal status	-1.207	-1.464 to -0.949	< 0.001	-0.984	-1.247 to -0.722	< 0.001
Unemployed	-0.441	-0.726 to -0.156	0.002	-0.305	-0.589 to -0.021	0.036
Married	0.319	0.034 to 0.604	0.028	0.135	-0.155 to 0.426	0.360
BPRS symptom clusters						
Depressive/anxiety subscale	0.130	0.021 to 0.239	0.019	0.017	-0.091 to 0.125	0.758
Mania subscale	-0.674	-0.794 to -0.554	< 0.001	-0.490	-0.620 to -0.360	< 0.001
Negative subscale	-0.190	-0.218 to -0.000	0.051	-0.005	-0.114 to 0.104	0.929
Positive subscale	-0.465	-0.557 to -0.372	< 0.001	-0.226	-0.330 to -0.122	< 0.001

being employed, and having a low level of manic and positive symptoms were associated with a more favourable appraisal of treatment. The variance explained by the multivariable model was R^2 =0.106, with manic symptoms, with a detained legal status of admission and mania symptoms explaining the greatest proportion.

In the subgroup analysis with data from the EUNOMIA study only, the effect of the variables remained broadly consistent, with all of the 95% confidence intervals falling inside those from the full sample and with the same variables significant and in the same direction. The only exception to this was 'employment status' (B = -0.305, 95%, CI = -0.589 to -0.021, P = 0.036 in the full sample, in comparison to B = -0.230, 95%, CI = -0.561 to 0.100, P = 0.172).

3.3. Legal status of admission, symptom severity and initial treatment evaluation

The next stage of the analysis explored whether symptom levels had a similar predictive value in voluntary and involuntary patients. An interaction effect between legal status and each symptom subscale score was separately entered into the existing multivariable model (presented in Table 2). Such interaction effect was found only for manic symptoms (B=0.372, 95%, CI=0.067-0.677, P=0.017). A post hoc analysis was conducted in order to identify the direction of the interaction. In detained patients the association between more manic symptoms and less positive treatment appraisal was stronger (R^2 =0.062, P<0.001) than in voluntary patients (R^2 =0.012, P=0.009).

4. Discussion

4.1. Main findings

Mania, depression/anxiety and positive BPRS subscale scores were significantly associated with patients' initial appraisal of treatment in the univariate analysis. When the influence of all tested variables was considered however, only two types of symptoms remained significant: patients with higher levels of mania and positive symptoms tended to appraise hospital treatment more negatively. Younger age, female gender, being unemployed and being detained involuntarily were also linked with less positive appraisals in multivariable analyses.

4.2. Strengths, limitations and methodological considerations

Whilst previous research emphasised the importance of patients' initial appraisal of treatment as a predictor of treatment outcome, this is the first large study focusing on understanding the factors that determine more or less favourable initial appraisals of hospital treatment. The sample size (2105 patients) provides sufficient statistical power to detect even small effect sizes, thus allowing the interpretation of negative findings. We tested observer-ratings of symptoms as predictors of the patient-rated initial appraisal of treatment. This avoided obtaining associations as a mere effect of a generalised tendency to provide more or less positive ratings in different self-reporting measures (see Hansson et al., 2007; Fakhoury et al., 2002). The same measures of symptoms and treatment appraisal were used in all studies from which the data were pooled, and that all ratings were provided by well trained researchers who were not involved in treatment.

The major limitation of the study is a potential selection bias. In two of the three studies (InvolvE and EUNOMIA), voluntary patients were included only if they expressed a level of being coerced. In the third study (EDEN), there were no involuntary patients. Also, not all eligible patients could be recruited in the three studies. This may have influenced the absolute levels of symptoms and initial treatment appraisals, however associations between variables are assumed to be more robust against a selection bias, and exploring associations was the aim of this analysis. Another limitation is that we did not consider in the analysis what exact treatment was administered during the first days of hospital treatment and whether specific treatment components were linked with patients' appraisal of treatment. We did not have consistent data on what type and dose of antipsychotic medication patients had taken at the time of the interview, so that medication was not examined as a possible influencing or mediating factor. Finally, the multivariable model explained only about 10% of the variance of the initial assessment of treatment. Whilst such an amount of explained variance is common for this type of research, it still leaves about 90% of the variance unexplained.

4.3. Comparison with previous research

Patients' initial statements about the appropriateness of their psychiatric treatment have been the subject of research for more than 30 years. They have been referred to using terms such as initial subjective response (Van Putten and May, 1978), early subjective reactions (Priebe, 1987), and initial treatment satisfaction (Priebe et al., 2009). In this study, we used initial appraisal.

It reflects the type of questions asked in the CAT and mirrors the nature of assessing a treatment that has just started and is unlikely to have yet had a major effect. However, when comparing the findings with the literature, one has to consider that other studies may have used a different terminology.

The literature on treatment satisfaction in psychiatry, i.e. not initial but later appraisals of treatment, suggests that mood symptoms have the greatest impact on patients' ratings. Patients who are more depressed tend to express a more negative view of their treatment. It has been suggested that this reflects a negative rating bias of patients with high levels of depression rather than a specific experience of treatment (Priebe et al., 1998; Hansson et al., 2007; Fakhoury et al., 2002; Fakhoury and Priebe, 2002). In this study, the multivariable analysis did not identify depression, but mania and positive symptoms as influential. This is unlikely to be explained by a mood dependent rating bias, but may reflect how patients with different symptoms of schizophrenia experience the in-patient setting. Given patients level of insight into their disorder has been found to negatively correlate with both positive and negative symptoms (Minz et al., 2003; De Hert et al., 2009), whilst in our study negative symptoms were found to have no impact on treatment appraisal, the finding is also unlikely to be attributable to simply how aware the patients were of their illness and the need for treatment.

The finding is partly consistent with the results of Richardson and colleagues (2010) who reported a less positive initial treatment appraisal for patients with higher levels of mania, positive and depression/anxiety symptoms. That analysis however had a different focus, used a smaller sample of exclusively involuntary patients, a different factor analysis of the BPRS to define the subsyndromes, and did not adjust for other symptoms and patient characteristics.

Patients with higher levels of manic and positive symptoms may have greater difficulty in coping with being placed in the confined environment of a hospital ward. In the ward they have reduced autonomy and limited space. They have to conform to the organisational requirements of the ward and are surrounded by other patients who are also acutely ill, many of whom can be noisy and aggressive. Manic and positive symptoms may lead to less satisfactory interactions with other service users and staff members and greater conflict, which in turn may negatively impact upon their perception of treatment still further. Patients who exhibit 'disturbed behaviour' or experience acute positive symptom exacerbation are also more likely to receive combined anti-psychotics and high doses of antipsychotic medication (Paton et al., 2008), which can lead to a higher side effect burden (Centorrino et al., 2004), which could result in a lower appraisal of the treatment they receive. In a qualitative review of experiences of involuntary inpatient treatment (Katsakou and Priebe, 2007) a number of features were highlighted that can negatively impact upon a patient's experience of admission, and these may particularly resonate with those suffering from more intense manic and positive psychotic symptoms. Patients experiencing greater levels of positive symptoms such as paranoia or hallucinations may struggle more in an unfamiliar setting which may feel frightening or insecure. Patients with severe manic symptoms are likely to be more agitated, and struggle more with being contained in an enclosed space with rigid rules and limited opportunities for activities. The finding that manic symptoms influence patients' appraisal even more in those involuntarily admitted, who therefore would experience an even greater restriction to their autonomy, lends further support to this argument.

4.4. Implications

The findings suggest that patients with different symptom profiles experience and respond to psychiatric hospital treatment in different ways and, as a result, express different appraisals within a few days after admission. Patients with schizophrenia experiencing more severe manic and positive symptoms, and manic symptoms in particular if detained involuntarily are more likely to report a more negative initial evaluation of treatment. This suggests that it is symptom characteristics which may relate specifically to being in a ward environment that have a greater impact on the initial appraisal of treatment, as opposed to mood which has been found to be the most important in predicting a range of other patient self-reported outcomes (Priebe et al., 1998; Fakhoury et al., 2002).

The findings underline the importance of assessing patients' initial appraisal of treatment as a relevant process variable in both research and routine practice. Further studies may explore environmental factors, processes of interaction between service users and staff, treatment components and mediating processes as an explanation of the association of manic and positive symptoms with more negative appraisals of hospital treatment after only a few days. In a recent review Millon (2009) has suggested that acute inpatient settings are at present too custodial and over reliant on medication, and suggests that providing more psychosocial interventions as an alternative may be a way to try and address these issues. If these are problems which disproportionally affect those experiencing higher positive and manic symptoms, then this possible solution may go some way in addressing the more negative experiences these particular patients report.

Acknowledgement

The authors would like to thank Algirdas Dembinskas, Anastasia Karastergiou, Andrzej Kiejna, Pětr Nawka, George Onchev, Jiri Raboch, Torres-González, Tim Amos, Richard Morris, Til Wykes and their research teams for their part in the data collection, along with all the clinical staff and patients who took part. The authors would also like to thank Stephen Bremner for his advice on the presentation of the analysis. The data collection was funded by a Grant from the Policy Research Programme of the Department of Health UK, and the European Commission (Quality of Life and Management of Living Resources Programme). The funding source had no further role in the current study design, analysis, write up of the paper, or the decision to submit this paper for publication.

References

- Awad, A.G., Hogan, T.P., Voruganti, L.N., Heslgrave, R.J., 1995. Patients' subjective experiences on antipsychotic medications: implications for outcome and quality of life. International Clinical Psychopharmacology 10 (3), 123–132.
- Bartkó, G., Herzog, I., Békésy, M., 1987. Predicting outcome of neuroleptic treatment of the basis of subjective response and early clinical improvement. Journal of Clinical Psychiatry 48, 363–365.
- Berger, G.K., Calsyn, R.J., Morse, G.A., Klinkenberg, W.D., Trusty, M.L., 1997. Factor structure of the brief psychiatric rating scale. Journal of Clinical Psychology 43 (5), 451–454.
- Bröker, M., Röhricht, F., Priebe, S., 1995. Initial assessment of hospital treatment by patients with paranoid schizophrenia: a predictor of outcome. Psychiatry Research 58, 77–81.
- Centorrino, F., Goren, J.L., Hennen, J., Salvatore, P., Kelleher, J.P., Baldessarini, R.J., 2004. Multiple verses single antipsychotic agents for hospitalised psychiatric patients: case control study of risks verses benefits. American Journal of Psychiatry 161, 700–706.
- De Hert, M.A.F., Simon, V., Vidovic, D., Franic, T., Wampers, M., Peuskins, J., van Winkel, R., 2009. Evaluation of the association between insight and symptoms in a large sample of patients with schizophrenia. European Psychiatry 8, 507–512.
- Fakhoury, W., Kaiser, W., Röeder-Wanner, U.U., Priebe, S., 2002. Subjective evaluation: is there more than one criterion? Schizophrenia Bulletin 28, 319–327.
- Fakhoury, W., Priebe, S., 2002. Subjective quality of life: its association with other constructs. International Review of Psychiatry 14, 219–225.

- Hansson, L., Björkman, T., Priebe, S., 2007. Are important patient-rated outcomes in community mental health care explained by only one factor? Acta Psychiatrica Scandinavica 116, 115–118.
- Harel, O., 2009. The estimation of R^2 and adjusted R^2 in incomplete data sets using multiple imputation. Journal of Applied Statistics 26 (10), 1109–1118.
- Kallert, T.W., Glockner, M., Onchev, G., Raboch, J., Karastergiou, A., Solomon, Z., Magliano, L., Dembinskas, A., Kiejna, A., Nawka, P., Torres-Gonzaléz, F., Priebe, S., Kjellin, L., 2005. The EUNOMIA project on coercion in psychiatry: study design and preliminary data. World Psychiatry 4 (3), 168–172.
- Kallert, T.W., Priebe, S., McCabe, R., Kiejna, A., Rymaszewska, J., Nawka, P., Ocvár, L., Raboch, J., Stárková-Kališová, L., Koch, R., Schützwohl, M., 2007. Are day hospitals effective for acutely ill psychiatric patients? A European multicenter randomised controlled trial. Journal of Clinical Psychiatry 68, 278–287.
- Kallert, T.W., Schützwohl, M., Mattes, C., EDEN-study group, 2000. The Client Socio-demographic and Clinical History Inventory. Faculty of Medicine, Dresden University of Technology, Dresden, Germany.
- Katsakou, C., Priebe, S., 2007. Patient's experiences of involuntary hospital admission and treatment: a review of qualitative studies. Epidemiologia e Psichiatria Sociale 16 (2), 172–178.
- Little, R.J.A., Rubin, D.B., 1987. Statistical Analysis with Missing Data. John Wiley & Sons, New York.
- Millon, A., 2009. Mental health nurses establishing psychosocial interventions within acute patient settings. International Journal of Mental Health Nursing 18 (2), 83–90.
- Minz, A.R., Dobson, K.S., Romney, D.M., 2003. Insight in schizophrenia: a metaanalysis. Schizophrenia Research 61 (1), 75–88.
- Paton, C., Barnes, T.R.E., Cavanagh, M., Taylor, D., Lelliott, P., 2008. High-dose and combination antipsychotic prescribing in acute adult wards in the UK: the challenges posed by p.r.n. prescribing. British Journal of Psychiatry 192, 435–439.
- Priebe, S., 1987. Early subjective reactions predicting the outcome of hospital treatment in depressive patients. Acta Psychiatrica Scandinavica 76, 134–138.
- Priebe, S., Barnicot, K., McCabe, R., Kiejna, A., Nawka, P., Raboch, J., Schützwohl, M., Kallert, T., 2010a. Patients' subjective initial response and the outcome of inpatient and day hospital treatment. European Psychiatry 26, 408–413.
- Priebe, S., Gruyters, T., 1994. Patients' and caregivers' initial assessments of day hospital treatment and the course of symptoms. Comprehensive Psychiatry 34, 234–238.
- Priebe, S., Gruyters, T., 1995a. The importance of the first three days: predictors of treatment outcome in depressed inpatients. British Journal of Clinical Psychology 34, 229–236.
- Priebe, S., Gruyters, T., 1995b. Patients' assessment of treatment predicting outcome. Schizophrenia Bulletin 21, 87–94.
- Priebe, S., Huxley, P., Knight, S., Evans, S., 1999. Application and results of the Manchester Short Assessment of Quality of Life (MANSA). International Journal of Social Psychiatry 45, 7–12.
- Priebe, S., Katsakou, C., Amos, T., Leese, M., Morriss, R., Rose, D., Wykes, T., Yeeles, K., 2009. Patients' views and readmissions 1 year after involuntary hospitalisation. British Journal of Psychiatry 149, 49–54.

- Priebe, S., Kaiser, W., Huxley, P., Röder-Wanner, U.U., Rudolf, H., 1998. Do different subjective evaluation criteria reflect distinct constructs? Journal of Nervous and Mental Disease 186, 385–392.
- Priebe, S., Katsakou, C., Glöckner, M., Dembinskas, A., Fiorillo, A., Karastergiou, A., Kiejna, A., Kjellin, L., Nawka, P., Onchev, G., Raboch, J., Schuetwohl, M., Solomon, Z., Torres-González, F., Wang, D., Kallert, T., 2010b. Patients' views of involuntary hospital admission in 1 and 3 months: prospective study in 11 European countries. British Journal of Psychiatry 196, 179–185.
- Priebe, S., Katsakou, C., Yeeles, K., Amos, T., Morriss, R., Wang, D., Wykes, T., 2010c. Predictors of clinical and social outcomes following involuntary hospital admission: a prospective observational study. European Archives of Psychiatry and Clinical Neuroscience 261 (5), 377–386.
- Richardson, M., Katsakou, C., Priebe, S., 2010. Association of treatment satisfaction and psychopathological sub-syndromes among involuntary patients with psychotic disorders. Social Psychiatry and Psychiatric Epidemiology 46, 695–702.
- Richardson, M., Katsakou, C., Torres-Gonzalez, F., Onchev, G., Kallert, T., Priebe, S., 2011. Factorial validity and measurement equivalence of the client assessment of treatment scale for psychiatric inpatient care—a study of three European countries. Psychiatry Research 188, 156–160.
- Rubin, D.B., 1987. Multiple Imputation for Nonresponse in Surveys. John Wiley & Sons, New York.
- Shafer, A., 2005. Meta analysis of the brief psychiatric rating scale factor structure. Psychological Assessment 17, 324–335.
- Spratt, M., Carpenter, J., Sterne, J.A.C., Carlin, J.B., Heron, J., Henderson, J., Tilling, K., 2010. Strategies for multiple imputation in longitudinal studies. American Journal of Epidemiology 172 (4), 478–487.
- Van der Does, A.J., Dingemans, P.M., Linszen, D.H., Nugter, M.A., Scholte, W.F., 1995. Dimensions and subtypes of recent-onset schizophrenia: a longitudinal analysis. Journal of Nervous and Mental Disease 183, 681–687.
- Van Putten, T., May, R., 1978. Subjective response as a predictor of outcome in pharmacotherapy: the consumer has a point. Archives of General Psychiatry 35, 477–480.
- Van Putten, T., May, R., Marder, R., Wittman, L., 1978. Subjective response to antipsychotic drugs. Archives of General Psychiatry 38 (2), 187–190.
- Velligan, D., Prihoda, T., Dennehy, E., Biggs, M., Shores-Wilson, K., Crismon, L.M., Rush, J.A., Miller, A., Suppes, T., TrivedI, M., Kashner, M.T., Witte, B., Troprac, M., Carmody, T., Chiles, J., Shon, S., 2005. Brief psychiatric rating scale expanded version: how do new items affect factor structure? Psychiatric Research 135, 217–228.
- Ventura, J., Lukoff, D., Nuechterlein, K.H., Liberman, R.P., Green, M.F., Shaner, A., 1993. A manual for the expanded brief psychiatric rating scale. International Journal of Methods in Psychicatric Research 3, 227–243.
- Ventura, J., Nuechterlein, K.H., Subotnik, K.L., Gutkind, D., Gilbert, E., 2000. Symptom dimensions in recent-onset schizophrenia and mania: a principle components analysis of the 24-item brief psychiatric rating scale. Psychiatric Research 97, 129–135.
- World Health Organisation, 1998. ICD-10 Classifications of Mental and Behavioural Disorder: Diagnostic Criteria for Research (DCR-10). World Health Organisation, Geneva.

Appendix III: Systematic review protocol

Systematic review protocol

Objectives:

- 1) What is the course of negative symptoms in stable schizophrenia patients over a standard treatment period (i.e. 10 weeks-36 months)?
- 2) On a study-group level, what changes in negative symptoms can actually be achieved?
- 3) Do any patient or treatment characteristics influence the course of negative symptoms?
- 4) Does the method of assessment impact upon how stable the course of negative symptoms is recorded to be?
- 5) Is there any variation between the courses of individual negative symptoms?

Search process:

Electronic databases: MEDLINE, PsychINFO, EMBASE, CENTRAL.

Reference lists from selected papers hand-searched

Key journal search: American Journal of Psychiatry, Acta Scandinavica, British Journal of Psychiatry, Schizophrenia Bulletin, Schizophrenia Research, The Lancet. Hand-searched, back to 1962:

Selection Criteria:

Database selection:

Diagnosis: 100% of sample diagnosed with Schizophrenia.

Patient status: assessments conducted exclusively with outpatients

Age: 18-65

Sample size: minimum 50 participants total.

Length of time between first and last assessment points: minimum 10 weeks.

Methods: Quantitative research only. Not reviews, case-studies, or papers which have not used a validated symptom measurement scale.

Assessment methods: Any peer reviewed assessment tool used to establish the severity of specific negative symptoms of schizophrenia (including motor retardation), or the severity of a general schizophrenia negative symptom subscale. Examples measuring negative symptom as a singular subscale include (but are not exclusive to): BPRS, PANSS, SANS, NSA-16, NSA-4, PNS-Q. Measures of individual symptoms include (but not excluded to): SANS, Emotional Blunting

scale, Chapman Anhedonia scales, TEPS, Fawcett-Clark Pleasure scale, Snaith-Hamilton Scale or individual items on the above subscale measures.

Publication date: Start from when standardised assessment of negative symptoms was initiated (i.e. BPRS, 1962).

Language: Include non-English papers where possible (German, Italian, French, Spanish). Articles written in non-Latin characters excluded

Abstracts and 'Grey literature' excluded.

Search terms:

1 + 2 + 3

Medline search terms:

1. Disorder	2. Symptoms	3. Treatment /Duration
*Schizophren\$	Negative symptoms	Change\$
Deficit syndrome	Reduced affect	Effect\$
	Flattened affect	Therap*
	Blunted affect	Intervention /Intervention studies
	Emotional experience	Efficacy
	Emotional expression	Impact
	Alogia	Treatment / *therapeutics
	Anhedonia	Medication
	Avolition	*Longitudinal
	Asociality	*Follow up /*Follow up studies
	Motor retardation	course
	Amotivation	Stability
	Apathy	Time
		Progress\$
		Persist\$
		Year\$ */treatment outcome

* denotes an exploded term

\$ denotes open

/ denotes a meshed term

PsychINFO variation in mesh terms:

Efficacy: efficacy, * Treatment effect evaluation Intervention: intervention, /intervention Treatment: treatment, /treatment Medication: Medication, *drug therapy

EMBASE variation in mesh terms:

Deficit syndrome, *negative syndrome/ Blunted affect, /blunted affect Motor retardation, /motor retardation Anhedonia, /anhedonia Effect, *therapy effect/ Intervention, *intervention study/ Efficacy, *drug efficacy/ Treatment, /treatment outcome* Medication, /drug therapy*

CENTRAL full search terms list:

(schizophren* OR deficit syndrome) AND (negative symptoms OR reduced affect OR flattened affect OR blunted affect OR emotional experience OR emotional expression OR alogia OR anhedonia OR avolition OR asociality OR motor retardation OR amotivation OR apathy) AND (change* OR effect* OR therap* OR intervention OR efficacy OR impact OR treating OR medication OR longitudinal OR follow up OR course OR stability OR time OR progres* OR persist* OR year*):ti,ab,kw in Clinical Trials

-Screening (3 stage process: title, abstract, and full paper screen)

Exclude if:

- Non-human, neuro-imaging or neurobiology study -If the title indicates it is clearly not relevant (i.e. nothing to do with topic area, schizophrenia, negative symptoms etc.) -Study published in a non-Latin character language -The study is a case report, letter to the editor, conference abstract, book chapter, qualitative study, or review[†] -Clearly no repeated assessments (i.e. straight comparison between two samples in one time period) -includes inpatients -No repeated assessments of schizophrenia symptoms OR repeat defined by outcome rather than standardised time-point (i.e. re-admission) -<50 participants total -A structured assessment of negative symptoms was not undertaken, using a psychometrically validated assessment tool. -No details on exclusively schizophrenic samples presented -total study length <10 weeks, or no details reported within 3 year interval -Child/older adult (ie >65 years old) included in sample

* Note; if the study is a review concerning the longitudinal course of negative symptoms then source data should be obtained, and reference list screened.

Appendix IV: Systematic review extraction sheet

Inde	Indentification features S									cteristics			
#	Screener	date	Authors	Year	Title	Journal	Included?	Reason for exclusion	Design	Inclusion/e	xclusion cri	teria	
										1	2		3

							Country	Total	Method ofDx	Randomised?	participants blinded
4	5	6	7	8	9	10		participants		if so, how?	

			Data at each lev	el: measure 1						
Assessors blinded?	Multiple Raters?	management of		intervention	recruited	T1				
			Symptoms			std				
		drop out?	measured		number	timepoint	assessment method	mean	dev	no# parts

					Data at each level: measure							
	Т			Symptoms	intervention	recruited	ted T1					
timepoint	assessmen	mean	std dev	no# parts	measured		number	timepoint	assessment method	mean	std dev	no# parts

					Participant characteristics							
		Т				Age	Gender	Ethnicity %				
timepoint	assessmen	mean	std dev	no# parts	Mean	std dev	(% male)	1- White	2- Black Carribean	3- Black African	4- Black other	5- Indian

				Group level			Co-morbidity %	
6-Pakistani	7- Bangladeshi	8- Chinese	9-Other	co-morbidities	Anxiety Disorder	Depressive disorder	Substance abuse disorder	Personality Disorder

% employed	length of	education	Illness	duration	Age o	ofonset	no# pre	vadmis	Anti-psychotic med	Stated conflict of
	mean	std dev	mean	std dev	mean	std dev	mean	std dev	(chlorprom equiv, mg)	Interests?